

CITY COUNCIL REGULAR MEETING AGENDA Monday, May 13, 2024 at 7:00 PM

T (937) 848-4666 | www.cityofbellbrook.org

- 1. CALL TO ORDER
- 2. PLEDGE OF ALLEGIANCE
- 3. ROLL CALL
 - A. Motion to excuse the absence of T.J. Hoke
- 4. APPROVAL OF THE MINUTES
 - A. Approval of the April 22, 2024 Regular City Council Meeting Minutes.
- 5. MAYOR'S ANNOUNCEMENTS AND SPECIAL GUEST
 - A. Recognition and Celebration of Jerry Burgan's Retirement After More Than 37 Years Service to Bellbrook.
- 6. CITIZEN COMMENTS
- 7. CITIZENS REGISTERED TO SPEAK ON AGENDA ITEMS
- 8. PUBLIC HEARING OF PROPOSED ORDINANCES
 - A. Ordinance 2024-O-4 APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR DOWNTOWN BELLBROOK (Havens)
 - B. Ordinance 2024-O-5 APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR THE WEST BUSINESS DISTRICT OF BELLBROOK (Ashley)
 - C. Ordinance 2024-O-6 REPEALING TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE IN ITS ENTIRETY AND ENACTING A NEW TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE (Greenwood)

9. INTRODUCTIONS OF ORDINANCES

A. Ordinance No. 2024-O-7 AUTHORIZING THE CITY MANAGER TO SUBMIT THE TAX BUDGET FOR FISCAL YEAR 2025 TO THE GREENE COUNTY AUDITOR, AND DECLARING AN EMERGENCY (Cyphers)

10. ADOPTION OF RESOLUTIONS

A. Resolution 2024-R-12 DETERMINING IT NECESSARY TO PROCEED WITH SUBMITTING THE QUESTION OF LEVYING AN ADDITIONAL TAX IN EXCESS OF THE TEN-MILL LIMITATION TO THE ELECTORS OF THE CITY, PURSUANT TO OHIO REVISED CODE SECTION 5705.19 AND SUBSECTION 5705.19(JJ), AS AMENDED, WHICH LEVY SHALL BE TWO AND TWO TENTHS (2.2) MILLS AND SHALL RUN FOR A CONTINUING PERIOD OF TIME PURSUANT TO OHIO REVISED CODE SECTION 5705.19, AND REQUESTING THE COUNTY AUDITOR TO CERTIFY MATTERS IN **CONNECTION THEREWITH (Harding)**

11. OLD BUSINESS

- A. Dissemination of master priorities list rankings.
- 12. NEW BUSINESS
- 13. CITY MANAGER'S REPORT
- 14. COMMITTEE REPORTS
 - A. Safety Committee
 - B. Service Committee
 - C. Finance/Audit Committee
 - D. Community Affairs Committee

- **15. CITY OFFICIAL COMMENTS**
- **16. EXECUTIVE SESSION**
- 17. ADJOURNMENT

File Attachments for Item:

A. Approval of the April 22, 2024 Regular City Council Meeting Minutes.

RECORD OF PROCEEDINGS

Bellbrook City Council Meeting April 22, 2024

1. CALL TO ORDER:

Mayor Schweller called the Regular Meeting of the Bellbrook City Council to order at 7:00pm

2. PLEDGE OF ALLEGIANCE:

Mayor Schweller led the Council in the Pledge of Allegiance.

3. ROLL CALL

PRESENT

Mr. Logan Ashley

Mrs. Katherine Cyphers

Mr. Forrest Greenwood

Mr. Ernie Havens

Mayor Mike Schweller

ABSENT

Mr. Brady Harding

Mr. T.J. Hoke

Mayor Schweller asked for a Motion to excuse Mr. T.J. Hoke and Mr. Brady Harding. Motion made by Mr. Havens, Seconded by Mrs. Cyphers.

Voting Yea: Mr. Ashley, Mrs. Cyphers, Mr. Greenwood, Mr. Havens, Mayor Schweller Motion Carries

4. APPROVAL OF THE MINUTES

A. Approval of the April 8, 2024, Regular Council Meeting Minutes.

Mayor Schweller asked if anyone had comments or corrections to the minutes of the April 8, 2024, meeting. Hearing none, the minutes were declared to be approved.

5. MAYOR'S ANOUNCEMENTS AND SPECIAL GUESTS

- A. Representatives from Premier Health recognized emergency medical efforts from the Bellbrook Police and Fire personnel which resulted in the saving of a life.
- B. Mayor Schweller thanked the Sugar Maple committee and chairperson Logan Ashley for the great turn out for the festival. He also, thanked the police, fire, and admin departments for all their help.

6. CITIZEN COMMENTS

None

RECORD OF PROCEEDINGS

Bellbrook City Council Meeting April 22, 2024

7. CITIZENS REGISTERED TO SPEAK ON AGENDAS ITEMS

None

8. PUBLIC HEARING OF PROPOSED ORDINANCES

None

9. INTRODUCTIONS OF ORDINANCES

A. <u>Ordinance 2024-O-4:</u> Approving the application to create a community entertainment district for downtown Bellbrook (Havens)

Mayor Schweller noted there will be a Public Hearing on this matter scheduled for May 13, 2024.

Mr. Foster spoke on the CED in brief, noting a presentation will be provided for the public hearing.

Mayor Schweller asked for a Motion to introduce Ordinance 2024-O-4.

Motion to Introduce Ordinance 2024-O-4.

Motion made by Mr. Havens, Seconded by Mr. Greenwood.

Voting Yea: Mr. Ashley, Mrs. Cyphers, Mr. Greenwood, Mr. Havens, Mayor Schweller.

Motion Carries

B. Ordinance 2024-O-5: Approving the application to create a community entertainment district for the West business district of Bellbrook (Ashley)

Mayor Schweller asked for a Motion regarding Introduction of Ordinance 2024-O-5.

Motion to Introduce Ordinance 2024-O-5.

Motion made by Mr. Ashley, Seconded by Mrs. Cyphers.

Voting Yea: Mr. Ashley, Mrs. Cyphers, Mr. Greenwood, Mr. Havens, Mayor Schweller. Motion Carries

C. <u>Ordinance 2024-O-6:</u> Repealing title four "zoning" of part twelve "planning and zoning code" of the Bellbrook municipal code in its entirety and enacting a new title four "zoning" of part twelve "planning and zoning code" of the Bellbrook municipal code (Greenwood)

Mr. Foster explained this is the zoning code re-write and he will provide a presentation of the

Item A.Section 4, Item

RECORD OF PROCEEDINGS

Bellbrook City Council Meeting April 22, 2024

main changed at the public hearing.

Mayor Schweller asked for a Motion regarding Introduction of Ordinance 2024-O-6.

Motion to Introduce Ordinance 2024-O-6.

Motion made by Mr. Greenwood, Seconded by Mr. Havens.

Voting Yea: Mr. Ashley, Mrs. Cyphers, Mr. Greenwood, Mr. Havens, Mayor Schweller.

Motion Carries

10. ADOPTION OF RESOLUTIONS

None

11. **OLD BUSINESS**

None

12. **NEW BUSINESS**

None

13. CITY MANAGER REPORT

Mr. Schommer spoke on the upcoming Audit scheduled for April 29, 2024, and April 30, 2024, where the auditors will be onsite.

He also thanked the Bellbrook Police and Fire staff for all their hard work, noting the high level of service the community receives from two of the best public safety departments in the area.

Mr. Schommer mentioned plans for the 2025 budget starting with the tax budget.

14. **COMMITTEE REPORT**

A. Safety committee

None

B. Service Committee

Mr. Greenwood spoke on Earth Day, PFAS, and the upcoming EPA hearing fro Tom's Mulch.

C. Finance/Audit Committee

RECORD OF PROCEEDINGS

Bellbrook City Council Meeting April 22, 2024

None

D. Community Affairs Committee

Mr. Havens provided details on the process for the Bellbrook Sugarcreek Park District Commissioners position.

Mayor Schweller asked for a Motion to recommend to the Common Pleas Court the appointment of Greg Meredith to the position of Bellbrook Sugarcreek Park District Commissioner and authorizing the mayor to send such recommendation in letter form.

Motion made by Mr. Havens, Seconded by Mrs. Cyphers. Voting Yea: Mr. Ashley, Mrs. Cyphers, Mr. Greenwood, Mr. Havens, Mayor Schweller. Motion Carries

15. CITY OFFICIAL COMMENTS

Mr. Ashley- Thanked the City employees for all the help with the Sugar Maple Festival

Mrs. Cyphers- Thanked the Sugar Maple Festival Committee

Mr. Greenwood- Noted this was an awesome Sugar Maple Festival & thanked all the employees and festival staff. He also mentioned Mr. Jerry Burgan's upcoming retirement and how he will be greatly missed Congrats to the Police and Fire departments on their great work with the Bellbrook residents.

Mr. Havens- Thanked Mr. Ashley for his work with the Sugar Maple Festival. Also mentioned Police and blocking street with vehicle to protect the community.

16. EXECUTIVE SESSION

None

17. **ADJOURMENT**

Hearing no further business coming before the Council, Mayor Schweller declared the meeting adjourned at 8:07pm

Michael Schweller, Mayor	
Robert Schommer Clerk of Council	

File Attachments for Item:

A. Ordinance 2024-O-4 APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR DOWNTOWN BELLBROOK (Havens)



MEMORANDUM

TO: Bellbrook City Council FROM: Mayor Michael Schweller

Subject: Community Entertainment District Applications

City of Bellbrook

15 East Franklin Street Bellbrook, Ohio 45305

T (937) 848-4666 F (937) 848-5190

www.cityofbellbrook.org

I have reviewed the two submitted applications to establish both an East and West Community Entertainment District within the city limits of Bellbrook. Upon the review, I found the applications contained all the required elements in accordance to the applicable State Laws.

One application has been submitted by Matthew Borszsz c/o Treig LLC representing ownership of property at 4441 S.R. 725 within the proposed CED. The second application has been submitted by City Manager Rob Schommer at the direction of City Council on behalf of the City representing ownership of property within the proposed CED including 12 W. Franklin Street.

I agree the establishment of these Community Entertainment Districts are in the best interest of the community and will allow for current and future businesses and property owners within the City the opportunity for growth and enhancement. Without the consideration of establishing these CED's, the City of Bellbrook would significantly limit pending economic development opportunities as well as limit future opportunities for businesses.

As Bellbrook continues its excellent work towards revitalizing Downtown and enhancing services to the community, this growth and economic development initiative is vital to our continued success.

Considering all aspects of the submitted applications, and the benefit to the community, I recommend Council approve the applications and take the necessary actions to establish both East and West Community Entertainment Districts.

Signed,

Michael Schweller, Mavor



Monday May 13, 2024 7:00 pm City Council Chambers 15 E. Franklin Street

There will be an open Public Hearing by the Bellbrook City Council regarding Ordinance No.2024-O-4

A Public Hearing will be held by Bellbrook City Council, regarding Ordinance No 2024-O-4

Ordinance 2024-O-4 APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR DOWNTOWN BELLBROOK

Monday May 13, 2024 7:00 pm in the Council Chambers 15 E. Franklin Street. The public is welcome to attend or send comments to the Clerk of Council at clerk@cityofbellbrook.org. A copy of the Ordinance is attached.

Agenda and additional meeting information available at www.cityofbellbrook.org

May 13, 2024

Ordinance No. 2024-O-4

City of Bellbrook State of Ohio

Ordinance No. 2024-O-4

APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR DOWNTOWN BELLBROOK

WHEREAS, Ohio Revised Code Section 4301.80 allows the City of Bellbrook to create a Community Entertainment District (CED) on application to the Mayor which district would encompass in excess of 20 acres; and

WHEREAS, The City has applied for the creation of such district on lands encompassing areas within Downtown Bellbrook which lands exceed 20 acres; and

WHEREAS, the application is complete and meets the requirements of Ohio Revised Code 4301.80; and

WHEREAS, notice of such application to designate the Community Entertainment District was published and noticed as required by Ohio Revised Code 4301.80; and

WHEREAS, Mayor Michael Schweller has reviewed the application and has formally recommended approval of the application based upon the application meeting criteria set forth in Ohio Revised Code 4301.80.

NOW, THEREFORE, THE CITY OF BELLBROOK HEREBY ORDAINS:

Section 1. City Council hereby accepts the recommendation of the Mayor as required by Ohio Revised Code 4301.80 and further finds that the application requesting the creation of a Community Entertainment District (CED) does in fact contain those requirements as provided for in the referenced statute.

Section 2. City Council specifically finds as follows:

- A. The map drawing and Zoning Description incorporated by reference is in sufficient detail to identify the boundaries of the district as set forth in Exhibit "A"
- B. The application contains a general statement of the nature and types of establishments to be located within the proposed CED.
- C. That the proposed Community Entertainment District will substantially contribute to entertainment, retail, and social opportunities for the community.
- D. The proposed time frame for development of the establishments for the CED is in accordance with the City of Bellbrook zoning and development plans.
- E. The proposed CED is in excess of twenty contiguous acres and is within the City of Bellbrook which was incorporated as a Village prior to 1880 and is within Greene County where at least one CED already exists.

RECORD OF ORDINANCES

Item A.Section 8, Item

Ordinance No. 2024-O-4 May 13, 2024

Section 3. After due discussion and deliberation, review of the application, and after taking into account the recommendation of the Mayor with regard to the application for the creation of a Community Entertainment District, this Council hereby finds that the application meets the needs of the City and the requirements of Ohio Revised Code 4301.80. The City Manager is hereby authorized to take whatever actions are necessary to implement the application with any affected state agencies, or otherwise.

Section 4. That it is found and determined that all formal actions of the City Council relating to the adoption of this Ordinance were adopted in an open meeting of this Council, and that all deliberations of this Council and any of its committees that resulted in such formal action, were in meetings open to the public, in compliance with all legal requirements, including §121.22 of the Revised Code of the State of Ohio.

PASSED BY City Council this day of	2024.
Yeas;Nays.	
AUTHENTICATION:	
Michael W. Schweller, Mayor	
Robert Schommer, Clerk of Council	
APPROVED AS TO FORM:	

Stephen McHugh, Municipal Attorney

2



City of Bellbrook

15 East Franklin Street Bellbrook, Ohio 45305

T (937) 848-4666 F (937) 848-5190

www.cityofbellbrook.org

Community Entertainment District Application March 2024

Applicant's Name and Address

City of Bellbrook c/o Rob Schommer, City Manager 15 East Franklin Street Bellbrook, Ohio 45305 (937) 848-4666

Detailed Map of the Community Entertainment District (CED)

The attached exhibit "A" serves as the map of the proposed district that encompasses a total of 45.5 acres and includes privately owned land, publicly owned land, and public rights-of-way. The applicant is the property owner of the land listed below inside the proposed CED.

- 1. The City of Bellbrook L35000100020004200
- 2. The City of Bellbrook L35000200050004700
- 3. The City of Bellbrook L35000200050004500
- 4. The City of Bellbrook L35000200050004300
- 5. The City of Bellbrook L35000200050004900
- 6. The City of Bellbrook L35000200050004400
- 7. The City of Bellbrook L35000200050004200
- 8. The City of Bellbrook L35000100020005000
- 9. The City of Bellbrook L35000100020004900
- 10. The City of Bellbrook L35000100020004800
- 11. The City of Bellbrook L35000100020004600

Attached as Exhibit A.

Nature and Type of Establishments

The Ohio Revised Code, in Section 4301.80(A), a Community Entertainment District is defined as follows:

- (A) As used in this section, "community entertainment district" means a bounded area that includes or will include a combination of entertainment, retail, educational, sporting, social, cultural, or arts establishments within close proximity to some or all of the following types of establishments within the district, or other types of establishments similar to these:
- (1) Hotels;
- (2) Restaurants;
- (3) Retail sales establishments;
- (4) Enclosed shopping centers;
- (5) Museums;
- (6) Performing arts theaters;
- (7) Motion picture theaters;
- (8) Night clubs;
- (9) Convention facilities;
- (10) Sports facilities;
- (11) Entertainment facilities or complexes;
- (12) Any combination of the establishments described in division (A)(1) to (11) of this section that provide similar services to the community.

The proposed "Bellbrook East" CED encompasses Bellbrook's Historic Old Village District, the vibrant Downtown corridor of the city. Bellbrook's historic core is built upon its rich history as a regional tourism destination dating back to as early as 1816, to Village incorporation in 1832, through over a century of development to its entertainment, retail, social, and cultural connections. The historic core includes the acclaimed Blueberry Café, Macintosh Tavern, Veli's Pizzeria, and Bellbrook Brewing Company. In addition to specialty retail boutiques, Winter's Public Library, Bellbrook's Historical Museum, and many other small businesses are included in the proposed CED. A new full-service restaurant and a unique craft cocktail lounge are currently proposed in the proposed district as well. The city has also approved and began work on the "Downtown Streetscape Project". This project includes improvements to walkability, pedestrian safety, infrastructure, and the overall aesthetics of Downtown.

Collectively, the current and planned private and public development projects within the proposed CED reinforce the destination qualities of the area, and along with a dynamic mix of walkable uses, offer a unique attraction and catalyst for future development and investment within the district.

Timeframe for Completion of Establishments

The proposed CED is fully developed. However, redevelopment of several properties will begin in 2024.

Evidence that Land in the CED is Consistent with the City's Comprehensive Plan

The land uses in the proposed CED are consistent with the City's Comprehensive Plan.

Attached as Exhibit B.

Certificate from Surveyor/Engineer that CED Contains at Least Twenty Contiguous Acres

Attached as Exhibit C

A processing Fee of Ten Dollars (\$10) made payable to the City of Bellbrook was submitted with the original application.

Exhibit A



----- Indicates outline of proposed Community Entertainment District

----- Indicates property owned by applicant



City of Bellbrook

15 East Franklin Street Bellbrook, Ohio 45305

T (937) 848-4666 F (937) 848-5190

www.cityofbellbrook.org

Exhibit B

March 19, 2024

City of Bellbrook 15 East Franklin Street Bellbrook, Ohio 45305

RE: Community Entertainment District-Bellbrook East

Dear Sir or Madam:

Please be advised that the proposed Community Entertainment District-Bellbrook East for which boundaries are defined on the attached map (Exhibit A) are consistent with the City's Comprehensive Plan and Zoning Map as required in Ohio Revised Code Section 4301.80(B)(5).

City of Bellbrook

Jason Foster, OhioCED

Community Development Administrator



2500 Newmark Drive Miamisburg, OH 45342 | (866) 552-3536 | <u>JAine com</u>

March 20, 2024

Mayor Michael Schweller City of Bellbrook 15 East Franklin Street Bellbrook, OH 45305

Re: Community Entertainment District – Bellbrook East

Dear Mayor Schweller:

Please be advised that the proposed Community Entertainment District - Bellbrook East, for which boundaries are outlined in blue as defined on the attached map (Exhibit A), contains no less than twenty (20) contiguous acres as required in the Ohio Revised Code Section 4301.80(B)(6).

Sincerely,

LJB Inc.

Daniel Hoying, P.E., P.S., STP

Project Development, Infrastructure















File Attachments for Item:

B. Ordinance 2024-O-5 APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR THE WEST BUSINESS DISTRICT OF BELLBROOK (Ashley)



MEMORANDUM

TO: Bellbrook City Council FROM: Mayor Michael Schweller

Subject: Community Entertainment District Applications

City of Bellbrook

15 East Franklin Street Bellbrook, Ohio 45305

T (937) 848-4666 F (937) 848-5190

www.cityofbellbrook.org

I have reviewed the two submitted applications to establish both an East and West Community Entertainment District within the city limits of Bellbrook. Upon the review, I found the applications contained all the required elements in accordance to the applicable State Laws.

One application has been submitted by Matthew Borszsz c/o Treig LLC representing ownership of property at 4441 S.R. 725 within the proposed CED. The second application has been submitted by City Manager Rob Schommer at the direction of City Council on behalf of the City representing ownership of property within the proposed CED including 12 W. Franklin Street.

I agree the establishment of these Community Entertainment Districts are in the best interest of the community and will allow for current and future businesses and property owners within the City the opportunity for growth and enhancement. Without the consideration of establishing these CED's, the City of Bellbrook would significantly limit pending economic development opportunities as well as limit future opportunities for businesses.

As Bellbrook continues its excellent work towards revitalizing Downtown and enhancing services to the community, this growth and economic development initiative is vital to our continued success.

Considering all aspects of the submitted applications, and the benefit to the community, I recommend Council approve the applications and take the necessary actions to establish both East and West Community Entertainment Districts.

Signed,



Monday May 13, 2024 7:00 pm City Council Chambers 15 E. Franklin Street

There will be an open Public Hearing by the Bellbrook City Council regarding Ordinance No.2024-O-5

A Public Hearing will be held by Bellbrook City Council, regarding Ordinance No 2024-O-5

Ordinance 2024-O-5 APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR THE WEST BUSINESS DISTRICT OF BELLBROOK

Monday May 13, 2024 7:00 pm in the Council Chambers 15 E. Franklin Street. The public is welcome to attend or send comments to the Clerk of Council at clerk@cityofbellbrook.org. A copy of the Ordinance is attached.

Ordinance No. 2024-O-5

May 13, 2024

City of Bellbrook State of Ohio

Ordinance No. 2024-O-5

APPROVING THE APPLICATION TO CREATE A COMMUNITY ENTERTAINMENT DISTRICT FOR THE WEST BUSINESS DISTRICT OF BELLBROOK

WHEREAS, Ohio Revised Code Section 4301.80 allows the City of Bellbrook to create a Community Entertainment District (CED) on application to the Mayor which district would encompass in excess of 20 acres; and

WHEREAS, Treig LLC has applied for the creation of such district on lands encompassing areas within the West area of Bellbrook including 4441 S.R. 725 which lands in total exceed 20 acres; and

WHEREAS, the application is complete and meets the requirements of Ohio Revised Code 4301.80; and

WHEREAS, notice of such application to designate the Community Entertainment District was published and noticed as required by Ohio Revised Code 4301.80; and

WHEREAS, Mayor Michael Schweller has reviewed the application and has formally recommended approval of the application based upon the application meeting criteria set forth in Ohio Revised Code 4301.80.

NOW, THEREFORE, THE CITY OF BELLBROOK HEREBY ORDAINS:

Section 1. City Council hereby accepts the recommendation of the Mayor as required by Ohio Revised Code 4301.80 and further finds that the application requesting the creation of a Community Entertainment District (CED) does in fact contain those requirements as provided for in the referenced statute.

Section 2. City Council specifically finds as follows:

- A. The map drawing and Zoning Description incorporated by reference is in sufficient detail to identify the boundaries of the district as set forth in Exhibit "A"
- B. The application contains a general statement of the nature and types of establishments to be located within the proposed CED.
- C. That the proposed Community Entertainment District will substantially contribute to entertainment, retail, and social opportunities for the community.
- D. The proposed time frame for development of the establishments for the CED is in accordance with the City of Bellbrook zoning and development plans.
- E. The proposed CED is in excess of twenty contiguous acres and is within the City of Bellbrook which was incorporated as a Village prior to 1880 and is within Greene County where at least one CED already exists.

RECORD OF ORDINANCES

Item B.Section 8, Item

Ordinance No. 2024-O-5 May 13, 2024

Section 3. After due discussion and deliberation, review of the application, and after taking into account the recommendation of the Mayor with regard to the application for the creation of a Community Entertainment District, this Council hereby finds that the application meets the needs of the City and the requirements of Ohio Revised Code 4301.80. The City Manager is hereby authorized to take whatever actions are necessary to implement the application with any affected state agencies, or otherwise.

Section 4. That it is found and determined that all formal actions of the City Council relating to the adoption of this Ordinance were adopted in an open meeting of this Council, and that all deliberations of this Council and any of its committees that resulted in such formal action, were in meetings open to the public, in compliance with all legal requirements, including §121.22 of the Revised Code of the State of Ohio.

PASSED BY City Council this day or	t 2024.
Yeas;Nays.	
AUTHENTICATION:	
Michael W. Schweller, Mayor	_
Robert Schommer, Clerk of Council	_
APPROVED AS TO FORM:	

Stephen McHugh, Municipal Attorney

2

Community Entertainment District Application

March 2024

Applicant's Name and Address

Treig LLC c/o Matthew Borszcz 4441 State Route 725 Bellbrook, Ohio 45305 (937) 475-6579

Detailed Map of the Community Entertainment District (CED)

The attached exhibit serves as the map of the proposed district that encompasses a total of twenty-three (23) acres. The applicant is the property owner of the land listed below inside the proposed CED.

1. Treig LLC – L35000200130017900

Attached as Exhibit A.

Nature and Type of Establishments

The Ohio Revised Code, in Section 4301.80(A), a Community Entertainment District is defined as follows:

- (A) As used in this section, "community entertainment district" means a bounded area that includes or will include a combination of entertainment, retail, educational, sporting, social, cultural, or arts establishments within close proximity to some or all of the following types of establishments within the district, or other types of establishments similar to these:
- (1) Hotels;
- (2) Restaurants;
- (3) Retail sales establishments;
- (4) Enclosed shopping centers;
- (5) Museums;
- (6) Performing arts theaters;
- (7) Motion picture theaters;
- (8) Night clubs;
- (9) Convention facilities;
- (10) Sports facilities;
- (11) Entertainment facilities or complexes;
- (12) Any combination of the establishments described in division (A)(1) to (11) of this section that provide similar services to the community.

The proposed "Bellbrook West" CED contains a diverse group of businesses engaged in commercial activity, including restaurants, retail sales establishments, service establishments, and many other complimentary businesses. The proposed district encompasses the City of

Bellbrook's western most commercial centers and an undeveloped 7.36-acre tract along the western boundary line of the city.

Timeframe for Completion of Establishments

The proposed CED is fully developed with the exception of two parcels on the western boundary line totaling 7.36 acres. These parcels could be developed as early as 2025.

Evidence that Land in the CED is Consistent with the City's Comprehensive Plan

The land uses in the proposed CED are consistent with the City's Comprehensive Plan.

Attached as Exhibit B.

Certificate from Surveyor/Engineer that CED Contains at Least Twenty Contiguous Acres

Attached as Exhibit C

A processing Fee of Ten Dollars (\$10) made payable to the City of Bellbrook was submitted with the original application.

Exhibit A



----- Indicates outline of Community Entertainment District-Bellbrook West

----- Indicates property owned by applicant



City of Bellbrook

15 East Franklin Street Bellbrook, Ohio 45305

T (937) 848-4666 F (937) 848-5190

www.cityofbellbrook.org

Exhibit B

March 19, 2024

City of Bellbrook 15 East Franklin Street Bellbrook, Ohio 45305

RE: Community Entertainment District-Bellbrook East

Dear Sir or Madam:

Please be advised that the proposed Community Entertainment District-Bellbrook West for which boundaries are defined on the attached map (Exhibit A) are consistent with the City's Comprehensive Plan and Zoning Map as required in Ohio Revised Code Section 4301.80(B)(5).

City of Bellbrook

Jason Foster, OhioCED

Community Development Administrator



2500 Newmark Drive Miamisburg, OH 45342 | (866) 552-3536 | Lilling com

March 20, 2024

Mayor Michael Schweller City of Bellbrook 15 East Franklin Street Bellbrook, OH 45305

Re: Community Entertainment District – Bellbrook West

Dear Mayor Schweller:

Please be advised that the proposed Community Entertainment District - Bellbrook West, for which boundaries are outlined in blue as defined on the attached map (Exhibit A), contains no less than twenty (20) contiguous acres as required in the Ohio Revised Code Section 4301.80(B)(6).

HOYING

Sincerely,

LJB Inc.

Daniel Hoying, P.E., P.S., STP

Project Development, Infrastructure















File Attachments for Item:

C. Ordinance 2024-O-6 REPEALING TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE IN ITS ENTIRETY AND ENACTING A NEW TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE (Greenwood)



Monday May 13, 2024 7:00 pm City Council Chambers 15 E. Franklin Street

There will be an open Public Hearing by the Bellbrook City Council regarding Ordinance No. 2024-O-6

A Public Hearing will be held by Bellbrook City Council, regarding Ordinance No 2024-O-6

Ordinance 2024-O-6 REPEALING TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE IN ITS ENTIRETY AND ENACTING A NEW TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE

Monday May 13, 2024 7:00 pm in the Council Chambers 15 E. Franklin Street. The public is welcome to attend or send comments to the Clerk of Council at clerk@cityofbellbrook.org. A copy of the Ordinance is attached.

Agenda and additional meeting information available at www.cityofbellbrook.org

RECORD OF ORDINANCES

Item C.Section 8, Item

Ordinance No. 2024-O-6 May 13, 2024

City of Bellbrook State of Ohio

Ordinance No. 2024-O-6

REPEALING TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE IN ITS ENTIRETY AND ENACTING A NEW TITLE FOUR "ZONING" OF PART TWELVE "PLANNING AND ZONING CODE" OF THE BELLBROOK MUNICIPAL CODE

WHEREAS, Article XVIII, Section 3 of the Constitution of the State of Ohio grants municipalities the authority to exercise all powers of local self-government and to enact and enforce local police, sanitary, and other similar regulations that are not in conflict with the general laws; and

WHEREAS, the City of Bellbrook previously adopted Part Twelve "Planning and Zoning Code" and incorporated it into the Bellbrook Municipal Code; and

WHEREAS, based upon the recommendation of the Planning Board, City Council wishes to repeal Title Four "Zoning" of Part Twelve "Planning and Zoning Code" of the Bellbrook Municipal Code in its entirety and enact a new Title Four "Zoning" of Part Twelve "Planning and Zoning Code" of the Bellbrook Municipal Code to update zoning regulations in the City of Bellbrook; and

WHEREAS, in updating Title Four of Part Twelve of the Bellbrook Municipal Code, City Council intends to repeal the only two chapters in Title Four (i.e., Chapters 1240 "Zoning Code" and 1244 "Zoning Permits and Certificates") in their entirety and adopt a new Chapter 1240 "Zoning Code".

NOW, THEREFORE, THE CITY OF BELLBROOK HEREBY ORDAINS:

- Section 1. The old Title Four "Zoning" of Part Twelve "Planning and Zoning" of the Bellbrook Municipal Code is hereby repealed in its entirety. This specifically includes Chapter 1240 "Zoning Code" and Chapter 1244 "Zoning Permits and Certificates".
- Section 2. A new Title Four "Zoning" of Part Twelve "Planning and Zoning" of the Bellbrook Municipal Code is hereby enacted, which includes Chapter 1240 "Zoning Code" as set forth in Exhibit A, which is attached hereto and incorporated herein by reference, with deletions shown by **strikethrough** and additions shown by **bold** and <u>underscored</u>.
- Section 3. It is hereby found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council, and that any and all deliberations of this Council that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including, but not limited to Section 121.22 of the Ohio Revised Code.

PASSED BY C	ity Council this	day of	2024
Yeas;	Nays.		

RECORD OF ORDINANCES

Item C.Section 8, Item

Ordinance No. 2024-O-6 May 13, 2024

Michael W. Schweller, Mayor Robert Schommer, Clerk of Council APPROVED AS TO FORM: Stephen McHugh, Municipal Attorney	AUTHENTICATION:	
APPROVED AS TO FORM:	Michael W. Schweller, Mayor	
	Robert Schommer, Clerk of Council	

ARTICLE 1 TABLE OF CONTENTS AND FEE SCHEDULE

ARTICLE 2 <u>INTENT AND PURPOSE</u>

ARTICLE 3 CONSTRUCTION OF LANGUAGE AND DEFINITIONS

Section 3.01 Construction of Language

Section 3.02 Definitions

ARTICLE 4 ZONING DISTRICTS AND MAPS

Section 4.01 Districts

Section 4.02 District Boundaries

Section 4.03 Uncertainty of Boundary Location

Section 4.04 Zoning Upon Annexation

ARTICLE 5 R-1AA, R-1A, R-1B, ONE FAMILY RESIDENTIAL DISTRICTS

Section 5.01 Purpose and Intent

Section 5.02 Permitted Principal Uses

Section 5.03 Accessory Uses

Section 5.04 Conditional Uses

Section 5.05 Yard Requirements

Section 5.06 Building Height Regulations

Section 5.07 Accessory Parking

Section 5.08 Signs

Section 5.09 Transient Rental Regulations

ARTICLE 6 R-2 TWO FAMILY RESIDENTIAL DISTRICT

Section 6.01 **Purpose and** Intent

Section 6.02 Permitted Principal Uses

Section 6.03 Accessory Uses

Section 6.04 Conditional Uses

Section 6.05 Yard Requirements

Section 6.06 Building Height Regulations

Section 6.07 Accessory Parking

Section 6.08 Signs

Section 6.09 Transient Rental Regulations

ARTICLE 7 R-3 MULTI-FAMILY RESIDENTIAL DISTRICT

Section 7.01 **Purpose and Intent**

Section 7.02 Permitted Principal Uses

Section 7.03 Accessory Uses

Section 7.04 Conditional Uses

Section 7.05 Yard Requirements

2024-O-6 Exhibit A

Section 7.06 Building Height Regulations

Section 7.07 Accessory Parking

Section 7.08 Signs

Section 7.09 Schedule of Yard & Lot Requirements

Section 7.09 Transient Rental Regulations

Section 7.10 Schedule of Yard and Lot Requirements

ARTICLE 8 <u>O-1 OFFICE BUILDING DISTRICT</u>

Section 8.01 Purpose and Intent

Section 8.02 Permitted Principal Uses

Section 8.03 Accessory Uses

Section 8.04 Conditional Uses

Section 8.05 Yard Requirements

Section 8.06 Building Height Regulations

Section 8.07 Accessory Parking

Section 8.08 Off-Street Loading

Section 8.09 Signs

Section 8.10 Landscaping

Section 8.11 Screening

Section 8.12 Lot Coverage

ARTICLE 9 B-1 HIGHWAY BUSINESS DISTRICT

Section 9.01 **Purpose and Intent**

Section 9.02 Permitted Principal Uses

Section 9.03 Accessory Uses

Section 9.04 Conditional Uses

Section 9.05 Yard Requirements

Section 9.06 Building Height Regulations

Section 9.07 Accessory Parking

Section 9.08 Off-Street Loading

Section 9.09 Signs

Section 9.10 Screening

ARTICLE 10 <u>B-2 CONVENIENCE SHOPPING DISTRICT</u>

Section 10.01 **Purpose and** Intent

Section 10.02 Permitted Principal Uses

Section 10.03 Accessory Uses

Section 10.04 Yard Requirements

Section 10.05 Building Height Requirements

Section 10.06 Accessory Parking

Section 10.07 Off-Street Loading

2024-O-6 Exhibit A

Section 10.08 Signs Section 10.09 Screening

ARTICLE 11

B-3 NEIGHBORHOOD BUSINESS DISTRICT

Section 11.01 Purpose and Intent

Section 11.02 Permitted Principal Uses

Section 11.03 Accessory Uses

Section 11.04 Yard Requirements

Section 11.05 Building Height Regulations

Section 11.06 Accessory Parking

Section 11.07 Off-Street Loading

Section 11.08 Signs

Section 11.09 Screening

ARTICLE 12 B-4 CENTRAL BUSINESS DISTRICT

Section 12.01 **Purpose and** Intent

Section 12.02 Permitted Principal Uses

Section 12.03 Accessory Uses

Section 12.04 Yard Requirements

Section 12.05 Building Height Regulations

Section 12.06 Accessory Parking

Section 12.07 Off-Street Loading

Section 12.08 Signs

Section 12.09 Screening

ARTICLE 13

WO, WP WELL FIELD PROTECTION DISTRICTS

Section 13.01 Purpose and Intent

Section 13.02 Definitions

Section 13.03 Determination of Applicability

Section 13.04 WO Wellhead Operation District

Section 13.05 WP Wellfield Protection Overlay District

ARTICLE 14

BELLBROOK OLD VILLAGE DISTRICT PLAN

Section 14.01 General

Section 14.02 Designation of District

Section 14.03 Permitted Principal Uses

Section 14.04 Accessory Uses

Section 14.05 Conditional Uses

Section 14.06 Non-Conforming Building

Section 14.07 Yard Requirements

Section 14.08 Parking Requirements

2024-O-6 Exhibit A

Section 14.09 Off-Street Loading

Section 14.10 Signs

Section 14.11 Standards

Section 14.12 Exclusions and Controls Not Appropriate

Section 14.13 Village Review Board

Section 14.14 Application for Old Village District Permit

and Information Required

Section 14.15 Review Procedure

Section 14.16 Permit Required

Section 14.17 Appeals Procedure

Section 14.18 Special Procedure for Demolition

Applications

Section 14.19 Review of Requests for A Variance

Appendix A-1-14 Guidelines for the Old Village District

Form VRB-1 Application for Certificate of

Appropriateness

Form VRB-2 Work Sheet for Demolition Request

ARTICLE 15

FLOOD DAMAGE PREVENTION

Section 15.01 General Provisions

Section 15.02 Definitions

Section 15.03 Administration

Section 15.04 Use and Development Standards for Flood Hazard

Reduction

Section 15.05 Appeal and Variances

Section 15.06 Enforcement

ARTICLE 16

A-1 AGRICULTURAL DISTRICT

Section 16.01 Purpose and Intent

Section 16.02 Permitted Principal Uses

Section 16.03 Accessory Uses

Section 16.04 Conditional Uses

Section 16.05 Yard & and Lot Requirements

Section 16.06 Building Height Regulations

Section 16.07 Accessory Parking

Section 16.08 Signs

ARTICLE 17

PLANNED UNIT DEVELOPMENT (PUD) DISTRICTS

Section 17.01 Purposes and Intent

Section 17.02 Types of Planned Development Mandatory PUD Applicability

Section 17.03 General Standards for Planned Developments Voluntary

PUD Applicability

Section 17.04 Procedure for Approval Types of Planned Unit

Development Districts

Section 17.05 PD-1 Planned Residential Development General

Provisions Governing Planned Unit Developments

Section 17.06 PD-2 Planned Business Development Approval

Procedures for Planned Unit Development (PUD) Zoning

Classification

Section 17.07 PD-3 Planned Industrial Development Approval

Procedures for Specific Site Plan

Section 17.08 Submissions Required for Planned Developments Planned

Unit Development Obligations and Binding and Enforceable Conditions

Section 17.09 Modifications to PUD Permitted Uses

Section 17.10 Similarity of Uses and Conditional Uses

Section 17.11 Modifications to Approved Specific Site Plans

Section 17.12 PD-1 Planned Residential Development Districts

Section 17.13 PD-2 Planned Business Development Districts

Section 17.14 PD-3 Planned Industrial Development Districts

Section 17.15 PD-4 Planned Mixed-Use Development Districts

Section 17.16 PD-5 Planned Public and Private Buildings and

Grounds Districts

Appendix A-1 Planned Development Procedure for

Approval - Preliminary Plan

Appendix A-2 Planned Development Procedure for

Approval - Final Development Plan

ARTICLE 18 <u>GENERAL PROVISIONS</u>

Section 18.01 General Regulations

Section 18.02 Yard Required for Corner & and Through Lots

Section 18.03 Reserved

Section 18.04 Lots Adjoining Alleys

Section 18.05A Accessory Buildings

Section 18.05B Swimming Pools

Section 15.05C Temporary Storage Units

Section 18.06 Fences & and Walls

Section 18.07 Removal of Soil, Sand, or Other Material

Section 18.08 Essential Services

Section 18.09 External Effects

2024-O-6 Exhibit A

Section 18.10 Outdoor Storage and Waste Disposal

Section 18.11 Projections into Required Yards

Section 18.12 Exceptions to Height Limitations

Section 18.13 Temporary Uses

Section 18.14 Major Street Setbacks

Section 18.15 Off-Street Loading Regulations

Section 18.16 Off-Street Parking Regulations

Section 18.17 Screening

Section 18.18 Reserved

Section 18.19 Septic Tanks or Wells

Section 18.20 Permitted Signs

Section 18.20A Permitted Signs (General)

Section 18.20B Permitted Signs (Old Village District)

Section 18.21 Non-Conformities

Section 18.22 Trailers, Commercial Semitrailers & and Trucks

Section 18.23 Drive-In Service Establishments

Section 18.24 Barriers to Encroachment

Section 18.25 Reserved

Section 18.26 Recreational Vehicles

Section 18.30 Adult Entertainment Facilities

Section 18.31 Light & and Glare

Section 18.40 Personal Wireless Service Sites

ARTICLE 19 <u>ENFORCEMENT AND PENALTIES</u>

Section 19.01 Zoning Permits Required

Section 19.02 Contents of Application for Zoning Permit

Section 19.03 Submission for Thoroughfare Plan Review

Section 19.04 Approval of Zoning Permit/Zoning Compliance Certificate

Section 19.05 Expiration of Zoning Permit/Zoning Compliance Certificate

Section 19.06 Certificate of Health Officer

Section 19.07 Application for Zoning Permit/Zoning Compliance

Certificate

Section 19.08 Failure to Obtain Zoning Permit/Zoning Compliance

Certificate

Section 19.09 Record of Zoning Permits/Certificates of Zoning

Compliance

Section 19.10 Complaints Regarding Violations

Section 19.11 Other Action

Section 19.12 Penalties

Section 19.13 Affected Parties

Section 19.14 Violation, Nuisance Per Se: Abatement

Section 19.15 Schedule of Fees, Charges, and Expenses

Section 19.16 Waiver

ARTICLE 20 <u>ADMINISTRATION</u>

Section 20.01 Office of Zoning Inspector Community Development

Administrator Created

Section 20.02 Duties of **Zoning Inspector Community Development**

Administrator

Section 20.03 Planning Board Created

Section 20.04 Duties of Planning Board

Section 20.05 Proceedings of Planning Board

Section 20.06 Board of Zoning Appeals Created

Section 20.07 Duties of the Board of Zoning Appeals

Section 20.08 Proceedings of the Board of Zoning Appeals

Section 20.09 Hearings of the Board of Zoning Appeals

Section 20.10 Action by the Board of Zoning Appeals

Section 20.11 Duties of **Zoning Inspector Community Development**

Administrator, Board of Zoning Appeals, Legislative Authority & and

Courts on Matters of Appeal

Section 20.12 Procedures & and Requirements for Administrative

Reviews

Section 20.12.1 Administrative Review

Section 20.12.2 Stay of Proceedings

Section 20.13 Procedures & and Requirements for Variances

Section 20.13.1 Procedure for Consideration of Petitions for Variance

Section 20.13.2 Application and Standards for Variance

Section 20.13.3 Supplementary Conditions and Safeguards

Section 20.14 Procedures & and Requirements for Approval of

Conditional Use Permits

Section20.14.1 General

Section 20.14.2 Contents of Application for Conditional Use Permit

Section 20.14.3 General Standards Applicable to All Conditional

Uses

Section 20.14.4 Supplemental Safeguards and Conditions

Section 20.14.5 Expiration of Conditional Use Permit

Section 20.15 Determination of Similar Uses

Section 20.16 Determination of District Boundary Location

Section 20.17 Amendment Limitation

ARTICLE 21 AMENDMENT

Section 21.01 Right of Petition-Referral to Planning Board

Section 21.02 Contents of Application

2024-O-6 Exhibit A

Section 21.03 Reviews by Officials and Agencies

Section 21.04 Council Action on Recommendation

Section 21.05 Scheduling Public Hearing by Council

Section 21.06 Notice of Property Owners

Section 21.07 Amendment Limitation

Section 21.08 Annexation

Section 21.09 Fee

ARTICLE 22 <u>VALIDITY AND SEVERABILITY</u>

ARTICLE 23 INTERPRETATION AND CONFLICT

ARTICLE 24 <u>REPEAL OF CONFLICTING ORDINANCES</u>

ARTICLE 25 <u>EFFECTIVE DATE</u>

<u>ARTICLE 2</u> <u>INTENT AND PURPOSE</u>

The City's This Ordinance Zoning Code is based on a Comprehensive Plan, the purpose of which is to lessen the congestion on the public streets, to reduce undue hazards due to flooding, and to safeguard and promote the public health, safety, and general welfare. This Comprehensive abovementioned pPlan has been formulated with due consideration among other things, to the character of each district of the City municipality and its peculiar suitability for uses to the conservation of property values; to the general trend and character of building and population development; to the prevention of undue concentration of population; to the advancement of social and economic stability; and to the facilitation of adequate provision of public transportation, streets, highways, sewers, water mains, schools, recreation areas and other public facilities. It is the further purpose of this Ordinance to safeguard the public health, safety, and general welfare.

ARTICLE 3 CONSTRUCTION OF LANGUAGE AND DEFINITIONS

3.01 **CONSTRUCTION OF LANGUAGE**

For the purposes of this **Zoning Code** Ordinance, certain terms or words used herein shall be interpreted as follows:

The word "person" includes a firm, association, organization, partnership, trust, company, or corporation, as well as an individual; the present tense includes

the future tense, the singular number includes the plural, and the plural number includes the singular; the word "shall" is mandatory, and the word "may" is

permissive; the words "used" or "occupied" include the words "intended", "designed" or "arranged to be used or occupied"; the word "building includes the word "structure," and the word "dwelling" includes the word "residence"; the word "lot" includes the words "plot" or "parcel". In case of any difference of meaning or implication between the text of this Ordinance and any caption or illustration, the text shall control.

Terms not herein defined shall have the meaning customarily assigned to them.

3.02 <u>DEFINITIONS</u>

Accessory Use or Building: Is a A use or building on the same lot width and of a nature customarily incident or subordinate to those of the main use or building. For yard setback and maximum allowable lot coverage purposes only, a private in-ground swimming pool shall be considered an accessory building or structure.

All temporary and/or manufactured carports shall be considered detached accessory buildings, even if attached or anchored to a main structure.

Accessory Dwelling Unit: A second dwelling unit subordinate to the principal dwelling that shares ownership and utility connections with the principal unit on a single-family zoned lot.

<u>Adult Entertainment Facility</u>: A facility having <u>a Ssignificant portion of its function as adult entertainment which includes the following listed categories:</u>

Adult Book/Video Store: An establishment which deals in books, magazines or other periodicals, or video tapes of which at least ten percent, as measured by publicly accessible display area, are distinguished, or characterized by an emphasis on depictions or descriptions relating to specified sexual activities or specified anatomical areas, as defined below, or an establishment having an area devoted primarily to the sale or display of such materials.

Adult Entertainment Theater: A facility used to a substantial extent for presenting material distinguished or characterized by an emphasis on

matter depicting, describing, or relating to specified sexual activities, or specified anatomical areas as herein defined for the observation by patrons therein.

Adult Entertainment Business: Any establishment involved in the sale of services or products characterized by salacious conduct appealing to prurient interest for the observation or participation in by patrons, the exposure or presentation of specified anatomical areas or physical contact of live males or females, and characterized by including but not limited to photography, dancing, stripping, reading, massage, male or female impersonation, and similar functions which utilize activities as specified below.

Specified Sexual Activities:

- (a) Human genitals in a state of sexual stimulation or arousal;
- (b) Acts, real or simulated, of human masturbation, sexual intercourse, sodomy, cunnilingus, or fellatio; and
- (c) Fondling or other erotic touching of human genitals, pubic region, buttock, or female breasts.

Specified Anatomical Areas:

- (a) Less than completely and opaquely covered human genitals, pubic region, buttock, and female breasts below a point immediately above the top of the areola; and
- (b) Human male genitals in a discernable turgid state even if completely and opaquely covered.

Agriculture: See "Farm" definition below.

<u>Alley:</u> Any dedicated public way affording a secondary means of access to abutting property, and not intended for general traffic circulation.

<u>Alteration:</u> Is a<u>A</u>ny change, addition, or modification in construction or type of occupancy, any change in the structural members of a building, such as walls or partitions, columns, beams or girders, the consummated act of which may be referred to herein as "altered" or "reconstructed".

<u>Antenna (Dish):</u> A dish-shaped antenna to be used for receiving communication signals from satellites.

Apartment: A suite of rooms or a room in a multi-family building arranged and intended for a place of residence of a single family or a group of individuals

living together as a single house hold. keeping unit as herein defined.

Apartment Hotel: A building designed to contain both dwelling units and individual guest rooms or suites of rooms, which building may include accessory uses such as a cigar store, coffee shop, etc., when such uses are accessible only from the lobby.

Auto Service Station: Is a A place where gasoline, or any other automobile engine fuel (stored only in underground tanks), kerosene, or motor oil and lubricants or grease (for operation of motor vehicles) are retailed directly to the public on the premises, which includes ing the sale of minor accessories and the servicing of and general repair and engine rebuilding of automobiles, but does not includeing storage of inoperable vehicles.

Auto Repair Station: Is a A place where, along with the sale of engine fuels, the following services may be carried out: general repair, engine rebuilding, rebuilding, or reconditioning of motor vehicles; collision service, such as body frame, or fender straightening and repair; overall painting and under-coating of automobiles.

<u>Basement:</u> Theat portion of a building which is partly or wholly below grade but so located that the vertical distance from the average grade to the floor is greater than the vertical distance from the average grade to the ceiling. A basement shall not be counted as a story, <u>unless otherwise</u> <u>provided by this Code.</u> except as provided in paragraph 105.

Block: Is a A parcel of land bound on all sides by a street or streets.

Board of Zoning Appeals: Means & The Board of Zoning Appeals of Bellbrook, Ohio, which Board also serves as the Property Review Commission.

Boarding House/Rooming House: (Rooming House) a $\underline{\mathbf{A}}$ building, other than a hotel, where for compensation and by pre—arrangement for definite periods, meals or lodging and meals are provided for three (3) or more persons, but not exceeding ten (10) sleeping rooms. A $\underline{\mathbf{rR}}$ coming $\underline{\mathbf{hH}}$ ouse or a furnished-room house shall be deemed a $\underline{\mathbf{bB}}$ oarding $\underline{\mathbf{hH}}$ ouse for the purposes of this Ordinance.

<u>Brewpub: An e</u>Establishment primarily engaged in the retail sale of prepared food for consumption, which includes the brewing of beer as an accessory use. The brewing operation processes the ingredients to make beer and ale by mashing, cooking, and fermenting. The brewing operation does not include the production of any other alcoholic beverage.

- (a) A brewpub shall not produce more than **four thousand** (4,000) barrels of beer or ale per year.
- (b) No brewing equipment or storage is permitted on the exterior of the building.

<u>Building:</u> Is a Any structure, either temporary or permanent, having a roof supported by columns or walls, and intended for the shelter, or enclosure of persons, animals, chattels, or property of any kind.

<u>Building Height:</u> Is t<u>T</u>he vertical distance measured from the established grade to the highest point of the roof surface for flat roofs; to the deck line of mansard roofs; and to the average height between eaves and ridge for gable, hip, and gambrel roofs. Where a building is located on sloping terrain, the height may

be measured from the average ground level of the grade at the building wall.

<u>Building Line</u>: <u>Is a A_line</u> formed by the face of the building, and for the purposes of this Ordinance, the $\frac{\mathbf{b}\mathbf{B}}{\mathbf{b}\mathbf{B}}$ uilding $\frac{\mathbf{l}\mathbf{L}\mathbf{i}}{\mathbf{l}\mathbf{b}}$ is the same as a $\frac{\mathbf{f}\mathbf{F}}{\mathbf{r}}$ ront $\mathbf{s}\mathbf{S}$ etback $\frac{\mathbf{l}\mathbf{L}}{\mathbf{l}\mathbf{b}}$ ine.

Carport:

(a) <u>Temporary</u>: A building with a rigid roof comprised of metal, wood, or fiberglass, open on two (2) or more sides for the purpose of storing of vehicles and RVs. All temporary freestanding carports

and/or manufactured carports shall be considered detached accessory buildings, even if attached or anchored to a main building. No tarpaulins or colored plastic covers are permitted to cover a vehicle housed in a temporary carport.

(b) <u>Permanent</u>: A carport may be built as a permanent addition to a residence and formed by extension of a main building, comprised of the same building materials, and having the same characteristics as the main building.

<u>Carry-Out:</u> A place of business where beverages and prepared food are sold for consumption off the premises.

<u>Channel:</u> A natural or artificial depression of perceptible extent, with definite bed and banks to confine and conduct continuously or periodically flowing water.

<u>Clinic:</u> An establishment where human patients who are not lodged overnight are admitted for examination and treatment by a group of physicians, dentists, or similar profession<u>al</u>s.

<u>Club:</u> <u>Is aA</u>n organization of persons for special purposes or for the promulgation of sports, arts, sciences, literature, politics, or the like, but not operated for profit.

<u>Commercial Vehicles:</u> Any vehicle licensed by the State of Ohio as a commercial vehicle or used for a commercial purpose.

(a) <u>Commercial Semi-tTrailer:</u> Any commercial vehicle designed or used for carrying persons or property with another and separate motor vehicle so that in operation a part of its own weight or that of its

load, or both, rests upon and is carried by another vehicle, also defined as a vehicle having wheels only at the rear, the front resting on a tractor or towing vehicle.

(b) Commercial Tractor: Every motor vehicle having motive power designed or used for drawing other vehicles and not so constructed as to carry any load thereon, or designed or used for drawing other vehicles while carrying a portion of such other vehicles or the load thereon, or both.

- (c) <u>Commercial Trailer:</u> Any commercial vehicle designed or used for carrying persons or property wholly on its structure and for being drawn by a motor vehicle.
- (d) <u>Commercial Truck:</u> Any automotive vehicle designed to carry loads, and which exceeds ten thousand pounds (10,000 lbs.) in weight unloaded.

Conditional Use: A conditional use is a use permitted only after review of an application by the Board of Zoning Appeals, such review being necessary because the provisions of this Ordinance covering conditions, precedent, or subsequent, are not precise enough to all applications without interpretation, and such review is required by the Ordinance. A conditional use does not require "undue hardship" to be allowable. The conditional uses that are found in this Ordinance appear as "special approval" on recommendation by the Planning Board and review by the Board of Zoning Appeals. These land uses could not be logically allocated to one (1) zone or another, or the effects of such uses could not be foreseen as of a given time.

Conditional Use Permit: This is a permit issued by the Board of Zoning Appeals to allow certain specific developments that would not otherwise be allowed in that particular zoning district where the land is located. These permits are issued only after the applicant has followed the procedures as stated in this Ordinance. Development under a Conditional Use Permit differs from a zoning change in that it is much more specific. The applicant submits plans and if approved, he the applicant must follow those plans exactly or re-apply for a permit before deviating from that plan.

<u>Convalescent or Nursing Home:</u> An establishment which specializes in providing necessary services to those unable to care for themselves.

<u>Council:</u> <u>Means tThe Municipal City</u> Council of Bellbrook, Ohio.

<u>Deck:</u> An open platform projecting from a wall of a building, surrounding a pool or free-standing, which is supported by structural pillars or posts at grade or by

the principal building structure itself. Free-standing decks shall be considered accessory buildings for setback purposes only.

<u>Development:</u> Any man-made change to improved or unimproved real estate, including but not limited to buildings or other structures, mining, dredging, filling, grading, paving, excavating, or drilling operations.

<u>District:</u> <u>Is a A portion of the incorporated area of the municipality City</u> within which certain regulations and requirements or various combinations thereof apply under the provisions of this Ordinance.

<u>Drive-In:</u> Is a $\underline{\mathbf{A}}$ business establishment so developed that its retail or service character is dependent on providing a driveway approach or parking spaces for

motor vehicles to serve patrons while in the motor vehicle, or within a building or structure on the same premises and devoted to the same purpose as the drive-in service.

<u>Dwelling Unit:</u> Is a <u>A</u> building, or a portion thereof, designed for occupancy of one (1) family for residential purposes and having cooking facilities.

<u>Dwelling</u>, <u>One-Family</u>: <u>Is a A</u> building designed exclusively for and occupied exclusively by one (1) family.

<u>Dwelling</u>, <u>Two-Family</u>: <u>Is a</u> A building designed exclusively for occupancy by two (2) families living independently of each other.

<u>Dwelling, Multiple-Family:</u> <u>Is a A</u> building, or a portion thereof, designed exclusively for occupancy by three (3) or more families living independently of each other.

Enclosed Patio: An enclosed area immediately adjacent and connected to a structure which is not constructed nor intended as livable area. Such structures may be enclosed by screens or windows which shall compose at least fifty percent (50%) of any outside wall.

<u>Encroachment Lines:</u> Lines marking the limits of the floodway and the limit of encroachment of fill into the floodplain.

<u>Erected:</u> <u>The term "erected"</u> <u>Fi</u>ncludes built, constructed, altered, reconstructed, moved upon, or any physical operations on <u>the a premises</u>, which are required for the construction. Excavation, fill, drainage, and the like, shall be considered a part of erection.

<u>Essential Services:</u> <u>Is t</u>The erection, construction, alteration or maintenance by public utilities or municipal departments of underground, surface, or overhead

gas, electrical, telephone, telegraph, steam, fuel or water transmission or distribution systems, collection, communication, supply, or disposal

systems, including towers, poles, wires, mains, drains, sewers, pipes, conduits, cables, fire alarm and police call boxes, traffic signals, hydrants, and similar accessories in

connection therewith, but not including buildings which are necessary for the furnishing of adequate service by such utilities or municipal departments for the general health, safety, or welfare.

Excavation: Is a Any breaking of ground, except common household gardening and ground care.

<u>Family:</u> Is one or two persons or parents, with their direct lineal descendants and adopted children together with no more than two persons not so related, or a group of not more than three persons who need not be related, living together as a single housekeeping unit in a dwelling unit.

<u>Farm:</u> All the contiguous neighboring or associated land operated as a single unit on which bona fide farming is carried on directly by the owner-operator,

Manager, or tenant farmer, by his own labor or with the assistance of members of his household or hired employees; provided, however, that land to be considered a farm hereunder shall include a continuous parcel of five (5) acres or more in area; provided, further, farms may be considered as including establishments operated as bona fide greenhouses, nurseries, orchards, chicken

hatcheries, poultry farms, and apiaries; but establishments keeping furbearing animals, or operate riding or boarding stables, commercial dog kennels, noncommercial stone quarries or gravel or sand pits, shall not be considered farm hereunder unless combined with bona fide farm operations on the same continuous tract of land of not less than twenty (20) acres. No farms shall be operated as piggeries, or for the disposal of garbage, sewage, rubbish, offal or rendering plants, or for the slaughtering of animals except such animals as have been raised on the premises or have been maintained on the premises for at least a period of one (1) year immediately prior thereto and for the use and consumption by persons residing on the premises.

<u>Fence:</u> Any structure other than part of a building which encloses or partially encloses any premises.

<u>Filling:</u> Is <u>t</u>The depositing or dumping of any matter onto or into the ground, except common household gardening and ground care.

Flood Control: The construction of dikes, river embankments, channels, or dams for protection from floods.

<u>Flood Plain:</u> A channel and the areas adjoining a channel which may be flooded.

<u>Flood Profile:</u> A graph, chart, or longitudinal plot along a stream or river of maximum water surface elevations of a flood.

<u>Flood Proofing:</u> A combination of structural and nonstructural additions, changes or adjustments to properties and structures subject to flooding primarily for the reduction or elimination of flood damage to properties, water and sanitary facilities, and contents of buildings. Walls must be water-tight and structural components must be able to resist hydrostatic and hydrodynamic loads and effects of buoyancy.

<u>Floodway Fringe:</u> The part of the regulatory <u>Flood Plain</u> which is outside the floodway. One (1) boundary of the <u>Floodway</u> Fringe is the <u>eEncroachment lLine</u>, and the other boundary is the <u>Landward Boundary</u> of the <u>Regional Flood</u>.

<u>Floor Area:</u> For the purposes of computing the minimum allowable floor area in a residential dwelling unit, the sum of the horizontal areas of each story of the

building shall be measured from the interior faces of the exterior walls. The floor area measurement is exclusive of areas of basements, unfinished attics, attached garages, breezeways, and enclosed and unenclosed porches, except basement areas designed and used for dwelling or business purposes.

<u>Floor Area Gross:</u> The sum of the gross horizontal areas of all the several floors of a building or buildings, including interior balconies and mezzanines. All

horizontal measurements are to be made between the exterior faces of walls including the walls of roofed porches having more than one (1) wall. The gross floor area of a building shall include the floor area of accessory buildings, on the same lot, measured the same way.

Floor Area: (Ffor the purpose of computing parking): Theat area used for or intended to be used for the sale of merchandise or services, or for use to serve patrons, clients, or customers. Such floor area which is used or intended to be used principally for the storage or processing of merchandising, such as hallways, stairways, and elevator shafts, or for utilities or sanitary facilities, shall be excluded from this computation of

"Floor Area-" **for the purpose of computing parking.** Measurements of useable floor area shall be the sum of the horizontal areas of the several floors of the building, measured from the interior faces of the exterior walls.

Garage: Parking: A space or structure or series of structures for the temporary storage or parking of motor vehicles, not primarily of commercial vehicles or for

dead storage of vehicles, having no public shop or service in connection therewith, other than for the supplying of motor fuels and lubricants, air, water, and other operating commodities wholly within the buildings, to the patrons of the garage only and not readily visible from other properties nor advertised for sale on the exterior of the building.

Garage: Private: An accessory building or portion of a main building designed or used solely for the storage of motor-driven vehicles, boats, and similar vehicles owned or used by the occupants of the building to which it is accessory.

<u>Garage: Service:</u> Any premises used for the storage or care of motor-driven vehicles, or where any such vehicles are equipped for operation, repair or kept for remuneration, hire, or sale.

<u>Grade (Ground level):</u> The average of the finished ground level at the center of all walls of a building. In the case where walls are parallel to and within five

(5) feet of a sidewalk, the above ground level shall be measured at the sidewalk, unless otherwise defined herein.

<u>Hedge:</u> Any planting which encloses or partially encloses any premises.

<u>Home Occupation:</u> An activity, office, or occupation conducted in a dwelling unit where such use is clearly incidental and secondary to the residential use. <u>Please</u>

refer to Article #5, Sec. 5.03, Accessory Uses, (2), Home Occupations.

<u>Hospital or Sanitarium:</u> A public or semi-public facility that provides accommodations and continuous service for the sick and injured including obstetrical, medical, and surgical care.

<u>Hotel:</u> A building occupied as the more-or-less temporary abiding place of individuals who are lodged with or without meals in which there are ten (10) or

more sleeping rooms and no provision made for cooking in any individual room or apartment. A hotel may include a restaurant or cocktail lounge, public banquet halls, ballrooms, or meeting rooms.

Junk Yards/(Salvage Yards): Is an An open area where waste, used, or secondhand materials are bought and sold, exchanged, stored, baled, packaged, disassembled, or handled including, but not limited to, scrap iron and

On the metals, paper, rags, rubber tires, and bottles. A "Junk Yard" or "Salvage Yard" includes automobile wrecking yards and includes any area of more than two hundred (200) square feet for storage, keeping, or abandonment of junk, but does not include uses established entirely within enclosed buildings. Two (2) or more inoperative or unlicensed vehicles shall be construed to be a junk yYard.

Kennel:

- (a) any lot or premises used for the sale, boarding, or breeding of dogs, cats, or other animals for commercial purposes; or
- (b) of any lot or premises upon which are kept more than four (4) animals which are over the age of four (4) months, and which are either dogs or cats. Kennels are not included as permitted principal or accessory uses in any Rresidential Zzoning Ddistricts and therefore, so are not allowed in those Ddistricts;
 - (1) an excess number of dogs and/or cats on premises that exist as of the effective date of this end of the premises. No additional dogs and/or cats is achieved on the premises. No additional dogs and/or cats may be added to any such lot or premises until the number of dogs and cats is reduced to three (3) or less.

Loading Space: Is a An off-street space on the same lot with a building, or group of buildings, for the temporary parking of a commercial vehicle while loading and unloading merchandise or material.

Lot: Is a A parcel of land occupied, or to be occupied by a main building or a group of such buildings and accessory buildings, or utilized for the principal use and uses accessory thereto, together with such open spaces as are required under the provisions of this Ordinance. Every lot shall abut upon and have permanent access to a public street and have a minimum frontage of forty (40) feet thereon.

Lot, Area: Is the The total horizontal area within the lot lines of the lot.

Lot, Corner: A lot which has at least two (2) contiguous sides, each abutting upon a street for its full length.

<u>Lot, Interior:</u> Is any **Any** lot other than a corner lot.

<u>Lots, Through</u>: Is any <u>Any</u> interior lot having frontages on two (2), more or less, parallel streets as distinguished from a corner lot. In the case of a row of double frontage lots, all sides of said lots adjacent to streets shall be considered frontage, and front yards shall be provided as required.

<u>Lot, Coverage:</u> Is the <u>The</u> part or percent of the lot occupied by buildings including accessory buildings.

Lot, Line: The lines bounding a lot as defined herein.

<u>Front Lot Lines:</u> In the case of an interior lot, <u>a Front Lot Line</u> is that line separating said lot from the street. In the case of a corner lot, or double frontage lot, <u>a Front Lot Line</u> it is that line separating said lot from either street.

Rear Lot Lines: Is that The lot line opposite the front lot line. In the case of a lot pointed at the rear, the #Rear | Lot | Line shall be an imaginary line parallel to the #Front | Lot | Line, not less than ten (10) feet long lying farthest from the #Front | Lot | Line and wholly within the lot. In the case of a eCorner | Lot, the #Rear | Lot | Line is opposite the #Front | Lot | Line of least dimension.

<u>Side Lot Line:</u> Is any <u>Any</u> lot line other than the front lot line or rear lot line. A side lot line separating a lot from a street is a side street lot line. A side lot line separating a lot from another lot, or lots is an interior side lot line.

<u>Lot Width:</u> <u>Is the The</u> horizontal distance between the <u>sSide lLot lLines</u>, measured at the two <u>(2)</u> points where the building line or setback line intersects the side lot lines.

Lot of Record: Is a A parcel of land, the dimensions of which are shown on a document or map on file with the County Recorder or in common use by Municipal City or County Officials, and which actually exists exists as so shown, or any part of such parcel held in a record ownership separate from that of the remainder thereof.

<u>Main Building:</u> Is a <u>A</u> building in which is conducted the principal use of the lot upon which it is situated.

<u>Main Use:</u> Is the <u>The</u> principal use to which the premises are devoted and the principal purpose for which the premises exist.

Major Thoroughfare: Is an An arterial street which is intended to serve as a large volume traffic way for both the immediate municipal area and the region beyond, and may be designated a major thoroughfare, parkway, freeway, expressway, or equivalent term to identify those streets comprising the basic structure of the street plan. Any street with a width, existing or proposed, of eighty (80) feet shall be considered a major thoroughfare.

<u>Major Thoroughfare Plan:</u> The official plan, as adopted by the Planning Board, of the major highways and streets, on file in the Office of the County Recorder,

including all amendments and supplements subsequently adopted.

<u>Master Plan:</u> Is <u>t</u>The comprehensive plan approved by the Planning Board including graphic and written proposals indicating the general location for streets, parks, schools, public buildings, and all physical development of the <u>City</u> municipality and includes any unit or part of such plan, and any amendment to such plan or parts thereof.

<u>Mezzanine</u>: Is an <u>An</u> intermediate floor in any story occupying not to exceed two-thirds (2/3) of the floor area of such story.

Microbrewery, Microwinery, Microdistillery: Establishments engaged in on-site production of beer, wine, spirituous liquor, or other alcoholic beverages and sales of the same by the glass for on-premise consumption. These establishments are primarily used for the production of beer, wine, spiritous liquor, or other alcoholic beverages and may include retail and food service as an accessory use. The brewing operation processes the

ingredients to make beer and ale by mashing, cooking, and fermenting. The brewing operation does not include the production of any other alcoholic beverage. The distilling operation is a small boutique-style distillery which produces relatively small quantities usually done in small batches.

- (a) A microbrewery shall not produce more than **fifteen thousand** (15,000) barrels of beer or ale per year.
- (b) No production equipment or storage is permitted on the exterior of the building.

<u>Mobile Home:</u> Is a <u>A</u> detached single-family dwelling to be located on foundation supports, designed for long_term occupancy, and containing sleeping accommodations, a flush toilet, a tub or shower bath, and kitchen facilities, including major appliances and furniture, with plumbing and electrical connections provided for attachment to outside systems; and designed to be

transported after fabrication on its own wheels, or in flatbed or other trailers or detachable wheels.

<u>Motel:</u> Is a <u>A</u> series of attached, semi-detached, or detached rental units containing a bedroom, bathroom, and a closet space. Units shall provide for overnight lodging and are offered to the public for compensation and shall cater primarily to the public traveling by motor vehicle. It may include all facilities specified under the definition of "Hotel". in Item 58.

Non-Conforming Building: Is a <u>A</u> building, or portion thereof, lawfully existing at the effective date of this Ordinance, or amendments thereto, and that does not

conform to the use regulations of the district in which it is located.

Nursery:

- (a) Plant Material: Is a A space including accessory building or structure, for the growing or storage of live trees, shrubs, or plant materials not offered for retail sale on the premises, including products used for gardening or landscaping.
- (b) **<u>Retail:</u>** Is a <u>A</u> space including accessory building or structure, or combination thereof, for the storage of live trees, shrubs, or plants offered for retail sale on the premises, including products used for gardening or landscaping.

<u>Off-Street Parking Lot:</u> Is a $\underline{\mathbf{A}}$ facility providing vehicular parking spaces along with adequate drives and aisles for maneuvering, so as to provide access for entrance and exit for the parking of more than two (2) vehicles.

Open Space: Theat part of a zoning lot, including courts or yards, which is open and unobstructed from its lowest level to the sky, accessible to all tenants upon the zoning lot.

<u>Open Space (Common)</u>: Theat area either dedicated to the public or commonly owned and/or available to all the residents of a $\underline{\mathbf{p}}$ -Planned $\underline{\mathbf{U}}$ $\underline{\mathbf{u}}$ nit $\underline{\mathbf{d}}$ -Development $\underline{\mathbf{a}}$ -Area.

Parking Space: Pursuant to this Ordinance, a "Parking Space" shall Is hereby determined to be a minimum area of two hundred (200) square feet, said area shall be exclusive of drives, aisles or entrances giving access thereto and shall be fully accessible for the storage or parking of permitted vehicles.

Patio: An open-sided, ground level improvement of the surface of a lot without roof which covers the vegetation and ground with concrete, wood, stone, gravel, or any other non-living building material.

Patio Cover: A projection out from the building to cover an area used for outdoor living or entry way. No vehicle, car, or similar vehicle may be stored or parked under this projection.

<u>Personal Wireless Services:</u> Services as defined in Title 47, Section 332(c)(7), of the United States Code.

<u>Personal Wireless Service Antenna:</u> A device for transmitting or receiving electromagnetic signals which carry information essential to the provision of a

Personal Wireless Service.

<u>Personal Wireless Service Facility:</u> Facilities for the provision of Personal Wireless Services.

<u>Personal Wireless Service Site:</u> A place where Personal Wireless Service Facilities are maintained and operated.

<u>Personal Wireless Service Tower:</u> A structure for supporting a Personal Wireless Service Antenna.

<u>Planned Development:</u> Land under unified control, planned and developed as a whole according to comprehensive and detailed plans, including streets, utilities, lots or building sites, site plans and design principles for all buildings as intended to be located, constructed, used, and related to each other, and for

other uses and improvements on the land as related to buildings. Development may be a single operation or a definitely programmed series of development operations including all lands and buildings, with a program for provision, operations and maintenance of such areas, improvements, and facilities necessary for common use by the occupants of the development.

Planning Board: Means the The Planning Board of Bellbrook, Ohio.

Pool, Swimming: A structure constructed or placed below ground or above ground, which is capable of containing water in excess of eighteen inches (18") of depth. For yard placement and setback purposes only, in-ground and

above ground swimming pools are considered accessory buildings.

<u>Public Utility:</u> <u>Is any Any</u> person, firm, or corporation, municipal department, board, or commission duly authorized to furnish and furnishing under state or

municipal regulations to the public: gas, steam, electricity, sewage disposal, communication, telegraph, telephone, transportation, or water.

Recreational Vehicle (or RV): A vehicle which is used primarily for pleasure and/or recreational purposes. For purposes of determining the length, width, and/or height of a recreational vehicle, the dimensions as stated on an official Certificate of Title will be used. Types of recreational vehicles include **but are not limited to,** boats, motor homes, travel trailers, campers, jet skis, snowmobiles, off-road vehicles, unattached truck caps, and trailers of any kind.

Regional Flood: A flood having an average frequency of occurrence on the order of once in **one hundred** (100) years, although such a flood may occur in any year.

Regulatory Flood Plain: The channel and areas adjoining a channel which may be covered by the regional flood.

Restaurant: An establishment whose primary business is serving food and beverages to patrons for consumption inside the building.

Row House or Town House: A two (2) story row of three (3) or more attached one (1) family dwellings, each unit of which extends from the basement to the roof.

Story: Is that The part of a building, except a mezzanine as defined herein, included between the surface of one (1) floor and the surface of the next floor, or if there is no floor above, then the ceiling next above. If the floor level directly above a basement is more than six (6) feet above grade, such basement shall be considered a story.

Story (Half): Is an An uppermost story lying under a sloping roof having an area of at least two hundred (200) square feet, with a clear height of seven feet six

inches (7'6"). For the purposes of this Ordinance, the usable floor area is only that area having at least four feet (4') clear height between floor and ceiling.

<u>Street:</u> <u>Is a A</u> public thoroughfare which affords the principal means of access to abutting property.

<u>Structure:</u> Is a Anything constructed or erected, the use of which requires location on or in the ground or attachment to something having location on or in the ground.

<u>Taproom/Wine Bar</u>: An establishment that serves malt beverages and/or wine with a minimum of <u>fifty percent (50%)</u> of the gross floor area devoted to the retail sales of malt beverages, wine, and related products. Taprooms/Wine Bars do not include the service or sale of distilled spirits.

<u>Temporary Use or Building:</u> <u>Is a A</u> use or building permitted by the Board of Zoning Appeals to exist during periods of construction of the main building or use, or for special events.

Temporary Storage Unit: A structure manufactured primarily for temporary or moveable storage, including but not limited to PODS®.

<u>Tent:</u> Any structure used for living or sleeping purposes, or for sheltering a public gathering constructed wholly or in part from canvas, tarpaulin, or other similar materials. <u>Tents include</u>, <u>but may not be limited to</u>, <u>and shall include</u>: <u>movable</u> shelter provided for circuses, carnivals, side shows, revival meetings, camp meetings, and <u>all other</u> similar meetings or exhibitions. <u>in temporary structures</u>.

<u>Trailer Court, (Mobile Home Park):</u> Any plot of ground upon which two (2) or more trailer coaches or mobile homes occupied for dwelling or sleeping purposes may be located.

<u>Transient Occupancy:</u> The right to use, occupy or possess, the use, occupancy, or possession of the following: dwelling unit; dwelling, one (1) family; dwelling, two (2) family; or a dwelling, multiple family; or a portion of any of the aforementioned, for a period of <u>twenty-five</u> (25) consecutive calendar days or less.

<u>Transient Rental:</u> The renting, letting, subletting, leasing, or subleasing of a:—dwelling unit; a dwelling, one family; <u>a</u> dwelling, two family; or a dwelling,

multiple family; or a portion of any thereof for "transient occupancy".

<u>Use:</u> Is the <u>The</u> purpose for which land or a building is arranged, designed, or intended, or for which land or a building is or may be occupied.

<u>Variance</u>: <u>Is a A modification of the literal provisions of the Zoning <u>Code</u> Ordinance granted when strict enforcement of the Zoning <u>Code</u> Ordinance would cause practical difficulty owing to circumstances unique to the individual property on which the variance is granted.</u>

<u>Wall:</u> Any solid structure other than part of a building which encloses or partially encloses any premises.

<u>Watercourse:</u> A natural or artificially constructed channel in which a flow of water occurs either continuously or intermittently in a definite direction.

<u>Yards:</u> The open spaces on the same lot with a main building, unoccupied and unobstructed from the ground upward except as otherwise provided in this Ordinance, as defined herein:

- (a) Front Yard: Is an An open space extending the full width of the lot, the depth of which is the minimum horizontal distance between the front lot line and the nearest point of the main building.
- (b) Rear Yard: Is an An open space extending the full width of the lot, the depth of which is the minimum horizontal distance between the rear lot line and the nearest point of the main building.
- (c) <u>Side Yard:</u> <u>Is an An</u> open space between a main building and the side lot line, extending from the front yard to the rear yard, the

width of which is the horizontal distance from the nearest point of the side lot line to the nearest point of the main building.

ARTICLE 4 ZONING DISTRICTS AND MAPS

4.01 **DISTRICTS**

In order to carry out the intent and purpose of this **Zoning Code** Ordinance, the City of Bellbrook is hereby divided into the following districts:

A-1 Agricultural District

R-1AA One Family Residential District

R-1A One Family Residential District

R-1B One Family Residential District

R-2 Two Family Residential District

R-3 Multi-Family Residential District

O-1 Office Building District

B-1 Highway Business District

B-2 Convenience Shopping District

B-3 Neighborhood Business District

B-4 Central Business District

I-1 Industrial District

F-1 Floodway District (Overlying District)

F-2 Floodway Fringe District (Overlying District)

WO Wellhead Operation District

WP Well Field Protection Overlay District

Old Village District (Overlying District)

PD-1 Planned Residential Development Districts

PD-2 Planned Business Development Districts

PD-3 Planned Industrial Districts

PD-4 Planned Mixed-Use Development Districts

PD-5 Planned Public and Private Buildings and Grounds Districts

4.02 **DISTRICT BOUNDARIES**

The <u>B</u>boundaries of the zoning districts listed above in Section 4.01 are shown on the <u>Official</u> Zoning Map of Bellbrook, Ohio. This map together with all explanatory data thereon, including all changes thereof as hereinafter provided, shall be <u>are</u> incorporated <u>into</u>-and made a part of this Ordinance.

The <u>O</u>efficial <u>Z</u>zoning <u>M</u>map shall be identified by the signature of the <u>President of City's Mayor Council</u>, attested <u>to</u> by the Clerk of Council, and <u>shall</u> bearing the seal of the <u>Municipality City</u> under the following words: "This seal is to certify that this is the Official Zoning Map referred to in Section 4.02 of the Zoning <u>Code</u> Ordinance of the City of Bellbrook, Ohio

(including date of adoption). If, in accordance with the provisions of this **Zoning Code** Ordinance, changes are made in district boundaries or other matter portrayed on the Official Zoning Map, the Official Zoning Map shall be updated by the Community Development Administrator ("CDA") or the CDA's designee within five (5) business normal working days after the amendment has been approved by the Council and shall include together with an entry on the Official Zoning Map as follows: "On (date), by official action of the Council, the following change(s) were made: (brief description with reference number to Council proceedings)."

The original and one copy of the The Official Zoning Map is to shall be maintained and kept current, up to date—with one (1) copy on public display in the Municipal City Building, one (1) copy posted on the City's website, and the original in the Clerk's office, accessible to the public. and The Official Zoning Map is shall be the final authority as to the current zoning district designations status of lands, building and other structures in the municipality City.

4.03 <u>UNCERTAINTY OF BOUNDARY LOCATION</u>

Where uncertainty exists with respect to the boundaries of the various districts as shown on the <u>Official</u> Zoning Map, the following rules shall apply:

- (1) Boundaries indicated as approximately following the center lines of streets, highways, or alleys shall be construed to follow such center lines;
- (2) Boundaries indicated as approximately following platted lot lines shall be construed as following such lot lines;
- (3) Boundaries indicated as approximately following municipal limits shall be construed as following municipal limits;
- (4) Boundaries indicated as following railroad lines shall be construed to be midway between the main tracks;
- (5) Boundaries indicated as following the shorelines shall be construed to follow such shorelines, and in the event of change in the shoreline shall be construed as moving with the actual shoreline; boundaries indicated as approximately following the center line of streams, rivers, canals, lakes, or other bodies of water shall be construed to follow such center lines;

- (6) Boundaries indicated as parallel to the extension of features indicated in subsections <u>one</u> (1) through <u>five</u> (5) above shall be so construed. Distances not specifically indicated on the Official Zoning Map shall be determined by the scale of the map; <u>and</u>
- (7) Where physical or cultural features existing on the ground are at variance with those shown on the Official Zoning Map, or in other circumstances not

covered by subsections <u>one</u> (1) through 5 <u>six</u> (6) above, the <u>City Council</u> Board of Zoning Appeals shall interpret the district boundaries.

4.04 **ZONING UPON ANNEXATION**

Whenever any area is annexed to the Municipality City of Bellbrook, one of the following conditions shall apply:

- (1) Land that is zoned <u>before</u> annexation shall be classified as being in whichever <u>zoning</u> district <u>of the City</u> of the Ordinance that most closely conforms with the zoning that existed prior to annexation; such classification to be recommended <u>by the Planning Board to Council after public hearing</u>, for an interim period, by the Planning Board to the Council and the <u>Board Council</u> shall approve the same after a public hearing.
- (2) Land not zoned prior to annexation shall be classified in the same manner into whichever **zoning** district of **the City** this Ordinance that most closely conforms with the existing use of the annexed area or in accordance with the **City's Comprehensive** master **pP**lan in the case of vacant land.
- (3) In all cases, within three (3) months after the effective date of annexation, the Planning Board shall recommend the appropriate permanent zoning districts

for such the annexed area to the Council after public hearing, and the Official Zoning Mmap shall be amended upon approval by Council pursuant according to the prescribed procedure set forth in Article 21.

ARTICLE 5 R-1AA, R-1A, R-1B ONE FAMILY RESIDENTIAL DISTRICTS

5.01 **PURPOSE AND INTENT**

These districts are the most restrictive of the residential districts. The purpose and intent <u>are</u> to provide for an environment of predominantly low-density, single_unit dwellings; to allow plus certain other facilities which serve the residents living in the district; and to regulate transient occupancy within the <u>eC</u>ity to protect and preserve the permanency of residential neighborhoods, protect property values, and ensure the comfort and safety of the residents.

5.02 **PERMITTED PRINCIPAL USES**

- (1) One-family dwellings, not including trailers or tent dwellings.
- (2) Churches and similar places of worship₃: **p**rovided such use is adjacent to a school or commercial area and/or access is by means of roads designated on primary or secondary thoroughfares by the <u>City's</u> Major Thoroughfare Plan.
- (3) Parks and playgrounds.
- (4) Public and private elementary, junior, and senior high schools.
- (5) Parish houses and convents in conjunction with churches or schools.
- (6) Nursery schools operated and housed in a church building or other established house of worship.

(7) Accessory Dwelling Units, provided such use shall meet the following:

- (a) An accessory dwelling unit may be located within a principal single-family detached dwelling or a detached accessory structure on the same lot as a principal dwelling.
- (b) The accessory dwelling unit shall share all public utilities with the principal dwelling unit. Accessory dwelling units shall not be metered separately.
- (c) A minimum of one (1) off-street parking space shall be provided on the lot for the accessory dwelling unit in addition to

the off-street parking spaces required for the principal dwelling unit.

- (d) The accessory dwelling unit shall be limited in size to a maximum of fifty percent (50%) of the total living area of the principal dwelling or eight hundred (800) square feet, whichever is less, in the R-1AA Zoning District. The accessory dwelling unit shall be limited in size to a maximum of fifty percent (50%) of the total living area of the principal dwelling or six hundred (600) square feet in the R-1A and R-1B Zoning Districts.
- (e) The accessory dwelling unit shall contain a living area, a minimum of one (1) bathroom, and a cooking facility.
- (f) No new access points or driveways shall be created or installed for access to the accessory dwelling unit.

5.03 <u>ACCESSORY USES</u>

- (1) Private garage for storage of vehicles of residents and employees.
- (2) A home occupation is an occupation conducted in a dwelling unit, provided that:
 - (a) A home occupation shall be conducted entirely within the principal residential structure with no more than <u>twenty-five</u> <u>percent</u> (25%) of the total floor area to be used in connection with the home occupation. No home occupation shall be conducted in any accessory structure, nor shall the storage of equipment related to the home occupation be contained within any accessory structure;
 - (b) There shall be no change in the external appearance of the building or premises resulting from the home occupation nor shall there be any storage of products, equipment, or materials on the lot. No more
 - than one (1) sign, not to exceed one (1) square foot in total sign area, is permitted on the building, which shall be non-illuminated and mounted flat against the wall of the principal building;
 - (c) Only members of the household residing on the premises where the home occupation occurs may be employed full or part-time in the home occupation. In no case shall a home occupation be permitted to operate earlier than 8 <u>a.m.</u> AM or later than 6 <u>p.m.</u> PM.

An exception to this requirement is for direct sale of consumer products where parties for the purpose of taking orders or selling merchandise are held;

- (d) Traffic generated by the home occupation shall not create safety hazards nor be substantially greater in vehicular size nor exceed on any continual basis the average number of vehicle trips that would normally be expected. Vehicles having a cargo area greater than twenty-two (22) feet in length or having dual rear axles are prohibited;
- (e) No area on the residential property may be created to park any vehicles or equipment related to the home occupation. Any home occupation which requires the creation of parking in addition to the parking areas already provided for the single-family residence shall not be operated within a residential zoning district. The use of a home occupation shall not reduce or render unusable areas provided for

required off-street parking for the dwelling unit;

- (f) A home occupation shall not produce heat, cold or dampness, create noise, vibration, fire, explosion, light, glare, dust, smoke, fumes, or odor, electric or electronic disturbances detectable to normal sensory perception by a person located off the premises or beyond the walls of the dwelling; and
- (g) A home occupation shall be conducted in accordance with all other eity regulations of the City and the county as well as nd all state and federal laws and licensing requirements.
- (3) Pools: Swimming pools for use by residents and guests only.: See Accessory Buildings (**Zoning Code** Section 18.05**B**) **and**; and Fences **and** Walls and Hedges (**Zoning Code** Section 18.06).
- (4) Signs Permitted as Accessory Uses. See Permitted Signs (**Zoning Code** Section 18.20).

5.04 <u>CONDITIONAL USES</u>

The following uses are allowed in any R-1 <u>R</u>residential <u>D</u>district provided a conditional use permit is granted by the Board of Zoning Appeals as provided in <u>Zoning Code</u> Section 20.<u>1408</u> of this Ordinance, and further provided, that all buildings allowed by such conditional use permit shall be

set back from all lot lines a minimum of three (3) feet for each one (1) foot of building height.

- (1) Public recreation and community center buildings and recreation areas or buildings operated by membership clubs for the benefit of their members and not for gain, provided that any principal building, accessory building, or outdoor
- swimming pool shall be located not nearer than <u>two hundred</u> (200) feet from any adjoining land zoned for a residential use.
- (2) Public owned or leased buildings, public utility buildings, telephone exchanges, transformer stations and sub-stations, and garages and maintenance buildings.
- (3) Institutions of higher learning and libraries.
- (4) Cemeteries, when extension of existing cemeteries.
- (5) Golf courses.
- (6) Public and private school facilities (excluding buildings used for instructional purposes, which are permitted principal uses in R-1AA, R-1A, R-1B, R-2, and R-3

 Delistricts).

5.05 YARD REQUIREMENTS

See **Zoning Code** Section 7.1009, Schedule of Yard and Lot Requirements.

5.06 <u>BUILDING HEIGHT REGULATIONS</u>

In any R-1 residential district, nNo building shall be erected in excess of two and one half (2 $\frac{1}{2}$) stories or thirty-five (35) feet in height <u>in any R-1</u> Zoning District.

5.07 <u>ACCESSORY PARKING</u>

Two (2) car spaces for each dwelling unit is required. Parking for other uses; see See Zoning Code Section 18.16 for additional uses and parking regulations.

All driveways and parking spaces located in the front, side, and rear yards shall be required to have a paved surface which shall be defined as a hard,

smooth surface made principally of asphalt, concrete, Portland cement, brick, or pavers that will bear travel. Existing unpaved driveways or parking spaces located in the front or side yards must be paved if additions or extensions are proposed to be made to any part of the unpaved driveway or parking space in the front or side yards. The parking of vehicles on the lawn or in other areas not designated for parking, shall be is prohibited.

No more than two (2) recreational vehicles may be parked upon any lot within these residential zoning districts at any time.

5.08 **SIGNS**

See **Zoning Code** Section 18.20 for sizes and location of permitted signs.

5.09 TRANSIENT RENTAL REGULATIONS

No dwelling unit or dwelling, one family, or a portion of any thereof in the R-1AA, R-1A, or R-1B **D**districts, shall be used as a transient rental.

ARTICLE 6 R-2 TWO-FAMILY RESIDENTIAL DISTRICTS

6.01 **PURPOSE AND INTENT**

The purpose and intent of this district recognizes the existence of older residential areas of the <u>City</u> municipality where larger houses have been or can be converted from single-family to two-family residences to extend the economic life of these structures and allow the owners to justify the expenditures for repairs and modernization. This district also allows the construction of new two-family residences where slightly greater densities are permitted. This Article also regulates transient occupancy within this district and protects and preserves the permanency of residential neighborhoods, protects property values, and ensures the comfort and safety of the residents.

6.02 **PERMITTED PRINCIPAL USES**

- (1) Two-family dwellings.
- (2) Those uses permitted in R-1 Districts.

6.03 ACCESSORY USES

Those accessory buildings and accessory uses customarily incidental to the **p**Permitted **Pp**rincipal **Uu**ses in this district.

6.04 **CONDITIONAL USES**

- (1) Those conditional uses permitted in R-1 **Zoning** Districts.
- (2) Mobile Home Courts.
- (3) Nursery schools, provided that there is compliance with <u>all sS</u>tate <u>laws</u>, <u>regulations</u>, <u>and licensure</u> requirements. <u>regarding space for play area per child</u>.

6.05 **YARD REQUIREMENTS**

See **Zoning Code** Section 7.**10**, 09 Schedule of Yard and Lot Requirements.

6.06 **BUILDING HEIGHT REGULATION**

In any R-2 District, nNo building shall be erected in excess of <u>two and one-half</u> (2 ½) stories or <u>thirty-five</u> (35) feet in height <u>in any R-2 Zoning District</u>.

6.07 <u>ACCESSORY PARKING</u>

- (1) There shall be provided $\underline{\mathbf{fF}}$ our (4) off-street parking spaces for each two-family dwelling $\underline{\mathbf{is}}$ required.
- (2) There shall be provided $\underline{\mathbf{t}}$ Two (2) parking spaces for each one-family dwelling $\underline{\mathbf{is}}$ required.
- (3) All other uses—See **Zoning Code** Section 18.16 **for additional uses** and parking regulations.

6.08 **SIGNS**

See **Zoning Code** Section 18.20 for size and location of permitted signs.

6.09 TRANSIENT RENTAL REGULATIONS

No dwelling unit or dwelling, two_family or a portion of any thereof in the R-2 **Zoning D**districts, shall be used as a transient rental.

ARTICLE 7 <u>R-3 MULTI-FAMILY RESIDENTIAL DISTRICT</u>

7.01 <u>PURPOSE AND INTENT</u>

The purpose and intent of this district is to allow construction of apartment buildings and condominiums and to regulate transient occupancy within this district to protect and preserve the permanency of residential neighborhoods, protects property values, and ensure the comfort and safety of the residents.

7.02 <u>PERMITTED PRINCIPAL USES</u>

- (1) Those uses permitted in R-1 and R-2 **Zoning** Districts.
- (2) Two family dwellings.
- (3) Apartment **h**ouse**s** and **Aa**partment H**h**otels.
- (4) Multiple dwellings.
- (5) Churches and similar places of worship.
- (6) Parish houses and convents in conjunction with churches or schools.
 - (7) Public community center buildings, parks, playgrounds, and golf courses.
 - (8) Public and parochial schools.
 - (9) Public libraries.
 - (10) Row houses.
 - (11) Town houses.

7.03 <u>ACCESSORY USES:</u>

- (1) Garages shall be permitted for storage purposes only with no repair facilities.
- (2) Those accessory buildings and accessory uses customarily incidental

to the <u>p</u>Permitted <u>p</u>Principal <u>u</u>Uses in this district.

7.04 <u>CONDITIONAL USES:</u>

Districts.

- (1) Those conditional uses permitted in R-1 and R-2 **Zoning**
- (2) Convalescent or Nnursing Hhomes.
- (3) Dormitories and Ggroup Hhousing.
- (4) Fraternities, <u>Ssororities</u>, <u>Cclubs</u>, <u>Llodges</u>, <u>Ssocial</u> or <u>Rrecreational buildings or properties not for profit.</u>
- (5) Hospitals, Cclinics, Ssanitariums for Hhuman Ccare.
- (6) Mortuaries.
- (7) Motels.
- (8) Offices of architects, engineers, and artists.
- (9) Offices of surgeons, physicians, dentists, and other similar professional persons concerned with the community health and medical treatment of persons.
- (10) Offices in which the personnel will be employed for work in executive, administrative, legal, writing, clerical, stenographic, accounting, insurance, or similar enterprises.

7.05 <u>YARD REQUIREMENTS</u>

See **Zoning Code** Section 7.**10**, 09, Schedule of Yard and Lot Requirements.

7.06 <u>BUILDING HEIGHT REGULATIONS:</u>

In any R-3 District, nNo building shall be erected in excess of two and one-half $(2 \frac{1}{2} \frac{1}{2})$ stories or thirty-five (35) feet in height in any R-3 Zoning District.

7.07 <u>ACCESSORY PARKING</u>

- (1) In any R-3 District, there shall be provided <u>t</u>Two (2) off_street parking spaces per dwelling unit <u>are required in any R-3 Zoning District</u>.
- (2) There shall be provided oone (1) off_street parking space for each two (2) roomers is required in any R-3 Zoning District.
- (3) For parking space required for other than residential uses, sSee Zoning Code Section 18.16 for additional uses and parking regulations.

7.08 **SIGNS**

See **Zoning Code** Section 18.20 for size and location of permitted signs.

7.09 TRANSIENT RENTAL REGULATIONS

No dwelling unit or dwelling, multi-family, or a portion of any thereof in the R-3 **Zoning D**elistrict, shall be used as a transient rental.

7.10 SCHEDULE OF YARD AND LOT REQUIREMENTS

(Measurements stated in feet unless otherwise noted)

District	Min. Lot Area (SF)	Min. Lot Width	Min. Corner Lot Width	Min. Front Yard*	Min. Rear Yard	Min. Side Yard	Total Side Yards	Types of Bldgs.	Min. Floor Area per Unit (SF)
R <u>-</u> 1AA	5 acres	200	200	100	100	20	50	All	1500
R <u>-</u> 1A	20,000	100	125	40	50	8	20	1/1 ½	1200
						10	25	ST	1200
								2/2 ½	
								ST	
R <u>-</u> 1B	16,000	100	125	30	40	8	20	1/1 ½	1200
						10	25	ST	1200
								2/2 ½	
								ST	
R <u>-</u> 2	14,000	100	125	30	40	8	20	1/1 ½	850
						10	25	ST	850
								2/2 1/2	
								ST	

2024-O-6 Exhibit A

R <u>-</u> 3	10,000**	100	125	30	30	8 (50)	20	1	***
				(50)	(50)	10	(100)	STORY	****
				, ,		(50)	25	2	
							(100)	STORY	

^{*} Figures shown in parenthesis are the setbacks required when an R-3 Zoning District abuts a single-family residential district and applies to all buildings (including accessory buildings).

**** One hundred fifty (150) square feet for each additional bedroom.

NOTE: Whenever the frontage of one (1) side of a block is more than forty percent (40%) developed at the time of the enactment of this Ordinance, the required setback for new construction or alteration shall be the average of the established setbacks. However, in no event shall, provided, this regulation shall not be so interpreted so as to require a front yard or more than fifty (50) feet for residential use.

^{**} Minimum lot area for a two-family dwelling in this district. For each additional dwelling unit, there shall be an additional **two thousand** (2,000) square feet of lot area. At no time shall the lot areas be less than **five thousand** (5,000) square feet per unit.

^{***} Seven hundred fifty (750) square feet for two (2) bedrooms; five hundred seventy (570) square feet for one (1) bedroom.

ARTICLE 8 <u>O-1 OFFICE BUILDING DISTRICTS</u>

8.01 **PURPOSE AND INTENT**

The purpose <u>and intent</u> of this district is to provide for the maintenance and alteration of existing buildings and for new construction of business office and professional buildings.

8.02 **PERMITTED PRINCIPAL USES**

- (1) Offices of surgeons, physicians, dentists, and other similar professional persons concerned with the community health and medical treatment of persons.
- (2) Facilities for human care such as hospitals, medical clinics, sanitariums, rest, and convalescent homes.
- (3) Offices of architects, engineers, artists, and others employed in the graphic arts.
- (4) Offices in which the personnel will be employed for work in executive, administrative, legal, writing, clerical, stenographic, accounting, insurance, or similar enterprises.
- (5) Public and <u>sSemi-pPublic</u> <u>bBuildings</u>, not including storage yards, for materials or equipment.
- (6) Dental Llaboratories.
- (7) Art or <u>a</u>Antique <u>s</u>Shops, <u>p</u>Photographic <u>s</u>Studios, <u>i</u>Interior <u>d</u>Decorating <u>s</u>Studios.

8.03 <u>ACCESSORY USES</u>

Those uses and buildings customarily incidental to the **permitted Pp**rincipal uses **permitted** in this district.

8.04 <u>CONDITIONAL USES</u>

Research <u>Ll</u>aboratories.

8.05 <u>YARD REQUIREMENTS</u>

In the <u>O-1</u> Office Building Districts, the minimum yard areas of the B-3 Neighborhood Business District shall apply.

8.06 **BUILDING HEIGHT REGULATIONS**

The building height in this district shall not exceed two and one-half (2 ½ 1/2) stories or thirty-five (35) feet except that in the an O-1 Zoning District that is contiguous to the a B-4 Central Business District; the may have a building height shall not exceeding three (3) stories or forty (40) feet.

8.07 <u>ACCESSORY PARKING</u>

Accessory parking sSpaces shall be provided in accordance with the provisions of as required in Zoning Code Section 18.16 of thisOrdinance.

8.08 <u>OFF-STREET LOADING</u>

<u>Off-street loading sSpaces shall be provided as required in in accordance with the provisions of Zoning Code</u> Section 18.15.

8.09 **SIGNS**

See **Zoning Code** Section 18.20 for size and location of permitted signs.

8.10 LANDSCAPING

All yard areas not used for buildings, parking₂ or ways for pedestrians or vehicles shall be planted with grass and/or shrubbery so as to provide attractive green areas.

8.11 <u>SCREENING</u>

See **Zoning Code** Section 18.**17** for screening regulations for uses adjoining residential districts.

8.12 **LOT COVERAGE**

Principal buildings shall be limited to thirty percent (30%) of the lot area and accessory buildings, parking areas, and drives shall also be limited to thirty percent (30%) of the lot area.

2024-O-6 Exhibit A

ARTICLE 9 <u>B-1 HIGHWAY BUSINESS DISTRICTS</u>

9.01 **PURPOSE AND INTENT**

This <u>purpose and intent of this</u> district is <u>designed</u> to provide for highway-oriented uses and less intensive business types to serve the market of the urban area rather than the neighborhood. <u>and is These districts are</u> typically mapped along major traffic arteries and/or adjacent to the Central Business District.

9.02 **PERMITTED PRINCIPAL USES**

- (1) Those uses permitted in all residential districts.
- (2) Automobile service stations including light repairs and accessory sales and installation.
- (3) Drive-In Restaurants.
- (4) Motels and Hotels.
- (5) Restaurants.

9.03 <u>ACCESSORY USES</u>

Uses customarily accessory to the above<u>-listed permitted principal</u> uses.

9.04 <u>CONDITIONAL USES</u>

building.

- (1) Auto wash facilities which are at least partially enclosed in a
- (2) Building services and supplies, including lumber yards.
- (3) Mortuary establishments (<u>which may include</u> a caretaker's residence may be provided within the main building of <u>the</u> mortuary establishment).

9.05 <u>YARD REQUIREMENTS</u>

In a B-1 Highway Business District, the following yard area shall be provided:

- (1) Front yards: A <u>twenty-five (25)</u> foot front yard shall be required excepting where the frontage on one side of the block is divided between B-1 Highway Business District and a residential district.

 In that event, the front yard of the residential district shall apply to the area in the B-1 Highway Business District.
- (2) Side Yards: When a B-1 Highway Business District abuts a residential district, a side yard of not less than ten (10) feet in width is required. In all other instances, a side yard is not required in a B-1 Highway Business District yards shall not be required; except a yard of not less than 10 feet in width shall be provided where a side lot line of the B-1 Highway Business District abuts a residential district.
- (3) Rear Yards: When a B-1 Highway Business District abuts a residential district, a rear yard of thirty (30) feet is required. In all other instances, a rear yard is not required in a B-1 Highway Business District. Rear yards shall not be required except where a rear lot line of the B-1 Highway Business District abuts a residential district; a rear yard of 30 feet shall be provided.

9.06 **BUILDING HEIGHT REGULATIONS**

No building in the B-1 Highway Business District shall exceed <u>two</u> and one-half (2 ½) stories or <u>thirty-five</u> (35) feet in height.

9.07 <u>ACCESSORY PARKING</u>

<u>Parking Sspaces</u> shall be provided in accordance with the provisions of **Zoning Code** Section 18.16.

9.08 <u>OFF-STREET LOADING</u>

<u>Off-Street loading</u> <u>sS</u>paces shall be provided in accordance with the provisions of **Zoning Code** Section 18.15.

9.09 **SIGNS**

See **Zoning Code** Section 18.20 for size and location of permitted signs.

9.10 <u>SCREENING</u>

See **Zoning Code** Section 18.17 for screening regulations for uses adjoining residential districts.

ARTICLE 10

B-2 CONVENIENCE SHOPPING DISTRICT:

10.01 **PURPOSE AND INTENT**

The <u>purpose and intent of this</u> district is <u>designed</u> to provide for a limited range of convenience goals to supply the emergency needs of those <u>citizens</u> living in the immediate vicinity. The <u>sSize</u> <u>of this</u> <u>district</u> will be approximately one (1) acre and <u>shall include contain</u> one (1) to four (4) uses to serve <u>about</u> <u>approximately two thousand</u> (2,000) people.

10.02 <u>PERMITTED PRINCIPAL USES</u>

- (1) Those uses permitted in all residential districts.
- (2) Barber and beauty shops.
- (3) Drug Stores.
- (4) Food Stores.
- (5) Laundry and dry-cleaning pick-up service and coin operated washing and dry cleaning facilities.

10.03 <u>ACCESSORY USES</u>

Uses customarily accessory to the above<u>-listed permitted principal</u> uses.

10.04 <u>YARD REQUIREMENTS</u>

The yard requirements for this zoning district are the same as the Those yard requirements of the adjacent and surrounding residential districts.

10.05 **BUILDING HEIGHT REGULATIONS**

Maximum permitted height for buildings in this district shall be 2 ½ stories or 35 feet. No building in the B-2 Convenience Shopping

<u>District shall exceed two and one-half (2 ½) stories or thirty-five</u> (35) feet in height.

10.06 <u>ACCESSORY PARKING</u>

As required in Section 18.16 of this Ordinance. Parking Sspaces shall be provided in accordance with the provisions of Zoning Code Section 18.16.

10.07 **OFF-STREET LOADING**

Off-Street loading sSpaces shall be provided as required in Zoning Code Section 18.15.

10.08 **SIGNS**

See **Zoning Code** S**S**ection 18.20 for size and location of permitted signs.

10.09 <u>SCREENING</u>

See **Zoning Code** Section 18.2017 for screening regulations for uses adjoining residential district.

ARTICLE 11

B-3 NEIGHBORHOOD BUSINESS DISTRICT:

11.01 <u>PURPOSE AND INTENT</u>

The purpose <u>and intent</u> of this district is to provide an integrated collection of structures and uses designed to supply a majority of the daily needs of those <u>citizens</u> living in the immediate vicinity. Ideally, these districts should be spaced no less than one (1) mile apart.

11.02 **PERMITTED PRINCIPAL USES**

- (1) Those uses permitted in all residential districts.
- (2) Auto service station, for only the sale of gas, lubricants, coolants and accessories and the performance of incidental services such as tire installation and automobile washing, cleaning, and polishing, but not major overhaul, bumping, or painting.
- (3) Automobile sales and service (new or used).
- (4) Bakery in which the manufacture is limited to goods retailed only.
- (5) Barber and beauty shops.
- (6) Book and stationary stores.
- (7) Bowling <u>a</u>Alleys.
- (8) Candy, <u>or</u> confectionery store<u>s</u> in which the manufacture is limited to goods retailed only.
- (9) Carry-outs.
- (10) Clothing Sstores.
- (11) Dairy <u>bBars</u> <u>and</u>, <u>sSoft</u> <u>iIce</u> <u>Ccream shops</u>.
- (12) Department Sstores.
- (13) Drug Sstores.

- (14) Dry cleaning, laundry pickup service, and coin-operated washing and drycleaning facilities.
- (15) Dry goods and notions stores.
- (16) Eating and drinking restaurants or other places serving food and/or beverages.
- (17) Banks, Ffinance, and Lloan Aagencies.
- (18) Floral shop, fruit, nursery stock, and produce sales.
- (19) Food Mmarkets.
- (20) Furniture and appliances, including rugs, floor coverings, drapery sewing machine shops, used furniture, office equipment, supplies and similar uses.
- (21) Gift Sshops.
- (22) Hardware <u>S</u>stores.
- (23) Household appliance repair.
- (24) Jewelry Sstores.
- (25) Photographic <u>Ss</u>tudio<u>s</u>.
- (26) Physicians' offices, dentists' offices, and private clinics.
- (27) Plumbing and heating shops.
- (28) Radio, TV, and music stores.
- (29) Shoe <u>Sstores</u>.
- (30) Shoeshine and shoe repair shops.
- (31) Tailors and dress makers' shops.
- (32) Uses similar to the above uses.

11.03 <u>ACCESSORY USES</u>

Uses customarily accessory to the above<u>-listed permitted</u> <u>principal</u> uses.

11.04 <u>YARD REQUIREMENTS</u>

In a $n-\underline{B-3}-\underline{N}$ eighborhood $\underline{b}\underline{B}$ usiness $\underline{d}\underline{D}$ istrict, the following minimum yard areas shall be provided are required:

(1) Front yards: No front yard shall be required except:

When the frontage on one side of a block is divided between a $\underline{\mathbf{N}}$ neighborhood $\underline{\mathbf{B}}$ business $\underline{\mathbf{D}}$ district and a residential district, or is across the street from any residential district, or is across street from any residential district, the front yard requirement of the residential district shall apply to the area in the $\underline{\mathbf{n}}\underline{\mathbf{N}}$ eighborhood $\underline{\mathbf{b}}\underline{\mathbf{B}}$ usiness $\underline{\mathbf{d}}\underline{\mathbf{D}}$ istrict.

- (2) Side Yards: Side yards shall not be required excepting:
 - (a) A yard not less than <u>ten</u> (10) feet in width shall be provided where a side lot line of a n \underline{N} eighborhood $\underline{b}\underline{B}$ usiness $\underline{d}\underline{D}$ istrict abuts a residential district.
 - (b) In all other cases, no side yards shall be required for a business, but if such a yard is voluntarily provided, it shall be not less than six (6) feet in width.
- (3) Rear Yards: Rear yards shall not be required excepting where a rear lot line of a <u>nN</u>eighborhood <u>bB</u>usiness <u>dD</u>istrict abuts a residential district. In such instances, there shall be a rear yard of twenty-five (25) feet for a one_ (1) or two_ (2) story building. Such yards may be measured from the centerline of an intervening alley. An additional one (1) foot of rear yard shall be provided for each two (2) feet of height over twenty-five (25) feet.

A wall or decorative fence at least five feet six inches (5'6") high shall be placed along the boundary line of a rear yard abutting a residential district.

BUILDING HEIGHT REGULATIONS

Maximum permitted height for buildings in this district shall be thirty five (35) feet in height. No building in the B-3 Neighborhood Business District shall exceed two and one-half (2 ½) stories or thirty-five (35) feet in height.

11.06 **ACCESSORY PARKING**

In a neighborhood business district parking shall be provided as required in Section 18.16 of this Ordinance. Parking spaces shall be provided in accordance with the provisions of Zoning Code Section 18.16.

11.07 <u>OFF-STREET LOADING</u>

<u>Off-street loading sSpaces</u> shall be provided as required in <u>Zoning</u> <u>Code</u> Section 18.15.

11.08 <u>SIGNS</u>

See **Zoning Code** Section 18.20 for size and location of permitted signs.

11.09 SCREENING

See **Zoning Code** Section 18.17 for screening regulations for uses adjoining residential districts.

11.10 **CONDITIONAL USES**

Conditional uses are as follows:

The following use is permitted pursuant to Article #20, Section 20.14, Procedures and Requirements for Approval of Conditional Use Permits and subject to compliance with applicable standards and conditions set forth in Article #18, Section 18.30 and to all other applicable requirements of the Ordinances of this City:

Adult entertainment facilities, as defined in Article #18, Section 18.30.

ARTICLE 12

B-4 CENTRAL BUSINESS DISTRICT

12.01 <u>INTENT</u>

This district is designed to provide for a restricted variety of retail stores and related activities and for office buildings and service establishments which occupy the prime retail frontage in the Central Business District, and which serve the comparison, convenience, and service needs of a consumer population well beyond the corporate boundaries of the <u>City</u> <u>Municipality</u>. The district regulations are also designed to provide for a centrally located major shopping complex which will be serviced with conveniently located off-street parking compounds and safe pedestrian movement, but to exclude non-retail uses which generate a large volume of truck traffic.

12.02 <u>PERMITTED PRINCIPAL USES</u>

- (1) Any generally recognized retail business, service establishments, or processing uses as follows:
 - (a) Those uses permitted in all residential districts.
 - (b) Apparel shops, including specialty shops of all sorts, shoe

Sstores, and similar uses.

- (c) Shops selling automobile parts and accessories exclusively.
- (d) Banks, loan offices, stock exchange offices, and other financial institutions.
- (e) Commercial recreation facilities such as bowling alleys or movie theaters.
- (f) Department <u>Ss</u>tores.
- (g) Drug <u>Ss</u>tores.
- (h) Eating and drinking restaurants or other places serving food and/or beverages.

- (i) Food stores including supermarkets and all types of specialty food stores such as bakeries, candy stores₂ and similar uses.
- (j) Furniture and appliances, including rugs, floor coverings, drapery, sewing machine shops, used furniture, office equipment, supplies, and similar uses.
- (k) Gift shops, camera shops, record shops, and similar uses.
- (l) Hardware and related stores **<u>such</u>** as paint, wallpaper, and similar uses.
- (m) Hotels and motels.
- (n) Professional and other offices drawing a large number of clients and/or customers such as, but as not restricted to:
 - (1) Chamber of Commerce, <u>Aa</u>utomobile <u>Cc</u>lubs.
 - (2) Doctors, dentists, lawyers, architects.
 - (3) Insurance **companies**, realtors, unions.
 - (4) Post offices.
 - (5) Utility Ooffices.
- (o) Publishing and printing.
- (p) Repair shops such as shoe and watch repair.
- (q) Service shops **<u>such</u>** as barber, beauty, laundry, cleaner, and similar uses.
- (r) Travel agencies.
- (s) Variety <u>S</u>stores.
- (t) Brewpubs, **<u>Tt</u>**aprooms, and Wwine <u>B</u>**b**ars.

- (2) Public and semi-public buildings and privately-owned schools such as but not restricted to:
 - (a) Churches.
 - (b) Fraternal organizations.
 - (c) Librar<u>iesy</u>.
 - (d) Municipal Ooffices.
 - (e) Parking garages.
 - (f) Nursery schools, provided that there is compliance with State of Ohio requirements regarding spaces for play areas per child either on site or in a public play area no more than one (1) block from the facility.
- (3) Other uses, which in the opinion of the Planning Board are similar to the above uses indicated as being permitted. The Planning Board shall receive a written recommendation from the <u>Old</u> Village <u>District</u> Review Board when considering other or additional uses in the Old Village District. The following uses are expressly prohibited:
 - (a) Adult <u>Ee</u>ntertainment <u>F</u><u>f</u>acilities, (See Article 18, Section 18.30);
 - (b) Auto service stations;
 - (c) Mechanized car wash facilities; and
 - (d) New or used auto sales lots or showrooms.
- (4) Off-street parking facilities provided according to the provision of **Zoning Code** Section 18.16 excluding multi-story parking garages.

12.03 ACCESSORY USES

Accessory structures and uses customarily incidental to the above permitted uses.

12.04 **YARD REQUIREMENTS**

In a Central Business District Zone, the following yard areas shall be provided:

- (1) Front Yards: No front yard shall be required except where the frontage on one (1) side of a block is divided between a Central Business District and a residential district, or across the street from a residential district, the front yard requirement of the residential zone shall apply to the area in the Central Business District.
- (2) Side Yards: Side yards shall not be required excepting:

A yard not less than six (6) feet in width shall be required where a side lot line of Central Business District abuts a residential district. One (1) additional foot of yard space shall be added for each additional two (2) feet of building height above twenty-five (25) feet.

(3) Rear Yards: Rear yards shall not be required excepting:

Where a rear lot line of a Central Business District abuts a residential district, there shall be a rear yard of twenty-five (25) feet for one (1) or two (2) story buildings, an additional one (1) foot of rear yard shall be provided for each two (2) feet of building height over twenty-five (25) feet, such yard may be measured from the centerline of an intervening alley. Where there is no alley, a wall or decorative fence at least five feet six inches (5' 6") high shall be placed along the boundary line of a rear yard abutting a residential district.

12.05 **BUILDING HEIGHT REGULATIONS**

There shall be no specific height limitation in a B-4 District provided, however, that prior to the issuance of a zoning permit for any structure which is planned to exceed thirty-five (35) feet in height, the Old Village **District** Review Board shall make a finding that any such excessive height will not be detrimental to the public safety, **or**; to the light, ; air, or privacy of any other structure or use currently existing or approved for construction.

12.06 <u>ACCESSORY PARKING</u>

2024-O-6 Exhibit A

In the B-4 Central Business District, there are no requirements for providing a minimum number or amount of parking spaces. **However, t**The General Provisions of Section 18.16(1) of this **Zoning Code** Ordinance shall apply, however, to accessory parking provided in this district. The Old Village District Plan shall be used as a guide in locating accessory parking lots.

12.07 <u>OFF-STREET LOADING</u>

<u>Off-street loading sSpaces</u> shall be provided <u>as required by</u> in accordance with the provisions of **Zoning Code** Section 18.15.

12.08 **SIGNS**

See **Zoning Code** Section 18.20 for sizes and location of permitted signs.

12.09 <u>SCREENING</u>

See **Zoning Code** Section 18.17 for screening regulations for uses adjoining residential districts.

12.10 <u>CONDITIONAL USES</u>

(1) Microbreweries, Amicrowineries, and Amicrodistilleries.

ARTICLE 13

WO, WP WELL FIELD PROTECTION DISTRICTS

13.01

PURPOSE AND INTENT

It is the <u>purpose and</u> intent of the WO Wellhead Operation District and the WP Well Field Protection Overlay District to safeguard the public health, safety, and welfare of the customers of the protected public water supplies and to protect the community's potable water supply against contamination by regulating land use and the storage, handling, use and/or production of Regulated Substances as defined <u>below</u> in Section 13.02 of this Article of the Zoning Code. These districts will be shown on the City of Bellbrook Zoning Map at the location of any existing or proposed public wells and their capture areas.

13.02

DEFINITIONS

- "Aquifer" means a glacial formation, group of glacial formations, or part of a glacial formation that contains enough saturated permeable material to yield significant quantities of water.
- "<u>Deminimis Quantity</u>" means a specified amount Regulated Substances, expressed in gallons and/or pounds, that is excluded from the provisions of the WO Wellhead Operation District and the WP Well Field Protection Overlay District regulations. Any use of Regulated Substances in excess of the deminimis quantities and not explicitly subject to exclusion is considered nonconforming.
- "<u>Direct Recharge Area</u>" means that portion of a drainage basin in which water infiltrating vertically from the surface will intercept the water table.
- "<u>Non-Routine Maintenance</u>" means activities necessary not more frequently than every twenty-four (24) months to keep structures and equipment in good repair.
- "One Year Capture Area" means the area around protected public water supply well fields delineated by the one-year travel time contour.
- "Overlay District" means a district described on the zoning map within which, through super-imposition of a special designation,

certain regulations and requirements apply, in addition to those of the

underlying zoning districts to which such designation is added.

- "<u>Potable Water</u>" means water that is satisfactory for drinking, culinary and domestic purposes, meeting current drinking water standards.
- "Protected Public Water Supply" means a public water system which services as least fifteen (15) service connections used by year-round residents or regularly serves at least twenty-five (25) year-round residents, and having a one-year capture area defined through appropriate hydrologic studies.
- "Recharge Lagoon" means a body of water designed and maintained by man to add water to the groundwater at a rate greater than that occurring naturally.
- "Regulated Substances" means chemicals and mixtures of chemicals which are health hazards. Materials packaged for personal or household use as food or drink for man or other animals are not Regulated Substances. Regulated Substances include:
 - (1) Chemicals, including but not limited to PFAS chemicals, for which there is scientific evidence that acute or chronic health effects may result from exposure including carcinogens, toxic and highly toxic agents, reproductive toxins, irritants, corrosives, sensitizers, hepatoxins, nephrotoxins, neurotoxins, agents which act on the hematopoietic system, and agents which damage the lungs, skin, eyes, or mucous membranes.
 - (2) Mixtures of chemicals which have been tested as a whole and have been determined to be a health hazard.
 - (3) Mixtures of chemicals which have not been tested as a whole but <u>but</u> which contain any chemical which has been determined to be a health hazard and which comprises one percent (1%) or greater of the composition on a weight per unit weight basis, and mixtures of chemicals which include carcinogen if the concentration of the carcinogen in the mixture is one tenth of one percent (.1) or greater of the composition on a weight per unit weight basis.

- (4) Ingredients of mixtures prepared within the WP Well Field Protection Overlay District in cases where such ingredients are health hazards but comprise less than one tenth of one percent (0.1) of the mixture on a weight per unit weight basis if carcinogenic, or less than one percent (1%) of the mixture on a weight per unit weight basis if non-carcinogenic.
- (5) Petroleum and non-solid petroleum derivatives (except non-PCB dielectric fluids).

"<u>Travel Time Contour</u>" means a locus of points from which water takes an equal amount of time to reach a given destination such as a well or well field.

"<u>Underground Storage Tank</u>" means any one or combination of tanks (including underground pipes connected thereto) which is used to contain an accumulation of Regulated Substances and the volume of which (including the volume of underground pipes connected thereto) is

ten percent (10%) or more beneath the surface of the ground. Flowthrough process tanks and septic tanks approved by the Health Department or State Environmental Protection Agency, as applicable, are excluded from the definition of underground storage tanks.

"Well Field" means a tract of land that contains a number of wells for supplying water.

"Zone of Influence" means a zone delineated by iso-travel time contours around well fields. The zone is calculated, based on the rite of movement of groundwater in the vicinity of well with an allowance for the dispersion of a pollutant entering into and moving with the groundwater.

DETERMINATION OF APPLICABILITY

(1) Responsibility

It shall be the responsibility of any person owning real property and/or owning or operating a business within the City of Bellbrook to make a determination of the applicability of this Article as it pertains to the property and/or business and failure to do so shall not excuse any violations of this Article.

(2) Applicability Based on Regulated Substances

Applicability based on deminimis quantities and/or limited exclusions pertaining to Regulated Substances as set forth in this Article shall be determined separately for each use, as **that term is** defined in Article 3, Section 3.02, Paragraph 107, associated with the property.

13.04 <u>WO WELLHEAD OPERATION DISTRICT</u>

(1) Permitted Principal Uses:

- (a) Municipal water supply, treatment, and operations facilities in accordance with the City of Bellbrook's plan(s) for water supply and treatment.
- (b) Public parks, playgrounds, and community centers.

(2) Conditional Uses:

- (a) Public utility uses including electric and telephone substations; gas regulator and meter station buildings; and electric and communications transmission towers and structures.
- (b) Existing single family detached residences.

(3) Groundwater Protection Standards:

- (a) Use of Regulated Substances in conjunction with public water supply and treatment activities shall not be restricted by this paragraph.
- (b) Use of Regulated Substances in conjunction with public parks, playgrounds and community centers shall be in accordance with the City of Bellbrook's management plan for Maintenance of Sensitive Areas.

- (c) Use of Regulated Substances in conjunction with conditional uses in this district shall be limited to:
 - (1) The aggregate of Regulated Substances in use may not exceed twenty (20) gallons or one hundred and sixty (160) pounds at any time.
 - (2) The total use of Regulated Substances may not exceed fifty (50) gallons or four hundred (400) pounds in any twelve (12) month period.
- (d) A limited exclusion from the provisions of Section 13.04, Paragraph (3)(c) is authorized for non-routine maintenance or repair of property or equipment. The use of Regulated Substances under this exclusion shall be limited to:
 - (1) The aggregate of Regulated Substances in use may not exceed fifty (50) gallons or four hundred (400) pounds at any time.
 - (2) The total use of Regulated Substances may not exceed one hundred (100) gallons or eight hundred (800) pounds in any twelve (12) month period.
- (e) Storage of regulated Substances in conjunction with public water supply and treatment activities shall not be restricted by this paragraph.
- (f) Underground storage of fuel and lubricants for vehicle operations in conjunction with permitted and conditional uses in this district shall be in tanks placed above the floor surface
- of a below grade vault. Said vault shall allow access for physical inspection of the tank for leakage and the interior of the vault shall be continuously monitored and alarmed to provide for
- automatic and immediate detection of any release from the tank.

- (g) Notwithstanding other provisions of this Article, nonconforming uses in this district presently utilizing underground storage tanks for fuel and lubricants for vehicle operations shall be permitted to replace existing tanks with those constructed as per the specifications of Section 13.04, Paragraph (3)(f) and not exceeding the capacity of existing tanks. Replacement of underground tanks for Regulated Substances other than fuel and lubricants for vehicle operations is not permitted.
- (h) Storage of Regulated Substances other than fuel and lubricants for vehicle operations in conjunction with permitted and conditional uses in this district is prohibited.
- (i) As part of the findings required under Article 19 of the Zoning Code prior to issuance of a Zoning Permit/Certificate of Zoning Compliance, the Zoning Officer Community Development Administrator shall utilize the Hazard Potential Ranking System, identified in Section 13.05, Paragraph (9) to assist in the determination of intensity of use within this district. No substitutions of a non-conforming use shall be permitted which result in an increase of the Hazard Potential Ranking on a parcel within this district.
- (j) All uses within this district shall be connected to the public wastewater disposal system within a three (3) year period from the effective date of this Ordinance or have a wastewater

disposal system approved by the Greene County Combined General Health District.

13.05 WP WELL FIELD PROTECTION OVERLAY DISTRICT

(1) <u>Applicability of Well Field Protection Overlay District to Underlying Zoning Districts</u>:

The provisions of this Section shall be applicable to all lands shown as being located within the boundaries of the WP Well Field Protection Overlay District on the Zoning Map and shall be supplemental to the regulations of the underlying zoning district. Where the requirements of this section are in conflict with the regulations of the underlying district, the more restrictive regulations shall apply.

(2) Permitted Principal Uses:

The permitted uses within the WP Well Field Protection Overlay District shall be those of the underlying zoning district.

(3) Accessory Uses:

Accessory uses and buildings within the WP Well Field Protection Overlay District shall be those customarily incident to the permitted principal uses of the underlying zoning district.

(4) Conditional Uses:

The requirements of this paragraph shall be in addition to any applicable regulations found elsewhere within the Zoning Code.

- (a) The conditional uses within the WP Well Field Protection Overlay District shall be those of the underlying zoning district, except as specified in Section 13.05, Paragraphs (4)(b) and (7).
- (b) When listed as a permitted or conditional use within the underlying zoning district, the excavation, extraction, mining, or processing of
- sand, gravel, and limestone from the earth for resale shall remain as conditional uses in the WP Well Field Protection Overlay District, subject to Board of Zoning Appeals approval of an excavation and facilities plan that includes, but is not limited to:
 - (1) An existing site plan with topographic detail at two (2) foot contour intervals, all planimetric information, depth to groundwater, and flood plain characteristics where applicable.
 - (2) The proposed extent and depth of excavations.
 - (3) Slope angle of excavation walls (any final slopes shall be at the angle of repose for the remaining material).
 - (4) Use and disposition of the spoil and/or overburden materials from the excavations including a landscaping and vegetation plan to stabilize any disturbed material.
 - (5) Surface drainage plan:

- (A) Drainage into on-site excavation from proximate off-site transportation facilities such as roadways and roadbeds and off-site watercourses is prohibited unless the applicant provides a plan which otherwise protects the excavation from off-site waterborne Regulated Substances.
- (B) The final on-site grading shall minimize all surface drainage into the excavations.
- (6) A post-excavation and operation land use plan.
- (7) A security plan (unauthorized access shall be strictly prohibited as long as any excavations remain on site).

(5) Non-Conforming Uses:

Uses using, storing, handling and/or producing Regulated Substances in amounts in excess of the quantities specified in Section 13.05, Paragraphs (8)(a) through (8)(f) are non-conforming uses subject

to the requirements of this section.

(6) Yard Requirements and Building Height Regulations:

Yard requirements and building height regulations shall be those of the underlying zoning district.

(7) Prohibited Uses:

Sanitary landfills, drywells, landfills comprised of demolition debris or other non-approved matter and junkyards are prohibited within the WP Well Field Protection Overlay District.

(8) Groundwater Protections Standards:

- (a) Use, storage, handling and/or production of Regulated Substances in conjunction with permitted and conditional uses in this district shall be limited to:
 - (1) Aggregate of Regulated Substances: The aggregate of Regulated Substances in use, storage,

handling and/or production may not exceed twenty (20) gallons or one hundred and sixty (160) pounds at any time.

- (2) <u>Total Use of Regulated Substances</u>: The total use, storage, handling and/or production of Regulated Substances may not exceed fifty (50) gallons or four hundred (400) pounds in any twelve (12) month period.
- (b) A limited exclusion from the provisions of Section 13.05, Paragraph (8)(a) is authorized for non-routine maintenance or repair of property or equipment. These, storage, handling and/or production of Regulated Substances under this exclusion shall be limited to:
 - (1) The aggregate of Regulated Substances in use, storage, handling and/or production may not exceed fifty (50) gallons or four hundred (400) pounds at any time.
 - (2) The total use, storage, handling and/or production of Regulated Substances may not exceed one hundred (100) gallons or eight hundred (800) pounds in any twelve (12) month period.
 - (3) The application of USEPA approved agricultural chemicals by licensed personnel using USEPA best recommended practices. Below ground applications in excess of one hundred (100) gallons or eight hundred (800 pounds in any twelve (12) month period shall require 72-hour prior notice to the City of Bellbrook Administrative Officer Community Development Administrator or his designee.
- (c) A limited exclusion from the provisions of Section 13.05, Paragraph (8)(a) is authorized for each medical and research laboratory use, provided however, Regulated Substances shall be stored, handled, or used in containers not to exceed five (5) gallons or forty (4) pounds of each substance and the aggregate inventory of Regulated Substances shall not

exceed two hundred and fifty (250) gallons or two thousand (2,000) pounds.

- (d) A limited exclusion from the provisions of Section 13.05, Paragraph (8)(a) is authorized for Regulated Substances which are cleaning agents, provided, however, such cleaning agents are packaged for personal or household use or are present in the same form and concentration as a product packaged for use by the general public, and provided the aggregate inventory of such cleaning agents shall not exceed one hundred (100) gallons or eight hundred (800) pounds at any time. In no case shall Regulated Substances claimed under this exclusion include hydrocarbon or halogenated hydrocarbon solvents.
- (e) A limited exclusion from the provisions of Section 13.05, Paragraph (8)(a) is authorized for on-site storage of a maximum one-year supply of agricultural chemicals to be used for routine onsite agricultural operations, provided such substances are stored in standard approved packaging and such chemicals are applied to cropland under Best Management Practices as indicated by soil tests, agricultural experts or label directions approved by the United States Environmental Protection Agency (EPA) or the Ohio Department of Agriculture. This limited exclusion also applies to the application of agricultural
- chemicals to cropland where such chemicals are brought in from other locations. This provision does not exempt such agricultural chemicals either stored on-site or brought in from other locations from the inventory, spill reporting and underground storage tank protection requirements of the WP Well Field Protection Overlay District.
- (f) With the exception of residential use of heating fuels in tanks having a capacity equal to or less than five hundred (500) gallons, the underground storage of fuel and lubricants for

vehicle operation and fuel for building and/or process heating in conjunction with permitted and conditional uses in this district shall be in tanks secondarily contained and monitored. Such installations shall be subject to approval by the Zoning Inspector Community Development Administrator ("CDA") or his the CDA's designee.

(g) Notwithstanding other provisions of this Article, nonconforming uses in this district presently utilizing underground storage tanks for fuel and lubricants for vehicle operations and

fuel for building and/or process heating shall be permitted to replace existing tanks with those constructed as per the specifications of Section 13.05, Paragraph (8)(f) and not exceeding the capacity of existing tanks. Replacement of underground tanks for Regulated Substances other than the above noted fuels and lubricants is not permitted.

(h) As part of the findings required under Article #-19 of thise Zoning Code, Ordinance prior to issuance of a Zoning Permit or a Certificate of Zoning Compliance, the Zoning Inspector Community Development Administrator or the CDA's his

designee shall utilize the Hazard Potential Ranking System in Section 13.05, Paragraph (9) to assist in the determination of intensity of use within this district. No substitutes of a non-conforming use shall be permitted which result in an increase of the Hazard Potential Ranking on a parcel within this district. If the quantities of Regulated Substances initially exceed the deminimis quantities above, they shall be considered legally non-conforming. Such legally non-conforming quantities cannot be increased.

(9) <u>Hazard Potential Rating System</u>:

In order to assess the risk for potential groundwater contamination, a hazard rating has been developed for various activities categorized by their Standard Industrial Classification (SIC) Code. This rating is based on the kind of materials commonly associated with each use looking only at the most critical hydrologic factors.

Table 1 below lists the site hazard potential by land use activity (source) on a scale of one (1) to nine (9), with one (1) being a low hazard and nine (9) a very high hazard. This rating is based on the intrinsic hazards posed by different land uses and is related to the materials commonly used or stored on the site or the types and amounts of wastes commonly discharged.

Table II below lists the hazard potential determined on the basis of materials known to be used, stored, or disposed of at a specific site.

If the two tables referenced above indicate different site hazard potential ratings for the SIC-coded land use activity and the materials found on-site, the higher of the two scores is the rating for the site.

(10) Enforcement Provisions:

(a) Scope:

- (1) The provisions of this <u>s</u>Section shall be effective within the WP Well Field Protection District, except as otherwise provided. This Ordinance provides for pollution control pertaining to the public water supply.
- (2) Nothing contained in this section shall be construed so as to interfere with any existing or future lawful requirements that

may be, or heretofore were, imposed by any other public body authorized to enact sanitary, health, or water pollution abatement restrictions so long as such requirements are consistent with, or more stringent than, the stated purpose of this Ordinance.

(b) Administration:

Except as otherwise provided herein, the Administrative Officer Community Development Administrator for the City of Bellbrook or the CDA's his designee, hereinafter referred to as the Administrative Officer, shall administer, implement, and enforce the provisions of this section.

(c) Notice of Violation:

(1) Any person found in violation of any provisions of this <u>s</u>Section, or any order, requirement, rule, or regulation

Hissued under the authority of this Section will be served with a written notice stating the nature of the violation and providing reasonable time for compliance; provided however, written notice of violation may be dispensed with under the conditions described In Section 13.05, Paragraph (10)(i)(2) and provided further, that if the Administrative Officer Community Development Administrator or the CDA's designee has previously promulgated a schedule of compliance or issued an order addressing the same type of or a similar violation and the time for compliance has passed, the Administrative Officer Community Development Administrator or the CDA's designee may dispense with establishing another time period for compliance.

(2) The notice shall be served in the manner provided by law for the service of civil process. Where the address of the violator

is unknown, service may be made upon the owner of the property involved at the tax mailing address of the owner as shown on the **c**County tax record.

(d) Inspections: Subject to applicable provisions of law, the Administrative Officer Community Development Administrator or the CDA's authorized designee bearing proper identification, shall be permitted to enter private property at any reasonable time, with reasonable cause or with prior notification, for such purposes as inspection, observation, measurement, sampling, and records examination pertaining to the requirements of this Ordinance to ensure that activities are in accordance with the provisions of this sSection. Upon request of the entity, which is the subject of the iInspection, and if permitted by the State Ohio Public Records Law, information obtained as a result of the inspection shall be maintained as confidential. If the owner or tenant does not consent to the entry Administrative Officer Community Development Administrator or the CDA's designee for the above-stated purposes, the Administrative Officer Community Development Administrator or the CDA's designee may apply to a court of competent jurisdiction for an appropriate warrant or other authority to enter said property.

(e) Vandalism: No person shall maliciously, willfully, or negligently break, damage, destroy, uncover, deface, or tamper with any structure,

appurtenance, property, or equipment which Is is a part of or used in conjunction with water facilities of the City of Bellbrook and/or any

other protected public water supply, or which results In the violation of the provisions of this Ordinance.

- (f) Severability: A finding by any court or other jurisdiction that any part or provision of this Ordinance is invalid shall not affect the validity of any other part or provision of this Ordinance which can be given effect without the Invalid parts or provisions.
- (g) Handling of Regulated Substances:
 - (1) No person shall place, deposit, or permit to be deposited, store, process, use, produce, dispose of, transport, or discharge, hereinafter referred to as "handle,' any Regulated Substance on public or private property within the City of Bellbrook or in any area under the jurisdiction of the City of Bellbrook except as provided by law, statute, ordinance, rule, or regulation.
 - (2) Any violation of this paragraph is hereby determined to be a nuisance.
- (h) Underground Storage Tanks Declared to Constitute Dangerous Nuisances:
 - (1) With the exception of the residential use of home heating fuel in tanks having a capacity equal to or less than five hundred (500) gallons, and underground storage systems for accessory vehicle fuel, vehicle lubricants, fuel for building heating and fuel for process heating, any storage of Regulated Substances In underground storage tanks within the WP Well Field Protection Overlay District shall be deemed to constitute a dangerous nuisance. Every such nuisance must and shall be abated no later than five (5) years from the effective date of this Ordinance. Any tank not removed under Section (1) shall meet the requirements of Section (2) within five (5) years.

(2) With the exception of residential use of home heating fuel in tanks having a capacity equal to or less than five hundred (50O)

gallons, any underground storage tank systems for vehicle fuel and lubricants and fuel for building and/or process heating within the WP Well Field Protection Overlay District not removed within five (5) years from the effective date of this Ordinance must be secondarily contained and monitored in accordance with plans submitted to and approved by the Administrative Officer Community Development Administrator and the City of Bellbrook Fire Department.

(i) Reporting Requirements:

(1) Regulated Substance Activity Inventory:

(A) Applicability:

- (1) Except as provided in Section 13.05, Paragraph (10)(i)(1)(B), any owner or occupant of any land in the WP Well Field Protection Overlay District at the effective date of this Ordinance, shall file a Regulated Substance Activity Inventory Report with the Administrative Officer Community Development Administrator. Said Report shall be filed within one hundred and eighty (180) days of the effective date of this Ordinance and at twenty-four (24) month intervals thereafter.
- (2) Except as provided in Section 13.05, Paragraph (10)(i)(l)(B), any new owner or occupant of any land in the WP Well Field Protection Overlay District shall file a Regulated Substance Activity Inventory Report prior to receipt of a Zoning Permit or a Certificate of Zoning Compliance and at twenty-four (24) month intervals following the date of occupancy. For purposes of this paragraph, "new" shall be defined as

subsequent to the effective date of this Ordinance.

- (3) Where a person owns, operates, or occupies more than one location, Regulated Substance Activity Inventory Reports shall be made for each location.
- (4) Agricultural uses shall file a Regulated Substance Act Inventory Report within one hundred eighty (180) days of the effective date of this Ordinance and at twelve (12) month intervals thereafter. Regulated Substance Inventory Activity Reports for agricultural uses shall include total annual on-site application of Regulated Substances for the reporting property.

(B) Exclusions to Activity Inventory Reporting:

- (1) Any exclusion set forth in this subsection shall apply provided that said exclusion does not substantially increase any risk or hazard to the public health or water supply, wells or well fields; and provided further that any spill, leak, discharge, or mishandling shall be subject to the provisions of Section 13.05, Paragraph (10)(i)2. Any exclusions granted herein shall not remove or limit the liability and responsibility of any person or activity Involved.
- (2) A limited exclusion from Regulated Substance activity Inventory reporting Is hereby authorized for **!i**ncidental uses of Regulated Substances provided the uses are limited as follows:
 - (a) The aggregate of Regulated Substances in use may not exceed

- twenty (20) gallons or one hundred and sixty (160) pounds at any time.
- (b) The total use of Regulated Substances may not exceed fifty (50) gallons or four hundred (400) pounds In any twelve (12) month period.
- (3) A limited exclusion from Regulated Substance activity inventory reporting is hereby authorized for non-routine maintenance or repair of property or equipment In the WP Well Field Protection Overlay District provided the uses are limited as follows:
 - (a) The aggregate of Regulated Substances In use may not exceed fifty (50) gallons or four hundred (400) pounds at any time.
 - (b) The total use of Regulated Substances may not exceed one hundred (100) gallons or eight hundred (800) pounds in any twelve (12) month period.
- (4) A limited exclusion from Regulated Substance activity Linventory reporting is hereby authorized for Regulated Substances which are cleaning agents, provided however, such cleaning agents are packaged for personal or household use or are present in the same form and concentration as a product packaged for use by the general public, and provided the aggregate inventory of such cleaning agents shall not exceed one hundred (100) gallons or
- shall not exceed one hundred (100) gallons or eight hundred (800) pounds at any time. In no case shall Regulated Substances claimed under the exclusion Include hydrocarbon or halogenated hydrocarbon solvents.

- (5) A limited exclusion from Regulated Substance activity Inventory reporting is hereby
- authorized for medical and research laboratory uses in the WP Well Field Protection Overlay District, provided however, Regulated Substances shall be stored, handled, or used in containers not to exceed five (5) gallons or forty (40) pounds of each substance and the aggregate Inventory of Regulated Substances shall not exceed two hundred and fifty (250) gallons or two thousand (2,000) pounds.
- (6) An exclusion from Regulated Substance activity inventory reporting is hereby authorized for the transportation of Regulated Substances through the WP Well Field Protection Overlay District provided that the transportation vehicle is in compliance with applicable local, Sstate, and Federal laws and regulations; provided that the regulated substance is fueling the transportation vehicle; and provided that the transportation vehicle is in continuous transit,
- making delivery or is stopped for a period of time not to exceed seventy-two (72) hours.
- (7) A limited exclusion from Regulated Substance activity **!i**nventory reporting is hereby authorized for owners and occupants of single- or two-family residences, provided however, the storage and use of Regulated Substances are related to the maintenance of the residence or vehicles under control of the occupant and provided waste Regulated Substances, are appropriately disposed of to a permitted solid waste facility or a permitted publicly-owned wastewater treatment works.
- (2) Spills, Leaks or Discharges:

- (A) Any person with direct knowledge of a spill, leak, or discharge of a Regulated Substance within the WP Well Field Protection Overlay District shall, if such spill, leak, or discharge escapes containment or contacts a non-impervious ground surface and is not immediately and completely remediated, give notice to the Public Works Director of the City of Bellbrook or the operator on duty at the affected or potentially affected water treatment facility by telephone within thirty (30) minutes. The notification shall include at a minimum, the location of the incident, name and telephone number, date and time thereof, type of substance(s), concentration and volume, and control or corrective action taken. Such notification shall in no way alleviate other local, Sstate, and Ffederal reporting obligations as required by law.
- (B) The application of agricultural chemicals, fertilizers, mineral acids, organic sulphur-compounds, etc. used in routine agricultural operations, including plant nutrients and crop protection materials, applied under Best Management Practices as indicated by soil tests, agricultural experts, or label directions approved by the United States EPA or the Ohio Department of Agriculture, shall not be considered a spill, leak, or discharge subject to the reporting provisions of this paragraph.
- (C) Any entity or person who spills, leaks or discharges said substance(s) shall be liable for any reasonable expense, loss, or damages incurred by the City of Bellbrook in response to such an incident, in addition to the amount of any lines imposed on account thereof under Sstate and Ffederal law; said The entity or person shall document and maintain sufficient records so as to reflect accurately the circumstances related to any such incident and develop and implement procedures to substantially eliminate the likelihood of recurrence of such spills, leaks, or discharges as soon as practicable following the incident, but no later than one hundred eighty (180) days after the incident.

(3) Falsifying Information:

No person shall make any false statement, representation, or certification in any report or other document filed or required to be maintained pursuant to this Ordinance.

(4) Retention of Records:

Any reports or records compiled or submitted pursuant to this paragraph shall be maintained by the user for a minimum of five (5) years, or so long as enforcement or judicial proceedings are being pursued, whichever is longer.

- (j) Public Water Supply Protection Authorities:
 - (1) Application: If any activity or use of Regulated Substance is deemed by the Administrative Officer Community Development Administrator to pose a real and present danger of contaminating surface and/or groundwater which would normally enter the public water supply, the Administrative Officer Community Development Administrator, in accordance with Section 713.13 of the Ohio Revised Code, is authorized to:
 - (A) Cause cessation of said activity or use of the Regulated Substance;
 - (B) Require the provision of administrative controls and/or facilities sufficient to mitigate said danger; and/or
 - (C) Cause the provision of pollution control and/or abatement activities.
 - (2) Considerations: When considering the exercise of any of the above authorities or actions, the Administrative Officer Community Development Administrator shall consult with the appropriate administrative official of any potentially affected protected public water supply. Such consultation shall determine what measures need to be taken to ensure the public water supply is reasonably and adequately protected from contamination for the present and the future. The Administrative Officer Community Development Administrator may take into consideration any evidence represented by the entity regarding cost effectiveness and the economic effectiveness and the economic impact imposed by the requirements or actions.
 - (3) Exemption of Certain Regulated Substances:

The Administrative Officer Community Development Administrator is authorized to exclude certain Regulated Substances, that pose no threat to groundwater, from the provisions of these regulations. Prior to authorizing the exemption of any Regulated Substance, the Administrative Officer Community Development Administrator shall have such request for exemption reviewed by the Environmental Advisory Board. The recommendation of the Environmental Advisory Board shall be binding on the administrative officer Community Development Administrator.

(4) Technical Consultants: Upon application for a Certificate of Zoning Compliance and/or Occupancy Certificate for a use within the WP Well Field Protection Overlay district, the Administrative Officer Community Development Administrator may employ such technical expertise as needed to ensure compliance with the provisions of these regulations. All costs incurred in the compliance review process shall be passed through to the applicant and shall be in addition to those fees normally charged by the City of Bellbrook to review an application for a Certificate of Zoning Compliance and/or Occupancy Certificate.

(k) Well Field Protections Board:

- (1) Appeals: Any person may appeal an action of the Administrative Officer Community Development Administrator made pursuant to this Ordinance as provided for under Article #20 by filing a Notice of Intent to Appeal within twenty (20) days of said action and a Statement of Appeal within thirty (30) days of the date that the order being appealed was issued. A Notice of Intent to Appeal shall include at a minimum, name, address, telephone number, date, and a statement of intent of appeal. A Statement of Appeal shall include all information contained in the Notice of Intent to Appeal, a description of the nature of the appeal, and any pertinent documentation.
- (2) Appeals Board: The Board of Zoning Appeals of the City of Bellbrook shall hear Well Field Protection Overlay District appeals. Said <u>The</u> Board shall have the authority to take appeals, investigate matters related to said appeals, deny, uphold, or otherwise modify or waive, the <u>Administrative Officer's</u> <u>the Community</u> **Development Administrator's** actions on a case-by-case basis.

Item C.Section 8, Item

(1) Penalties for Violations:

Vandalism: A violation of the provisions of Section 13.05, Paragraph (10)(e), beyond the time limit for compliance set forth by the Administrative Officer, Community Development Administrator's Notice of Violation, or the compliance schedule established by the administrative officer, Community Development Administrator shall constitute a minor misdemeanor punishable as provided in Article #19 of the Zoning Code.

NEED TO INSERT ARTICLE 13, TABLE I AND II HERE – Table is currently on the City's website with Article 13

ARTICLE 14 BELLBROOK OLD-VILLAGE PLAN DISTRICT

14.01 GENERAL

This section is intended to create the Bellbrook Old Village District Plan and to establish procedures for control, development, and regulation of this district along with a master design plan and architectural controls specific to the designated area. These controls are to be in addition to existing planning and zoning controls and do not replace the underlying zoning pattern but are supplemental in nature. Where there is conflict between these controls and those of the underlying district, the regulations of this section shall prevail.

It is also the intent of this <u>O</u>rdinance to recognize the history of the area as well as human values and basic property rights. No attempt is made to create a situation of undue financial strain upon any business or resident.

As such, the intent of this **O**ordinance is to regulate demolition and guide new development and construction in the <u>Old</u> Village District. In no manner is this <u>O</u>ordinance to be interpreted to interfere with the day-to-day activities and normal periodic maintenance of any home or structure.

Council shall adopt a <u>resolution</u> containing guidelines (hereinafter referred to as the "Guidelines") for implementing this <u>O</u>erdinance and carrying out its intent. A board to be known as the <u>Old</u> Village <u>District</u> Review Board, established in accordance with <u>Section</u> <u>Article</u> 14.13 <u>below</u> <u>hereof</u>, shall administer this <u>O</u>rdinance in accordance with the Guidelines as adopted by Council and as amended from time to time. In case of conflict between the Guidelines and any provision of this <u>O</u>erdinance, the <u>O</u>erdinance provision<u>s</u> shall prevail.

14.02 <u>DESIGNATION OF DISTRICT</u>

The area to be covered by these regulations is shown on the <u>Official Z</u>zoning <u>M</u>map. The boundaries are generally described as follows: bounded on the west by Little Sugarcreek, the south by the centerline of West South Street and East South Street, the east by the existing corporate line and the north by a line running along the southern and western boundary of Bellbrook Farms Subdivision just south of Upper Hillside Drive, extending across Main Street approximately 1300 feet west of Main Street then south to a point approximately 500 feet north of the centerline of Franklin Street, then west to the Little Sugarcreek.

Where any question arises as to exact location of the district boundaries, the City Council shall interpret the district boundaries.

14.03 PERMITTED PRINCIPAL USES

All uses permitted within the underlying zoning district.

14.04 <u>ACCESSORY USES</u>

All accessory uses allowed in the underlying zoning districts are permitted.

14.05 CONDITIONAL USES

All conditional uses allowed in the underlying zoning districts are permitted.

14.06 NON-CONFORMING BUILDING

The lawful use of any dwelling, building or structure and of any land or premises, as existing and lawful at the time of enacting this zZoning Code ordinance or amendments thereto, may be continued, although such use does not currently conform with the provisions of this Zoning Code such ordinance or amendments thereto, but if any such non-conforming use is voluntarily discontinued, for two one years—or more, any future use of such land shall must be in conformity with Article 14 of the Zoning Code Regulations.

14.07 YARD REQUIREMENTS

The regulations of the underlying zoning districts shall apply.

14.08 PARKING REQUIREMENTS

<u>Parking Sspaces</u> shall be provided in accordance with the provisions of <u>Zoning</u> Code Section 18.16.

14.09 OFF-STREET LOADING

<u>Off-street loading sSpaces</u> shall be provided in accordance with the provisions of <u>Zoning Code</u> Section 18.15.

14.10 **SIGNS**

18.20B.

Signs shall be regulated in accordance with provisions of **Zoning Code** Section

14.11 STANDARDS

The significant architecture of the Bellbrook Old Village District is generally classified as a simple Victorian with modifications.

There are numerous buildings that do not fall within this broad description, but the general architectural character of Bellbrook the Old Village District is Victorian in nature and the standards are aimed at preserving existing examples and establishing perimeters parameters which will allow acceptable and pleasing contemporary interpretation of the standards.

(1) Structural Proportion:

Bellbrook the Old Village District architecture is characterized by two-story buildings with tall, narrow doors and windows and in earlier examples, slightly squatter proportions. New buildings shall conform generally to the proportions of adjoining buildings and all buildings, either new or remodeled shall have windows and doors of proportions similar to adjoining buildings. Roof slopes shall also be similar to adjoining buildings.

(6) Foundations:

Foundation walls of the mid-19th century were of local limestone. Acceptable foundation materials shall be limited to limestone, brick, poured concrete or other material which is similar in appearance to the original foundation material.

(6) Exterior Walls:

The exterior walls of the period were of brick or narrow wood siding. Wall surface construction shall employ these materials or a modern construction material which creates the same effect. Cosmetic treatment using such materials as asphalt shingles which simulate brick or stone is prohibited.

(4) Roof Materials:

Roofing materials may consist of metal, slate, asphaltic shingles, or wood shingles. Prohibited materials include aluminum shingles formed to look like wood, asphalt roll roofing and simulated cosmetic materials.

(5) Chimneys:

Chimneys and flues shall be of brick or stucco masonry or other material which is compatible with the architecture of the period. Metal chimneys simulating masonry construction are prohibited.

(6) Windows and Door Lintels:

Original lintels were stone, wood, or brick. Where feasible, lintels shall be restored or compatible stone, brick, or wood shall be used in remodeling, additions, or new construction(s).

(7) Porches and Outbuildings:

The construction of porches or outbuildings which do not conform to these standards or to the character of the main building is prohibited.

14.12 EXCLUSIONS AND CONTROLS NOT APPROPRIATE

Nothing in this $\underline{\mathbf{O}}$ ordinance shall be construed to prevent the ordinary maintenance or repair of any property, area, or object within the $\underline{\mathbf{Old}}$ Village District. For the purposes of this $\underline{\mathbf{O}}$ ordinance, a modification of a window or door which causes the area of the opening to vary by no more than ten percent $\underline{(10\%)}$ from its original size shall be ordinary maintenance or repair and is not regulated.

Neither the exterior color nor the interior arrangement of the building and the use to be made thereof, shall be considered by the Village Review Board in determining whether an application complies with the provisions of this **O**erdinance, provided such use is permitted by the applicable zoning designation.

14.13 <u>OLD VILLAGE DISTRICT REVIEW BOARD</u>

The <u>Old Village District</u> Review Board <u>(the "Board")</u> shall consist of five <u>(5)</u> members to be appointed by Council.

- (1) One (1) member shall be a resident of said district.
- (2) One (1) member shall own real property at a location within the district and operate a business at that location. This member need not be a <u>City</u> resident. If a qualified person cannot be found to fill this position, then Council may fill the position by appointing any City resident who has the requisite interest and skill.
- (3) Three (3) members shall be residents of the City who shall serve in atlarge positions. They need not live in the district.

The <u>City</u> Manager shall retain the services of a registered architect to advise the <u>Old</u> Village <u>District</u> Review Board. The <u>City</u> Manager shall confer with the <u>Old</u> Village <u>District</u> Review Board and obtain the approval of Council before making the appointments.

The first members appointed, and each succeeding member shall be appointed to specific capacities on the Board to be designated as: at-large member, resident member, business member.

The terms shall be three (3) years. However, the first members appointed shall serve respectively for the following terms: two (2) for one (1) year; two (2) for two (2) years and two (2) for three (3) years.

Vacancies on the <u>Old</u> Village <u>District</u> Review Board shall be filled for the unexpired terms of the members whose offices have become vacant in the same manner as is provided for the appointment of such members.

In the event a Board member ceases to meet his the designated qualifications of his for appointment, his the member's position membership on the Board is automatically terminated and a successor member meeting those qualifications shall be appointed by the Council. Board members serve at the pleasure of Council.

The Old Village District Review Board shall make recommendations to the Board of Zoning Appeals on applications for variance from the guidelines and purposes of this Ordinance. The Old Village District Review Board shall not have quasi-judicial jurisdiction over applications for variance.

14.14 <u>APPLICATION FOR OLD VILLAGE DISTRICT PERMIT AND INFORMATION REQUIRED</u>

(1) Application for **Old** Village District Permit:

Old Village District permits are required as set forth in Section 14.16 hereof. This Old Village District permit shall be obtained by application to the Zoning Inspector Community Development Administrator prior to commencement of construction and is required in addition to any other required zoning permit or building permit.

(2) In addition to any other information required by the **Zoning Code and related** regulations, Regulations, a scale drawing in duplicate of all affected exterior elevations accompanied by a written description setting forth details of work to be done, materials to be used, brick or accurate color photographs or facsimiles to be used in the modifications and an appropriate presentation showing such proposed details as door and window designs seven sets of scale drawings of all affected exterior elevations accompanied by a written description setting forth the scope of work to be done, materials to be used, color renderings of modifications, and an appropriate presentation of such details as door and window design shall accompany the applications.

These documents as amended and accepted shall be official. Any deviation without written permission by the Village Review Board Community Development Administrator shall be cause for revocation of all permits including any building permit. Such presentation of material shall be no larger than 18" X 24" but may contain any number of such sized units as deemed necessary by the applicant.

14.15 REVIEW PROCEDURE

(1) Application Review by Zoning Inspector Community Development Administrator:

The Zoning Inspector Community Development Administrator shall determine whether the subject property meets all other requirements of this Ordinance and shall present his or her findings to the Old Village District Review Board.

Signs meeting the provisions of Article #18, Section 18.20B, Permitted Signs (Old Village District) shall be exempt from a public hearing but shall require the written approval of the **Community Development Administrator.**

- (2) Application Review by <u>Old</u> Village <u>District</u> Review Board:
 - (a) Hearing: The <u>Old</u> Village <u>District</u> Review Board shall notify the applicant of a time and place of a hearing on the application not less than ten (10) days before the date of the hearing. The hearing shall be held thirty (30) days after delivery of the application and supportive materials to the <u>Zoning Inspector</u>. The hearing shall be held at the next available, regularly scheduled meeting not more than thirty (30) days following the delivery of the application and supportive materials to the <u>Zoning Inspector Community Development Administrator</u>. Public notice shall be required.
 - (b) Action on Application: The Board is encouraged to decide on the night of the hearing; however, it is required that
 - (1) Within seven (7) ten (10) business days after the hearing the Board shall approve, approve with modifications, or disapprove the application recommend approval, recommend approval with modifications, or recommend denial of the application. This decision recommendation shall be based upon a careful review of the record and application, considering the Guidelines and purposes of this Ordinance, and it shall comply with the following standards:
 - (A) If the Village Review Board finds that the proposed changes conform with the Guidelines then the application shall be approved recommended for approval.
 - (B) If the Village Review Board finds that the proposed changes are not in conformance with the Guidelines and that strict application of the Guidelines would <u>cause practical</u> <u>difficulty in the viable use of the property, deprive the applicant of substantially all economically viable use of the property.</u>

property, then the Village Review Board shall approve the application recommend the approval of the application with such modifications as will permit reasonable use of the property while conforming as closely with the Guidelines as prudence permits.

- (C) If the Village Review Board finds that the proposed changes are not in conformance with the Guidelines and that full conformance would not cause the applicant <u>practical</u> <u>difficulty in the</u> to lose substantially all economically viable use of the property, then it shall <u>disapprove</u> <u>recommend</u> <u>denial of</u> the application.
- (2) Written notification of such action recommendation shall be given to the Zoning Inspector Community Development Administrator, applicant, and Clerk of Council. In the case of approval recommendation of approval with modifications, a list of these modifications shall be forwarded to the Zoning Inspector Community Development Administrator and the Clerk of Council. If the applicant accepts these suggested modifications, he the applicant shall revise his the application and supportive material accordingly and file them with the Zoning Inspector Community Development Administrator who shall then issue the necessary permits.
- (3) A majority of the decision-making members of the Village Review Board shall constitute a quorum; however, no application shall be denied recommended for denial except by the affirmative vote of a majority of the entire Board.
- (4) If an application is denied <u>recommended for denial</u>, the Board shall set forth the reasons for such denial <u>recommendation</u> in writing. The <u>Village Review</u> Board shall also make recommendations as to what changes would be required for <u>recommended</u> approval.
- (5) If the application is denied <u>recommended for denial</u>, the application may be refiled upon written certification by the applicant that such resubmission is based upon modification of the proposal with the objective of:

- (A) Conforming with <u>the Old</u> Village <u>District</u> Review Board recommendations, or
- (B) Proposing significant change in conditions or other pertinent factors which were not present during the previous reviews.

Such resubmission shall be heard at the earliest convenience by the Village Review Board at the next available, regularly scheduled meeting not more than thirty (30) days following the delivery of the application and supportive materials to the Community Development Administrator or, alternatively, the application may be forwarded to the Board of Zoning Appeals ("BZA") for consideration with no modifications at the request of the applicant.

14.16 PERMIT REQUIRED

Permits are required for the following activities within the **Old** Village District:

- (1) Demolition of all or part of a main building.
- (2) Structural modifications which substantially destroy a significant exterior architectural feature of a main building constructed prior to 1945.
- (3) Construction of a new main building.
- (4) Erection of any sign regulated by Article 18, Section 18.20B of these Zoning Regulations.

14.17 APPEALS PROCEDURE

- (1) If the application is denied or approved recommended for denial or recommended for approval with modifications, the applicant may request a hearing before City Council. Second hearing before the Old Village District Review Board to show modifications complying with the Guidelines and purposes of this Ordinance or the applicant may choose to forward the application to the Board of Zoning Appeals for a public hearing. The appeal and hearing request shall be made in writing and delivered directly to the Community Development Administrator within thirty (30) days of the Old Village District Review Board's recommendation. The request for a hearing shall be made in writing directly to the Clerk of Council within 45 days of the Village Review Board's decision. The City Council may reverse, affirm, or modify the action of the Village Review Board by a simple majority vote of the entire Council.
- (2) The Board of Zoning Appeals shall schedule a public hearing at its next regularly scheduled meeting, which shall not be more than thirty (30) days

after the hearing request unless otherwise agreed by City Staff and the applicant. Council shall attempt to schedule appeal hearings at a time reasonably convenient to the applicant. Unless the applicant shall otherwise request, all appeals hearings shall be held within thirty (30) days of receipt of the hearing request.

(3) An applicant who is dissatisfied with the <u>BZA's decision</u> action of <u>Council may</u> appeal the BZA's determination to the Greene County Common Pleas Court, within thirty (30) days of the decision, pursuant to Ohio Revised Code Sections <u>2505 and 2506</u>. Have further appeal to the courts, as provided by the general law of the State of Ohio.

14.18 SPECIAL PROCEDURE FOR DEMOLITION APPLICATIONS

In its deliberations upon an application for demolition of all or part of a main building within the <u>Old</u> Village District, the <u>Old</u> Village <u>District</u> Review Board shall consider alternatives to demolition. If the Village Review Board finds cause to believe that there are feasible and prudent alternatives to demolition, then the Village Review Board shall so inform the applicant and delay its decision on the application for no more than 180 days. During the period of the delay, the applicant shall make a good faith investigation of such alternatives and submit a written report to the <u>Old</u> Village <u>District</u> Review Board. Thereafter, the <u>Village Review</u> Board shall consider the entire record and make its <u>decision recommendation to the Community Development Administrator</u>. If the <u>Old</u> Village <u>District</u> Review Board finds that there are feasible and prudent alternatives to demolition and that the proposed demolition would be contrary to the intent of this <u>O</u>erdinance, then the application shall be <u>recommended for denial</u>. Otherwise, the application shall be <u>recommended for approval</u>.

14.19 REVIEW OF REQUESTS FOR A VARIANCE

The <u>Old</u> Village <u>District</u> Review Board shall review all applications for a variance from underlying zoning regulations for properties within the district and shall advise the Board of Zoning Appeals regarding the applicability of the Guidelines. The <u>Old</u> Village <u>District</u> Review Board shall also give its recommendation regarding disposition of the request. If the request requires a permit under Section 14.16 <u>above</u> <u>hereof</u>, then that requirement is cumulative to the requirement for a variance, and work may not proceed until approvals have been received from <u>both</u> the Village Review Board and the Board of Zoning Appeals.

GUIDELINES FOR THE OLD VILLAGE <u>DISTRICT</u> BELLBROOK, OHIO

1.0 Introduction

The "Old Village of Bellbrook", which is also known as the Old Village District, is one of the most attractive villages in southwestern Ohio, an identity which has been important in attracting new residents and businesses. Facing rapid growth and expansion possibilities, the City of Bellbrook wants to retain that identity while encouraging a strong, healthy economy and a secure, desirable place to live. Through the review process embodied in Article 14 of the Zoning Code and these <u>G</u>guidelines, the <u>e</u>City accomplishes both goals. Bellbrook has maintained a review process since adopting the 1972 Zoning Code; these guidelines are part of the updating of those original zoning provisions.

The following <u>gG</u>uidelines are the interpretive explanation and the decision-making guide for Bellbrook Zoning Code, Article 14. They are designed to assist owners, residents, architects, interested citizens, and the <u>Old</u> Village <u>District</u> Review Board <u>in</u> determin<u>inge</u> the appropriateness and the direction of the physical growth in a very special district: the Old Village <u>of Bellbrook</u> District. Important to these goals is the proper maintenance of existing properties, guidance in new construction, and the coherent development of the district.

2.0 The Old Village of Bellbrook District

The Old Village District is a distinct geographical area (see map) set roughly between Little Sugar Creek, the flood plains, and the northern hill. This district contains a harmonious combination of 1800's and post-World War II development that forms the unique character of Bellbrook.

3.0 The Old Village District Review Board

The <u>Old Village District Review</u> Board <u>("OVDRB")</u> consists of five (5) people who are convened to serve two general functions:

- (1) provide guidance and assistance in the development of the district and the accomplishment of the purposes listed in Article 14; , and
- 2) preserve the historic architecture found within the district. The Village Review Board ODVRB is distinct from the Planning and Zoning Commission Board and the Board of Zoning Appeals ("BZA") which serve other purposes. Both the OVDRB and the City's Zoning Inspector Community Development Administrator ("CDA") are ready with help and assistance for all property owners.

4.0 Certificate of Appropriateness

The <u>Guidelines adopted by Council in conjunction with Article 14 of the Zoning Code</u>
Old Village Ordinance requires the issuance of a Certificate of Appropriateness prior to
undertaking certain types of activities within the Old Village District. The <u>Zoning Inspector</u>
<u>Community Development Administrator</u> makes an initial determination of the necessity for a
Certificate of Appropriateness at the time that an application is made for a zoning permit. If the
<u>Zoning Inspector Community Development Administrator</u> believes that a Certificate of
Appropriateness is required, then <u>the Community Development Administrator</u> will provide the
applicant with <u>the requisite form(s)</u> a form VRB-1 (See Appendix for Copies of Forms VRB-1
and VRB-2) and will explain the requirements. The Certificate <u>of Appropriateness</u> is required for
the following activities in the Old Village District:

- Demolition of all or part of a main building.
- Structural modifications which substantially destroy a significant exterior architectural feature of a main building constructed prior to 1945.
- Construction of a new main building.
- Erection of any sign regulated by Article 18.20B of the Zoning Regulations.

4.1 <u>Demolition</u>

Under Section 19.01 of the Zoning Code, no demolition is permitted within the City of Bellbrook without a zoning permit. When an application for such a zoning permit indicates that the property is located within the Old Village District, the Zoning Inspector Community Development Administrator will inform the applicant that a Certificate of Appropriateness is required from the Old Village District Review Board.

It is the intent of these Guidelines and the related provisions of the Zoning Code Old Village Ordinance to discourage demolitions in the Old Village District when possible. If the Old Village District Review Board finds cause to believe that there are feasible and prudent alternatives to demolition, then the ODVRB will recommend to the Community Development Administrator that the request for a Certificate of Appropriateness will be denied. The Old Village District Review Board and/or the Community Development Administrator may consult with an appraiser, a building contractor, and/or other professionals person, for information required in making this its-determination. The fForm(s) that the applicant must submit to the Community Development Administrator when a Certificate of Appropriateness is required VRB-2 is available as an optional form to may assist the Old Village District Review Board in making its recommendation that regard. If the forms and information submitted by the applicant in connection with the application for a Certificate of Appropriate supplied on the Form VRB-2 indicates suggest that the property could be repaired and leased at an annual rent which would produce a gross return of ten percent (10%) on the owner's investment, then the Old

Village <u>District</u> Review Board may conclude that there is at least one feasible and prudent alternative to demolition.

If the <u>Old</u> Village <u>District</u> Review Board determines that there are feasible and prudent alternatives to demolition, it will so inform the applicant <u>accordingly</u> and will afford the applicant a period of 180 days to file a written response. Upon receiving that response, the <u>Old</u> Village <u>District</u> Review Board shall reconsider the matter based upon the entire record then on file. If the <u>Old</u> Village <u>District</u> Review Board makes a final determination that there are no feasible and prudent alternatives to demolition, then it will inquire into any new use to which the applicant may propose to put <u>to</u> the property following demolition. If the <u>Old</u> Village <u>District</u> Review Board determines that the objectives of Article 14 of the Zoning Code would be served best by a conversion of the land to the proposed <u>new</u> use, then it may <u>approve recommend</u> demolition, subject to the posting of a bond in such amount as the <u>Old</u> Village <u>District</u> Review Board deems adequate for ensuring that the conversion will occur.

4.2 Structural Modifications

A Certificate of Appropriateness is required for structural modifications of main buildings constructed in the Old Village District prior to 1945, if a significant exterior architectural feature will be substantially destroyed. At the time of an application for a zoning permit, the Zoning Inspector Community Development Administrator will make an initial determination as to whether a Certificate of Appropriateness is required. If it is deemed to be required, then the Zoning Inspector Community Development Administrator will provide the requisite form(s) to the applicant issue a Form VRB—I for completion by the applicant. Issuance of the zoning permit will be withheld pending approval of the Certificate of Appropriateness.

The <u>Old</u> Village <u>District</u> Review Board will <u>only recommend</u> approv<u>ale</u> <u>of</u> the request for <u>structural modifications</u> only in case <u>if</u> it finds that a denial <u>of the request would likely cause practical difficulty in applicant's deprive the applicant of substantially all economically viable use of the property. In such a case the <u>Old</u> Village <u>District</u> Review Board will <u>recommend</u> approv<u>ale</u> <u>of</u> the request with such conditions as it believes appropriate for minimizing the loss to the Old Village <u>District</u> Review Board finds that the proposed change would not substantially destroy a significant exterior architectural feature of a main building, it will find that a Certificate of Appropriateness is not required.</u>

A modification to a window or door which would cause the area of the opening to vary by more than ten percent (10%) from its original construction size will be deemed to substantially destroy a significant exterior architectural feature and therefore requires a Certificate of Appropriateness. It will be assumed that the original size was the same as the size at present unless there is evidence to show otherwise.

If the <u>Old</u> Village <u>District</u> Review Board should <u>recommend refusal of</u> refuse a request for a structural modification, it **shall notify the Community Development Administrator**

<u>accordingly</u>, and the Community Development Administrator shall provide the applicant with a written finding describing the feature involved and stating how the proposed modification would destroy that feature.

4.3 Construction of a New Main Building

Upon receipt of a request for a zoning permit for construction of a new main building in the Old Village District, the Zoning Inspector Community Development Administrator will furnish the applicant with the requisite form(s) a Form VRB—I and a copy of these Gguidelines. The applicant will be advised regarding the mandatory standards of Section 14.11 of the Bellbrook Zoning Code, (as set forth in Section 4.3.1 below hereof, as well as and the voluntary guidelines set forth in Section 4.3.2 below. If the applicant's request meets the mandatory standards, then the Old Village District Review Board shall recommend will approvale. the Certificate of Appropriateness.

4.3.1 Mandatory Standards

The general architectural character of Bellbrook is Victorian, and the standards are aimed at preserving existing examples and establishing parameters which will allow acceptable and pleasing contemporary interpretation of the standards.

1. Structural Proportions

Bellbrook architecture is characterized by two_story buildings with tall, narrow doors and windows and in earlier examples slightly squatter proportions. New buildings shall conform generally to the proportions of adjoining buildings and all buildings, either new or remodeled shall have windows and doors (or] of proportions similar to adjoining buildings. Roof slopes shall also be similar to adjoining buildings.

2. Foundations

Foundation walls of the mid-19th century were of local limestone. Acceptable foundation materials shall be limited to limestone, brick, poured concrete, or other material which is similar in appearance to the original foundation material.

3. Exterior Walls

The exterior walls of the period were of brick or narrow wood siding. Wall surface construction shall employ these materials or a modem construction material which creates the same effect.,—Cosmetic treatment using such materials as asphalt shingles which simulate brick or stone are prohibited.

4. Roof Materials

2024-O-6 Exhibit A

Roofing materials may consist of metal, slate, asphaltic shingles, or wood shingles. Prohibited materials include aluminum shingles formed to look like wood, asphalt roll roofing, and simulated cosmetic materials.

5. Chimneys

Chimneys and flues shall be of brick or stucco masonry or other material which is compatible with the architecture of the period. Metal chimneys simulating masonry construction are prohibited.

6. Windows and Door Lintels

Original lintels were stone, wood, or brick. Where feasible, lintels shall be restored, or compatible stone, brick or wood shall be used in remodeling, additions, or new construction.

7. Porches and Outbuildings

The construction of porches or outbuildings which do not conform to these standards or to the character of the main building are prohibited.

4.3.2 Voluntary Compliance

Owners of properties within the Old Village District are encouraged to observe the following voluntary guidelines when building or remodeling.

1. Scale.

The buildings within the Old Village Historic District are small to medium-sized residential, institutional, and commercial structures. Bellbrook, as a particularly pedestrian community, needs to maintain the sizing of building elements that enhances the "human" scale of the district. The scale of new work and its constituent parts should be compatible with the original buildings of the town and the rest of the district.

2. Massing

The buildings within the district are situated on small to medium-sized lots. New construction and work should be compatible with the existing forms found in the older buildings. Care should be taken to avoid crowding too large of a building on the lot when the aesthetic elements of orientation and spacing will be sacrificed. A similar ratio of building footprint to lot area should be maintained from property to property. The wall surface/opening ratio found in older buildings is encouraged and desirable. Glass curtain walls ("picture windows" or "display windows") should be avoided as they detract from the character of the district. Similarly, large, flat walls which are unbroken by openings or

setbacks/articulations on the front facade are not desirable and are much less likely to enhance the characteristics of the district.

3. Height

The height of an addition should not exceed the height of the original building, and usually, additions should be lower in height than the main building. The height of a newly constructed building should be comparable to the height of existing! adjacent buildings. New buildings constructed in undeveloped areas should complement and be relatively consistent with the height of existing buildings in the district such that the character and appearance of the district is maintained and extended.

4. Setback

The setback of new buildings or new work should be comparable to the streetscape setback of the original adjacent buildings. For example, if the building on lot A has a setback of five (5) feet from the sidewalk and the building on lot C has a setback of fifteen (15) feet from the sidewalk, a new building constructed on lot B should be set either: (1) equal to that of lot A; , or (2) equal to that of lot C; , or 3) between five (5) and fifteen (15) feet from the sidewalk. The building on lot B should not be set directly at the sidewalk line or at a distance greater than fifteen (15) feet from the sidewalk. The main entrance of a building should "face" and open onto the main thoroughfare; hHistorically, buildings in Bellbrook were accessed from the streets, therefore, buildings which cannot be oriented to the main street are discouraged.

5. Architectural Detail

New buildings should use architectural details that are similar to surrounding buildings, or which were featured on buildings in Bellbrook historically. Basically, architectural details need to have a historic reason for being used, whether they are being added to an existing building or are part of new construction. In some cases, an old photograph of the property in question will supply the necessary information about what details were appropriate to the building. In other cases, an owner will have to refer to photographs of other buildings to determine what elements will maintain a sense of continuity with the surrounding architecture.

6. Materials

Building materials should be of similar detailing, texture, scale, and appearance to those used historically on a particular building or within the district. Most of the <u>Dd</u>istrict's buildings are wood frame with wooden siding. The use of such natural materials is preferred and should be encouraged <u>- hH</u> owever, where natural appearing materials, such as vinyl or aluminum siding, are the only prudent alternative for the landowner, they may be used. Generally, if the materials appear to be synthetic or are highly reflective, they are

inappropriate in the district. When possible, masonry/brick colors should be matched to that existing on a building. Siding can cause problems (mildew, water, damage) even though the aesthetic problems are hidden. When only a portion of a wall needs to be resided, materials matching the original should be used, instead of covering the whole building in new siding.

7. Integrity of the District and its /bBuildings

Compatibility of new work with the original work is highly desirable. New work should look new; it should not pretend to be historic. At the same time, it should complement, not contradict, the existing structures around it. Where an addition meets an existing building, the connection should be carefully designed so that the new work does not detract from what was already there (e.g., Example: a new addition to an existing brick building may use vinyl siding, doors, and windows). However, a skylight on the front of the addition would be inappropriate (e.g., Example: if original openings are filled in, the outline of the original opening should remain apparent by setting new in-fill material back from the surface and leaving original sills and lintels in place. Door openings should be treated similarly.

8. New Construction—Additions.

In general, additions should follow the <u>G</u>guidelines above relating to scale, massing, height, detail, materials, and integrity. The design of an addition should respond or reflect the architectural style of the original building. The design should also respond, in a general way, to surrounding buildings. The additions should respect the original and not overpower it in any way. The connection between <u>an</u> addition and <u>the</u> original <u>structure</u> should be designed so that it does not detract from either structure. Significant or prominent architectural features of the original building should not be destroyed, removed, or obscured by the addition.

9. New Construction—Infill

The general aim of the <u>G</u>guidelines for new construction is to encourage compatibility with the character and quality found in 19th and early 20th century buildings of the district. Infill construction projects should follow the guidelines above relating to scale, massing, height, detail, materials, and integrity.

10. Site Improvements—Fences

The district is characterized by open space and landscaping. Constricting the open space by erecting fences is discouraged. Privacy fences are not characteristic of the district and are particularly offensive. If a privacy fence is necessary, it should **be in compliance with all City ordinances and** be contained to the rear of the principal building on the property. All new fences should not extend beyond the front edge of the principal building. Existing

retaining walls and fences should be repaired and retained whenever possible. All fences should consider style, color, massing, and the other considerations listed above. Historically, Bellbrook was not developed as or with "gated" properties. The complete enclosure of a property or grouping of properties should be avoided.

11. Site Improvement—Landscaping

The district is characterized by landscaping and green space. Reducing green space by adding additional pavement for driveways or parking areas should be limited whenever possible. Parking areas in front yards should be permitted in extreme situations only. New driveways and parking areas should respect existing land and street contours or minimize the view of parked cars. Landscaping design should be compatible with the surrounding buildings and landscape elements. Landscaping, special lighting, seating, and decorative paving should be designed to be simple and complimentary to the district. Antique or historic reproductions may be used with discretion. Mature trees should be retained as should other significant features, such as large rocks or groundswells.

12. Alterations—Doors and Windows

Possibly the most important features of any building are the doors and windows. The size and location of openings are an essential part of the overall design and an important element of architectural styling. Original openings should not be altered. Original doors and window sashes should be repaired rather than replaced, whenever possible. When replacement is necessary, the new door or window should match the original in size and style as closely as possible. Metal or plastic window frames are generally unacceptable unless they are painted. Screens and storm windows should be as inconspicuous as possible. Metal storm windows or doors are not appropriate unless they are painted.

13. Alterations—Roofs

The existing roof line and architectural features which give the building its character (towers, roof shape, dormers, cornices, brackets, and chimneys) should be retained whenever possible. The addition of features such as vents, skylights, decks, and roof top utilities should be avoided or should be inconspicuously placed and screened where necessary. Generally, rolled roofing, built-up tar and gravel, plastic, or fiberglass roofing material is inappropriate. However, on flat or low-pitched roofs which are not visible from public areas, these roof materials may be considered. Where original roofing materials are existent, efforts should be made to retain as much as possible if repairs will meet the needs of the owner.

14. Alterations—Building Details

Architectural features such as window hoods, cornices, brackets, bay windows, door surrounds (side lights), railings, shutters, porches, and other ornamental elements should

be preserved. Porches should not be enclosed in a permanent fashion. Missing materials should be replicated and replaced if possible; however, it is better for a building to be missing one decorative element than to remove those remaining for the sake of appearances. If replacement of an element with new is necessary, every effort should be made to match the original as closely as practicable. These kinds of distinctive features help identify and distinguish the character of the Old Village **District** of Bellbrook.

15. Alterations—Utility Installation

The installation of utility and mechanical systems, such as water or gas meters, antennas and central air conditioning units should be inconspicuously placed, avoiding installation on the street façade whenever possible. Television reception antennas and satellite dishes should be located where they are not visible, on the front façade or other public areas. Mechanical equipment on the ground should be screened with a fence or plant materials or housed in a structure that is in harmony with the surroundings. Mechanical equipment attached to the side or roof of a building should be kept as low as possible and covered, or painted, to blend with the background. Wall or window air conditioning units should not be placed on the street façade if possible.

4.4 Signs

The Village Review Board reviews all requests for erection of new signs within the Old Village District. The Board applies the standards Signs meeting the provisions of Article #18., Section-18.20B, Permitted Signs (Old Village District), shall be exempt from a public hearing but shall require written prior approval of the Community Development Administrator. in its deliberations and approves a Certificate of Appropriateness when those standards have been met.

4.4.1 **Voluntary Guidelines for Signs**

- 1. Signs should be designed for clarity, legibility, and compatibility with the building or property on which they are located. They should be neat and "professional" in appearance.
- 2. Signs should be located above the storefront, on storefront windows or on awnings. Signs should not cover any architectural features. They should capitalize on the special character of the building and reflect the nature of the business contained therein. (Billboards, internally illuminated signs, and temporary illuminated signs are prohibited by the Zoning Code. ordinance.
- 3. Free-standing signs should not be sized in such a way as to obstruct views and detract from the character of the district. Signs made of natural materials (such as wood) are

Item C.Section 8, Item

2024-O-6 Exhibit A

encouraged; plastic and other synthetic materials are inappropriate; and . Sstandardized pole signs should be avoided.

4. Other outside attachments to the building, such as light fixtures, should <u>be</u> compatible. Ultra-contemporary and bright lights directly against the building surface are less desirable. Distinctive lights which stand away from the building wall and are only as bright as necessary are preferred.

ARTICLE 15 FLOOD DAMAGE PREVENTION

15.01 <u>GENERAL PROVISIONS</u>

- (A) Statutory Authorization. Article XVIII, Section 3, of the Ohio Constitution grants municipalities the legal authority to adopt land use and control measures for promoting the health, safety, and general welfare of its citizens. Therefore, the City Council of the City of Bellbrook, State of Ohio, does ordain as follows:
- (B) Findings of Fact. The City of Bellbrook has special flood hazard areas that are subject to periodic inundation which may result in loss of life and property, health and safety hazards, disruption of commerce and governmental services, extraordinary public expenditures for flood protection and relief, and impairment of the tax base. Additionally, structures that are inadequately elevated, floodproofed, or otherwise protected from flood damage also contribute to flood loss. In order to minimize the threat of such damages and to achieve the purposes hereinafter set forth, these regulations are adopted.
- (C) Statement of Purpose. It is the purpose of these regulations to promote the public health, safety, and general welfare, and to:
 - (1) Protect human life and health;
- (2) Minimize expenditure of public money for costly flood control projects;
 - (3) Minimize the need for rescue and relief efforts associated with flooding and generally undertaken at the expense of the general public;
 - (4) Minimize prolonged business interruptions;
 - (5) Minimize damage to public facilities and utilities such as water and gas mains, electric, telephone, and sewer lines, streets, and bridges located in areas of special flood hazard;
 - (6) Help maintain a stable tax base by providing for the proper use and development of areas of special flood hazard so as to protect property and minimize future flood blight areas;
 - (7) Ensure that those who occupy the areas of special flood hazard assume responsibility for their actions;
 - (8) Minimize the impact of development on adjacent properties within and near flood prone areas;
 - (9) Ensure that flood storage and conveyance functions of the floodplain are maintained;
 - (10) Minimize the impact of development on the natural, beneficial values of the floodplain;
 - (11) Prevent floodplain uses that are either hazardous or environmentally incompatible; and
 - (12) Meet community participation requirements of the National Flood Insurance Program.

(D) Methods of Reducing Flood Loss

In order to accomplish its purposes, these regulations include methods and provisions for:

- (1) Restricting or prohibiting uses which are dangerous to health, safety, or property due to water hazards, or which result in damaging increases in flood heights or velocities;
- (2) Requiring that uses vulnerable to floods, including facilities which serve such uses, be protected against flood damage at the time of initial construction;
- (3) Controlling the alteration of natural flood plains, stream channels, and natural protection barriers, which help accommodate or channel flood waters;
- (4) Controlling filling, grading, dredging, excavating, and other development which may increase flood damage; and
- (5) Preventing or regulating the construction of flood barriers, which will unnaturally divert flood waters or which may increase flood hazards in other areas.

(E) Lands to Which these Regulations Apply

These regulations shall apply to all areas of special flood hazard within the jurisdiction of the City of Bellbrook as identified in Section 15.01(F), including any additional areas of special flood hazard.

(F) Basis for Establishing the Areas of Special Flood Hazard

For the purposes of these regulations, the following studies and/or maps are adopted:

- (1) Flood Insurance Study, Greene County, Ohio and Incorporated Areas, and Flood Insurance Rate Map, Greene County, Ohio, and Incorporated Areas, both effective March 8, 2022.
- (2) Other studies and/or maps, which may be relied upon for establishment of the flood protection elevation, delineation of the 100-year floodplain, floodways, or delineation of other areas of special flood hazard.

- (3) Any hydrologic and hydraulic engineering analysis authored by a registered Professional Engineer in the State of Ohio which has been approved by the City of Bellbrook as required by Section 15.04 (C) Subdivisions and Other New Developments.
 - (a) Any revisions to the aforementioned maps and/or studies are hereby adopted by reference and declared to be a part of these regulations. Such maps and/or studies are on file at 15 East Franklin Street, Bellbrook, OH 45305.

(4) Overlay District Designation

- (a) The areas of special flood hazard identified [on the Official Zoning Map] shall be shown as an overlay district. The overlay district shall be designated as a Flood Hazard (FH) District. Within the Flood Hazard District there shall be a Floodway Component. The floodway portion of the Flood Hazard District shall adhere to all requirements for the areas of special flood hazard, as well as those sections which specifically address the floodway.
- (b) The overlay FH District shall be superimposed over the Official Zoning Map. The underlying zoning district as shown on the Official Zoning Map shall hereafter be called the base district. Uses and minimum requirements shall be determined by the base district. However, if the provisions governing the overlay district are stricter than those of the base district, the provisions of this Ordinance shall supersede those of the base district.

(5) District Boundary Changes

(a) The delineation of the Flood Hazard District may be revised by the City of Bellbrook where natural or man-made changes have occurred and/or more detailed studies conducted or undertaken by a federal, state, local, or other qualified agency or where individual documents support such changes. However, prior to any such change, approval must be obtained from the Federal Emergency Management Agency.

(G) Abrogation and Greater Restrictions

(1) These regulations are not intended to repeal any existing ordinances including subdivision regulations, zoning, or building codes. In the event of a conflict between these regulations and any other ordinance, the more restrictive shall be

followed. These regulations are not intended to repeal, abrogate, or impair any existing easements, covenants, or deed restrictions. However, where this **O** or overlap, whichever imposes the stronger restrictions shall prevail.

(H) Interpretation

In the interpretation and application of these regulations, all provisions shall be:

- (1) Considered as minimum requirements;
- (2) Liberally construed in favor of the governing body; and
- (3) Deemed neither to limit nor repeal any other powers granted under state statutes. Where a provision of these regulations may be in conflict with a state or federal law, such state or federal law shall take precedence over these regulations.

(I) Warning and Disclaimer of Liability

(1) The degree of flood protection required by these regulations is considered reasonable for regulatory purposes and is based on scientific and engineering considerations. Larger floods can and will occur on rare occasions. Flood heights maybe increased by man-made or natural causes. These regulations do not imply that land outside the areas of special flood hazard or uses permitted within such areas will be free from flooding or flood damage. These regulations shall not create liability on the part of the City or Bellbrook, any officer or employee thereof, or the Federal Emergency Management Agency, for any flood damage that results from reliance on these regulations, or any administrative decision lawfully made thereafter.

(J) Severability

(1) Should any section or provision of these regulations be declared by the courts to be unconstitutional or invalid, such decision shall not affect the validity of the regulations as a whole, or any part thereof, other than the part so declared to be unconstitutional or invalid.

15.02 <u>DEFINITIONS</u>

Unless specifically defined below, words or phrases used in this $\underline{\mathbf{O}}$ or dinance shall be interpreted so as to give them the meaning they have in common usage and to give this $\underline{\mathbf{O}}$ ordinance the most reasonable application.

(A) Accessory Structure: A structure on the same lot with, and of a nature customarily incidental and subordinate to, the principal structure.

- (B) <u>Appeal</u>: A request for a review of the floodplain administrator's interpretation of any provisions of these regulations or a request for a variance.
- (C) <u>Base Flood</u>: The flood having a one percent chance of being equaled or exceeded in any given year. The base flood may also be referred to as the 1% chance annual flood or one-hundred (100) year flood.
- (D) <u>Base (100-Year) Flood Elevation (BFE)</u>: The water surface elevation of the base flood in relation to a specified datum, usually the National Geodetic Vertical Datum of 1929 or the North American Vertical Datum of 1988, and usually expressed in Feet Mean Sea Level (MSL). In Zone AO areas, the base flood elevation is the lowest adjacent natural grade elevation plus the depth number (from 1 to 3 feet).
- (E) <u>Basement</u>: Any area of the building having its floor subgrade (below ground level) on all sides.
- (F) Development: Any manmade change to improved or unimproved real estate, including but not limited to buildings or other structures, mining, dredging, filling, grading, paving, excavation or drilling operations, or storage of equipment materials.
- (G) <u>Enclosure Below the Lowest Floor</u>: See "Lowest Floor."
- (H) Executive Order 11988 (Floodplain Management): Issued by President Carter in 1977, this order requires that no federally assisted activities be conducted in or have the potential to affect identified special flood hazard areas, unless there is no practicable alternative.
- (I) <u>Federal Emergency Management Agency</u> (FEMA): The agency with the overall responsibility for administering the National Flood Insurance Program.
- (J) Fill: A deposit of earth material placed by artificial means.
- (K) <u>Flood or Flooding</u>: A general and temporary condition of partial or complete inundation of normally dry land areas from:
 - (1) The overflow of inland or tidal waters; and/or
 - (2) The unusual and rapid accumulation or runoff of surface waters from any source.
- (L) <u>Flood Hazard Boundary Map (FHBM)</u>: Usually the initial map, produced by the Federal Emergency Management Agency or U.S. Department of Housing

- and Urban Development, for a community depicting approximate special flood hazard areas.
- (M) Flood Insurance Rate Map (FIRM): An official map on which the Federal Emergency Management Agency or the U.S. Department of Housing and Urban Development has delineated the areas of special flood hazard.
- (N) <u>Flood Insurance Risk Zones</u>: Zone designations on FHBMs and FIRMs that indicate the magnitude of the flood hazard in specific areas of a community. Following are the zone definitions:
 - (1) Zone A: Special flood hazard areas inundated by the 100-year flood in any given year; base flood elevations are not determined.
 - (2) Zone A1-30 and Zone AE: Special flood hazard areas inundated by the 100-year flood in any given year; base flood elevations are determined.
 - (3) Zone AO: Special flood hazard areas inundated by the 100-year flood in any given year; with flood depths of 1 to 3 feet (usually sheet flow on sloping terrain); average depths are determined.
 - (4) Zone AH: Special flood hazard areas inundated by the 100-year flood in any given year; flood depths of 1 to 3 feet (usually areas of ponding); base flood elevations are determined.
 - (5) Zone A99: Special flood hazard areas inundated by the 100-year flood to be protected from the 100-year flood by a Federal flood protection system under construction; no base flood elevations are determined.
 - (6) Zone B and Zone X (shaded): Areas of 500-year flood; areas subject to the 100-year flood with average depths of less than 1 foot or with contributing drainage area less than 1 square mile; and areas protected by levees from the base flood.
 - (7) Zone C and Zone X (unshaded): Areas determined to be outside the 500-year floodplain.
- (O) <u>Flood Insurance Study (FIS)</u>: The official report in which the Federal Emergency Management Agency or the U.S. Department of Housing and Urban Development has provided flood profiles, floodway boundaries (sometimes shown on Flood Boundary and Floodway Maps), and the water surface elevations of the base flood.

- (P) <u>Floodproofing</u>: Any combination of structural and non-structural additions, changes, or adjustments to structures which reduce or eliminate flood damage to real estate or improved real property, water and sanitary facilities, structures, and their contents.
- (Q) <u>Flood Protection Elevation</u>: The Flood Protection Elevation, or FPE, is the base flood elevation plus two [2] feet of freeboard. In areas where no base flood elevations exist from any authoritative source, the flood protection elevation can be historical flood elevations, or base flood elevations determined and/or approved by the floodplain administrator.
- (R) <u>Floodway</u>: A floodway is the channel of a river or other watercourse and the adjacent land areas that have been reserved in order to pass the base flood discharge. A floodway is typically determined through a hydraulic and hydrologic engineering analysis such that the cumulative increase in the water surface elevation of the base flood discharge is no more than a designated height. In no case shall the designated height be more than one foot at any point within the community. The floodway is an extremely hazardous area, and is usually characterized by any of the following: Moderate to high velocity flood waters, high potential for debris and projectile impacts, and moderate to high erosion forces.
- (S) <u>Freeboard</u>: A factor of safety usually expressed in feet above a flood level for the purposes of floodplain management. Freeboard tends to compensate for the many unknown factors that could contribute to flood heights greater than the height calculated for a selected size flood and floodway conditions, such as wave action, obstructed bridge openings, debris and ice jams, and the hydrologic effect of urbanization in a watershed.
- (T) Historic structure: Any structure that is:
 - (1) Listed individually in the National Register of Historic Places (a listing maintained by the U.S. Department of Interior) or preliminarily determined by the Secretary of the Interior as meeting the requirements for individual listings on the National Register;
 - (2) Certified or preliminarily determined by the Secretary of the Interior as contributing to the historical significance of a registered historic district or a district preliminarily determined by the Secretary to qualify as a registered historic district;
 - (3) Individually listed on the State of Ohio's inventory of historic places maintained by the Ohio Historic Preservation Office; or

- (4) Individually listed on the inventory of historic places maintained by Bellbrook's historic preservation program, which program is certified by the Ohio Historic Preservation Office.
- (U) <u>Hydrologic and hydraulic engineering analysis</u>: An analysis performed by a professional engineer, registered in the State of Ohio, in accordance with standard engineering practices as accepted by FEMA, used to determine flood elevations and/or floodway boundaries.
- (V) <u>Letter of Map Change (LOMC)</u>: A Letter of Map Change is an official FEMA determination, by letter, to amend or revise effective Flood Insurance Rate Maps, Flood Boundary and Floodway Maps, and Flood Insurance Studies. LOMCs are broken down into the following categories:
 - (1) <u>Letter of Map Amendment (LOMA)</u>: A revision based on technical data showing that a property was incorrectly included in a designated special flood hazard area. A LOMA amends the current effective Flood Insurance Rate Map and establishes that a specific property is not located in a special flood hazard area.
 - (2) <u>Letter of Map Revision (LOMR)</u>: A revision based on technical data that, usually due to manmade changes, shows changes to flood zones, flood elevations, floodplain and floodway delineations, and planimetric features. One common type of LOMR, a LOMR-F, is a determination concerning whether a structure or parcel has been elevated by fill above the base flood elevation and is, therefore, excluded from the special flood hazard area.
 - (3) Conditional Letter of Map Revision (CLOMR): A comment by FEMA regarding a proposed project that would, upon construction, affect the hydrologic or hydraulic characteristics of a flooding source and thus result in the modification of the existing regulatory floodway, the effective base flood elevations, or the special flood hazard area. A CLOMR does not amend or revise effective Flood Insurance Rate Maps, Flood Boundary and Floodway Maps, or Flood Insurance Studies.
- (W) <u>Lowest Floor</u>: The lowest floor of the lowest enclosed area (including basement) of a structure. This definition <u>excludes</u> and "enclosure below the lowest floor" which is an unfinished or flood resistant enclosure usable solely for parking of vehicles, building access or storage, in an area other than a basement area, provided that such enclosure is built in accordance with the

- applicable design requirements specified in these regulations for enclosures below the lowest floor.
- (X) Manufactured home: A structure, transportable in one or more sections, which is built on a permanent chassis and is designed for use with or without a permanent foundation when connected to the required utilities. The term "manufactured home" does not include a "recreational vehicle". For the purposes of these regulations, a manufactured home includes manufactured homes and mobile homes as defined in Chapter 4781 of the Ohio Revised Code.
- (Y) Manufactured home park: As specified in the Ohio Adm. Code 4781-12-01(K), a manufactured home park means any tract of land upon which three or more manufactured homes, used for habitation are parked, either free of charge or for revenue purposes, and includes any roadway, building, structure, vehicle, or enclosure used or intended for use as part of the facilities of the park. A tract of land that is subdivided and the individual lots are not for rent or rented, but are for sale or sold for the purpose of installation of manufactured homes on the lots, is not a manufactured home park, even though three or more manufactured homes are parked thereon, if the roadways are dedicated to the local government authority. Manufactured home park does not include any tract of land used solely for the storage or display for sale of manufactured homes.
- (Z) Mean sea level: For purposes of the National Flood Insurance Program, the National Geodetic Vertical Datum (NGVD) of 1929 or other datum, to which base flood elevations shown on a community's Flood Insurance Rate Map are referenced.
- (AA) National Flood Insurance Program (NFIP): The NFIP is a Federal program enabling property owners in participating communities to purchase insurance protection against losses from flooding. This insurance is designed to provide an insurance alternative to disaster assistance to meet the escalating costs of repairing damage to buildings and their contents caused by floods. Participation in the NFIP is based on an agreement between local communities and the Federal government that states if a community will adopt and enforce floodplain management regulations to reduce future flood risks to all development in special flood hazard areas, the Federal government will make flood insurance available within the community as a financial protection against flood loss.
- (BB) New construction: Structures for which the "start of construction" commenced on or after the effective date of a floodplain regulation adopted by Bellbrook and includes any subsequent improvements to such structures. For the purposes of determining insurance rates,

- structures for which the "start of construction" commenced on or after the effective date of an initial FIRM [June 1, 1977] and includes any subsequent improvements to such structures.
- (CC) <u>Person</u>: <u>Includes aAny</u> individual or group of individuals, corporation, partnership, association, or any other entity, including state and local governments and agencies. An agency is further defined in the Ohio Rev. Code §111.15(A)(2) as any governmental entity of the state and includes, but is not limited to, any board, department, division, commission, bureau, society, council, institution, state college or university, community college district, technical college district, or state community college. "Agency" does not include the general assembly, the controlling board, the adjutant general's department, or any court.
- (DD) <u>Recreational vehicle</u>: A vehicle which is (1) built on a single chassis, (2) 400 square feet or less when measured at the largest horizontal projection, (3) designed to be self-propelled or permanently towable by a light duty truck, and (4) designed primarily not for use as a permanent dwelling but as temporary living quarters for recreational, camping, travel, or seasonal use.
- (EE) <u>Registered Professional Architect</u>: A person registered to engage in the practice of architecture pursuant to Ohio Rev. Code §4703.01 and 4703.19.
- (FF) <u>Registered Professional Engineer</u>: A person registered as a professional engineer pursuant to Ohio Rev. Code Chapter 4733.
- (GG) <u>Registered Professional Surveyor</u>: A person registered as a professional surveyor pursuant to Ohio Rev. Code Chapter 4733.
- (HH) Special Flood Hazard Area or Areas of Special Flood Hazard: Also known as "Areas of Special Flood Hazard", IIt is the land in the floodplain subject to a one percent or greater chance of flooding in any given year. Special flood hazard areas are designated by the Federal Emergency Management Agency on Flood Insurance Rate Maps, Flood Insurance Studies, Flood Boundary and Floodway Maps and Flood Hazard Boundary Maps as Zones A, AE, AH, AO, A1-30, or A99. Special flood hazard areas may also refer to areas that are flood prone and designated from other federal state or local sources of data including but not limited to historical flood information reflecting high water marks, previous flood inundation areas, and flood prone soils associated with a watercourse.
- (II) <u>Start of Construction</u>: The date the building permit was issued, provided the actual start of construction, repair, reconstruction, rehabilitation, addition,

placement, or other improvement was within 180 days of the permit date. The actual start means either the first placement of permanent construction of a structure on a site, such as the pouring of slab or footings, the installation of piles, the construction of columns, or any work beyond the stage of excavation; or the placement of a manufactured home on a foundation. Permanent construction does not include land preparation, such as clearing, grading, and filling; nor does it include the installation of streets and/or walkways; nor does it include excavation for a basement, footings, piers, or foundations or the erection of temporary forms; nor does it include the installation on the property of accessory buildings, such as garages or sheds not occupied as dwelling units or not part of the main structure. For a substantial improvement, the actual start of construction means the first alteration of any wall, ceiling, floor, or other structural part of a building, whether or not that alteration affects the external dimensions of a building.

- (JJ) <u>Structure</u>: A walled and roofed building, manufactured home, or gas or liquid storage tank that is principally above ground.
- (KK) <u>Substantial Damage</u>: Damage of any origin sustained by a structure whereby the cost of restoring the structure to the 'before damaged' condition would equal or exceed 50 percent of the market value of the structure before the damage occurred.
- (LL) <u>Substantial Improvement</u>: Any reconstruction, rehabilitation, addition, or other improvement of a structure, the cost of which equals or exceeds 50 percent of the market value of the structure before the "start of construction" of the improvement. This term includes structures which have incurred "substantial damage," regardless of the actual repair work performed. The term does not, however, include:
 - (1) Any project for improvement of a structure to correct existing violations of state or local health, sanitary, or safety code specifications which have been identified by the local code enforcement official and which are the minimum necessary to assure safe living conditions; or
 - (2) Any alteration of a "historic structure," provided that the alteration would not preclude the structure's continued designation as a "historic structure."
- (MM) <u>Variance</u>: A grant of relief from the standards of these regulations.
- (NN) <u>Violation</u>: The failure of a structure or other development to be fully compliant with these regulations.

15.03 ADMINISTRATION

- (A) Designation of the Floodplain Administrator: The Zoning Administrator (Community Development Administrator) is hereby appointed to administer and implement these regulations and is referred to herein as the Floodplain Administrator.
- (B) Duties and Responsibilities of the Floodplain Administrator. The duties and responsibilities of the Floodplain Administrator shall include but are not limited to:
 - (1) Evaluate applications for permits to develop in special flood hazard areas.
 - (2) Interpret floodplain boundaries and provide flood hazard and flood protection elevation information.
 - (3) Issue permits to develop in special flood hazard areas when the provisions of these regulations have been met, or refuse to issue the same in the event of noncompliance.
 - (4) Inspect buildings and lands to determine whether any violations of these regulations have been committed.
 - (5) Make and permanently keep all records for public inspection necessary for the administration of these regulations including Flood Insurance Rate Maps, Letters of Map Amendment and Revision, records of issuance and denial of permits to develop in special flood hazard areas, determinations of whether development is in or out of special flood hazard areas for the purpose of issuing floodplain development permits, elevation certificates, floodproofing certificates, variances, and records of enforcement actions taken for violations of these regulations.
 - (6) Enforce the provisions of these regulations.
 - (7) Provide information, testimony, or other evidence as needed during variance hearings.
 - (8) Coordinate map maintenance activities and FEMA follow-up.
 - (9) Conduct substantial damage determinations to determine whether existing structures, damaged from any source and in special flood hazard areas identified by FEMA, must be the development standards of these regulations.

(C) Floodplain Development Permits

(1) It shall be unlawful for any person to begin construction or other development activity including but not limited to filling, grading, construction, alteration, remodeling, or expanding any structure; or alteration of any watercourse wholly within, partially within or in contact with any identified special flood hazard area, as established in Section 1.6, until a floodplain development permit is obtained from the Floodplain Administrator. Such floodplain development permit shall show that the proposed development activity is in conformity with the provisions of these regulations. No such permit shall be issued by the Floodplain Administrator until the requirements of these regulations have been met.

(D) Application Required

- (1) An application for a floodplain development permit shall be required for all development activities located wholly within, partially within, or in contact with an identified special flood hazard area. Such application shall be made by the owner of the property or his/her authorized agent, herein referred to as the applicant, prior to the actual commencement of such construction on a form furnished for that purpose. Where it is unclear whether a development site is in a special flood hazard area, the Floodplain Administrator may require an application for a floodplain development permit to determine the development's location. Such applications shall include, but not be limited to:
 - (a) Site plans drawn to scale showing the nature, location, dimensions, and topography of the area in question; the location of existing or proposed structures, fill, storage of materials, drainage facilities, and the location of the foregoing.
 - (b) Elevation of the existing, natural ground where structures are proposed.
 - (c) Elevation of the lowest floor, including basement, of all proposed structures.
 - (d) Such other material and information as may be requested by the Floodplain Administrator to determine conformance with, and provide enforcement of these regulations.
 - (e) Technical analyses conducted by the appropriate design professional registered in the State of Ohio and submitted with an application for a floodplain development permit when applicable:
 - (i) Floodproofing certification for non-residential floodproofed structure as required in Section 15.04 (E).
 - (ii) Certification that fully enclosed areas below the lowest floor of a structure <u>not</u> meeting the design requirements of Section 15.04(D)(5) are designed to automatically equalize hydrostatic flood forces.

- (iii) Description of any watercourse alteration or relocation that the flood carrying capacity of the watercourse will not be diminished, and maintenance assurances as required by Section 15.04(I)(3).
- A hydrologic and hydraulic (iv) analysis demonstrating that the cumulative effect of proposed development, when combined with all other existing and anticipated development will not increase the water surface elevation of the base flood by more than one foot in special flood hazard areas where the Federal Emergency Management Agency has provided base flood elevations but no floodway as required by Section 15.04(I)(2).
- (v) A hydrologic and hydraulic engineering analysis showing impact of any development on flood heights in an identified floodway as required by Section 15.04 (I)(1).
- (vi) Generation of base flood elevation(s) for subdivision and other new developments as required by Section 15.04(C).
- (f) A Floodplain Development Permit Application Fee set by the Schedule of Fees adopted by the City of Bellbrook.
- (E) Review and Approval of a Floodplain Development Permit Application

(1) Review

(a) After receipt of a complete application, the Floodplain Administrator shall review the application to ensure that the standards of these regulations have been met. No floodplain development permit application shall be reviewed until all information required in Section 3.4 has been received by the Floodplain Administrator.

- (b) The Floodplain Administrator shall review all floodplain development permit applications to assure that all necessary permits have been received from those federal, state or local governmental agencies from which prior approval is required. The applicant shall be responsible for obtaining such permits as required including permits issued by the U.S. Army Corps of Engineers under Section 10 of the Rivers and Harbors Act and Section 404 of the Clean Water Act, and the Ohio Environmental Protection Agency under Section 401 of the Clean Water Act.
- (F) Inspections: The Floodplain Administrator shall make periodic inspections at appropriate times throughout the period of construction in order to monitor compliance with permit conditions.
- (G) Post-Construction Certifications Required. The following as-built certifications are required after a floodplain development permit has been issued:
 - (1) For new or substantially improved residential structures, or nonresidential structures that have been elevated, the applicant shall have a *Federal Emergency Management Agency Elevation Certificate* completed by a registered professional surveyor to record as-built elevation data. For elevated structures in Zone A and Zone AO areas without a base flood elevation, the elevation *certificate* may be completed by the property owner or owner's representative.
 - (2) For all development activities subject to the standards of Section 15.03(K)(1) a Letter of Map Revision.
 - (3) For new or substantially improved nonresidential structures that have been floodproofed in lieu of elevation, where allowed, the applicant shall supply a completed Floodproofing Certificate for Non-Residential Structures completed by a registered professional engineer or architect together with associated documentation.
- (H) Revoking a Floodplain Development Permit. A floodplain development permit shall be revocable, if among other things, the actual development activity does not conform to the terms of the application and permit granted thereon. In the event off the revocation of a permit, an appeal may be taken to the Appeals Board in accordance with Section 15.05 of these regulations.

- (I) Exemption from Filing a Development Permit. An application for a floodplain development permit shall not be required for maintenance work such as roofing, painting, and basement sealing, or for small nonstructural development activities (except for filling and grading) valued at less than \$2,500.
- (J) State and Federal Development
 - (1) Development that is funded, financed, undertaken, or preempted by state agencies shall comply with minimum NF1P criteria.
 - (2) Before awarding funding or financing or granting a license, permit, or other authorization for a development that is or is to be located within a 100-year floodplain, a state agency shall require the applicant to demonstrate to the satisfaction of the agency that the development will comply with the minimum NW criteria and any applicable local floodplain management resolution or ordinance as required by Ohio Revised Code Section 1521.13. This includes, but is not limited to:
 - (a) Development activities in an existing or proposed manufactured home park that are under the authority of the Ohio Department of Commerce and subject to the flood damage reduction provisions of the Ohio Administrative Code Section 4781:12.
 - (b) Major utility facilities permitted by the Ohio Power Siting Board under Section 4906 of the Ohio Revised Code.
 - (c) Hazardous waste disposal facilities permitted by the Hazardous Waste Siting Board under Section 3784 of the Ohio Revised Code.
 - (3) Development activities undertaken by a federal agency and which are subject to Federal Executive Order 11988 Floodplain Management.
 - (a) Each federal agency has a responsibility to evaluate the potential effects of any actions it may take in a floodplain; to ensure that its planning programs and budget request reflect consideration of flood hazards and floodplain management; and to prescribe procedures to implement the policies and requirements of EO 11988.
- (K) Map Maintenance Activities. To meet National Flood Insurance Program minimum requirements to have flood data reviewed and approved by FEMA, and to ensure that the City of Bellbrook flood

maps, studies and other data identified in Section 15.01 (F) accurately represent flooding conditions so appropriate floodplain management criteria are based on current data, the following map maintenance activities are identified:

- (1) Requirement to Submit New Technical Data
- (a) For all development proposals that impact floodway delineations or base flood elevations, the community shall ensure that technical data reflecting such changes be submitted to FEMA within six months of the date such information becomes available. These development proposals include:
 - (i) Floodway encroachments that increase or decrease base flood elevations or alter floodway boundaries;
 - (ii) Fill sites to be used for the placement of proposed structures where the applicant desires to remove the site from the special flood hazard area;
 - (iii) Alteration of watercourses that result in a relocation or elimination of the special flood hazard area, including the placement of culverts; and
 - (iv) Subdivision or other new development proposals requiring the establishment of base flood elevations in accordance with Section 15.04.
 - (b) It is the responsibility of the applicant to have technical data, required in accordance with Section 15.03 (K)(1) prepared in a format required for a Conditional Letter of Map Revision or Letter of Map Revision, and submitted to FEMA. Submittal and processing fees for these map revisions shall be the responsibility of the applicant.
 - (c) The Floodplain Administrator shall require a Conditional Letter of Map Revision prior to the issuance of a floodplain development permit for:
 - (i) Proposed floodway encroachments that increase the base flood elevation; and
 - (ii) Proposed development which increases the base flood elevation by more than one foot in riverine areas where FEMA has provided base flood elevations but no floodway.
 - (d) Floodplain development permits issued by the Floodplain Administrator shall be conditioned upon the

applicant obtaining a Letter of Map Revision from FEMA for any development proposal subject to Section 15.03 (K)(1)(a).

- (2) Right to Submit New Technical Data: The Floodplain Administrator may request changes to any of the information shown on an effective map that does not impact floodplain or floodway delineations or base flood elevations, such as labeling or planimetric details. Such a submission shall include appropriate supporting documentation made in writing by the City Manager of Bellbrook, and may be submitted at any time.
- (3) Annexation/Detachment: Upon occurrence, the Floodplain Administrator shall notify FEMA in writing whenever the boundaries of the City of Bellbrook have been modified by annexation or the community has assumed authority over an area, or no longer has authority to adopt and enforce floodplain management regulations for a particular area. In order that the City of Bellbrook's Flood Insurance Rate Map accurately represent the City of Bellbrook boundaries, include within such notification a copy of a map of the City of Bellbrook suitable for reproduction, clearly showing the new corporate limits or the new area for which the City of Bellbrook has assumed or relinquished floodplain management regulatory authority.
- (L) Data Use and Flood Map Interpretation. The following guidelines shall apply to the use and interpretation of maps and other data showing areas of special flood hazard:
 - (1) In areas where FEMA has not identified special flood hazard areas, or in FEMA identified special flood hazard areas where base flood elevation and floodway data have not been identified, the Floodplain Administrator shall review and reasonably utilize any other flood hazard data available from a federal, state, or other source.
 - (2) Base flood elevations and floodway boundaries produced on FEMA flood maps and studies shall take precedence over base flood elevations and floodway boundaries by any other source that reflect a reduced floodway width and/or lower base flood elevations. Other sources of data showing increased base flood elevations and/or larger floodway areas than are shown on FEMA flood maps and studies, shall be reasonably used by the Floodplain Administrator.
 - (3) The Floodplain Administrator shall make interpretations, where needed, as to the exact location of the flood boundaries and areas of special flood hazard. A person contesting the determination of the location of the boundary shall be given a reasonable opportunity to appeal the interpretation as provided in Section 15.05, Appeals and Variances.

- (4) Where an existing or proposed structure or other development is affected by multiple flood zones, by multiple base flood elevations, or both, the development activity must comply with the provisions of this Oerdinance applicable to the most restrictive flood zone and the highest base flood elevation affecting any part of the existing or proposed structure; or for other developments, affecting any part of the area of the development.
- (M) Use of Preliminary Insurance Flood Rate Map and/or Flood Insurance Study Data

(1) Zone A:

- (a) Within Zone A areas designated on an effective FIRM, data from the preliminary FIRM and/or FIS shall reasonably utilized as best available data.
- (b) When all appeals have been resolved and a notice of final food elevation determination has been provided in a Letter of Final Determination (LFD), BFE and floodway data from the preliminary FIRM and/or FIS shall be used for regulating development.

(2)Zones Alt, A1-30, AH, and AO:

- (a) BFE and floodway data from a preliminary FIS or FIRM restudy are not required to be used in lieu of BFE and floodway data contained in an existing FIS and FIRM. However,
 - (i) Where BFEs increase in a restudied area, communities have the responsibility to ensure that new or substantially improved structures protected. are Communities are encouraged reasonably utilize preliminary FIS or FIRM data in instances where BFEs increase and floodways are revised to ensure that the health, safety, and property of their citizens are protected.
 - (ii) Where BFEs decrease, preliminary FIS or FIRM data should not be used to regulate floodplain development until the LFD has

been issued or until all appeals have been resolved.

(b) If preliminary FIRM or FIS has designated floodways where none had previously existed, communities should reasonably utilize this data in lieu of applying the encroachment performance standard of Section 15.04(I)(2) since the data in the draft or preliminary FIS represents the best data available.

(3)Zones B, C, an X:

- (a) Use of BFE and floodway data from a preliminary FIRM or FIS are not required for areas designated as Zone B, C, or X on the effective FIRM which are being revised to Zone AE, A1-30, All, or AO. Communities are encouraged to reasonably utilize preliminary FIS or FIRM data to ensure that the health, safety, and property of their citizens are protected.
- (N) Substantial Damage Determinations. Damages to structures may result from a variety of causes including flood, tornado, wind, heavy snow, fire, *etc*. After such a damage event, the Floodplain Administrator shall:
 - (1) Determine whether the damaged structures are located in special flood hazard areas;
 - (2) Conduct substantial damage determinations for damaged structures located in special flood hazard areas; and
 - (3) Require owners of substantially damaged structures to obtain a floodplain development permit prior to repair, rehabilitation, or reconstruction.

Additionally, the Floodplain Administrator may implement other measures to assist with the substantial damage determination and subsequent repair process. These measures include issuing press releases, public service announcements, and other public information materials related to the floodplain development permits and repair of damaged structures; coordinating with other federal, state, and local agencies to assist with substantial damage determinations; providing owners of damaged structures materials and other information related to the proper repair of damaged structures in special flood hazard areas; and assist owners of substantially damaged structures with Increased Cost of Compliance insurance claims.

15.04 USE AND DEVELOPMENT STANDARDS FOR FLOOD HAZARD REDUCTION

The following use and development standards apply to development wholly within, partially within, or in contact with any special flood hazard area as established in Section 15.01 (F), 15.03 (1)(1), or 15.03.

(A) Use Regulations

- (1) Permitted Uses. All uses not otherwise prohibited in this section or any other applicable land use regulation adopted by Bellbrook are allowed provided they meet the provisions of these regulations.
- (B) Water and Wastewater Systems. The following standards apply to all water supply, sanitary sewerage and waste disposal systems in the absence of any more restrictive standard provided under the Ohio Revised Code or applicable state rules:
 - (1) All new and replacement water supply systems shall be designed to minimize or eliminate infiltration of floodwaters into the systems;
 - (2) New and replacement sanitary sewerage systems shall be designed to minimize or eliminate infiltration of flood waters into the systems and discharge from the systems into flood waters; and,
 - (3) On-site waste disposal systems shall be located to avoid impairment to or contamination from them during flooding.

(C) Subdivisions and Other New Developments

- (1) All subdivision proposals and all other proposed new development shall be consistent with the need to minimize flood damage and are subject to all applicable standards in these regulations;
- (2) All subdivision proposals and all other proposed new development shall have public utilities and facilities such as sewer, gas, electrical, and water systems located and constructed to minimize or eliminate flood damage;
- (3) All subdivision proposals and all other proposed new development shall have adequate drainage provided to reduce exposure to flood damage; and
- (4) In all areas of special flood hazard where base flood elevation data are not available, the applicant shall provide a hydrologic and hydraulic

- engineering analysis that generates base flood elevations for all subdivision proposals and other proposed developments containing at least 50 lots or S acres, whichever is less.
- (5) The applicant shall meet the requirement to submit technical data to FEMA in Section 3.11(A)(1)(d) when a hydrologic and hydraulic analysis is completed that generates base flood elevations as required by Section 15.04 (C)(4).
- (D) Residential Structures. The requirements of Section 4.4 apply to new construction of residential structures and to substantial improvements of residential structures in zones A, A1-30, AE, AO, and All, when designated on the community's effective FIRM, and when designated on a preliminary or final FIRM issued by FEMA under the circumstances provided in Section 15.03 (M).
 - (1) New construction and substantial improvements shall be anchored to prevent flotation, collapse, or lateral movement of the structure resulting from hydrodynamic and hydrostatic loads, including the effects of buoyancy. Where a structure, including its foundation members, is elevated on fill to or above the base flood elevation, the requirements for anchoring (15.04 (D)(1)) and construction materials resistant to flood damage (15.04 (D)(2)) are satisfied.
 - (2) New construction and substantial improvements shall be constructed with methods and materials resistant to flood damage.
 - (3) New construction and substantial improvements shall be constructed with electrical, heating, ventilation, plumbing and air conditioning equipment and other service facilities that are designed and/or elevated so as to prevent water from entering or accumulating within the components during conditions of flooding.
 - (4) New construction and substantial improvement of any residential structure, including manufactured homes, shall have the lowest floor, including basement, elevated to or above the flood protection elevation.
 - (5) New construction and substantial improvements, including manufactured homes, that do not have basements and that are elevated to the flood protection elevation using pilings, columns, posts, or solid foundation perimeter walls with openings to allow the automatic equalization of hydrostatic pressure may have an enclosure below the lowest floor provided the enclosure meets the following standards:
 - (a) Be used only for the parking of vehicles, building access, or storage; and

- (b) be designed and certified by a registered professional engineer or architect to automatically equalize hydrostatic flood forces on exterior walls by allowing for the entry and exit of floodwaters; or
- (c) have a minimum of two openings on different walls having a total net area not less than one square inch for every square foot of enclosed area, and the bottom of all such openings being no higher than one foot above grade. The openings may be equipped with screens, louvers, or other coverings or devices provided that they permit the automatic entry and exit of floodwaters.
- (6) Manufactured homes shall be affixed to a permanent foundation and anchored to prevent flotation, collapse or lateral movement of the structure resulting from hydrodynamic and hydrostatic loads, including the effects of buoyancy. Methods of anchoring may include, but are not limited to, use of over-the-top or frame ties to ground anchors.
- (7) Repair or rehabilitation of historic structures upon a determination that the proposed repair or rehabilitation will not preclude the structure's continued designation as a historic structure and is the minimum necessary to preserve the historic character and design of the structure, shall be exempt from the development standards of Section 15.04(D).
- (E) Nonresidential Structures. The requirements of Section 15.04 (E) apply to new construction and to substantial improvements of nonresidential structures in zones A, A1-30, AE, AO, and AH, when designated on the community's effective FIRM, and when designated on a preliminary or final FIRM issued by TEMA under the circumstances provided in Section 1503 (M).
 - (1) New construction and substantial improvement of any commercial, industrial, or other nonresidential structure shall meet the requirements of Section 15.04 (D)(1)- (3) and (5)- (7).
 - (2) New construction and substantial improvement of any commercial, industrial, or other non-residential structure shall either have the lowest floor, including basement, elevated to or above the level of the flood protection elevation; or, together with attendant utility and sanitary facilities, shall meet all of the following standards:
 - (a) Be dry floodproofed so that the structure is watertight with walls substantially impermeable to

- the passage of water to the level of the flood protection elevation;
- (b) Have structural components capable of resisting hydrostatic and hydrodynamic loads and effects of buoyancy; and,
- (c) Be certified by a registered professional engineer or architect, through the use of a *Federal Emergency Management Agency Floodproofing Certificate*, that the design and methods of construction are in accordance with Section 15.04 (E)(2)(a) and (2).
- (F) Accessory Structures. Structures that are 600 square feet or less which are used for parking and storage only are exempt from elevation or dry floodproofing standards within zones A, A1-30, AE, AO, and AH designated on the community's FIRM. Such structures must meet the following standards:
 - (1) They shall not be used for human habitation;
 - (2) They shall be constructed of flood resistant materials;
 - (3) They shall be constructed and placed on the lot to offer the minimum resistance to the flow of floodwaters;
 - (4) They shall be firmly anchored to prevent flotation;
 - (5) Service facilities such as electrical and heating equipment shall be elevated or floodproofed to or above the level of the flood protection elevation; and
 - (6) They shall meet the opening requirements of Section 15.04 (D)(5)(C);
- (G) Recreational Vehicles. Recreational vehicles on sites within zones A, A1-A30, AE, AO, or AH must meet at least one of the following standards:
 - (1) They shall not be located on sites in special flood hazard areas for more than 180 days, or
 - (2) They must be fully licensed and ready for highway use, or
 - (3) They must be placed on the site pursuant to a floodplain development permit issued under Sections 15.03(C) and 15.03(D), and meet all standards of Section 15.04(D).
- (H) Gas or Liquid Storage Tanks
 - (1) Within Zone A, A1-A-30, AE, AO, or AH, new or substantially improved above ground gas or liquid storage tanks shall be anchored to prevent floatation or lateral movement resulting from hydrodynamic and hydrostatic loads.

- (I) Assurance of Flood Carrying Capacity. Pursuant to the purpose and methods of reducing flood damage stated in these regulations, the following additional standards are adopted to assure that the reduction of the flood carrying capacity of watercourses is minimized.
 - (1) Development in Floodways
 - (a) In floodway areas, development shall cause no increase in flood levels during the occurrence of the base flood discharge. Prior to issuance of a floodplain development permit, the applicant must submit a hydrologic and hydraulic analysis, conducted by a registered professional engineer, demonstrating that the proposed development would not result in any increase in the base flood elevation; or
 - (b) Development in floodway areas causing increases in the base flood elevation may be permitted provided all of the following are completed by the applicant:
 - (i) Meet the requirements to submit technical data in Section 15.03 (K)(1);
 - (ii) An evaluation of alternatives, which would not result in increased base flood elevations and an explanation why these alternatives are not feasible;
 - (iii)Certification that no structures are located in areas that would be impacted the increased flood elevation;
 - (iv) Documentation of individual legal notices to all impacted property owners within and outside the community, explaining the impact of the proposed action on their property; and
 - (v) Concurrence of the City Manager of Bellbrook and the Chief Executive Officer of any other communities impacted by the proposed actions.
 - (2) Development in Riverine Areas with Base Flood Elevations but No Floodways
 - (a) In riverine special flood hazard areas identified by FEMA where base flood elevation data are provided but no floodways have been designated, the cumulative effect of any proposed development, when combined with all other existing and anticipated development, shall not increase the base flood elevation more than 1.0 (one) foot at any point. Prior to issuance of a floodplain development permit, the applicant must submit a hydrologic and hydraulic

- analysis, conducted by a registered professional engineer, demonstrating that this standard has been met: or.
- (b) Development in riverine special flood hazard areas identified by FEMA where base flood elevation data are provided but no floodways have been designated causing more than one foot increase in the base flood elevation may be permitted provided all of the following are completed by the applicant:
 - (i) An evaluation of alternatives which would result in an increase of one foot or less of the base flood elevation and an explanation why these alternatives are not feasible:
 - (ii) Section 15.04 (I)(1)(b), items (i) and (iii)-(v).

(2) Alterations of a Watercourse

- (a) For the purpose of these regulations, a watercourse is altered when any change occurs within its banks. The extent of the banks shall be established by a field determination of the "bankfull stage." The field determination of "bankfull stage" shall be based on methods presented in Chapter 7 of the USDA Forest Service General Technical Report RM-245, Stream Channel Reference Sites: An Illustrated Guide to Field Technique or other applicable publication available from a Federal, State, or other authoritative source. For all proposed developments that alter a watercourse, the following standards apply:
 - (i) The bankfull flood carrying capacity of the altered or relocated portion of the watercourse shall not be diminished. Prior to the issuance of a floodplain development permit, the applicant must submit a description of the extent to which any watercourse will be altered or relocated as a result of the proposed development, and certification by a registered professional engineer that the bankfull flood carrying capacity of the watercourse will not be diminished.
 - (ii) Adjacent communities, the U.S. Army Corps of Engineers, and the Ohio Department of Natural Resources, Division of Water, must be notified prior to any alteration or relocation of a watercourse. Evidence of such notification must be submitted to the Federal Emergency Management Agency.
 - (iii)The applicant shall be responsible for providing the necessary maintenance for the

altered or relocated portion of said watercourse so that the flood carrying capacity will not be diminished. The Floodplain Administrator may require the permit holder to enter into an agreement with Bellbrook specifying the maintenance responsibilities. If an agreement is required, it shall be made a condition of the floodplain development permit.

(iv)The applicant shall meet the requirements to submit technical data in Section 15.03 (1)(1)(a)(iii) when an alteration of a watercourse results in the relocation or elimination of the special flood hazard area, including the placement of culverts.

SECTION 15.05 APPEALS AND VARIANCES

(A) Appeals Board Established

- (1) The Board of Zoning Appeals as established by the City of Bellbrook shall hear and decide appeals and requests for variances from the requirements of this **O**erdinance.
- (2) Records f the Appeals Board shall be kept and filed at 15 East Franklin Street, Bellbrook, OH 45305.

(3)

(B) Powers and Duties

- (1) The Appeals Board shall hear and decide appeals where it is alleged there is an error in any order, requirement, decision or determination made by the Floodplain Administrator in the administration or enforcement of these regulations.
- (2) Authorize variances in accordance with Section 15.05 (D) of these regulations.
- (C) Appeals. Any person affected by any notice and order, or other official action of the Floodplain Administrator may request and shall be granted a hearing on the matter before the Appeals Board provided that such person shall file, within [301 days of the date of such notice and order, or other official action, a brief statement of the grounds for such hearing or for the mitigation of any item appearing on any order of the Floodplain Administrator's decision. Such appeal shall be in writing, signed by the applicant, and be filed with the Floodplain Administrator. Upon receipt of the appeal, the Floodplain Administrator shall transmit said notice and all pertinent information on which the Floodplain Administrator's decision was made to the Appeals Board. Upon receipt of the notice of appeal, the Appeals Board shall fix a reasonable time for the appeal, give notice in writing

- to parties in interest, and decide the appeal within a reasonable time after it is submitted.
- (D) Variances. Any person believing that the use and development standards of these regulations would result in practical difficulty may file an application for a valiance. The Appeals Board shall have the power to authorize, in specific cases, such variances from the standards of these regulations, not inconsistent with Federal regulations, as will not be contrary to the public interest where, owning to special conditions of the lot or parcel, a literal enforcement of the provisions of these regulations would result in unnecessary hardship.

(1) Application for a Variance

- (a) Any owner, or agent thereof, of property for which a variance is sought shall make an application for a variance by filing it with the Floodplain Administrator, who upon receipt of the variance shall transmit it to the Appeals Board.
- (b) Such application at a minimum shall contain the following information: Name, address, and telephone number of the applicant; legal description of the property; parcel map; description of the existing use; description of the proposed use; location of the floodplain; description of the variance sought; and reason for the variance request.
- (c) All applications for a variance shall be accompanied by a Variance Application Fee set in the schedule of fees adopted by the City of Bellbrook.
- (2) Notice for Public Hearing. Procedures outlined in Zoning Code Article 20 Section 20.09.
- (3) Public Hearing. At such hearing the applicant shall present such statements and evidence as the Appeals Board requires. In considering such variance applications, the Appeals Board shall consider and make findings of fact on all evaluations, all relevant factors, standards specified in other sections of these regulations and the following factors:
 - (a) The danger that materials may be swept onto other lands to the injury of others.
 - (b) The danger to life and property due to flooding or erosion damage.

- (c) The susceptibility of the proposed facility and its contents to flood damage and the effect of such damage on the individual owner.
- (d) The importance of the services provided by the proposed facility to the community.
- (e) The availability of alternative locations for the proposed use that are not subject to flooding or erosion damage.
- (0 The necessity to the facility of a waterfront location, where applicable.
- (g) The compatibility of the proposed use with existing and anticipated development.
- (h) The relationship of the proposed use to the comprehensive plan and floodplain management program for that area.
- (i) The safety of access to the property in times of flood for ordinary and emergency vehicles.
- (j) The expected heights, velocity, duration, rate of rise, and sediment transport of the floodwaters and the effects of wave action, if applicable, expected at the site.
- (k) The costs of providing governmental services during and after flood conditions, including maintenance and repair of public utilities and facilities such as sewer, gas, electrical, and water systems, and streets and bridges.
- (4) Variances shall only be issued upon:
 - (a) A showing of good and sufficient cause.
 - (b) A determination that failure to grant the variance would result in practical difficulty due to the physical characteristics of the property. Increased cost or inconvenience of meeting the requirements of these regulations does not constitute an exceptional hardship to the applicant.
 - (c) A determination that the granting of a variance will not result in increased flood heights beyond that which is allowed in these regulations; additional threats to public safety; extraordinary public expense, nuisances, fraud on or victimization of the public, or conflict with existing local laws.
 - (d) A determination that the structure or other development is protected by methods to minimize flood damages.

- (e) A determination that the variance is the minimum necessary, considering the flood hazard, to afford relief.
- (f) Upon consideration of the above factors and the purposes of these regulations, the Appeals Board may attach such conditions to the granting of variances, as it deems necessary to further the purposes of these regulations.

(5) Other Conditions for Variances

- (a) Variances shall not be issued within any designated floodway if any increase in flood levels during the base flood discharge would result.
- (b) Generally, variances may be issued for new construction and substantial improvements to be erected on a lot of one-half acre or less in size contiguous to and surrounded by lots with existing structures constructed below the base flood level, providing items in Section 15.05 (D)(3)(a) to (i) have been fully considered. As the lot size increases beyond one-half acre, the technical justification required for issuing the variance increases.
- (c) Any applicant to whom a variance is granted shall be given written notice that the structure will be permitted to be built with a lowest floor elevation below the base flood elevation and the cost of flood insurance will be commensurate with the increased risk resulting from the reduced lowest floor elevation.

(E) Procedure at Hearings

- (1) All testimony shall be given under oath.
- (2) A complete record of the proceedings shall be kept, including all documents presented and a verbatim record of the testimony of all witnesses.
- (3) The applicant shall proceed first to present evidence and testimony in support of the appeal or variance.
- (4) The administrator may present evidence or testimony in opposition to the appeal or variance.
- (5) All witnesses shall be subject to cross-examination by the adverse party or their counsel.
- (6) Evidence that is not admitted may be proffered and shall become part of the record for appeal.
- (7) The Board shall issue subpoenas upon written request for the attendance of witnesses. A reasonable deposit to cover the cost of issuance and service shall be collected in advance.

- (8) The Board shall prepare conclusions of fact supporting its decision. The decision may be announced at the conclusion of the hearing and thereafter issued in writing or the decision may be issued in writing within thirty (30) days a reasonable time after the hearing.
- (F) Appeal to the Court. Those aggrieved by the decision of the Appeals Board may appeal such decision to the Greene County Court of Common Pleas, pursuant to Ohio Rev. Code Chapter **2505 and** 2506.
 - (1) which, if taken, will effect compliance with the provisions of these regulations;
 - (2) Specify a reasonable time for performance;
 - (3) Advise the owner, operator, or occupant of the right to appeal;
 - (4) Be served on the owner, occupant, or agent in person. However, this notice and order shall be deemed to be properly served upon the owner, occupant, or agent if a copy thereof is sent by registered or certified mail to the person's last known mailing address, residence, or place of business, and/or a copy is posted in a conspicuous place in or on the dwelling affected.
 - (C) Violations and Penalties. Violation of the provisions of these regulations or failure to comply with any of its requirements shall be deemed to be a strict liability offense, and shall constitute a minor misdemeanor. Any person who violates these regulations or fails to comply with any of its requirements shall upon conviction thereof be fined or imprisoned as provided by the laws of the City of Bellbrook. Each day such violation continues shall be considered a separate offense. Nothing herein contained shall prevent the City of Bellbrook from taking such other lawful action as is necessary to prevent or remedy any violation. The City of Bellbrook shall prosecute any violation of these regulations in accordance with the penalties stated herein.

SECTION 15.06 ENFORCEMENT

(A) Compliance Required

- (1)No structure or land shall hereafter be located, erected, constructed, reconstructed, repaired, extended, converted, enlarged or altered without full compliance with the terms of these regulations and all other applicable regulations which apply to uses within the jurisdiction of these regulations, unless specifically exempted from filing for a development permit as stated in Section 15.03 (I).
- (2) Failure to obtain a floodplain development permit shall be a violation of these regulations and shall be punishable in accordance with Section 15.06 (C).

- (3)Floodplain development permits issued on the basis of plans and applications approved by the Floodplain Administrator authorize only the use, and arrangement, set forth in such approved plans and applications or amendments thereto. Use, arrangement, or construction contrary to that authorized shall be deemed a violation of these regulations and punishable in accordance with Section 15.06 (C).
- (B) Notice of Violation. Whenever the Floodplain Administrator determines that there has been a violation of any provision of these regulations, he shall give notice of such violation to the person responsible therefore and order compliance with these regulations as hereinafter provided. Such notice and order shall:
 - (1) Be put in writing on an appropriate form;
 - (2) Include a list of violations, referring to the section or sections of these regulations that have been violated, and order remedial action, which, if taken, will effect compliance with the provisions of these regulations;
 - (3) Specify a reasonable time for performance;
 - (4) Advise the owner, operator, or occupant of the right to appeal;
 - (5) Be served on the owner, occupant, or agent in person. However, this notice and order shall be deemed to be properly served upon the owner, occupant, or agent if a copy thereof is sent by registered or certified mail to the person's last known mailing address, residence, or place of business, and/or a copy is posted in a conspicuous place in or on the dwelling affected.
- (C) Violations and Penalties. Violation of the provisions of these regulations or failure to comply with any of its requirements shall be deemed to be a strict liability offense, and shall constitute a minor misdemeanor. Any person who violates these regulations or fails to comply with any of its requirements shall upon conviction thereof be fined or imprisoned as provided by the laws of the City of Bellbrook. Each day such violation continues shall be considered a separate offense. Nothing herein contained shall prevent the City of Bellbrook from taking such other lawful action as is necessary to prevent or remedy any violation.

ARTICLE 16

A-1 AGRICULTURAL DISTRICT

16.01 <u>INTENT</u>

This district is composed of certain land being used for agricultural activities, publicly owned and operated buildings and facilities, open recreational uses, and other open land uses. Sub-marginal lands having no principal use also are included in this district. It is the intent of this district to protect the open area from the encroachment of scattered urban type uses until such time as the area is ready for more intense development.

16.02 **PERMITTED PRINCIPAL USES**

- (1) Municipal wells and water supply facilities.;
- (2) Agriculture and gardening including the sale of produce and plants raised on the premises.

16.03 ACCESSORY USES

- (1) Personal Wireless Service Sites. and
- (2) Any use customarily accessory or incidental to the permitted uses with the exception of uses listed under <u>Section</u> 16.04 (7) below.

16.04 <u>CONDITIONAL USES</u>

- (1) Cemeteries.;
- (2) Home occupations (See **Zoning Code** Section 3.02).;
- (3) One family dwellings.;
- (4) Open air commercial amusements.;
- (5) Public or private recreation facilities including parks, playgrounds, golf courses, boat docks, driving ranges, swimming pools, and customary accessory buildings.;
- (6) Supervised educational, religious, recreational, and physical fitness facilities with or without provisions for overnight short-

term residence such as boarding schools, band camps, religious retreats, scout camps, and physical fitness/weight control centers;

(7) Breeding, boarding, grooming₂ and retail sales of domestic pet birds. Retail sale of supplies related to the care of domestic pet birds. The business owner/operator must maintain his principal residence on property used for any business use outlined in <u>this</u> <u>subsection</u> (7).

16.05 **YARD AND LOT REQUIREMENTS**

- (1) Required yards:
 - (a) Front yards shall not be less than thirty-five (35) feet in

depth;

(b) Side yards shall not be less than fifteen (15) feet on each

side;

- (c) Rear yards shall not be less than **forty** (40) feet in depth.
- (2) Minimum Lot Area:

The minimum lot area shall not be less than **five** (5) acres.

16.06 <u>BUILDING HEIGHT REGULATIONS</u>

No structure shall exceed <u>two and one-half</u> (2 ½) stories or <u>thirty-five</u> (35) feet in height with the exception of personal wireless service towers which shall not exceed <u>two hundred</u> (200) feet in height.

16.07 <u>ACCESSORY PARKING</u>

In the Agricultural District parking shall be provided as required in Section 18.16 **of this Zoning Code**.

16.08 <u>SIGNS</u>

See **Zoning Code** Section 18.20 for size and location of permitted

signs.

Article 17 PLANNED UNIT DEVELOPMENT

17.01 PURPOSES-AND INTENT

The purpose and intent of the PUD districts is to provide a comprehensive development plan that is flexible and innovative when the development of a site by standard, more rigid, conventional zoning district regulations may produce less efficient use of the land and less amenities and benefits for the community and users of the development. Development under planned unit development provisions provides a means for encouraging ingenuity, imagination, and flexibility on the part of landowners, engineers, architects, site planners, and developers in the planning and design of land areas. It is not the intent of the planned unit development provisions to allow applicants to circumvent the intent of this Zoning Code or to allow development of land not in conformance with the Comprehensive Plan of the City.

Planned Unit Development zoning is of a substantially different character than other types of zoning. Because of the difference in character, special standards and procedures are hereby established to govern and guide the creation of Planned Development zoning districts.

Planned Development zoning is a privilege to be earned and not a right which can be claimed simply upon complying with all the standards established in this section. Planning Planning Board and/or Council require any reasonable condition or design consideration which will promote proper development of and benefit to the community. It is not intended that the Board and/or Council automatically grant the maximum use exceptions or density increases in the case of each Planned Development. The -Board -Council shall grant only such increase or latitude which is consistent with the benefit accruing to the City as a result of the Planned Development. As a condition for approval, each Planned Development must be compatible with the character and objectives of the zoning districts within which it is located, and each Planned Development shall be consistent with the objectives of the Bellbrook Comprehensive Plan.

Some specific purposes of the Planned Development procedures are:
(1) To take advantage of advances in technology, architectural design and functional land use design;
(2) To recognize the problems of population density, distribution and circulation and to allow a deviation from rigid established patterns of land uses, while maintaining control through application of defined policies and objectives;
(3) To produce a comprehensive development equal to or better than that resulting from traditional lot-by-lot land use development;

17.02 <u>TYPES OF PLANNED DEVELOPMENT MANDATORY PUD</u> APPLICABILITY

Proposals for development shall be developed under PUD district provisions

when the

<u>Community Development Administrator determines the development proposal exhibits one or more of the following listed characteristics or a characteristic of similar magnitude or nature:</u>

- (A) The total gross area of the proposed development is ten (10) acres or more in size, except:
 - (1) <u>Detached single-family developments submitted for development under R-1AA, R-1A, and/or R—1B Zoning District requirements;</u>
 - (2) <u>Developments proposed for A-1 zoned property and all of the uses proposed for the development are permitted by A-1 zoning; or</u>
 - (3) <u>Developments in which the zoning designation of the subject property is appropriate for the proposed development and such zoning designation was in effect and approved prior to adoption of this Ordinance.</u>
- (B) The proposed development involves uses permitted by standard zoning districts in accordance with the provisions of Section 6.01 through 12.09 of this Zoning Code, and such uses are permitted by one or more PUD districts, and without PUD district zoning classification the development would normally require the approval of more than one standard zoning district.
- PD-1 Planned Residential Development
- PD-2 Planned Business Development
- PD-3 Planned Industrial Development

17.03 <u>GENERAL STANDARDS FOR PLANNED DEVELOPMENTS</u> VOLUNTARY PUD APPLICABILITY

Applicants of proposals for development in which the development is determined, as per Section 17.02 above, do not meet the conditions of mandatory PUD applicability may voluntarily request to be considered for PUD zoning classification and development under PUD district provisions.

Approval for consideration shall not be construed or interpreted as approval of the requested PUD zoning.

(See 17.05, 17.06 and 17.07 for specific standards for respective Planned Development districts)

Before an application for Planned Development can be approved, the applicant must present evidence which clearly supports the following conclusions:

- (1) The proposed development advances the general welfare of the community and the immediate vicinity.
- (2) The plan is in conformity with the Comprehensive Plan.
- (3) The proposed development is consistent in all respects with the purpose and intent of this Zoning Ordinance.
- (4) The site will be accessible from public roads that are adequate to carry the traffic that will be imposed upon them by the proposed development, and the streets and driveways on the site of the proposed development will be adequate to serve the residents or occupants of the proposed development.
- (5) The development will not impose an undue burden on public services and facilities, such as fire and police protection.
- (6) The location and arrangement of structures, parking areas, walks, lighting, and appurtenant facilities shall be compatible with the surrounding land uses, and any part of a Planned Development not used for structures, parking and loading areas, or access ways, shall be landscaped or otherwise improved. The height of buildings shall be compatible with surrounding land uses as well as the general characteristics of the area to be developed and the surrounding area.
- (7) Natural features such as water courses, trees and rock outcrops will be preserved, to the degree possible, so that they can be incorporated into the layout to enhance the overall design of the Planned Development.
- (8) The Planned Development can be substantially completed within the period of time specified in the schedule of development submitted by the developer.
- (9) Special Review of B-3, B-4 and R-3 Zoning Districts:
 - (a) In cases where a parcel of land has been previously zoned or has been recommended for zoning to its highest business or residential use (B-4 Central Business District, B-3 Neighborhood Business District or R-3

- Multi-Family Residential District), the City Manager may recommend and the City Council may require that prior to the City's issuance of a Zoning Permit for new construction on any portion of the parcel the owner/developer shall follow the plan submissions procedures for the entire parcel as described under the Planned Development Article.
- (b) In such cases, because the Planning Board has already made a determination on the appropriate land use for the parcel and has recommended that the zoning be the highest possible business or residential use, a second Planning Board public hearing on the Planned Development land use shall not be required.
- (c) The Planned Development procedures concerning the submission of plans shall be followed and the Planning Board shall review the preliminary development plans and forward them to City Council following consultation with the City's engineering consultants.
- (d) The City Council shall then conduct the single Planned Development public hearing and then determine whether or not the PD-1 or PD-2 zoning and accompanying preliminary plans shall be approved.
- (e) If the zoning and plans are approved, approval of the Final Plans shall follow procedures as outlined under 17.082 Final Plan Stage.

17.04 PROCEDURE FOR APPROVAL TYPES OF PLANNED UNIT DEVELOPMENT DISTRICTS

Types of planned unit development districts include PD-1 (Planned Residential Development Districts); PD-2 (Planned Business Development Districts); PD-3 (Planned Industrial Development Districts); PD-4 (Planned Mixed-Use Development Districts); and PD-5 (Planned Public and Private Buildings and Grounds Districts). Each PUD district shall be governed by the requirements specified for each respective PUD district and the general provisions governing planned unit developments.

(1) Pre-application Conference:

Prior to filing a formal application for approval of a Planned Development, the developer shall request a pre-application conference with the Bellbrook Planning Board.

The purpose of such conference is to allow the developer to present a general concept of his proposed development prior to the preparation of detailed plans. For this purpose the presentation shall include, but not be limited to the following:

- (a) Written "letter of intent" from the developer establishing his intentions as to development of the land.
- (b) Topographic survey and location map.

- (c) Sketch plans and conceptual objectives regarding land use pattern, structure location and type, street and lot arrangement and tentative lot sizes.
- (d) Tentative proposals regarding water supply, sewage disposal, surface drainage, street improvements and flood control measures.

Planning Board shall then instruct the developer to file an application and advise the developer to familiarize himself with the Bellbrook Zoning requirements and Comprehensive Plan affecting the proposed development. A copy of Article #17, Planned Development, shall be made available by request.

(2) Preliminary Plan:

Seven (7) copies of the Preliminary Plan of the Planned Development and the application shall be filed with the Zoning Inspector, who shall in turn forward them to the Planning Board for consideration.

The required procedure for consideration and approval of the Preliminary Plan shall be:

- (a) An application for Zoning Amendment shall be secured from the Zoning Inspector. The completed application shall be filed with the Zoning Inspector accompanied by other subdivision requirements as specified in Section 17.081.
- (b) The Planning Board shall study materials received and confer with other agencies of government as appropriate in the case to determine general acceptability of the proposal as submitted. In the course of such preliminary consideration, the Planning Board may request and the applicant shall supply additional material needed to make specific determinations.
- (c) Following such study, the Planning Board shall hold a conference or conferences with the applicant to discuss desirable changes in the first or succeeding drafts of the Preliminary Development Plan and report.
- (d) Recommendations of the Planning Board to the applicant shall be in writing, and following any such conference, agreements between the applicant and the Planning Board as to changes in the Preliminary Plan or other matters are to be recorded and acknowledged by the Planning Board and the applicant.

 Items on which no agreement is reached or on which there is specific disagreement shall be recorded and the applicant may place in the record the reasons for any disagreement.
- (e) When the Preliminary Development Plan has been approved in principle (as a whole or with reservations specifically indicated) or when the applicant indicates in writing that no further negotiations with the Planning Board are desired before proceeding, and in any case, within sixty(60) days of the submission of the

preliminary plan application, the Planning Board shall schedule the proposed plan for a public hearing, following which it shall make its recommendations to Council. Such recommendations shall indicate approval, approval with specific reservations or disapproval with reasons.

(f) In the event Council upholds an unfavorable recommendation from Planning Board, a public hearing is not required. Otherwise, Council shall schedule a public hearing for the Preliminary Plan and respective Planned Development zone designation. Within sixty (60) days, the Council shall approve the proposal outright, approve subject to conditions, or deny the proposal.

Council approval shall be by ordinance.

If conditions are attached to Council approval, there shall be no change in the zoning map until such conditions are formally accepted in writing by the developer.

When approved, the area of land proposed for development shall be designated PD-1 (Planned Residential District) or PD-2 (Planned Business District) or PD-3 (Planned Industrial District) and shall be used only in accordance with the uses and densities shown on the Planned Development Preliminary Plan.

(3) Approval and Recording of Final Development Plan

The purpose of such recording is to designate with particularity the land subdivided into conventional lots, as well as the dimension of other lands not so treated, into common open areas and building areas, and also to define each building site as well as the use of the land in general.

The approval and recording of the Final Development Plan may be accomplished in stages if the stages have been specified in the Construction Schedule approved with the Preliminary Plan.

(a) Upon the designation of the Planned Development district by Council, the Final Development Plan shall be presented to the Zoning Inspector who shall see that all requirements of Section 17.082 have been complied with and that the Final Development Plan is in a form suitable for recording with the Greene County Recorder.

(b) The record plot (the Final Development Plan in a form suitable for re-zoning) shall then be forwarded to the Planning Board for review and, if approved, shall be forwarded to the City Manager who shall make appropriate arrangement with the applicant, in accord with 17.02(2), to ensure the accomplishment of public improvements. The City Manager shall consult with Council concerning any land to be publicly or commonly owned and shall arrange for the necessary legal deed(s).

(c) The approved record plot shall then be submitted to the Greene County Recorder for recording within six (6) months after approval by Council.

(d) No Final Development Plan within the corporate limits of Bellbrook shall be so recorded unless it shall have the approval of Council, as indicated by the signatures of the Mayor and the Clerk of Council, inscribed thereon.

(4) Zoning Permit:

No zoning permit shall be issued by the Zoning Inspector until the Final Development Plan has been approved and duly recorded. See Article # 19, Section 19.01 of the Bellbrook Zoning Ordinance.

(5) Certificate of Zoning Compliance:

The Zoning Inspector shall issue no Certificate of Zoning Compliance until all utilities have been accepted by the City of Bellbrook in accordance with the Final Development Plan.

If all utilities, including roadways, have not been accepted by the City of Bellbrook, the Zoning Inspector may issue a Temporary Certificate of Zoning Compliance upon certification by the City Manager that the existing public utilities are functionally acceptable and adequate for the use of the premises. Should the City Manager find, while the Temporary Certificate of Zoning Compliance is in force, that the existing utilities are no longer functionally acceptable and adequate, the City Manager may order the cancellation of the Temporary Certificate of Zoning Compliance and direct the developer to vacate the premises of occupants.

The Temporary Certificate of Zoning Compliance remains in effect until all utilities, including roadways, are accepted by the City of Bellbrook at which time the Zoning Inspector will issue a permanent Certificate of Zoning Compliance.

(6) Changes in the Planned Development:

A Planned Development shall be developed only according to the approved and recorded final plan, and all recorded amendments shall be binding on the applicants, their successors, grantees and assignees and shall limit and control the use of premises (including the general internal use of buildings and structures) and location of structures in the Planned Development as set forth therein. Any deviation from the approved and recorded final plan must have Planning Board approval. Planning Board shall decide whether a change is major or minor.

(a) Major Changes:

Changes which alter the concept or intent of the Planned Development, include but are not limited to: increases in the number of units per acre; change in location or amount of non-residential land use; more than fifteen percent (15%) modification in proportion of housing types; reductions of proposed open space; significant re-design of roadways, utilities or drainage.

These changes shall be approved only by submission of a new Preliminary Plan and supporting data, by following the "preliminary approval" steps and subsequent amendment of the final Planned Development Plan with the Greene County Recorder.

(b) Minor Changes:

A minor change is any change not defined as a major change. The City Zoning Inspector and the Planning Board may jointly approve minor change in the Planned Development which do not change the concept or intent of the development without repeating the "preliminary approval" steps. The Zoning Inspector shall enter in detail all minor changes on the official Final Development Plan on record with the Municipality of Bellbrook. The Zoning Inspector shall enter in detail all minor changes on the official Final Development Plan on record with the Municipality of Bellbrook.

(7) Schedule of Construction:

A modification of the schedule may be approved by the City Council if the developer shall present satisfactory evidence of reasonable effort toward meeting the initial schedule and justification for the modification.

If construction falls more than one (1) year behind schedule, as determined by the City Engineer, and the developer fails to justify the delay to Council's satisfaction, Council may proceed to complete all or any part of such improvements and recover the costs thereof by laying claim to the guarantee specified in 17.082(2).

(8) Effect of Denial of Planned Development:

If an application for a Planned Development is denied wholly or in part, then for a period of one (1) year from the date of submission thereof, the Planning Board need not consider any resubmission therefore, unless it finds substantive changes or a change in the conditions upon which the denial was based.

If a Preliminary Development Plan is approved and the Final Development Plan is thereafter disapproved, the applicant, or his successor in interest, may at any time,

submit one (1) or more new versions of the Final Development Plan, so long as the new versions are in full compliance with the approved Preliminary Development Plan, including any conditions attached to said plan.

17.05 <u>PD-1 PLANNED RESIDENTIAL DEVELOPMENT GENERAL</u> PROVISIONS GOVERNING PLANNED UNIT DEVELOPMENTS

- (A) Project Control and Ownership. Planned unit development district designation is intended to apply to development sites that are under single ownership or unified control to any comprehensive design, planning, and development of the site. The project land may be owned, leased, or controlled by either one or more persons, partnerships, corporations, or other appropriate business associations capable of satisfying the objectives and requirements of the planned unit development district. The City shall require proof of ownership, covenants, easements, and other forms of property rights or control to ensure satisfactory compliance with PUD objectives and requirements.
- (B) PUD Agreement. Upon approval of a specific site plan, the City, at its sole discretion, may require the PUD developer to enter into a PUD Agreement with the City, and to furnish a performance bond for the purposes of assuring satisfactory completion of all requirements for site improvements such as landscaping and drainage control facilities and any conditions and safeguards as may be set by the City Council and the adopted Ordinance approving the specific PUD. At its discretion, the City may accept a letter of credit or other form of performance guarantee and security in lieu of a performance bond. In all cases the sufficiency and adequacy of such bond, letter of credit or other form of guarantee or security shall be at the sole determination of the City.
- (C) Utility Requirements. All initial and all future expanded utility systems within the limits of all planned unit development districts are required to be placed underground. Utility systems subject to this requirement shall be those that primarily serve the development within the PUD district and shall include, but not be limited to telephone, cable television, and electrical systems. Appurtenances of these systems which can be effectively screened may be excluded from this requirement if the City Council determines that such appurtenances are essential for utility service to other areas of the community and that such exclusion will not violate the intent or character of the proposed planned unit development or any conditions, restrictions, or other requirements imposed upon the development.
- (D) Application Process and Sequence. Due to the flexible and special character of planned unit development districts and projects, and to assist the City in evaluating the merits and making findings of proposed PUD projects, the full PUD approval process generally involves either a two-stage review or a

combined review. The applicant may select either the two-stage process or the combined review process. Unless the applicant informs the Community Development Administrator at the time of submission of the PUD zoning classification application of selection of the combined review process, the applicant shall be considered to have selected the two-stage review process.

(1) Two-stage review process.

- (a) First stage. The first stage of the two-stage review process generally consists of submission of an application for PUD zoning classification in accordance with division (F) of this section and Section 17.06 (A) (C). The Planning Board then reviews and makes a recommendation on the PUD zoning classification application in accordance with Section 17.06 (D) (F). The City Council then, in accordance with Section 17.06 (G) and (H), reviews the PUD zoning classification application and makes a decision to either approve as submitted, approve with supplementary conditions, or disapprove the application. Approval of PUD zoning classification constitutes only approval of PUD district zoning for the subject property. A specific site plan shall be admitted and approved prior to the issuance of a City zoning permit for the subject property.
- Second stage. Unless the applicant selects the combined review process as per division (D)(2) of this section, the second stage of the two-stage PUD approval process begins at the initiation of the application and is effective only if the PUD application submitted in the first stage was approved or approved with supplementary conditions by City Council. The second stage generally consists of submission of a specific site plan for the subject property in accordance with division (F) of this section and Section 17.07 (A), (B), (F), and (G). An administrative review and recommendation upon the specific site plan application is then conducted by the Planning Board in accordance with Section 17.07 (H) – (J). The City Council then, in accordance with Section 17.07 (K) and (L), reviews the specific site plan application and makes an administrative decision to either approve the application as submitted, approve with supplementary conditions, or disapprove.

(2) Combined review process.

At the applicant's option the first and second stage of the two-stage review process may be combined for a combined review process. If the applicant elects such option, a PUD zoning classification application and a related specific site plan application, in accordance with Section

- 17.06 (C) and 17.07 (B), shall be simultaneously submitted by the applicant and both applications shall together proceed through a combined review process. In such case, the Planning Board shall concurrently review both applications, but shall first make a recommendation upon the PUD zoning classification prior to making a recommendation upon the specific site plan application. The City Council then shall concurrently review both applications, but shall first act upon the PUD zoning classification prior to taking administrative action upon the specific site plan application.
- (E) Application completeness and officially filed status. Planned unit development and specific site plan applications shall not be considered officially filed until the applicant has submitted all applicable filing fees, and submitted all information as may be required by the Community Development Department. Completeness of an application and submission of all necessary information is the responsibility of the applicant. Only after an application is determined by the Community Development Administrator to be complete and officially filed will application review procedures begin. Applications for specific site plan approval that do not adhere to the substantial conformance requirements of Section 17.07 may be considered to be incomplete, not officially filed, and not subject to City review procedures.
- (F) Site arrangement requirements. The buildings, circulation, open space, landscaping, and other elements of the proposed PUD shall be arranged, planned, and designed on the site to produce:
 - (1) Favorable relationships with the existing natural topography, bodies of water or water courses, existing desirable vegetation, exposure to significant views and exposure to sunlight and wind;
 - (2) Safety, convenience, and ease of pedestrian and vehicular movement on, about and throughout the site and between the site and the community;
 - (3) An overall positive visual quality throughout, into and from the development site;
 - (4) An efficient, functionally organized, and cohesive

development;

- (5) All areas designed for future expansion or not intended for immediate improvements or development shall be landscaped or otherwise maintained in a neat and orderly manner; and
- (6) Screening shall be as specified in Section 18.17, Screening.

- (G) Zoning map identification. Areas approved as a planned unit development district will be clearly marked and identified on the zoning map by indicating the specific PUD district upon which approval was granted, either PD-1, PD-2, PD-3, PD-4, or PD-5.
- (H) Multiple PUD projects. A proposed PUD project, either voluntary or mandatory, shall be considered a "multiple PUD project" when each individual use proposed for the project is permitted by at least one of the various PUD zoning districts (Example: PD-4), while at the same time all of the proposed uses for the project are not permitted by only one, single PUD zoning district (Example: PD-1 and PD-4). A multiple PUD project requires differing PUD district zoning classifications for respective, contiguous portions of the property proposed for development. When the proposed development is a multiple PUD project the applicant shall file a single PUD zoning application, identifying each specific PUD zoning district, incorporating a functional and general location phasing plan for each individual PUD, a functional and general locational sequencing schedule for the full multiple PUD project, and all other necessary application materials as per Section 17.06 (C). A multiple PUD differs from a PD-4 (Section 17.14) in that the differing uses (such as, residential vis-a-vis commercial) are located in the distinctive appropriate PUD District.

(I) Transition of PUD provisions and regulations.

- (1) Previous PUD approvals. All PUD approvals, including any approved modifications, amendments, or conditions, approved by the City prior to the effective date of this Ordinance, and where any time limitation for such approvals has not expired, shall be governed by the planned unit development provisions and regulations in effect immediately prior to the adoption this Ordinance. If any time limitation attached to an approval of a PUD, modification, amendment, or condition has expired, such approval shall be governed by the current provisions and regulations of Sections 17.01 through 17.16 and any other applicable sections of this Zoning Code.
- (2) Applications filed prior to effective date. All PUD applications and all requests to modify, change, amend, or alter a PUD that are officially filed, but not approved prior to the effective date of this Zoning Code shall be governed by the PUD provisions and regulations in effect on the official filing date of the application or request.
- (4) Applications filed after effective date. On and after the effective date of this Ordinance, all PUD applications, and all requests to modify, change, or alter a PUD shall be governed by the PUD provisions and regulations in effect at that time.
- (J) Appeal of Community Development Administrator's determination. In order to assure that Planning Board and City Council have all information pertinent to the decision-making process contained within this section, the Community Development Administrator has been given the responsibility to require additional information as necessary in order for

Planning Board and City Council to make an informed decision. Any applicant who objects to the provision of additional information may request a determination by Planning Board at its next regularly scheduled meeting. No further action shall take place regarding the application and submission until the Planning Board has rendered its decision, which shall be final.

(1) Policies Guiding Development:

This district is intended to provide flexibility in the arrangement and design of residential dwellings, based upon a unified development plan conceived and carried out for an entire area. Within this district, appropriate and reasonable population density is maintained while a variety of dwelling unit types is encouraged. Natural features such as topography, trees and drainage ways are encouraged to remain in their natural state to the degree possible. Such developments are generally characterized by a significant proportion of usable open space and a unified design concept with particular attention devoted to the periphery of the development, the overall objective being the compatibility of the development with its surroundings.

(2) <u>Permitted Uses:</u>

Permitted uses are those included as permitted and accessory uses in Articles #5, #6 and #7 of the Bellbrook Zoning Ordinance and convenience establishment as hereafter defined.

Convenience establishments as accessory uses are those established as necessary for the proper development of the community and to be so located, designated and operated to serve primarily the needs of the persons within the Planned Development Plot. Uses shall be generally limited to those uses permitted in the B-2 district, with no direct access or advertising signs for such uses to be visible from the exterior of the development. Such convenience establishments and their parking areas shall not occupy more than five percent (5%) of the total area of the development. No separate building or structure designed or intended to be used, in whole or part, for business purposes shall be constructed within a Residential Planned Development until not less than thirty percent (30%) of the dwelling units proposed in the development plan are certified for occupancy.

(3) Area Requirements:

The minimum land area required for a Planned Residential development shall be five (5) acres. This area requirement may be reduced at the discretion of the Planning Board and City Council if it can be demonstrated that a waiver is necessary to achieve an improved site design and that surrounding neighborhoods and public facilities will not be adversely affected.

(4) Density Requirements:

Any combination or cluster of housing units is permitted, provided that the average lot area per family or dwelling unit contained in the site, exclusive of the area of street rights of-way, parking areas and commercial areas, will not be less than eighty percent (80%) of the average lot area per family required in the surrounding non-agricultural districts. This density requirement may be varied at the discretion of the Planning Board and City Council if it can be demonstrated that a waiver is necessary to achieve an improved site design and that surrounding neighborhoods and public facilities will not be adversely affected.

(5) Site Design:

All housing shall be sited to preserve privacy and to ensure natural light.

Lot widths may be varied to permit a mixture of structural designs. Varied setback is encouraged.

Where feasible, housing units should be situated to abut common open space or similar areas. A clustering of dwellings is encouraged.

(6) Structure Spacing:

A minimum of fifteen (15) feet shall be maintained between principal structures.

<u>(7) Length:</u>

There shall be no continuous structure of town houses, attached dwellings or apartments containing more than twelve (12) dwelling units on ground floor level.

(8) Height:

The height of any residential structure within a Planned Development shall not exceed thirty-five (35) feet or 2 ¹/₂-stories. However, the Planning Board may grant an exception if it is demonstrated that additional height can be achieved with concurrent expansion of suitable open space to protect adjacent structures from adverse reduction of light and air.

(9) Setback and Screening:

A minimum setback of fifty (50) feet shall be provided along the entire perimeter of the development and retained in natural woods, or be suitably landscaped with grass and/or ground cover, shrubs and trees. Projects located adjacent to

commercial or industrially zoned areas shall provide suitable screening to the residential development as adjudged by the Planning Board. Screening shall not obscure

traffic visibility within fifty (50) feet of an intersection.

(10) Common Open Space:

A minimum of twenty-five percent (25%) of the total land in any Planned Residential Development shall be reserved for permanent common open space and recreational facilities for the residents or users of the area being developed. In extreme topographical conditions, at the discretion of the Planning Board, this requirement may be reduced. Only area having minimum dimensions of fifty (50) feet by one hundred (100) feet shall qualify for computation as usable open space.

(11) Parking Requirements:

See Article #18, Section 18.16 for required spaces and design criteria.

(12) <u>Signs</u>:

See Article #18, Section 18.20 for size and location of permitted signs.

(13) Utilities:

All utilities including electric, telephone, gas, water and sewer lines must be buried, except when deemed unfeasible as determined by the City Engineer.

(14) Transient Rental Regulations:

No dwelling unit or dwelling, one family, dwelling, two family or dwelling multi-family or a portion thereof in the PD-1 district shall be used as a transient rental.

17.06 <u>PD-2 PLANNED BUSINESS DEVELOPMENT APPROVAL</u> <u>PROCEDURES FOR PLANNED UNIT DEVELOPMENT (PUD) ZONING</u> CLASSIFICATION

- (A) *Notice to applicants.* Notice is hereby provided to all applicants that:
 - (1) Approval of PUD zoning classification shall not be deemed approval of a specific site plan or approval of a zoning permit;
 - (2) The issuance of a City zoning permit for all or any portion of a PUD project requires City Council approval of a specific site plan; and
 - (3) In accordance with Section 17.07 (C) and (D), unless an extended approval period is granted by City Council, approval of any specific site plan shall expire if actual construction has not started in the area of approval for the

respective specific site plan within two years from the effective date of City Council approval of the plan.

- (B) Pre-application meeting. Prior to filing an application for a planned unit development, the applicant or the applicant's agent is strongly encouraged to meet with the Community Development Administrator (the "CDA") or the CDA's designee. The purpose of this meeting is to informally and generally discuss the proposed development and the purpose, criteria, and standards of the planned unit development provisions and this Zoning Code, and to provide the prospective applicant an opportunity to ask questions about PUD requirements and the PUD review and approval process. The applicant shall note that no statement or representation of the CDA, the CDA's designee, or any representative of the City during this pre-application meeting and any subsequent pre-application meetings shall be binding upon the Planning Board, the City Council, or the City. In addition, the applicant is encouraged to engage in informal consultation with the City Engineer, entities providing energy or utility services necessary for the proposed PUD, and such other individuals, entities, or organizations as suggested by the CDA or the CDA's designee.
- (C) Application requirements. The applicant shall officially file, in accordance with Section 17.05(D), an application for planned unit development district classification with the Community Development Department and shall make payment to the City in an amount equal to the established filing fee applicable to the PUD application for the proposed development. The application shall contain an original and copies of all application materials, as required on the checklist on file at the time of application with the Community Development Department in a quantity specified by the Community Development Department.
- (D) Planning Board public hearing and notice. The Planning Board shall hold a public hearing on an officially filed planned unit development zoning application within 45 days after the application's official filing date. Before holding such public hearing, notice of the hearing shall be given as required by law. Notice shall also be provided by first class mail at least 15 days prior to the date of the hearing, and such notice shall be deemed given on the date of mailing. Mailed notices shall be provided to all owners of property within 300 feet from the exterior boundaries of the area associated with the proposed PUD zoning classification or PUD amendment proposal. The City, at its discretion, may provide notice by other alternative means such as hand delivery, newspaper publication, electronic media, and any forms of notice utilized in lieu of mailed notice, to owners outside of the 300-foot boundary. The notice shall set forth the time and place of the public hearing and the nature of the proposed development. The failure to deliver the notice as provided in this section, so long as it is not intentional, shall not invalidate the public hearing or any decision on the application.
- (E) Planning Board review and recommendation of PUD application. After a planned unit development application has been determined to be officially filed in accordance with Section 17.05 (F), the Planning Board shall review and study the application and any accompanying materials. In the course of its study, the Board may confer with other agencies of government, request additional information or clarification from the applicant, and

request additional study and comments from the Community Development Department staff. The Planning Board shall, by resolution, recommend to the City Council that the request for planned unit development zoning classification be approved as presented, approved with supplementary conditions or modifications, or disapproved. The Board shall then transmit to the City Council all papers constituting the record and the resolution containing the Board's recommendation. If the Board recommends approval with supplementary conditions, such conditions shall be fully expressed in the recommendation resolution. Any normally permitted or accessory use that is recommended to be included in the specific planned unit development shall be fully identified and expressed in the Board's resolution and based upon findings in accordance with Sections 17.12 (B), 17.13 (B), 17.14 (B), 17.15 (B), and 17.16 (B). If the Planning Board determines, in order to conduct a sufficient review and make its recommendation, that additional information is required from the applicant or additional study is required, the Board may table consideration of the application until such additional information is received by the Board or the Board's study is complete.

- (F) Criteria for Planning Board recommendation. Before making a recommendation for approval or approval with supplementary conditions or modifications, as per division (E) of this section, the Planning Board shall find that the facts submitted with the application and presented at the public hearing, and any modifications, amendments, or supplementary conditions, satisfy the standards and criteria for planned unit development approval as per division (I) of this section.
- (G) City Council public hearing and notice. Within 45 days after the receipt of the Planning Board recommendation and resolution, the Clerk of City Council shall set a date for a City Council public hearing, notify the applicant the date set, and such hearing shall be held as soon as reasonably possible thereafter, unless the applicant requests additional time. Before holding such public hearing, notice of the hearing shall be given as required by law. Notice shall also be provided by first class mail at least 15 days prior to the date of the hearing, and such notice shall be deemed given on the date of mailing. Mailed notices shall be provided to all owners of property within 300 feet from the exterior boundaries of the area associated with the proposed PUD zoning classification or PUD amendment proposal. The City, at its discretion, may provide notice by other alternative means such as hand delivery, newspaper publication, electronic media and/or any other forms of notice utilized in lieu of mailed notice, to owners outside of the 300 foot boundary. The notice shall set forth the time and place of the public hearing and the nature of the proposed development. The failure to deliver the notice as provided in this section, so long as it is not intentional, shall not invalidate the public hearing or any decision on the application.
- (H) Action by City Council. City Council shall, by an affirmative vote of at least four (4) of its members, either disapprove the recommendation of the Planning Board or introduce and hold the first reading of an Ordinance adopting or modifying such recommendation. Following the public hearing of the Ordinance, a planned unit development zoning classification request shall be either approved, approved with supplementary conditions, or disapproved by at least four (4) members of City Council. In the event the City Council approves the planned unit development, with or without

supplementary conditions, the Council shall find, by Ordinance, that the facts submitted with the application and presented at the public hearing, and any modifications, amendments or supplementary conditions satisfy the standards and criteria for planned unit development approval as per division (I) of this section. If City Council approves the planned unit development zoning classification, the approval Ordinance shall fully set forth any and all supplementary conditions for approval, and all standard district permitted or accessory uses within the specific PUD shall be fully expressed in the approval Ordinance and based upon findings in accordance with Sections 17.12 (B), 17.13 (B), 17.14 (B), and 17.15 (B), and 17.16 (B). If the City Council determines in order to conduct a sufficient review and make its recommendation that additional information is required from the applicant or additional study is required, the Council may table consideration of the application until such additional information is received by the Council or the Council's study is complete.

- (I) Standards and criteria for Planned Unit Development zoning approval. A planned unit development zoning classification shall only be approved when the following standards and criteria are satisfied.
 - (1) The planned unit development complies with the purpose and intent of this Zoning Code;
 - (2) The proposed development promotes the health, safety, and general welfare of the present and future inhabitants of the city;
 - (3) The proposed zoning and the conditions and requirements incorporated within the Ordinance approving the PUD zoning district provide for minimizing impacts on the surrounding development;
 - (4) The site will be accessible from current or planned public thoroughfares adequate to carry traffic which will be imposed upon them by the proposed development;
 - (5) Potential impacts on public services and facilities can be mitigated by site and building design and the benefits which will accrue to the city and the public;
 - (6) Existing and proposed utility services for the proposed residential population densities and nonresidential uses are or will be available to the project;
 - (7) The proposed development complies with applicable requirements and conditions of Section 17.05;
 - (8) Each individual section or subarea of the development, as well as the total development, can exist as a functionally independent environment. In the alternative and at the discretion of the City, adequate assurance has been

provided by the applicant and to the satisfaction of the City that such objective will be achieved;

- (9) Any permitted, conditional, or accessory uses excluded from the specific proposed planned unit development are based upon findings in accordance with Sections 17.12 (B), 17.13 (B), 17.14 (B), 17.15 (B), and 17.16 (B); and
- (10) The planned unit development can be substantially completed within the time specified in the schedule of development submitted by the applicant.

(1) Policies Guiding Development:

This district is provided in recognition that many commercial establishments seek to develop within unified commercial areas, usually under single ownership and control and typically called "shopping centers". Within the premises of the zone, such centers would have all the necessary services and facilities comprehensively provided in accordance with an approved development plan. Provisions of this zone are formulated to achieve harmoniously designed structures upon a well landscaped site, achieving a high degree of pedestrian vehicular separation, all of which would be compatible with surrounding land uses.

(2) Permitted Uses:

Uses included are those permitted, accessory and conditional uses in B-1, B-2, B-3 and B-4 Business Districts and O-1, Office Building District developed in accordance with the approved Development Plan but subject to approval of the Planning Board.

(3) Arrangement of Commercial Uses:

Commercial buildings and establishments shall be planned as groups having common parking areas and common ingress and egress points in order to reduce the number of potential accident locations at intersections with thoroughfares.

The plan of the project shall provide for the integrated and harmonious design of buildings and for adequate and properly arranged facilities as may be necessary to make the project attractive and efficient from the standpoint of the adjoining and surrounding existing or potential developments.

(4) Area Requirements:

The minimum land area for a Planned Business Development shall be two (2) acres.

(5) Structure Spacing:

A minimum of twenty (20) feet shall be required between adjacent principal buildings.

(6) Setback and Screening:

A setback of at least fifty (50) feet shall be provided along the entire perimeter of the development. However, where the Planned Development adjoins a Business or Industrial District, the setback and screening requirements shall be at the discretion of the Planning Board.

Where situated adjacent to a residentially zoned area, a minimum of twenty (20) feet along the exterior property line shall be planted with an evergreen hedge or dense planting of evergreen shrubs not less than three (3) feet in height at the time of planting. A landscaped mound adequate to provide screening may be substituted for hedge or shrubs.

In no case shall screening be placed within fifty (50) feet of a curb cut or intersection.

Vehicular access through such landscaped strip when leading from residential areas shall be permitted only for convenience of residents of adjoining residential areas, and not designed for use by the general public.

(7) Parking Requirements:

See Article #18, Section 18.16 for required parking criteria.

(8) Loading and Unloading Areas:

See Article #18, Section 18.15.

(9) Signs:

See Article #18, Section 18.20 for size and location of permitted signs.

(10) Utilities:

Wherever possible, utilities shall be housed in structures compatible with the development so as not to detract from the overall aesthetic design.

17.07 <u>PD-3 PLANNED INDUSTRIAL DISTRICT APPROVAL PROCEDURES</u> FOR SPECIFC SITE PLAN.

- (A) Notice to applicants. Notice is hereby provided to all applicants that:
 - (1) Approval of PUD zoning classification shall not be deemed approval of a specific site plan or approval of a zoning permit;
 - (2) The issuance of a City zoning permit for all or any portion of a PUD project requires City Council approval of a specific site plan; and

- (3) In accordance with divisions (C) and (D) of this section, unless an extended approval period is granted by City Council, approval of any specific site plan shall expire if actual construction has not started in the area of approval for the respective specific site plan within two years from the effective date of City Council approval of the plan.
- (B) Required specific site plan application. Prior to the issuance of a City zoning permit for any PUD project, a specific site plan application shall be officially filed in accordance with Section 17.05 (F), proceed through the review and approval process as per divisions (G) (K) of this section, and shall be approved by City Council. All specific site plan applications shall incorporate a minimum area of five acres. In cases in which the total gross area of the property approved for PUD zoning classification is less than five acres, the specific site plan shall incorporate 100% of the PUD property. In all cases the area incorporated in a specific site plan shall be contiguous.
- (C) Expiration of specific site plan approval. Unless an extended approval period is granted in accordance with division (D) of this section, approval or approval with supplementary conditions of any specific site plan shall expire if, in the judgment and determination of the City, the actual start of construction has not begun in the approved area of the respective specific site plan within two years from the effective date of City Council approval of the respective plan. In the event of expiration of an approved specific site plan, the applicant shall start afresh, submit a specific site plan application in accordance with Sections 17.05 (F) and division (E) of this section, and the application shall proceed through the review and approval process as per divisions (G) (K) of this section. All PUD site plans approved prior to the passage of this section shall not expire for five years from the date of approval.
- (D) Extension of specific site plan approval period. Upon request by the owner, a onetime only, administrative extension of the two-year approval period for a specific site plan may be granted by the City Council. In the event such an extension is granted, the period of extension shall not exceed one year. The owner shall submit a written request for an extended site plan approval period to the Community Development Administrator. The written request shall be submitted no sooner than nine months prior to, but no later than 90 days prior to the expiration date of the approved specific site plan which is the subject of the request for approval period extension. The written request shall include the necessity for the extension; submission of documentation and evidence that the owner has made a reasonable effort to begin the actual start of construction; the reason(s) why construction has not actually started to date and will not start prior to expiration of the specific site plan; the requested length of time to extend the specific site plan approval period (not to exceed one year); a description of the impact and major effects upon the full PUD project if the requested extension is not approved; a description of the major effects upon the full PUD project if the requested extension is approved, including changes in phasing or staging plans; and a revised time schedule showing the dates when construction will actually start for the area of the specific site plan and, if applicable, the remaining areas of the complete PUD project. Within 45 days from the Community Development Administrator's receipt of the

owner's written request the Community Development Administrator shall forward the owner's request and the Community Development Administrator's comments and recommendation to the City Council. Within 45 days after City Council receives the request and the Community Development Administrator's comments and recommendation, the Council shall take administrative action upon the request. The City Council shall, by resolution and administrative action, either deny the request for extension, approve the request for extension as submitted, or approve the request for extension for a lesser period of time than requested by the owner. Any extension of approval period for a specific site plan shall become effective and begin to run on the date of Council approval of such extension. Council shall approve an extended approval period for a specific site plan only when the following conditions are satisfied:

- (1) The owner has submitted a written request for an extended approval period for the specific site plan;
- (2) The subject property currently has a PUD zoning classification;
- (3) The approval period for the specific site plan has not been previously extended by City Council;
- (4) An extension of approval period for the specific site plan will not cause the City to breach any PUD Agreement made with the owner;
- (5) The owner, in the judgment of the Council, has taken reasonable steps and made reasonable efforts to actually start construction;
- (6) An extended approval period for the specific site plan will not violate the purpose and intent of Sections 17.01 through 17.15 or other applicable provisions of this Zoning Code;
- (7) An extended approval period for the specific site plan promotes the health, safety, and general welfare of the present and future inhabitants of the City; and
- (8) The period of extension for the specific site plan does not exceed one year.
- (E) Previously approved specific site plans. A specific site plan approved by City Council shall be deemed null and void upon approval of a subsequent specific site plan for all or part of the same property, unless otherwise conditioned by City Council.
- (F) Application requirements. The applicant shall, in accordance with Section 17.05 (G), officially file an application for a specific site plan with the Community Development Department, and shall make payment to the City in an amount equal to the established filing fee applicable to the specific site plan application for the proposed development. As per division (G) of this section, the specific site plan shall substantially conform to the PUD zoning plans, concepts, schedules, development information, and conditions as approved or

as approved with conditions by City Council. The application shall contain an original and copies of all application materials as required on the checklist on file at the time of application with the Community Development Department in a quantity specified by the Community Development Department.

(G) Specific site plan conformance with approved PUD.

- (1) An officially filed specific site plan shall substantially conform to the PUD zoning plans, concepts, schedules, and development information as approved by City Council. If the Community Development Administrator or the Planning Board determines that the officially filed specific site plan does not substantially conform, the applicant may either modify the specific site plan to the extent necessary for substantial conformance and resubmit for review, or start afresh and submit a new specific site plan for review, or apply for an amendment to the PUD zoning classification Ordinance. Notwithstanding Section 17.05 (F), a submitted specific site plan shall not be considered officially filed if the Community Development Administrator determines prior to Planning Board review that the submitted specific site plan does not adhere to the substantial conformance requirement.
- (2) At the discretion of the Planning Board, a submitted specific site plan that contains modifications to the plans, concepts, schedules, and development information as approved by City Council, may be determined to substantially conform. In no event shall a specific site plan be determined to substantially conform if the specific site plan involves changes in permitted uses as per Section 17.09, or changes which result in exceeding any limitation or any maximum amount imposed by an Ordinance originally granting or amending the PUD zoning classification for the subject property.
- (H) Planning Board public hearing and notice. The Planning Board shall hold a public hearing on an officially filed specific site plan application within 45 days after the application's official filing date. Before holding such public hearing, notice of the hearing shall be given as required by law. Notice shall also be provided by first class mail at least 15 days prior to the date of the hearing, and such notice shall be deemed given on the date of mailing. Mailed notices shall be provided to all owners of property within 300 feet from the exterior boundaries of the area proposed for specific site plan or modification approval. The City, at its discretion, may provide notice by other alternative means such as hand delivery, newspaper publication, electronic media and/or any other forms of notice utilized in lieu of mailed notice, to owners outside of the 300 foot boundary. The notice shall set forth the time and place of the public hearing and the nature of the proposed specific site plan. The failure to deliver the notice as provided in this section, so long as it is not intentional, shall not invalidate the public hearing or any decision on the application.
- (I) Recommendation by Planning Board. Planning Board shall take administrative action and, by resolution, recommend to the City Council that the submitted specific site plan be approved as presented, or approved with supplementary conditions or modifications,

or disapproved. The Board shall then transmit all papers constituting the record and the resolution containing the Board's recommendation to the Council. All resolutions for specific site plan recommendations to City Council shall indicate the Board is taking administrative action, and if the Board's recommendation is for approval with supplementary conditions, the resolution shall fully express such conditions. If the Planning Board determines in order to conduct a sufficient review and make its recommendation that additional information is required from the applicant or additional study is required, the Board may table consideration of the application until such additional information is received by the Board or the Board's study is complete.

- (J) Criteria for Planning Board recommendation. Before making a recommendation for approval or approval with supplementary conditions in accordance with division (I) of this section, the Planning Board shall find, by resolution, that the facts submitted with the specific site plan application and presented at the public hearing, and any modifications, amendments, or supplementary conditions, satisfy the standards and criteria for specific site plan approval as per division (M) of this section.
- (K) City Council public hearing and action. Within 45 days after the receipt of the Planning Board recommendation, the Clerk of the City Council shall set a date for a public hearing and such hearing shall be held as soon as reasonably possible thereafter, unless the applicant requests additional time. The public hearing shall be for the purposes of taking administrative action on the proposed specific site plan. Following the scheduled hearing the City Council shall, by motion, either approve, approve with supplementary conditions, or disapprove the proposed specific site plan. Approval of such motion requires an affirmative vote of four members of the Council. If the Council approves the specific site plan with supplementary conditions, the motion for approval shall fully set forth such conditions and be fully recorded in the meeting minutes. As the basis for City Council approval of a specific site plan, with or without supplementary conditions, Council shall find that the facts submitted with the application and any accompanying materials, and any modifications, amendments or supplementary conditions satisfy the standards and criteria for specific site plan approval as per division (M) of this section. If the standards and criteria of division (M) of this section are not satisfied, Council shall disapprove the specific site plan and shall make findings of fact indicating which particular standards and criteria are not satisfied. If the City Council determines in order to conduct a sufficient review and make its decision that additional information is required from the applicant or additional study is required, the Council may table consideration of the application until such additional information is received by the Council or the Council's study is complete.
- (L) Notice of City Council public hearing. Before holding the public hearing specified in division (K) of this section, notice shall be provided as required by law. Notice shall also be provided by first class mail at least 15 days prior to the date of the hearing, and such notice shall be deemed given on the date of mailing. Mailed notices shall be provided to all owners of property within 300 feet from the exterior boundaries of the area proposed for specific site plan or modification approval. The City, at its discretion, may provide notice by other alternative means such as hand delivery, newspaper, electronic media and/or any other forms of notice utilized in lieu of mailed notice, to owners outside of the 300-boundary. The

notice shall set forth the time and place of the public hearing and the nature of the proposed specific site plan. The failure to deliver the notice as provided in this section, so long as it is not intentional, shall not invalidate the public hearing or any decision on the application.

- (M) Standards and criteria for specific site plan approval. A specific site plan shall only be approved when the following standards and criteria are satisfied.
 - (1) The specific site plan complies with the purpose and intent of the specific PUD zoning;
 - (2) The proposed development carries out the purpose and intent of the City's Land Use Plan and Comprehensive Plan;
 - (3) The proposed development promotes the health, safety, and general welfare of the present and future inhabitants of the city;
 - (4) The specific site plan substantially complies with the substantial conformance requirement of division (G);
 - (5) The proposed development has no significant detrimental impact that outweighs the development's benefits to the community;
 - (6) The proposed streets and driveways on the site of the proposed development will be adequate to serve the residents, occupants or users of the proposed development, and the specific site plan, along with any necessary off-site vehicular circulation improvements, provides adequate vehicular ingress and egress and will be accessible from current or planned public thoroughfares adequate to carry traffic which will be imposed upon them by the proposed development;
 - (7) The proposed development minimizes conflicts between vehicular, pedestrian, and bicycle circulation patterns and movement;
 - (8) The benefits of the proposed development mitigate any burden on public services and facilities, including fire and police protection;
 - (9) Existing and proposed utility services for residential population densities and nonresidential uses are adequate for the projected demand during all phases of development and at full completion of development;
 - (10) The proposed development complies with applicable requirements and conditions of Section 17.05;
 - (11) Each individual section or subarea of the development, as well as the total development, can exist as a functionally independent environment. In the alternative and at the discretion of the City, adequate assurance has been provided

by the applicant and to the satisfaction of the City that such objective would be achieved;

- (12) The design and other amenities incorporated in the proposed development will provide increased benefits to the residents, occupants, users, and the community and such design and other amenities are in accord with the planned unit development provisions of this Zoning Code and other applicable Ordinances of the City;
- (13) The proposed development contains such covenants, easements, and other such forms of property rights and control as may reasonably be required for the maintenance and care of common, private facilities, and for the public health, safety, and welfare. If governmental ownership of common open space, recreational facilities, or other common facilities is planned, a copy of its acceptance has been filed with the application.

(1) Policies Guiding Development:

The provisions of this district are provided in recognition that many industrial establishments seek to develop within unified industrial areas having all necessary services and facilities comprehensively provided in accordance with a predetermined development plan. As in the Planned Business Zone (PD-2), provisions of this zone are formulated to encourage a high degree of coordinated development upon well landscaped premises. Particular attention is devoted to design of the periphery of the development with the objective of achieving compatibility with existing and potential surrounding land uses.

(2) Permitted Uses:

These uses will be permitted in accordance with the approved Development Plan:

- (a) Assembly plants except automobile assembly plants or plants of similar nature.
- (b) Automobile repair but no commercial wrecking, dismantling or salvage yard.
- (c) Auto service station.
- (d) Automobile, trucks, boat sales.
- (e) Bottling works.
- (f) Builders supply store.
- (g) Building and trades, including contractor's yard and utilities storage yard.

-Carpet cleaning, dry cleaning and dyeing, laundry. -Cold storage plant. Commercial greenhouse. Dairy products manufacture. -Fabrication, processing, packaging and/or manufacture of food products and condiments excluding fish products, slaughter houses and rendering and refining of fats, oils, fish, vinegar, yeast and sauerkraut. (m) Fabrication, processing, packaging and/or manufacture of cosmetics, drugs, perfumes, pharmaceuticals, and toiletries. Fabrication, processing, packaging and/or manufacture of articles or merchandise from the following previously prepared materials: bone, canvas, cellophane, cloth, cork, feathers, felt, fiber, horn, leather, paint, paper, plastics, precious or semi precious metals or stones, textiles, tobacco, wax, wood and yarn. Fabrication, processing, packaging, and/or manufacturing of musical instruments, toys, novelties, rubber or metal stamps. (p) Fabrication, processing, packaging and/or manufacture of ice, cold storage plant, bottling plant. Farm implements and contractor equipment sales and service. -Foundry casting light weight non-ferrous metals or electric foundry, not causing noxious fumes or odors. —Fuel or coal company. Furniture re-upholstering and repair. -Industrial research laboratories. -Inflammable liquids, underground storage only. Lumber yards including incidental millwork, coal, brick, stone. -Monument sales including incidental mechanical operations.

- (y) Motor freight depot or trucking terminal provided, the truck entrance and exits are on to streets whose pavement width is at least thirty (30) feet between curbs.
- (z) Painting, varnishing shops.
- (aa) Personal wireless service sites.
- (bb) Plumbing supply and contractingshops including storage yards.
- (cc) Poultry killing and dressing.
- (dd) Public garages, motor vehicle and bicycle repair shops, auto paint and body shops.
- (ee) Publishing and printing.
- (ff) Railroad freight stations, but not including switching, storage, freight yards, sidings or maintenance or fueling facilities.
- (gg) Repair, rental and servicing for appliances.
- (hh) Sign contractor.
- (ii) Stone grinding, dressing and cutting.
- (jj) Storage yard for building supplies and equipment, contractors equipment, food fabrics, hardware and similar goods when located entirely within a building, provided such buildings shall not be used for wrecking or dismantling of motor vehicles.
- (kk) Television and radio broadcasting towers.
- (11) Tin and sheet metal shop.
- (mm) Tool and die shop, wrought iron shop, blacksmith or machine shop, excluding drop hammers.
- (mm) Trailer rental and sales.
- (nn) Truck terminal.
- (00) Veterinary clinic or kennels, animal hospital, provided that all animals are housed in buildings or enclosures which are at least five hundred (500) feet from any "R" district.
- (pp) Warehouses.
- (qq) Wholesale distributors.

(rr) Uses similar to the above uses and any other manufacturing or industrial enterprise, operation or process whether making, assembling, repairing, buffing, finishing, plating, polishing, tempering, packing, shipping or storing; provided that any resulting cinders, dust, flashing, fumes, gas, noise, odor, refuse matter, smoke, vapor or vibration is no greater or more detrimental to the neighborhood than the specified uses, that no extra fire hazard is created, and the proposed use as determined by the Board of Zoning Appeals is similar in character to one of the specific uses in this section.

17.08 <u>SUBMISSION REQUIREMENTS FOR PLANNED DEVELOPMENT</u>

<u>PROPOSALS PLANNED UNIT DEVELOPMENT OBLIGATIONS AND</u>

BINDING AND ENFORCEABLE CONDITIONS.

If any portion of property included in a planned unit development is sold or leased, the PUD zoning classification and specific site plan obligations shall continue to be binding upon all subsequent owners or lessees, regardless of the acreage involved in the sale or lease. As a matter of procedure, all terms and conditions of any planned unit development zoning classification and specific site plan that were approved by the City Council in the past or may be approved in the future shall remain binding upon and enforceable against the subject tract of land, except to the extent modified in accordance with section 17.11 (A) through (C).

17.081 PRELIMINARY PLAN STAGE

(1) Application:

An application for Preliminary Planned Development shall be secured from, and the fully completed application shall be filed with the Zoning Inspector, together with the appropriate fee in accordance with the prevailing rate.

- (1) Material To Be Submitted with Applications:
- (a) Identification of all property owners within the proposed development, evidence of unified control of the entire area of the development, tentative agreement of all owners to proceed with development according to plan or to provide adequate sureties for completion.
 - (b) A map or maps indicating the relation of the proposed development to the surrounding area. As appropriate to the development proposed, such map or maps shall demonstrate access to major streets, and show the approximate location and sizes of existing public sewers, water lines and storm drainage systems.
- (c) Topographic data map drawn to a scale of one hundred (100) feet or less to one (1) inch by a registered surveyor and/or engineer showing:
 - (1) Boundary lines, bearing and distances;
 - (2) Easements, location, width and purpose;

- (3) Wooded areas, streams, lakes, marshes and other physical conditions affecting the site;
- (4) Ground elevations on the tract;
- (5) If deemed necessary by the City Engineer, subsurface conditions on the tract, including the location and results of tests made to ascertain the conditions of subsurface soil, rock and groundwater.
- (6) Name, address and phone number of registered surveyor, registered engineer and/or urban professional planner assisting in the preparation of the Preliminary Development Plan.

(3) Preliminary Development Plan and Report:

A Preliminary Development Plan and report shall accompany the application with maps at a scale of one hundred (100) feet or less to the inch, including as appropriate to the kind of planned development proposed; the following information, presented in generalized form:

- (a) Proposed land uses and appropriate height, bulk and location of principal structures sufficient to permit an understanding of the style of the development. Proposals containing residential units shall specify the number of housing units by size and type proposed within the initial phase of the proposal or within the overall development if the development is not to be staged.
- (b) Proposed automotive and pedestrian circulation patterns including streets by type (major, collector or minor) and width, public or private bicycle paths and pedestrian ways and existing or plotted streets proposed to be vacated.
- (c) Major off-street parking areas.
- (d) Proposed parks, playgrounds, school sites and other major open spaces as well as the general form of organization proposed to own and maintain any common open space.
- (e) General locations of utility installations and easements.
- (f) A Schedule of Construction which shall indicate the estimated date for the start of construction and the duration of the construction period in months. The commencement date of construction may be set relative to the Plan's approval date.

If development is to be in stages, indication as to order and timing of development and demonstration that each stage, when completed, would complement any completed earlier, and would form a reasonably independent unit even though succeeding stages were delayed.

(g)	-Proposa	ıls for	provisior	i of publ	ic facilitie	s, utilities	or service	es where
lacking	g or unli	kely to	be availa	ble when	needed fo	r the plann	ed develop	ment, or
for pro	oviding :	suitable	private	facilities.	, utilities c	or services.	A report	shall be
	_					d continuir		
-		- 1	ivate stre	-			O	

(h) The substance of covenants, grants and easements or other restrictions proposed to be imposed upon the use of the land, building and structures including proposed easements or grants for public utilities.

17.082 FINAL PLAN STAGE

(1) Requirements for the Final Development Plan:

(a) A map or maps in the form required by the governing Subdivision Regulations for recording of final plats or subdivisions, with such modifications and additions as required to achieve the design flexibility of the planned development concept. Similar modifications of standards contained in the governing Subdivision Regulations or in other regulations or policies applying generally may be reflected in such maps and reports if the Planning Board shall find and shall certify, after consultations with other agencies of government as appropriate in the specific case, that the public purposes of such regulations or policies are as well or better served by specific proposals of the formal plan and reports.

(b) A general site and land use plan for the Planned Development as a whole, indicating subareas for phase development if any, and showing location and use of structures and portions of structures in relation to building sites reserved for future use and uses for which sites are reserved, automotive and pedestrian circulatory networks, principal parking areas, open space not in building sites and the use for which it is intended, and such other matters as are required to establish a clear pattern of the relationships to exist between structures, uses, circulation and land.

(2) Agreements, Contracts, Deed Restrictions and Sureties:

(a) All agreements, contracts and deed restrictions shall be submitted in a form acceptable to the City. Acceptance of the documents shall be in the form of a letter from the City Manager certifying that all such requirements have been met.

The applicant shall guarantee the installation of the public improvements specified in the Final Development Plan through one of the following methods:

- (1) Filing a performance and labor and material payment bond in the amount of one hundred twenty five percent (125%) of the estimated construction cost as determined by the City;
- (2) Depositing or placing in escrow a certified check or cash or any other acceptable pledge, in the amount of one hundred twenty-five percent (125%) of the construction cost as approved by the City.

Acceptance of the form of guarantee selected shall be evidenced by a formal letter from the City Manager.

(b) The owners of the Planned Development or persons legally representing them shall execute a legally binding agreement providing for the maintenance of commonly owned open space, recreation areas and automotive and pedestrian circulatory networks. In addition, such owners or person(s) legally representing them shall authorize the Bellbrook Police Department and any other properly constituted law enforcement agency to exercise full powers of arrest law enforcement on the premises.

17.083 **DETAILED SITE PLANS**

Detailed plans for individual buildings or groups of buildings shall be submitted to the Planning Board for approval before each state of construction is completed. Such plans shall be in accord with the Final Development Plan and report as approved by the Planning Board and shall be in sufficient detail to permit determinations as to compliance with the requirements of this ordinance with respect to the particular planned development district and uses involved. The plans shall include:

- (1) Site plans for the building site or sites, indicating relationship to adjoining areas.
- (2) Floor plans of the buildings involved, indicating horizontal dimensions, uses of space, and floor areas.
- (3) Elevations of the buildings involved, indicating height and if required in determinations for the particular building or use, location and dimensions of all windows and other glassed areas.
- (4) Any modifications on single lots in PD zoning districts shall be heard by the Board of Zoning Appeals.

17.09 MODIFICATIONS TO PUD PERMITTED USES

The permitted uses for an approved PUD include all uses identified as permitted and accessory uses for the type of PUD district under which the PUD was approved, except those uses expressly identified as excluded uses in the Ordinance granting PUD zoning approval

for the subject property. Upon PUD zoning approval, an applicant is not required to develop all uses permitted for the applicant's specific PUD, however, an applicant shall only develop those uses permitted for the specific PUD. Any request to modify or change the permitted uses of a specific PUD, as such uses are identified in Sections 17.12 (B), 17.13 (B), 17.14 (B), 17.15 (B), and 17.16 (B), and as may be conditioned by the original Ordinance granting PUD zoning classification for the subject property, requires an amendment to such Ordinance. Upon a request to modify or change the permitted uses for a specific PUD, the City, at its discretion, shall require the applicant to either submit a new, full application in accordance with Section 17.06 (C), or submit a modified PUD zoning application, or submit supplementary materials to accompany the original PUD zoning application materials. In all cases, proposed amendments to the original zoning Ordinance shall be subject to review and approval procedures in accordance with Sections 17.06 (D) - (I).

17.10 SIMILARITY OF USES AND CONDITIONAL USES

(A) Similarity of uses. The Planning Board shall have the appropriate power and duty to determine if uses not specifically mentioned in this Zoning Code are similar to uses permitted in planned unit development districts. In no event shall a determination of similarity of use by the Planning Board overrule a City Council determination of exclusion of uses from a specific PUD, and the Ordinance originally granting or amending PUD zoning classification shall govern. If a PUD zoning classification application includes a list of intended uses which requires a determination of similarity, the Planning Board shall make such determination before its review and recommendation on the PUD zoning application or concurrently to its review and recommendation on the PUD zoning application.

(B) Conditional uses within PUDs. Planning Board shall have the appropriate power and duty to hear and decide upon applications for conditional uses which are requested to be approved uses within a PUD. The Planning Board shall approve, disapprove, or approve with conditions for PUD conditional use applications. Approval of a conditional use application shall expire one year after the effective date of Planning Board approval unless approval of a specific site plan for that conditional use has been approved. Upon approval of a specific site plan, the conditional use will expire with the expiration of the specific site plan.

17.11 MODIFICATIONS TO APPROVED SPECIFIC SITE PLANS

Requests to modify an approved specific site plan, when such requested modification does not involve a change of permitted uses for the specific PUD, may be approved by administrative action. Requests to modify shall be in writing and signed by the property owner. The Community Development Administrator shall determine the type and amount of any additional information necessary for consideration of the modification. Modifications shall be considered either major, minor or incidental, and shall be approved or disapproved by administrative action in accordance with divisions (A), (B) and (C) of this section. Requests to modify an approved specific site plan which involve a change of PUD permitted uses shall be considered a modification to PUD permitted uses and shall comply with Section 17.09.

- (A) Major modifications. Major modifications are defined as modifications which do not change the PUD's permitted uses and do not exceed any of the limitations and conditions of the PUD's approval Ordinance, and which result in an increase of 5% or more in building coverage; a 5% or greater increase in dwelling unit density; a significant redesign of roadways, or drainage; major redesign of a building which significantly alters the central architectural design or theme of the building; or modifications not considered to be of minor or incidental character. Major modifications to an approved specific site plan require resubmission of a specific site plan application in accordance with Section 17.07. At its discretion and in lieu of compliance with Section 17.07, the Planning Board may require the applicant to either submit a modified application or submit supplementary materials to accompany the original specific site plan application materials. In all cases major modifications to approved specific site plans shall be subject to review and approval procedures in accordance with Section 17.07 (G) (M).
- (B) Minor modifications. Minor modifications are defined as modifications which do not change the PUD's permitted uses and do not exceed any of the limitations and conditions of the PUD's approval Ordinance, and which result in an increase of less than 5% in building coverage; an increase of less than 5% in dwelling unit density; changes not exceeding 25 feet in building location not affecting front yard setbacks from streets or setbacks from exterior property lines of the PUD; an increase of not more than five feet in the height of a building or structure; minor redesign of a building such as significant alterations to exterior materials or colors, provided the redesign does not significantly alter the central architectural design or theme of the building; or changes not considered to be of major or incidental modification character. The Planning Board shall, by motion, approve or disapprove minor modifications to approved specific site plans. Such approval shall be based on a Board determination that the modifications are not in conflict with the intent and character of the approved specific site plan and such modifications do not change the permitted uses for the specific PUD. Minor modifications do not require a public hearing and Planning Board's decision shall be final.
- (C) Incidental modifications. Incidental modifications are defined as modifications which do not change the PUD's permitted uses and do not exceed any of the limitations and conditions of the PUD's approval Ordinance, and which result in less building coverage due to decreasing the size of structures; a decrease in the number of structures; a decrease in the number of dwelling units in a PD-1 or PD-4; minor redesign of PUD streets, such as adjustments in a turning radius; minor redesign, such as a realignment, of pedestrian circulation facilities or parking or loading areas; changes in landscaping or screening materials that do not alter the intended function of the landscaping or screening; an increase of five feet or less in the height of a building or structure; minor revisions of building elevations such as realignment of major building entrances or window placement, or significant alterations to exterior building materials or colors, provided the revisions do not alter the central design or architectural theme of a building; or changes not considered to be of minor or major modification character. The Community Development Administrator or shall approve or disapprove incidental modifications to approved specific site plans. Such approval shall be based on the Community Development Administrator's determination that the modifications are not in conflict with the intent and character of the approved specific

site plan and such modifications do not change the permitted uses for the specific PUD. In the alternative and at the discretion of the Community Development Administrator, the Community Development Administrator may request that incidental modifications be approved by the Planning Board.

- (D) Expiration of specific site plan modifications. Unless an extended approval period is granted in accordance with division (E) of this section, approval or approval with supplementary conditions of any modification shall expire if, in the judgment and determination of the city, the actual start of construction has not begun in the approved area of the respective specific site plan within two years from the effective date of Staff, Planning Board and/or City Council approval of the respective modification. In the event of expiration of an approved modification, the applicant shall start afresh, submit a modification application in accordance with Section 17.11.
- (E) Extension of specific site plan modification approval period. Upon request by the owner, a one-time only, administrative extension of the two-year approval period for a modification may be granted by Staff, Planning Board and/or City Council (depending on which entity approved the original modification). In the event such an extension is granted, the period of extension shall not exceed one year. The owner shall submit a written request for an extended modification approval period to the Community Development Administrator. The written request shall be submitted no sooner than nine months prior to, but no later than 90 days prior to the expiration date of the approved modification which is the subject of the request for approval period extension. The written request shall include the necessity for the extension; submission of documentation and evidence that the owner has made a reasonable effort to begin the actual start of construction; the reason(s) why construction has not actually started to date and will not start prior to expiration of the specific site plan; the requested length of time to extend the modification approval period (not to exceed one year); For minor or major modification extension requests, within 45 days from the Community Development Administrator's receipt of the owner's written request, the Community Development Administrator shall forward the owner's request and the Community Development Administrator's comments and recommendation to the Planning Board in the case of a minor mod, to City Council in the case of a major mod. Within 45 days after Planning Board or City Council receives the request and the Community Development Administrator's comments and recommendation, the Council shall take administrative action upon the request. The City Council shall, by resolution and administrative action, either disapprove the request for extension, approve the request for extension as submitted, or approve the request for extension for a lesser period of time than requested by the owner. Any extension of approval period for a modification shall become effective and begin to run on the date of Planning Board or Council approval of such extension. The Community Development Administrator (in the case of incidental mods) Planning Board and/or City Council shall approve an extended approval period for a modification only when the following conditions are satisfied:
 - (1) The owner has submitted a written request for an extended approval period for the modification;

- (2) The subject property currently has a PUD zoning classification;
- (3) The approval period for the modification has not been previously extended by City Council and/or Planning Board;
- (4) An extension of approval period for the modification will not cause the city to breach any PUD Agreement made with the owner;
- (5) The owner, in the judgment of the City Council and/or Planning Board, has taken reasonable steps and made reasonable efforts to actually start construction;
- (6) An extended approval period for the modification will not violate the purpose and intent of Section 17.01 through 17.16 and this Zoning Code;
- (7) An extended approval period for the modification promotes the health, safety and general welfare of the present and future inhabitants of the city; and
- (8) The period of extension for the modification does not exceed one year.
- (E) Previously approved modifications. A modification approved by City Council and/or Planning Board shall be deemed null and void upon approval of a subsequent specific site plan for all or part of the same property, unless otherwise conditioned by Planning Board and/or City Council.

17.12 PD-1 PLANNED RESIDENTIAL DEVELOPMENT DISTRICTS

(A) Intent. To provide flexibility in the arrangement and design of residential dwellings based upon a unified development plan conceived and carried out for the entire PUD tract. Within this district, appropriate and reasonable population density is maintained while a variety of dwelling unit types is permitted. Developers are to retain natural features such as topography, trees, and drainage ways. Such developments are generally characterized by a significant proportion of usable open space, a unified design concept with particular attention devoted to the periphery of the development, with the objective of compatibility of the development with its surroundings.

(B) Permitted, accessory and conditional uses.

(1) The permitted, conditional, and accessory uses in PD-1 districts are those uses included as permitted and accessory uses in residential districts, excluding manufactured home developments. One or more of these permitted and accessory uses may be excluded from the specific PD-1 when the City Council determines that the specific permitted or accessory use normally allowed in the listed residential districts is inappropriate for the specific R-

PUD. Such exclusion(s) shall be based upon at least one of the following findings that the specific excluded use:

- (a) Cannot be serviced by adequate public utilities; or presents the potential for significant environmental damage and a satisfactory plan of mitigation has not been provided by the applicant; or is inconsistent with the overall character and other uses of the proposed PUD; or is inappropriate for the topography of the site; or is incompatible with surrounding legal land uses or other approved land uses; or is inconsistent with the City's Land Use Plan or other approved plans of the city; or will create hazardous traffic conditions; or will impose an unmitigated burden on public services and facilities, such as fire and police protection; or
- (b) Will not promote the purpose and objectives of the planned unit development provisions of this code; or
- (c) Does not advance the general welfare of the community and the immediate vicinity and will adversely affect or impact adjoining or surrounding development.
- (2) Upon approval by City Council in accordance with Section 17.10 (B) those uses included as conditional uses in residential districts are allowed in a PD-1.
- (C) Land use intensity. For PD-1s, the coverage of the total gross area of the development shall not exceed 60% of the entire development. For the purposes of this section, coverage shall be interpreted to include the area covered by all buildings, all parking areas, all driveways, and all public and private streets and paved rights-of-way. Structures and other impervious surfaces included within, and developed as an integral part of an area of open space may, at the sole discretion of the Planning Board and City Council, be excluded from the coverage of the total gross area percentage. Developers of PD-1s are encouraged to achieve an approximate equal distribution of coverage for each individual phase or subarea of the total PUD.

(D) Land use density.

(1) The overall density of a PD-1 shall not exceed the maximum density allowed in the Land Use Plan for dwelling units per acre, excluding public rights-of-way that are subject to impact fee credits. At its discretion, City Council can increase the maximum density when such increase is determined to improve site design and the overall aesthetic of the development.

(2) Lot requirements:

(a) The minimum lot width, in one-family PD-1s shall be 60 feet.

- (b) When designing site plans for proposed, one-family PD-1s which will abut existing residential developments, the lot size for the proposed lots which are directly adjacent to the existing residential lots shall be, at a minimum, the same area as the minimum lot size requirement of the existing adjacent residential lots, or 20,000 square feet, whichever is less (i.e. if adjacent to R-1B, then minimum lot is size shall be 16,000 square feet).
- (c) There shall be a 50-foot buffer between multi-family projects that have buildings with more than two units abutting existing, or approved (but not yet constructed) single family projects.
- (E) Building height regulations. No building shall be erected in excess of fifty (50) feet in height, except, at the discretion of the City Council, the maximum height may be increased when such increase is determined to improve site design and the overall aesthetic quality of the development, does not adversely affect surrounding neighborhoods and public facilities, does not obscure light and air sources to immediately surrounding structures, and satisfies applicable fire and safety regulations.
- (F) Landscaping and screening. See Section 18.7, Screening, Buffering, and Landscaping. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (G) Exterior lighting. See Section 18.31, Light and Glare. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (H) Parking. Parking shall be provided in accordance with Section 18.16, Off-Street Parking Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (I) Off-street loading. Spaces shall be provided in accordance with Section 18.15, Off-Street Loading Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (J) Signs. In conjunction with and at the same time as submission of a specific site plan

application, the applicant may submit a proposed sign limitation and control package specifically fashioned for the PUD. The applicant may submit a proposed sign limitation and control package specifically fashioned for the PUD for separate approval as a minor modification to be considered and decided upon by the Planning Board.

(K) Variances. Variances within PD-1 districts shall be applied for in accordance with Section 20.13. They shall be considered on a case-by-case basis for individual parcels. The granting of a variance for an individual parcel shall not be considered approval of a variance for any other parcel within the same or any other district.

17.13 PD-2 PLANNED BUSINESS DEVELOPMENT DISTRICTS

(A) Intent. To provide for the development of business, office and commercial establishments within a unified commercial area and plan of development. Within this district, business, office and commercial establishments can be flexibly located to achieve compatible exterior physical design, to utilize the site more efficiently than allowed by standard zoning regulations, to take advantage of natural features of the site, to achieve a higher degree of pedestrian and vehicular separation, to eliminate undesirable features of strip commercial development, to reduce vehicular traffic conflicts within the site and with public rights-of-way, and to enhance compatibility of the development with its surroundings.

(B) Permitted, accessory, and conditional uses.

- (1) The permitted, conditional, and accessory uses in PD-2 zoning districts are those uses included as permitted, conditional and accessory uses in Office Building District (O-1), Highway Business District (B-1), Convenience Shopping District (B-2), Neighborhood Business District (B-3), and Central Business District (B-4). One or more of these permitted, conditional, and accessory uses may be excluded from the specific PD-2 when the City Council determines that the specific permitted, conditional or accessory use is inappropriate for the specific PD-2. Such exclusion(s) shall be based upon at least one of the following findings that the specific excluded use:
 - (a) Cannot be serviced by adequate public utilities; or presents the potential for significant environmental damage and a satisfactory plan of mitigation has not been provided by the applicant; or is inconsistent with the overall character and other uses of the proposed PUD; or is inappropriate for the topography of the site; or is incompatible with surrounding legal land uses or other approved land uses; or is inconsistent with the City's Comprehensive Plan or other approved plans of the City; or will create hazardous traffic conditions; or will impose an unmitigated burden on public services and facilities, such as fire and police protection; or
 - (b) Will not promote the purpose and objectives of the planned unit development provisions of this code; or

- (c) Does not advance the general welfare of the community and the immediate vicinity and will adversely affect or impact adjoining or surrounding development without satisfactory mitigation measures.
- (2) Wireless telecommunication facilities shall automatically be a conditional use permitted in a PD-2 district.
- (C) Building height regulations. The maximum height of any building in a PD-2 District shall be 70 feet, except, at the discretion of the City Council, the maximum height may be increased when such increase is determined to improve site design and the overall aesthetic quality of the development, does not adversely affect surrounding neighborhoods and public facilities, does not obscure light and air sources to immediately surrounding structures, and satisfies applicable fire and safety regulations.

(C) Landscaping and screening.

- (1) See Section 18.17, Screening, Buffering, and Landscaping. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (2) Any commercial development proposed to be located within 1,000 feet of any existing or proposed residential district or project shall present and implement plans for providing a barrier from noise generated by the activities of the commercial development. These activities include but are not limited to loading dock and delivery activities, truck movements, idling and parking, waste collection, compactor or shredding activities, roof mounted and ground mounted mechanical equipment, and parking lots and driveways. Methods which may be utilized include earthen mounding, solid masonry walls, permanent engineered sound barriers, or other methods or materials of equal affect as those listed above. The noise barrier shall be tall enough to exceed the height of any noise source such as truck engines and exhaust stacks.
- (E) Exterior lighting. See Section 18.31, Light and Glare. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (F) Parking. Parking shall be provided in accordance with Section 18.16, Off-Street Parking Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.

- (G) Off-street loading. Spaces shall be provided in accordance with Section 18.15, Off-Street Loading Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (H) Signs. In conjunction with and at the same time as submission of a specific site plan application, applicants may submit a proposed sign limitation and control package specifically fashioned for the PUD. The applicant may also submit a proposed sign limitation and control package specifically fashioned for the PUD for separate approval as a minor modification to be considered and decided upon by the Planning Board.

17.14 PD-3 PLANNED INDUSTRIAL DEVELOPMENT DISTRICTS

(A) Intent. To provide for the development of varied or similar industrial establishments within a unified industrial area and plan of development. Within this district, industrial establishments can be flexibly located to achieve compatible exterior physical design, to utilize the site more efficiently than allowed by conventional development standards, to take advantage of natural features of the site, to achieve a higher degree of pedestrian and vehicular separation, to comprehensively provide for necessary services and facilities in accordance with a predetermined plan, to reduce vehicular traffic conflicts within the site and with public rights-of-way, and to enhance compatibility of the development with its surroundings.

(B) Permitted, accessory, and conditional uses.

- (1) The permitted principal uses in PD-3 Districts include manufacturing, processing, warehousing, industrial service activities, and office and associated activities.
- (2) The accessory uses permitted in PD-3 Districts include uses that are customarily incidental to the permitted principal uses identified above, and temporary buildings and uses incidental to construction work, which buildings shall be removed upon the completion or abandonment of the construction work.
- (3) One or more of the permitted and accessory uses identified above may be excluded from a specific PD-3 District when the City Council determines that the specific permitted or accessory use that would otherwise be allowed is inappropriate for that specific PD-3. Such exclusion(s) shall be based upon at least one of the following findings that the specific excluded use:
 - (a) Cannot be serviced by adequate public utilities; or presents the potential for significant environmental damage and a satisfactory plan

of mitigation has not been provided by the applicant; or is inconsistent with the overall character and other uses of the proposed PUD; or is inappropriate for the topography of the site; or is incompatible with surrounding legal land uses or other approved land uses; or is inconsistent with the City's Land Use Plan or other approved plans of the City; or will create hazardous traffic conditions; or will impose an unmitigated burden on public services and facilities, such as fire and police protection; or

- (b) Will not promote the purpose and objectives of the planned unit development provisions of this Zoning Code; or
- (c) Does not advance the general welfare of the community and the immediate vicinity and will adversely affect or impact adjoining or surrounding development without satisfactory mitigation measures.
- (4) <u>Wireless telecommunication facilities shall automatically be a conditional use permitted in all PD-3 Districts.</u>
- (5) <u>Upon approval by City Council, specifically listed permitted and accessory uses allowed in commercial districts may be permitted as conditional uses in a PD-3.</u>
- (C) Building height regulations. The maximum height of any building in PD-3 district shall be 50 feet, except, at the discretion of the City Council, the maximum height may be increased when such increase is determined to improve site design and the overall aesthetic quality of the development, does not adversely affect surrounding neighborhoods and public facilities, does not obscure light and air sources to immediately surrounding structures, satisfies applicable fire and safety regulations, and is necessary for the operation of a proposed industrial use.

(D) Landscaping and screening.

- (1) See Section 18.17, Screening, Buffering, and Landscaping. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (E) Exterior lighting. See Section 18.31, Light and Glare. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (F) Parking. Parking shall be provided in accordance with Section 18.16, Off-Street Parking Regulations. Upon approval of the City Council, any of these requirements may be

modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.

- (G) Off-street loading. Spaces shall be provided in accordance with Section 18.15, Off-Street Loading Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (H) Signs. In conjunction with and at the same time as submission of a specific site plan application, applicants may submit a proposed sign limitation and control package specifically fashioned for the PUD. The applicant may submit a proposed sign limitation and control package specifically fashioned to the PUD for separate approval as a minor modification to be considered on a case-by-case basis for individual parcels.

17.15 PD-4 PLANNED MIXED-USE DEVELOPMENT DISTRICTS

(A) Intent. To provide for the development of business, office, and commercial establishments, and varied residential living opportunities within a unified plan of development. Within the planned development, selected residential uses may be integrated with commercial, business and office establishments, or developed as a subarea with residential neighborhood characteristics. This district provides for flexibility in locating uses, allows for utilization of the site more efficiently than standard zoning regulations, allows for advantageous use of natural features of the site, and allows for residential uses to be selectively integrated with or separated from business, commercial or office establishments.

(B) Permitted, accessory and conditional uses.

- (1) The permitted, conditional, and accessory uses in PD-4 zoning districts are those uses included as permitted, conditional, and accessory uses in residential and business districts, excluding manufactured home developments. One or more of these permitted, conditional, and accessory uses may be excluded from the specific PD-4 when the City Council determines that the specific permitted, conditional, or accessory use is inappropriate for the specific PD-4. Such exclusion(s) shall be based upon at least one of the following findings that the specific excluded use:
 - (a) Cannot be serviced by adequate public utilities; or presents the potential for significant environmental damage and a satisfactory plan of mitigation has not been provided by the applicant; or is inconsistent with the overall character and other uses of the proposed PUD; or is inappropriate for the topography of the site; or is incompatible with surrounding legal land uses or other approved land uses; or is inconsistent with the City's Land Use Plan or other approved plans of

- the city; or will create hazardous traffic conditions; or will impose an unmitigated burden on public services and facilities, such as fire and police protection; or
- (b) Will not promote the purpose and objectives of the planned unit development provisions of this code; or
- (c) Does not advance the general welfare of the community and the immediate vicinity and will adversely affect or impact adjoining or surrounding development without satisfactory mitigation measures.
- (2) Upon approval by City Council in accordance with Section 17.10, those uses included as conditional uses in residential and business districts are allowed in a PD-4.
- (C) PD-4 residential uses. Residential uses in PD-4 Districts are intended to be developed as separate, residential subareas of the PD-4. Residential uses shall only be developed in PD-4 Districts when the following conditions are satisfied.
 - (1) If a non-residential use is planned or developed within a horizontal distance of less than 100 feet from a residential use, residential uses shall be developed only at the second story or higher of buildings in which grade level use is a business, office or commercial PD-4 District permitted, or accessory or approved conditional use.
 - (2) The overall density of a PD-4 shall not exceed the maximum density allowed in the Land Use Plan for dwelling units per acre. When calculating density, the acreage portion of the calculation shall not include undevelopable areas where the natural or environmental conditions (i.e. steep slopes, floodways, wetlands, etc.) would prevent the construction of residential structures without an unreasonable investment to overcome these natural constraints, or areas dedicated as or to be dedicated as public rights-of-way and private streets.
 - (3) The minimum lot width in one-family portions of PD-4s, shall be 60 feet.
 - (4) When designing site plans for proposed, one-family portions of PD-4s which will abut existing residential developments, the lot size for the proposed lots which are directly adjacent to the existing residential lots shall be, at a minimum, the same size as the existing residential lots, or 20,000 square feet, whichever is less.
 - (5) There shall be a 50-foot buffer between multi-family projects that have buildings with more than two units abutting existing, or approved (but not yet constructed) single family projects.

(D) Building height regulations. The maximum height of any building in a PD-4 District shall be 50 feet, except, at the discretion of the City Council, the maximum height may be increased when such increase is determined to improve site design and the overall aesthetic quality of the development, does not adversely affect surrounding neighborhoods and public facilities, does not obscure light and air sources to immediately surrounding structures, and satisfies applicable fire and safety regulations.

(E) Landscaping and screening.

- (1) See Section 18.17, Screening, Buffering, and Landscaping. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, will not adversely affect surrounding neighborhoods and public facilities.
- (2) Any commercial development proposed to be located within 1,000 feet of any existing or proposed residential district or project shall present and implement plans for providing a barrier from noise generated by the activities of the commercial development. These activities include but are not limited to loading dock and delivery activities, truck movements, idling and parking, waste collection, compactor or shredding activities, roof mounted and ground mounted mechanical equipment, and parking lots and driveways. Methods which may be utilized include earthen mounding, solid masonry walls, permanent engineered sound barriers, or other methods or materials of equal affect as those listed above, which shall be approved by the Planning Department prior to installation. The noise barrier shall be tall enough to exceed the height of any noise source such as truck engines and exhaust stacks.
- (F) Exterior lighting. See Section 18.31, Light and Glare. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (G) Parking. Parking shall be provided in accordance with Section 18.16, Off-Street Parking Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (H) Off-street loading. Spaces shall be provided in accordance with Section 18,15, Off-Street Loading Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.

- (I) Signs. In conjunction with and at the same time as submission of a specific site plan application, applicants may submit a proposed sign limitation and control package specifically fashioned for the PUD. The applicant may also submit a proposed sign limitation and control package specifically fashioned to the PUD for separate approval as a minor modification to be considered on a case-by-case basis for individual parcels.
- (J) Variances. Variances of PD-4 districts (in the residential portions only) shall be applied for in accordance with Section 20.13. They shall be considered on a case-by-case basis for individual parcels. The granting of a variance for an individual parcel shall not be considered approval of a variance for any other parcel within the same or any other district.

17.16 PD-5 PLANNED PUBLIC AND PRIVATE BUILDINGS AND GROUNDS DISTRICTS

(A) Intent. To provide fo

- (B) Permitted, accessory, and conditional uses.
 - (1) The permitted principal uses in PD-5 Districts include the following:
 - (a) Churches and other places of worship, including parish and/or parsonage houses;
 - (b) Colleges, schools, and libraries;
 - (c) Public recreational buildings, playgrounds, and athletic fields;
 - (d) Golf courses, including country clubs and swimming pools, including beaches;
 - (e) Cemeteries;
 - (f) Fishing lakes and ponds;
 - (g) Hospitals, sanitariums, charitable institutions, and rest homes;
 - (h) Parks, community centers, and granges;
 - (i) Radio and television transmission stations;
 - (j) Airports;
 - (k) Parking lots:
 - (l) <u>Fraternal organizations</u>, <u>service clubs</u>, <u>and other nonprofit organizations and their buildings</u>, <u>grounds</u>, <u>and meeting places</u>;
 - (m) Nursery centers and daycare centers;
 - (n) Animal hospitals and veterinary clinics;
 - (o) Bingo establishments;
 - (p) Activities by the City, including but not limited to police buildings, fire stations, and other municipal structures and uses; and
 - (q) Banquet facilities with no on-site food preparation.
 - (2) <u>A conditional use permitted in PD-5 Districts include banquet facilities</u> with on-site food preparation.

- (3) The accessory uses permitted in PD-5 Districts include uses that are customarily incidental to the permitted principal uses identified above, and temporary buildings and uses incidental to construction work, which buildings shall be removed upon the completion or abandonment of the construction work.
- (4) One or more of the permitted, conditional, and/or accessory uses identified above may be excluded from a specific PD-5 District when the City Council determines that the specific permitted, conditional, and/or accessory use that would otherwise be allowed is inappropriate for that specific PD-5. Such exclusion(s) shall be based upon at least one of the following findings that the specific excluded use:
 - (a) Cannot be serviced by adequate public utilities; or presents the potential for significant environmental damage and a satisfactory plan of mitigation has not been provided by the applicant; or is inconsistent with the overall character and other uses of the proposed PUD; or is inappropriate for the topography of the site; or is incompatible with surrounding legal land uses or other approved land uses; or is inconsistent with the City's Land Use Plan or other approved plans of the City; or will create hazardous traffic conditions; or will impose an unmitigated burden on public services and facilities, such as fire and police protection; or
 - (b) Will not promote the purpose and objectives of the planned unit development provisions of this Zoning Code; or
 - (c) Does not advance the general welfare of the community and the immediate vicinity and will adversely affect or impact adjoining or surrounding development without satisfactory mitigation measures.
- (5) Wireless telecommunication facilities shall automatically be a conditional use permitted in all PD-5 Districts.
- (6) <u>Upon approval by City Council, specifically listed permitted and accessory uses allowed in commercial districts may be permitted as conditional uses in a PD-5.</u>
- (C) Building height regulations. The maximum height of any building in PD-5 district shall be 50 feet, except, at the discretion of the City Council, the maximum height may be increased when such increase is determined to improve site design and the overall aesthetic quality of the development, does not adversely affect surrounding neighborhoods and public facilities, does not obscure light and air sources to immediately surrounding structures, satisfies applicable fire and safety regulations, and is necessary for the operation of a proposed industrial use.

Item C.Section 8. Item

(D) Landscaping and screening.

- (1) See Section 18.17, Screening, Buffering, and Landscaping. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (E) Exterior lighting. See Section 18.31, Light and Glare. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (F) Parking. Parking shall be provided in accordance with Section 18.16, Off-Street Parking Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (G) Off-street loading. Spaces shall be provided in accordance with Section 18.15, Off-Street Loading Regulations. Upon approval of the City Council, any of these requirements may be modified, provided that such modification will improve the design and functionality of the site, increase benefits to the community that are derived from the development, and will not adversely affect surrounding neighborhoods and public facilities.
- (H) Signs. In conjunction with and at the same time as submission of a specific site plan application, applicants may submit a proposed sign limitation and control package specifically fashioned for the PUD. The applicant may submit a proposed sign limitation and control package specifically fashioned to the PUD for separate approval as a minor modification to be considered on a case-by-case basis for individual parcels.

ARTICLE 18

GENERAL PROVISIONS

18.01

GENERAL REGULATIONS

Except as herein provided, no building or structure shall be erected, converted, enlarged, reconstructed, moved, or structurally altered, nor shall any building or land be used:

- (1) Except for a purpose permitted in the district in which the building or land is located;
- (2) Except in conformance to the height or bulk limits established herein for the district in which the building or use is located;
- (3) Except in conformance to the yard and lot regulations of the district in which the building or use in located;
- (4) Except in conformance to the off-street parking and off-street loading space regulations of the district in which the building or use is located;
- (5) Unless such building or structure is located on a lot as herein defined and in no case shall there be more than one main building on a lot except as specifically provided hereinafter.

YARD REQUIRED FOR CORNER AND THROUGH LOTS

- (1) In any district the side yard of a corner lot that abuts the side street shall have the same setback requirements as the front yard.
- (2) A rear yard shall be provided parallel to and opposite from the front yard.
- (3) On through lots the front yard requirements shall apply to all street frontages.

RESERVED

LOTS ADJOINING ALLEYS

In calculating the area of a lot that adjoins an alley, for the purpose applying lot are requirement of this **O**erdinance, one-half

width of such alley abutting the lot shall be considered as part of

lot.

ACCESSORY BUILDINGS

18.02

18.03

18.04

of

(1/2) of the

such

18.05A

In a residential zone, no garage or other accessory building shall be erected within a required side yard or front yard.

When located at least sixty (60) feet from the front property line and completely to the rear of the main dwelling, the accessory building may be erected not less than eight (8) feet from the side or rear lot lines. Provided that when access to a garage is from an alley, such garage shall be located not less than ten (10) feet from the alley.

When located less than sixty (60) feet from the front property line and not completely to the rear of the dwelling, garages shall be constructed as a part of the main building or connected thereto by a covered breezeway.

Corner lot accessory buildings must be placed within the boundaries of the required rear yard as governed by Article 7, Sec. 7.<u>10</u> 09, Yard & <u>and</u> Lot Requirements. or Article # 17, Sec. 17.05, PD-1 Planned Residential Development and not less than 8' from the side and rear lot lines.

No detached accessory building measuring up to three hundred (300) square feet in Residential and B-1 Districts shall exceed one (1) story or twelve (12) feet in height and no detached accessory building in

Residential and B-1 Districts measuring from 301-600 square feet shall exceed one (1) story or 15' in height.

Detached accessory buildings (including garages) constructed on a one-family lot shall not exceed, in the aggregate, six hundred (600) square feet.

SWIMMING POOLS

18.05B

(1) Every outdoor swimming pool and the equipment and appurtenances thereof shall be completely surrounded by a fence or wall, or the entire rear yard of the property on which it is located, shall be completely fenced, or walled to prevent uncontrolled access. The fence or wall shall be not less than four five (54) feet in height, which shall be of a solid woven construction with no one opening dimension exceeding three (3) inches). And shall be maintained in good condition with a gate or device that shall be self-locking and secured when not in

use. Fences surrounding pools shall meet the requirements of Section 18.06(1)(a). A dwelling, house, or accessory building may be used as part of such enclosure. The walls of any above-ground pool shall not be deemed to constitute such a fence in and of themselves. All gates or doors opening through such enclosure shall be equipped with a lock and with a self-closing and self-latching device for keeping the gates or doors securely closed at all times when the pool is not in actual use or is unattended. ;

(2) Above ground pools, the circumference of which are decked and completely fenced, and the top of the fence is not less than **four** five (45) feet above the ground level are considered to meet the fence height requirements. All other provisions in Section 18.05(B)(1) apply.

18.05C

TEMPORARY STORAGE UNITS

A temporary storage unit may be permitted in any residential, commercial, or agriculturally zoned lot provided the following criteria are met:

- (a) A permit is required for the use and placement of a temporary storage unit, if the unit is in place for more than fourteen (14) days.
- (b) Only one unit may be permitted on any property at any time for a maximum of ninety (90) consecutive days, per calendar year.
- (c) Each use requires a permit. Permits can be issued, and units onsite, consecutively, but not concurrently.
- (d) Unit must be placed on a hard surface driveway in the front or side of the building, totally visible from the right-of-way.
- (e) Unit must be placed on private property, behind the sidewalk and out of the right-of-way.
- (f) Unit must maintain a minimum of two (2) feet of separation from any building located on the same parcel as the unit.

- (g) Unit shall remain closed when not being loaded or unloaded.
- (h) No substance may leak from inside the unit. No trash or refuse may be stored within the unit.
- (i) No unit shall be located closer to an adjacent parcel than the required minimum side or rear yard setback for accessory uses in the district the unit is located.
- (j) No unit shall be used for human or animal occupation.
- (k) No unit shall be located in such a manner on any property as to create a public nuisance.

18.06 <u>FENCES & AND WALLS</u>

- (1) Non corner lots:
 - (a) Side and rear yards:

No fence or wall shall exceed six (6) feet in height unless it is ten (10) or more feet from the property line. In residential zones, chain link, wire or wire mesh fences shall not exceed five (5) feet in height unless ten (10) or more feet from the property line. Side yard fences and walls shall not extend forward of the house or front setback line. All fences shall be constructed with supporting fence posts placed facing the interior of the lot.

(b) Front yard:

No fence or wall shall exceed four (4) feet in height above the ground; be made of wire or chain link; be closer than five (5) feet to the front property line.

- (2) Corner and double frontage lots:
 - (a) Fences and walls on both streets shall meet above front yard requirements.

- (3) Swimming pool fence See Section 18.05B Swimming Pool
- (4) Ornamental fences and those, which do not actually complete the enclosure of a yard, are considered an "Other" permit for purposes of the required fee.

18.07 **REMOVAL OF SOIL, SAND, OR OTHER MATERIAL**

The use of land for the removal of topsoil, sand or other material from the land other than materials from basement excavations is not permitted in any zone except under a temporary permit from the Board of Zoning Appeals; this permit may be denied or issued in appropriate cases after the filing of an application accompanied by a suitable agreement or bond that such removal will not cause stagnant water to collect or leave the surface of the land, at the expiration of such permit, in an unstable condition, or unfit for the growing of turf or for other land uses permitted in the district in which such removal occurs.

18.08 <u>ESSENTIAL SERVICES</u>

Essential services shall be allowed in any district insofar as permitted, authorized, or regulated by law or other ordinance. Buildings required in conjunction with an essential service may be permitted in any district when approved by the Planning Board. In granting such permission, the Planning Board shall take into consideration the location, size, use and effect such building will have on the adjacent land and buildings.

18.09 **EXTERNAL EFFECTS**

No land, building or structure in any district shall be used or occupied in any manner so as to create any dangerous, injurious, noxious or otherwise objectionable fire, explosive, or other hazard; noise, brilliant light, vibration; smoke, dust, fumes, odor or other form of air pollution; heat, cold, dampness, electrical or electronic disturbances, nuclear radiation, or any other condition, substance or element as per applicable federal standards to any person or property outside of the premises on which such building, structure or use is located; such uses when lawfully permitted under the provisions of this **O**ordinance shall be operated in a manner so as to insure that the property rights of all other parcels of land will not

2024-O-6 Exhibit A

be adversely affected to the extent of reducing the enjoyment of property rights thereon.

18.10 <u>OUTDOOR STORAGE AND WASTE DISPOSAL</u>

Every use shall be operated in accord with the following

provisions:

- (1) No highly flammable or explosive liquids, solids or gases shall be stored in bulk above ground except in an industrial district. Tanks or drums of fuel directly connected with heating devices or appliances located on the same lot as the tanks or drums of fuel are excluded from this provision;
- (2) All outdoor storage facilities for fuel, raw materials and products shall be enclosed by a fence, wall or planting to conceal such facilities from adjacent residential property;
- (3) No materials or wastes shall be deposited upon a lot in such form or manner that they may be transferred off the lot by wind, flood or natural causes or forces;
- (4) All materials or wastes which might cause fumes, dust or which constitute a fire hazard or which may be edible or attractive to rodents or insects shall be stored outdoors only in closed containers constructed of impervious material.

18.11 **PROJECTIONS INTO REQUIRED YARDS**

- (1) Chimneys, flues, sills, pilasters, cornices, eaves, gutters, and other similar features may project into a required side yard a maximum of eighteen (18) inches.
- (2) Front yards:

No structure may project into a required front yard; however, unroofed porches and steps may extend from the dwelling into the required front yard a maximum of ten (10) feet.

(3) No structure may project into a required side yard except that, where a single lot under one ownership existed in a residential district at the time of passage of this Ordinance, and such lot is of

insufficient width to meet the side yard requirements of this Ordinance, the Board of Zoning Appeals may grant a minimum variance to permit the construction of a one-family residence.

18.12 **EXCEPTIONS TO HEIGHT LIMITATIONS**

Chimneys, domes spires and necessary mechanical appurtenances and radio and television towers may exceed district height limitations.

- (1) Public, semi-public or public service buildings, hospitals, institutions or schools, where permitted, may be erected to a height not exceeding ninety (90) feet when the required side and rear yards are each increased by one (1) foot for each foot of additional building height above the height regulations for the district in which the building is located, provided, however, that prior to the issuance of a zoning permit for any structure which is planned to exceed thirty-five (35) feet in height, the Planning Board shall make a finding that any such excessive height will not be detrimental to the public safety; to the light, air or privacy of any other structure or use currently existing or approved for construction.
- (2) Commercial radio and television towers shall be located centrally on a continuous parcel having a dimension at least equal to the height of the tower measured from the center of the base of the tower to all points on each property line.
- (3) Except as otherwise provided herein, an amateur radio station antenna shall be erected at heights and dimensions to accommodate such service communications. Required permits must constitute the minimum practicable regulation necessary to accomplish the e<u>C</u>ity's legislative authority purposes.

18.13 <u>TEMPORARY USES</u>

In any district, subject to the conditions stated below, the Enforcing Officer Community Development Administrator or designee may issue a permit for the following temporary uses:

(1) Temporary building or yard for construction office, material, or equipment, provided such use is adjacent to the construction site and removed when construction is completed. Each permit shall be valid

for six (6) months and may be renewed if construction is underway and shall be removed when construction is completed or discontinued for more than thirty (30) days.

- (2) Temporary office incidental and necessary to real estate sales and rentals. Each permit shall be valid for one (1) year and may be renewed for one (1) additional year if conditions warrant such renewal.
- (3) Building and yard locations shall be subject to such conditions and safeguard as the Enforcing Officer Community Development Administrator or designee may deem necessary to preserve the character of the surrounding area.
- (4) Temporary tents for special events are permitted in residential districts for a maximum of one week per event. A zoning permit is required for temporary tents measuring 400 square feet or larger.

18.14 <u>MAJOR STREET SETBACKS</u>

Any building or structure shall hereinafter be constructed in accordance with the required front yard setback in the district in which it is to be located, measured from the required right-of-way line on major streets and secondary streets designated as such on the Major Thoroughfare Plan.

18.15 **OFF STREET LOADING REGULATIONS**

On the same premises with every building or structure or part thereof, erected and occupied for commerce, industry, public assembly, or other uses involving the receipt or distribution by vehicles of materials or merchandise, there shall be provided and permanently maintained adequate space for standing, loading, and unloading services in order to avoid undue interference with public use of the streets or alleys in conformance to the following:

- (1) General Provisions:
 - (a) Screening:

Off street loading spaces that adjoin or are across a street or alley from property zoned for residential use, shall have a dense evergreen planting, fence, masonry wall or such other screening, as may be determined by the Planning Board. The Planning Board shall also determine the height, location and density of screening used to provide adequate protection to adjoining property.

(b) Entrances and Exits:

Off-street loading spaces shall be provided with entrances and exits not less than twelve (12) feet in width and so located as to minimize traffic congestion.

(c) Dimensions:

Each off-street loading space shall be not less than ten (10) feet in width, twenty-five (25) feet in length and fifteen (15) feet in height, exclusive of access drives.

(d) Projections into yards:

Off street loading spaces may occupy all or any part of any required rear yard space.

(2) Amount of Loading Space Required:

The minimum amounts of off-street loading space shall be provided according to the table below. An area adequate for maneuvering, ingress and egress shall be provided in addition to required loading space.

Sq Ft of Gross Floor Area	Required # of Spaces
(a) Up to 10,000 sf	1
(b) 10,001 to 20,000 sf	2
(c) 20,001 to 40,000 sf	3
(d) 40,001 to 75,000 sf	4
(e) 75,001 to 125,000 sf	5
(f) For each add'1 50,000 sf	1 add'l space

18.16 **OFF STREET PARKING REGULATIONS**

Hereafter no building shall be erected or altered, and no land used unless there be provided adequate off-street parking space or spaces for the needs of tenants, personnel, and patrons together with means of ingress or egress.

(1) General Provisions:

- (a) Single Family Residential off-street parking spaces shall consist of a parking strip, driveway, garage, or combination thereof and shall be located on the premises they are intended to serve and subject to the provisions of Sections 18.03, 18.04 and 18.05, Accessory Buildings, of this Ordinance.
- (b) Any area once designated as required off street parking shall never be changed to any other use unless and until equal facilities are provided elsewhere.
- (c) Off street parking existing at the effective date of this **O** ordinance in connection with the operation of an existing building or use shall not be reduced to an amount less than hereinafter required for a similar new building or new use.

 (d) Two or more buildings or uses may collectively provide the required off street parking in which case the required number of parking spaces shall not be less than the sum of the requirements for the several individual uses computed separately.
- (e) In the instance of dual function of off-street parking spaces where operating hours of buildings do not overlap, the Board of Zoning Appeals may grant an extension.
- (f) The storage or sale of merchandise or the commercial repair of vehicles is prohibited.
- (g) For those uses not specifically mentioned, the requirements for off street parking facilities shall be in accord with a use which the Board of Zoning Appeals considers as being similar in type.
- (h) Ingress and Egress:

A suitable means of ingress or egress for vehicles to premises used for parking shall be provided and shall open directly from and to a public street, alley, or highway. The width of any exit, or entrance, adjoining property or opposite property zoned for residential uses shall be approved by the Traffic Engineer or Planning Board prior to obtaining any permit; therefore, the Traffic Engineer or the Planning Board may require the owner to provide acceleration and/or de-acceleration lanes where

(i) Fencing, wheel stops, or other physical barriers shall be provided for all boundaries of the parking area except at points of ingress and egress to prevent encroachment of

traffic volumes indicate the need.

vehicles.

(i) Pavement:

- (1) All parking lots or parking spaces shall be surfaced with a hard dust-free surface in conformance to the standards of the Bellbrook Service Department. For zoning purposes, compacted gravel is not considered a dust-free surface;
- (2) The parking of vehicles on grass or other areas not designated for parking shall be prohibited.

(k) Lighting:

If the parking lot is to be open for use after dark, it shall be provided with not less than one half (1/2) lumen of light per square foot of parking lot surface or the recommendation of the

City Engineer. Lights shall be shielded so as not to shine directly, or in an offensive manner, on the adjoining residential property.

(1) Screening:

When a parking lot abuts a residential zone, there shall be permanently maintained along such boundary, screening as provided in Section 18.17.

(m) Plot Plan to be Filed:

Prior to constructing an accessory parking lot, the owner or person in charge of the land to be used for parking shall submit a plot plan to the Zoning Inspector Community

Development Administrator who will submit same to the Planning Board, the Bellbrook Service Department,

Bellbrook Police Department or other agencies for their review and recommendations. Such plot plan shall show the boundaries of the property, location of adjacent houses, parking spaces, circulation patterns, drainage plan and construction plan for boundary walls and planting plan.

(n) Restricted Accessory Parking Areas:

The Board of Zoning Appeals may permit accessory parking within an adjacent lot zoned for residential uses, providing:

- (1) Such lot is necessary for the public convenience and will not have an adverse effect on adjacent properties.
- (2) A public hearing is held in accordance with the procedure given in Section 20.08.
- (3) Compliance with all provisions of Section 18.16 of this **O**erdinance.
- (4) No parking shall be permitted between the street line and the building line prevailing in the zone in which the proposed parking area is to be located. The resulting open area shall be planted in grass or otherwise landscaped to create a permanent green area.
- (5) A dense evergreen planting with a minimum height of four (4) feet and a mature height of at least

five feet, six inches (5'6"), or solidly constructed decorative fence shall be permanently maintained along the mutual boundary of the restricted accessory parking area and adjacent land zoned for residential uses except for the portion of such boundary located within a required front yard.

- (6) Whenever a lot located in a residential zone is used for accessory parking purposes and is located across the street from land in a residential zone, that portion of the lot used for parking purposes shall be screened from the street as specified in paragraph five (5) above, except for access drive. Such screening to be placed along the setback line.
- (7) Ingress and egress for vehicles to any premises used for parking under conditional use permit by the Board of Zoning Appeals shall be by means of streets or alleys through business or industrial areas, not by means of streets or alleys through residential areas.
- (o) When Off Street Parking Cannot be Located on the Lot with the Use:

Where off street parking required in connection with a use cannot be located on the lot with the use, such parking may be provided off the lot, subject to the following requirements:

- (1) If the use is multi-family, hotel, motel or tourist home, the off-lot spaces shall be within two hundred (200) feet of the principal entrance or the entrance for individual occupants for whom spaces are reserved.
- (2) If the use is other than multi-family, hotel, motel or tourist home, the furthest portion of the parking lot shall be within five hundred (500) feet of an entrance to the establishment.

- (3) Distances indicated above shall be measured along routes generally available to the pedestrians involved.
- (4) If and so long as satisfactory attendant parking service is provided, or if for employees of the establishments, parking spaces may be at greater distances than those indicated above.
- (5) Such off-lot spaces shall be located only in districts in which similar off-street parking is permitted.
- (6) The off-lot parking area shall be:
 - (A) Held in fee simple by the same owner as the use requiring the off-street parking space or,
 - (B) Under lease, rental, or other form of agreement satisfactory to the administrative official as assuring continuing availability for required off street parking for the use, or
 - (C) Established by the City for the purpose of providing such off-lot parking for specific areas and/or land uses and with number of spaces allocated by specific individual lots. Where, and to the extent that such lots serve other uses, spaces allocated may be similarly marked and reserved, in whole or in part, or may in whole or in part be made available for general public use.

(2) AMOUNT OF OFF-STREET PARKING SPACE REQUIRED:

The amount of required off street parking spaces for new uses or building additions thereto and additions to existing buildings, shall be determined in accordance with the following minimum parking provisions provided that no parking area shall project

into a required front yard in any residential district or be permitted between the curb line and building line in any district, except as hereinafter noted.

(a) One Family Dwelling:

Two (2) spaces for each family unit plus one (1) space for each roomer, one (1) of which may project into the required front yard.

(b) Two Family Dwelling:

Two (2) parking spaces for each family, plus one (1) space for each roomer.

(c) Multiple Family:

Two (2) parking spaces per dwelling unit plus one (1) space per each employee.

(d) Apartment Hotel:

One (1) space per apartment, plus one (1) for each employee.

(e) Housing for the Elderly:

One (1) parking space for each two (2) units, plus one (1) space for each employee.

(f) Boarding House:

One (1) parking space for each sleeping room.

(g) Hotel or Motel:

One (1) space per unit, plus one (1) for each employee.

(h) Churches:

One (1) parking space for each three (3) seats in the main building.

(i) Hospitals:

One (1) for each two (2) beds, plus one (1) for each staff doctor, plus one (1) for each two (2) full time employees on shift, including nurses.

(j) Sanitariums, Convalescent Homes or Children's Homes:

One (1) for each two (2) beds, plus one (1) for each two (2) employees.

(k) Elementary and Junior High Schools:

One (1) parking space for each employee, plus parking space for each eighty (80) square feet in the main auditorium, not containing fixed seats, or one (1) space for six (6) fixed seats in the main auditorium whichever is greater.

(l) Senior High Schools:

One (1) parking space for each employee, plus one (1) parking space for each ten (10) students, or one (1) parking space for each eighty (80) square feet of floor area in the main auditorium not containing fixed seats, or one (1) parking space for each six (6) fixed seats in the main auditorium, whichever is greater.

(m) College and Business University:

One (1) for each two (2) employees, plus one (1) for each three (3) students.

(n) Libraries, Museums or Art Galleries:

One (1) for each six hundred (600) square feet of floor area, plus one (1) for each three (3) employees.

(o) Post Office:

One (1) for each five hundred (500) square feet of floor area, plus one (1) for each three (3) employees.

(p) Private Clubs, Lodges:

One (1) parking space for each three (3) persons allowed by fire, health or building codes.

(q) Bowling Alleys:

Five (5 parking spaces for each alley.

- (r) Public Golf Courses:
- Six (6) parking spaces for each golf hole, plus one (1) space for each employee.
- (s) Sports Arenas (Indoor), Auditoriums, Theatres, Assembly Halls (other than in schools): One (1) parking space for each two and one half (2 ½) seats, plus one (1) for each two (2) employees.

(t) Stadium, Sports Arena or Similar Place of Outdoor Assembly:

One (1) parking space for each three (3) seats.

(u) Professional Offices, Medical Clinics:

One (1) parking space for each two hundred fifty (250) square feet of floor area with a minimum of twelve (12) spaces. When professional offices are maintained in a residence as a home occupation, one (1) parking space for each one hundred (100) square feet or major fraction thereof of the residence used as office area shall be provided in addition to that required for the residing family or families.

(v) Office Building:

One (1) parking space for each two hundred (200) square feet of gross floor area excluding any floor space used for parking.

(w) Banks, Dry Cleaning, Laundries and Similar Service Business:

One (1) parking space for each two hundred fifty (250) square feet of floor space.

- (x) Drive-In Banks with Inside Customer Service: Five (5) spaces for each teller window, plus one (1) for each employee; without inside customer service, one (1) space for each employee.
- (y) Auto Service Station: Six (6) spaces per bay.
- (z) Automobile Sales and Service Garage:

One (1) parking space for each two hundred (200) square feet of floor area in the main display room plus one (1) space for each employee.

(aa) Used Car Lot:

One (1) space for each fifteen hundred (1500) square feet of lot area.

(bb) Barber Shops and Beauty Parlors:

One (1) for each chair, plus one (1) for each employee.

(cc) Drive-In Eating Establishments and Fast-Food Establishments (where food is served directly to customers in parked motor vehicles, or food is served within the structure but may be consumed within the structure or on the premises outside the structure):

Three (3) parking spaces for each one hundred (100) square feet of gross floor area.

(dd) Furniture and Appliances, Household Equipment, Decorator, Electrician, Shoe Repair:
One (1) parking space for each eight hundred (800) square feet of floor area plus one (1) space for each two (2) employees.

(ee) Laundromats:

One (1) space for each two (2) washing machines.

(ff) Mortuaries or Funeral Homes:

One (1) parking space for each fifty (50) square feet of floor area in the slumber rooms, parlors, or individual funeral service rooms.

- (gg) Restaurants, Taverns, and Carry Outs Taprooms and Wine Bars: One (1) parking space for each one hundred (100) square feet of gross floor area.
- (hh) Retail Stores and Shopping Centers: Six (6) parking spaces for each one thousand (1000) square feet of gross floor area.
- (ii) Contractor Yards or Plant Storage Yard: One (1) space for each three (3) employees.
- (jj) Warehouses, Wholesale Stores: One (1) parking space for each eight hundred (800) square feet of floor area.
- (kk) Manufacturing Plants or Research Laboratories:

One (1) space for each one and one half (1 $\frac{1}{2}$) employees per largest work shift.

(ll) In the case of a use not specifically mentioned, the requirements for off street parking shall be the same as for a similar use specifically mentioned, similarly, to be determined by the Enforcing Officer Community

Development Administrator or designee.

(mm) Brewpubs, Microbreweries, Microwineries, and Microdistilleries: One (1) parking space for each one hundred (100) square feet of gross floor area for restaurant and/or retail service and one (1) parking space for each three hundred (300) square feet of gross floor area for production.

18.17

SCREENING, BUFFERING, AND LANDSCAPING

Hereafter no buildings or structures shall be erected, altered, or enlarged nor shall land be used for any non-residential use on a lot that adjoins or faces any residential district until a plan for screening has been submitted and approved by the <u>Community Development Administrator</u> <u>Enforcing Officer</u> or Planning Board.

- (1) Screening shall be provided for one (1) or more of the following purposes:
 - (a) A visual barrier to partially or completely obstruct the view of unattractive structures or activities;
 - (b) As an acoustic screen to aid in absorbing or deflecting noise;
 - (c) For the containment of debris and litter.
- (2) Screening may be one of the following or a combination of two (2) or more:
 - (a) A solid masonry wall;
 - (b) A solidly constructed decorative fence;
 - (c) Louvered fence;
 - (d) Dense evergreen plantings;
 - (e) Deciduous trees and shrubs.
- (3) Location of Screening:

Whenever any non-residential use abuts a residential district, a visual screening wall, fence, or a <u>dense</u> planting shall be erected or placed along such mutual boundary lines. <u>Where a new residential or commercial development abuts an existing, or approved but unconstructed residential development, a fifty-foot (50') landscape buffer shall be required. At its discretion, City Council can determine the appropriate landscape buffer in planned unit development districts.</u>

(4) Height of Screening:

Visual screening walls, fences or plantings shall be at least five feet, six inches (5'6") high except in required front yards when maximum height shall be not greater than two feet (2'0").

(5) Depth or Width of Screening:

Screening for purposes of absorbing or deflecting noise shall have a depth of at least fifteen (15) feet of dense plantings or a solid masonry wall in combination with decorative plantings.

(6) Protection:

Whenever required screening is adjacent to parking areas or driveways, such screening shall be protected by bumper blocks, posts or curbing to avoid damage by vehicles.

(7) Landscaping:

All yard areas not used for buildings, parking, or ways for pedestrians or vehicles, shall be planted with a mix of grass, shrubbery, shade trees, ornamental trees, evergreen trees, perennials, annuals, or ground covers, as to provide attractive green areas. This requirement applies to all zoning districts.

18.18 **RESERVED**

18.19 **SEPTIC TANKS OR WELLS**

Any residential construction utilizing wells and/or septic tanks shall be situated on a lot having a minimum site area of three fourth (3/4) of an acre.

18.20 **PERMITTED SIGNS**

Subsection 18.20A applies and governs signs in all districts. Subsection 18.20B modifies and delineates special controls for signs in the Old Village District.

18.20A PERMITTED SIGNS (GENERAL)

(1) PURPOSE:

The purpose of this sign section of thise $\underline{\mathbf{O}}$ ordinance is to:

- (a) Protect each person's Constitutional right to freedom of speech; and
- (b) Protect the public health, safety, convenience, comfort, prosperity, and general welfare.

(2) OBJECTIVES:

This section, 18.20A, regulates the time, place, and manner in which signs are displayed to achieve the following:

- (a) Primary Objectives:
 - (1) Permit non-commercial signs on any property within the $e\underline{\mathbf{C}}$ ity;
 - (2) Permit signs, which do not create a potential hazard to the public safety; and
 - (3) Permit commercial signs appropriate to the land use and/or zoning classification of each property within the e $\underline{\mathbf{C}}$ ity.
- (b) Secondary Objectives:
- (1) To create a more aesthetically pleasing $\underline{\mathbf{e}}\underline{\mathbf{C}}$ ity; and
 - (2) To eliminate visual clutter within the $e\underline{C}$ ity.

Commentary: A fundamental concept to understanding this sign section 18.20A of the Zoning Ordinance is the classification of a sign into the following two broad categories: (1) permanent versus temporary; and (2) commercial versus non-commercial.

(3) DEFINITIONS:

(a) Building Frontage:

Building frontage shall mean the maximum horizontal width of the ground floor of a building that approximately parallels and faces an adjacent public right-of-way of at least fifty (50) feet in width. In the case of a building with multiple occupants, the maximum horizontal width of the portion of the building where each occupant's main entrance is located shall be considered that occupant's separate and distinct building frontage.

In the case where the ground floor of a building is occupied by two (2) or more different tenants, the portion of the building frontage occupied by each tenant shall be considered a separate and distinct building frontage.

Corner lots and through lots shall be considered to have only one (1) distinct and separate building frontage.

(b) Commercial Message:

Any sign wording, logo, or other representation that, directly or indirectly, names, advertises, or calls attention to a business, product, service, or other commercial activity.

(c) Legible:

A message that is legible from a particular location is one that can be comprehended by a person with eyesight adequate to obtain a current Ohio driver's license standing in the public right-of-way way or other location from which legibility is to be determined. Where such facts are material, it shall be presumed that the observation takes place in daylight hours, and that the person making the observation is standing and is between five feet two inches (5'2") and six feet (6') tall.

(d) Localized Background:

A localized background is any distinctive material, pattern or color that provides a visual background for the sign message and that distinguishes a sign from a larger surface against which it is placed. (e) Sign:

A sign shall mean any visual communication device utilizing a letter, a word, a number, a symbol, a picture, an object, color, illumination or motion, the major function of such device being to convey visual information to or attract the visual attention of:

- (1) A person within a public right-of-way; and/or
- (2) A person not on the premises on which the visual communication device is located.

The term "sign" shall specifically include the following:

- (1) Any localized background which is a part of or is placed in conjunction with a sign for the purpose of improving the visibility of the sign; and,
- (2) Any artificial illumination device who's major function is to provide illumination of the visual information conveyed by a sign.

The following are types of signs:

(1) Directional Signs:

A directional sign is a sign that conveys information that pertains to the direction of traffic movement onto or within a premises.

(2) Ground Signs:

A ground sign is a sign not attached to a building.

(3) Projecting Sign:

A projecting sign is a sign supported by a building wall or column and extending a distance exceeding twelve (12) inches from the wall.

(4) Roof Sign:

A sign mounted vertically on a pitched roof which overhangs a wall at ground level.

(5) Wall-Sign:

A wall sign is a sign which is located on or formed by the surface of the wall of a building. A mansard roof facade on a building shall be considered part of the wall.

(6) Portable Sign (prohibited):

Any sign not attached to the ground or other permanent structure or a sign to be transported, including, but not limited to, signs designed to be transported by means of wheels; balloons used as signs, and signs attached to or painted on vehicles parked and visible from the public right-of-way, unless said vehicle is used in the normal day-to-day operations of the business. This definition does not include "sandwich board" signs.

(f) Sign Area:

The sign area shall mean the total area in square feet of all the visible information-conveying surface area(s) of a sign including all localized backgrounds but excluding all materials not conveying significant visual information and whose major function is providing structural support for the sign. An irregular shaped sign surface area shall be measured by calculating the surface area of a simple plane or solid geometric shape, which approximates the size and shape of the sign surface area. The area of individual elements of a sign placed against a non-localized background such as letters placed against a wall, shall be measured by calculating the area of the smallest single rectangle which would completely enclose all elements of the sign.

(g) Sign-Height:

Sign height shall mean the maximum vertical height in feet that the highest part of a ground sign extends above the surface of the ground underneath the sign.

(h) Sign - Permanent:

A permanent sign is a sign permitted by this Ordinance to be located on a premises which is permanently anchored for an unlimited period of time.

(i) Sign-Sandwich Board:

A temporary sign with two (2) hinged boards which can be placed on the ground. A sandwich board sign is not considered a portable sign under this <u>O</u>ordinance. <u>A</u> sandwich board shall only be displayed during the hours in which the business is open.

(j) Sign - Temporary:

A temporary sign is a movable sign permitted by this Ordinance to be located on a premises for a limited period of time. A temporary sign is not considered a portable sign under this **Oo**rdinance.

(a) any sign located on a premise in conjunction with any election is considered a temporary sign under this Ordinance and must conform to all standards of a temporary sign, except; any temporary sign in conjunction with any election shall only be displayed thirty (30) days prior to that election and must be removed within two (2) days after that election on any non-residential property.

(4) SIGNS EXEMPT FROM ORDINANCE:

The following signs are not subject to the provisions of this Ordinance:

(a) Signs Inside Buildings:

Any sign located inside or behind a window shall not be subject to any provision of the Ordinance, except the

Prohibited Signs and Sign Characteristics contained in this Ordinance.

(b) Other Signs:

Other signs that are not legible from the public right-of-way or from private property other than that on which the building is situated.

(5) SIGNS PARTIALLY EXEMPT FROM ORDINANCE:

The following signs may be erected or constructed without a permit, but may be subject to additional regulations under this section.

Where a sign is erected pursuant to a state statute or a court order, the sign may exceed the size standards of this **O**erdinance or otherwise deviate from the standards set forth in

this $\underline{\mathbf{O}}$ ordinance to the extent that the statute or court order expressly requires the larger size or other deviation. In all other respects, such signs shall conform to the standards of this

Oordinance:

- (a) Signs conforming to the Manual of Uniform Traffic Control Devices and bearing no commercial message;
- (b) Signs installed by employees or officials of a state or federal agency in the course of their governmental duties and bearing no commercial message;
- (c) Signs installed by employees or officials of the City of Bellbrook or of Greene County in the course of their governmental duties and bearing no commercial message;

- (d) Signs required by a state or federal statute;
- (e) Signs required by an order of a court of competent jurisdiction;
- (f) Signs installed by public utilities in their rightsof-way or on their facilities and bearing no commercial message other than such message as necessary to identify the use; and
- (g) Signs installed by a transit company with a franchise or other right to operate in the City of Bellbrook, where such signs are installed along its routes and relate to schedules or other information about the transit route.

(6) SIGNS PERMITTED IN PUBLIC RIGHT-OF-WAY.

The following signs, and only the following signs, shall be allowed in the public right-of-way:

(a) Signs conforming to the Manual of Uniform Traffic

Control Devices and bearing no commercial message;

(b) Signs installed by employees or officials of a state or

federal agency in the course of their governmental duties and bearing no commercial message;

- (c) Signs required by a state or federal statute;
- (d) Signs installed by employees or officials of the City
 of Bellbrook or Greene County;
 - (e) Signs installed by public utilities in their rightsof-way or on their facilities and bearing no commercial

message other than such message is necessary to identify the use;

(f) Signs installed by a transit company with a franchise

or other right to operate in Bellbrook, where such signs are installed along its routes and relate to schedules

or other information about the transit route; and

(g) In the Old Village District only, sandwich board signs located on sidewalks in conformance with Sec.

located on sidewalks in conformance with Sec. 18.20B. For display of non-commercial sandwich board signs, see Section 18.20A(9)(h).

(7) SIGNS ALLOWED WITHOUT A PERMIT:

The following signs shall be exempt from the permit requirements of this $\underline{\mathbf{O}}$ ordinance but shall be subject to all other standards of this $\underline{\mathbf{O}}$ ordinance:

- (a) Signs installed by employees or officials of the City of Bellbrook that do not fall under one of the broader exemptions of this section;
- (b) Detached signs smaller than two (2) square feet in area and less than four (4) feet in height, and containing no commercial message;
- (c) Wall signs smaller than two (2) square feet in area and containing no commercial message; and
- (d) In residential districts only, temporary signs allowed by this **O** ordinance that have no more than six (6) square feet of sign area per side or (twelve (12) square feet total sign area.

(8) PROHIBITED SIGNS AND SIGN CHARACTERISTICS:

The following signs are prohibited in the eCity:

- (a) Any sign erected at or near any intersection of any street in such a manner as to obstruct free and clear vision between the height of three (3) feet from the ground and the height of nine (9) feet from the ground;
- (b) Any sign which by reason of any combination of location, position, shape, or color, may interfere with, obstruct the view of, or be confused with any authorized traffic sign, signal, or device, or which makes use of the word "stop"; "look"; or "danger"; or other word phrase or symbol in such a manner as to interfere with, mislead, or confuse traffic;
- (c) Any lighter-than-air or inflatable sign situated on, attached or tethered to a premises;
- (d) Any sign, which conveys at any time the visual sensation or appearance of motion, or presents a non-constant visual image to the eye of an observer;
 - (1) 'Visual sensation or appearance of motion' shall not include a change of message that is completed in less than one (1) second. The change of message shall appear as a seamless transition from one message to the next and shall include no special visual effects: and
 - (2) A message display period of five minutes or longer shall constitute a constant visual image;
- (e) Any sign with lighting directed out or away from the sign so that the light is cast directly onto or toward a public street or sidewalk or toward private property other than the premises on which the sign is located;
- (f) Any sign utilizing a fixed or mobile beacon, strobe light, search light, signaling light, spotlight, or similar apparatus, equipment or device, which is directed above or outside of a premises in such a manner so as to attract an unusual amount of visual attention of persons not located on the premises;

- (g) Any sign or part thereof which utilizes flame as a source of light;
- (h) Any sign utilizing an energized lamp bulb where the surface of the bulb is directly visible to persons not located on the premises;
- (i) Any sign mounted on a roof, attached to a roof or integrated into a roof except as permitted for certain special cases as allowed under §18.20A(12)(c)(2),and for certain buildings in the B-4 zoning district of the Old Village District, and as permitted more fully set forth in §18.20B(3)(d),and
- (j) Any portable sign.

(9) GENERAL PROVISIONS:

A sign shall be designed, erected, altered, reconstructed, moved, and maintained in accordance with the provisions of this section of this Ordinance unless specifically modified by another section of this Ordinance.

(a) Permits Required:

A Zoning Permit shall be obtained for erection, construction, relocation, or alteration of any permanent or temporary sign unless exempted from this Ordinance or from permit requirements by the express terms of this Ordinance. Installation of any sign shall conform to all e<u>C</u>ity zoning, building, electrical and fire codes.

(b) Non-Conforming Signs:

See: Sec. 18.20B, Non-Conforming Signs, Non-Conforming Lots, Non-Conforming Uses of Land, Non-Conforming Structures; Non-Conforming Uses of Structures and Premises and Non-Conforming Characteristics of Use, Section 18.21 of this Ordinance.

(c) Maintenance of Signs:

Every sign, which requires a sign permit, shall be maintained in a safe, presentable, and good structural condition at all times, including the replacement of a defective part, painting, cleaning, and other acts required for the maintenance of said sign. Permits shall not be required for:

- (1) Routine maintenance of any sign, not involving structural changes to the sign; and
- (2) Changes of message, either manually or electronically, on an electronic message sign, changeable copy sign, subject to limitations of this **O**ordinance on the frequency of message changes.
- (d) Dangerous or Defective Signs:

A sign in dangerous or defective condition shall not be permitted on any premises. Any such sign shall be removed or repaired promptly upon receipt of notice from the City Manager or his/her designee. The City Manager may immediately remove or cause to be removed any dangerous or defective sign which, in the opinion of the City Manager, creates an immediate or potential danger to persons or property due to structural deficiencies, inadequate maintenance, or because of the location of the sign.

(e) Location of Signs:

All signs shall be located on private property and a minimum of 15 feet from the paved edge of the road. Signs shall not be permitted in the public right-of-way nor in a median, sidewalk or tree lawn area. Signs shall not be affixed to any improvement (utility pole, traffic control device, bridge, guardrail, or other safety barrier) within such right-of-way or within required sight distance of or on eCity-owned property or buildings, except within lawful proximity of polling places on Election Day, under rules established by the Greene County Board of Elections.

Any sign erected in the public right-of-way or on public property in violation of this **O**erdinance shall be deemed abandoned and may be removed by the City Manager or his/her designee without notice or compensation to the

owner. Removal by a <u>eC</u>ity official shall not affect the penalties applicable for the unlawful erection or placement of a sign in the public right-of-way or on public property.

(f) Duration of a Permanent Sign:

Any sign permitted in this Ordinance shall be considered to be a permanent sign unless otherwise stated in this Ordinance.

(g) Sign Location with Respect to Frontages:

Allowed signs in Business, Industrial or other non-residential Districts are determined in part by the frontage of the premises on particular roads; where the sign area is calculated based on the frontage on a particular road or street, the allowed sign shall be located along that road or street frontage, regardless of the fact that the premises may have frontage on other streets or roads.

(h) Non-Commercial Messages:

Any sign allowed under this **O** ordinance or a predecessor ordinance, by sign permit, by conditional use permit, or by variance, may contain, in lieu of any other message or copy, any non-commercial message that does not direct attention to a business operated for profit, or to a product, commodity, or service for sale or lease, or to any other commercial interest or activity, so long as said sign complies with the size, height, area, and other requirements of this **O** ordinance. The substitution of a non-commercial message may be made without any additional approval or permitting.

(i) Commercial Messages:

All commercial information conveyed by any sign permitted under this Ordinance must pertain to commercial activities lawfully conducted on the premises on which the sign is located. For purposes of this provision, the offering of a

property for sale or lease shall be considered a commercial activity conducted on the premises.

(j) Ground Sign:

(1) Setback:

Unless otherwise stated in this **O**erdinance, any temporary or permanent ground sign or any part thereof shall be set back a minimum distance of ten (10) feet from any right-of-way.

(2) Lot Frontage:

A ground sign shall be allowed only on a lot which has lot frontage on a public right-of-way. Corner lots and through lots shall have only one (1) lot frontage.

(3) Landscaping Requirements:

A permanent ground sign shall require a single continuous landscaped area to be maintained beneath

the sign in accordance with the following standards:

(A) The minimum size of landscaped area shall

be equal to the total area of the sign; and

(B) The landscaped area shall include living plantings aesthetically located and maintained.

The use of concrete, asphalt, or any other paved surface inside the required landscaped area beneath the sign shall be prohibited.

(k) Projecting Signs:

Where projecting signs are allowed, they shall conform with

the following standards. A projecting sign:

- (1) Shall have vertical clearance of at least ten (10) feet above a sidewalk and fifteen (15) feet above a driveway;
- (2) Shall be attached to the building wall at an angle of ninety (90) degrees and no part of the sign shall project more than four (4) feet from the wall; and
- (3) Shall not extend higher on the wall than the bottom height of any second story window.

(1) Wall Signs:

Where wall signs are allowed, they shall conform with the following standards. A wall sign:

- (1) Shall not extend more than twelve (12) inches from the wall of the building upon which it is mounted;
- (2) Shall be inclined from the vertical only to the extent necessary for conformity to the general contour of the wall to which the sign is mounted;
- (3) Shall not extend above the top of the wall and shall not extend beyond the limits of any wall to which they are attached;
- (4) Shall not obscure an architectural feature, such as but not limited to doors, windows or trim; and
- (5) Shall have hidden structural supports.

(m) Temporary Signs:

The illumination of a temporary sign shall be prohibited.

(10) SIGNS PERMITTED FOR AGRICULTURAL ZONES:

(a) Permanent Signs, Generally:

- (1) Permanent signs requiring a permit shall be prohibited on vacant or undeveloped land in agricultural zoning districts; and
- (2) Permanent signs shall be allowed on a premises with a permanent building designed for human use or

occupancy (not including storage or equipment sheds)

in accordance with the further provisions of this subsection.

(b) Occupied Land:

(1) Permanent Signs:

One (1) permanent ground sign shall be allowed for each occupied premises in the agricultural zoning district, subject to the following standards:

(A) The permitted sign area shall not exceed

one (1) square foot of sign area per ten (10) linear feet of lot frontage and shall not exceed thirty-two (32) square feet per side

or

sixty-four (64) square feet total sign area;

- (B) The sign shall not exceed six (6) feet in height;
- (C) The sign shall be subject to the setback standards applicable to ground signs in other zoning districts, as set forth under "General Provisions"; and
- (D) The sign may bear any non-commercial message or a commercial message related to agricultural activity conducted or agricultural products sold on the premises.

(2) Temporary Signs:

Temporary signs shall be allowed in the agricultural zoning district subject to the following:

- (A) Temporary signs shall be ground signs and shall not be attached to buildings, fences, or walls;
- (B) Such temporary signs shall not exceed twelve (12) square feet in area per side, or twenty-four (24) square feet total sign area, and shall not exceed six (6) feet in height;
- (C) One (1) temporary sign bearing a commercial message related to the sale, lease or rental of the premises is allowed per frontage at any time;
- (D) On occupied property, one (1) temporary sign may bear a commercial message related to temporary commercial activities lawfully conducted on the premises, including the occasional sale of personal property through a garage or yard sale, but no premises shall have a temporary sign bearing a commercial message (other than one related to the sale or lease of the property) for more than ninety (90) days per calendar year; and
- (E) The sign shall be located outside any public right-of-way but shall not be subject to the general ground sign setback provision.

(11) SIGNS PERMITTED FOR RESIDENTIAL ZONES:

(a) Permanent Signs Generally:

- (1) Permanent signs, other than neighborhood entrance signs that conform to Section 11(f), requiring a permit shall be prohibited on vacant or undeveloped land in residential zoning districts; and
- (2) Permanent ground signs are prohibited in all residential districts with the exception of permanent neighborhood entrance signs (f) and churches,

schools

or institutions (d); permanent wall signs are allowed in such districts in accordance with the further provisions of this section (11).

(b) Occupied Lots in Single- or Two-Family Residential Zones:

(1) Generally:

- (A) No sign may be directly illuminated;
- (B) Projecting signs shall be prohibited; and
- (C) The provisions of this subsection (b) apply to individual occupied residential lots in these districts; subsection(d), (e) and (f) of this section apply to signs in other locations in these districts.
- (2) Temporary Ground Signs:

No permanent ground signs shall be allowed. The following rules apply to temporary ground signs:

- (A) Temporary signs shall be ground signs and shall not be attached to buildings, fences, or walls;
- (B) Such temporary signs shall not exceed six (6) square feet per side, or twelve (12) square feet total sign area, and shall not exceed four (4) feet in height;

- (C) One (1) temporary sign bearing a commercial message related to the sale, lease or rental of the premises is allowed per frontage per premises at any time;
- (D) One (1) temporary sign may bear a commercial message related to temporary commercial activities lawfully conducted on the premises, including the occasional sale of personal property through a garage or yard sale, but no premises shall have a temporary sign bearing a commercial message (other than one related to the sale or lease of the property) for more than thirty (30) days per calendar year;
- (E) The sign shall be located outside any public right-of-way but shall not be subject to the general ground sign setback provision.

(3) Wall Sign:

Only one (1) wall sign shall be permitted on each dwelling unit and each sign shall not exceed one (1) square foot. The wall sign may bear any non-commercial message or a commercial message pertaining to a commercial activity lawfully conducted on the premises.

- (c) Multi-Family Residential Zones:
 - (1) Generally:
 - (A) No sign may be directly illuminated;
 - (B) Projecting signs shall be prohibited; and
 - (C) The provisions of this sub-, (c) apply to individual occupied residential premises in these districts; sub-sections (e),(e) and (f) of this section apply to signs in

other locations in these districts.

(2) Temporary Ground Signs:

No permanent ground signs shall be allowed. The following rules apply to temporary ground signs:

(A) Temporary signs shall be ground signs and shall not be attached to buildings, fences, or walls;

- (B) Such temporary signs shall not exceed six
- (6) square feet per side, or twelve (12) squarefeet total sign area, and shall not exceed four(4) feet in height;
- (C) One (1) temporary sign bearing a commercial message related to the sale, lease or rental of the premises is allowed per frontage per premises at any time;
- (D) One (1) temporary sign may bear a commercial message related to temporary commercial activities lawfully conducted on the

premises, including the occasional sale of personal property through a garage or yard sale, but no premises shall have a temporary sign bearing a commercial message (other than

one related to the sale or lease of the property) for more than thirty (30) days per calendar year;

(E) The sign shall be located outside any public right-of-way but shall not be subject to

the general ground sign setback provision.

(3) Wall Sign:

Only one (1) wall sign shall be permitted on each dwelling unit and each sign shall not exceed one (1) square foot. The wall sign may bear any non-commercial message or a commercial message pertaining to a commercial activity lawfully conducted on the premises.

(d) Signs for Institutional Uses:

Any school, house of worship, recreation center or other institutional use, including the Old Village District, may have the same signage allowed for any other use in the same zoning district in which it is located or the signs listed in paragraphs (1), (2) and (3), in this subsection:

- (1) One (1) detached ground sign, not to exceed thirty-six (36) square feet per side or seventy-two (72) square feet total sign area, or six (6) feet in height. Such signs may be illuminated;
- (2) The number of wall signs shall be limited to one per each wall of the building and the total combined area of all wall signs shall not exceed one (1) square foot of sign area per linear foot of building frontage; and
- (3) Signs conforming to the *Manual of Uniform Traffic Control Devices* and containing no commercial message are permitted in required off-street parking areas.

(e) Temporary Subdivision Signs:

As a temporary use accessory to the permitted activity of lawful subdivision development, one temporary sign per subdivision is permitted. There shall in no case be more than one (1) such sign for each fifty (50) lots in a proposed subdivision. Such sign shall not be illuminated and shall not exceed thirty-two (32) square feet per side, or sixty-four

- (64) square feet total sign area. Such signs shall be removed upon the sale of ninety percent (90%) of the lots in the subdivision.
- (f) Permanent Neighborhood Entrance Signs:

Permanent neighborhood or multi-family monument signs, either illuminated or non-illuminated, are permitted. Such signs may include a masonry wall, landscaping or other similar materials or features. Such signs shall be located at the principal entrance(s) to the neighborhood on private property and not in the public right-of-way. Such signs may not exceed thirty-two (32) square feet per side or sixty-four (64) square feet total sign area, and six (6) feet in height.

(12) SIGNS PERMITTED FOR BUSINESS, INDUSTRIAL, OR OTHER NON-RESIDENTIAL ZONES (EXCLUDING THE OLD VILLLIAGE DISTRICT):

(a) General-Provisions:

- (1) All permanent signs may be illuminated. Temporary signs shall not be illuminated; and
- (2) Signs for business, industrial, or other non-residential uses which are lawfully located in a residential zoning district shall be subject to the standards of this subsection (12).

(b) Ground or Projecting Signs:

- (1) Only one (1) ground or projecting sign per street frontage shall be permitted on a premises;
- (2) The ground or projecting sign shall not exceed one (1) square foot of sign area per one (1) linear foot of lot frontage not to exceed thirty-two (32) square feet per side or sixty-four (64) square feet total sign area; and
- (3) Ground signs shall be a maximum height of six(6) feet at the minimum sign setback line and for
- every additional five (5) feet of sign setback, two
- (2) feet may be added to the sign height not to

exceed a total sign height of sixteen (16) feet. (4) See (e)"Directional Signs".

(c) Wall Signs:

- (1) The number of wall signs shall be limited to one (1) per each wall of the building and the total combined area of all wall signs shall not exceed one
- (1) square foot of sign area per linear foot of building frontage; and
- (2) In the case of a building with a pitched roof that overhangs a wall at the ground floor level in a manner that precludes the practical placement of a wall sign on any part of the wall of the building, that portion of the roof surface that overhangs the wall shall be considered part of the wall and a wall sign may be located on that overhanging roof surface provided that no part of the sign extends more than three (3) feet from the roof surface and that the sign height does not exceed two (2) feet. An application for a permit for a sign under this paragraph shall include a detailed drawing by a licensed architect or engineer showing how the sign will be placed and showing sufficient dimensional details to indicate that such placement will conform with the requirements of this paragraph.

(d) Temporary Signs:

- (1) Temporary ground or wall signs shall be permitted.
 - (A) The sign area shall not exceed sixteen (16) square feet per side or thirty-two (32) square feet total sign area;
 - (B) The sign height shall not exceed six (6) feet;
 - (C) Only one temporary sign bearing a commercial message shall be allowed for each

business address at one time;

(D) Temporary ground or wall signs with commercial messages are permitted for four (4) separate occurrences per business for a total of thirty (30) days per quarter of a calendar year; and

(e) Directional Signs:

In addition to any other permanent or temporary sign permitted elsewhere in this Ordinance, permanent or temporary sign(s) which convey information which pertains to wayfinding onto or within a premises shall be permitted provided that:

- (1) The sign area shall not exceed two (2) square feet per side or four (4) square feet total sign area;
- (2) Sign height shall not exceed three (3) feet if located within twenty-five (25) feet of a public right-of-way or six (6) feet in height in any other location;
- (3) The sign shall be located outside any public right-of-way but shall not be subject to the general ground sign setback provision;
- (3) No commercial message on such a sign shall be legible from the public right-of-way or from private property other than the premises on which the sign is located;
- (4) Any commercial message, including the name or Logo of a business establishment, shall pertain to the premises on which it is located;
- (5) For purposes of the previous two paragraphs, in the case of a shopping center or other multi-tenant property with a single site plan, the entire area under one site plan shall be considered a single premises, regardless of the fact that some occupants

of the center may own their individual sites or spaces; and

(6) When located in the Old Village District, the sign shall comply with all the general design standards and general provisions applicable to signs located within the Old Village District.

18.20.B <u>PERMITTED SIGNS (OLD VILLAGE-DISTRICT)</u>

(1) SCOPE-OF-REGULATIONS:

The intent of this section is to modify sign requirements and to delineate special standards for signs in the Old Village District as defined in Appendix H of the Comprehensive Plan of Bellbrook (adopted by Ordinance No. 74-32) to ensure that such signs are compatible with the general architectural character of the Old Village. These regulations are in addition to regulations contained in Section 18.20.A, but where there is conflict with Section 18.20.A, the regulations contained herein shall take precedence. A Review Board named <u>Old</u> Village <u>District</u> Review Board shall be established by City Council to administer this section of the Zoning Regulations.

(2) NUMBER OF SIGNS PERMISSIBLE:

(a) Single Occupant-Buildings:

Single occupant buildings are permitted one (1) sign of each sign type (free-standing, wall/roof or projecting) no larger than the permissible size (see Section (4) below) for each frontage on a public street. One (1) additional sign no greater than one (1) square foot in area is permitted for each entrance, when flush mounted near the entrance for purposes of pedestrian information. No other exterior signs are permitted unless they are identified as a permitted exception in Section (7) "Exceptions."

(b) Multi-occupant Buildings:

Signs on buildings having more than one (1) occupant shall be controlled in the following manner:

- (1) Each occupancy facing a public street or public access driveway with a door opening directly from the occupancy onto the public street or access shall be allowed one (1) sign of each sign type (free-standing, wall/roof or projecting) of the permitted size, for each street frontage.
- (2) One sign of the permissible size (see Section (4) below) is permitted for each shared entrance. On this sign, all the building's occupants sharing the entrance may be listed.
- (3) One (1) additional sign no greater than one (1) square foot in area is permitted for each building entrance, when flush mounted near the entrance for purposes of pedestrian information. No other exterior signs are permitted unless they are identified as a permitted except in Section (7) "Exceptions".

(3) LOCATION OF SIGNS:

(a) Projecting:

In no case should the sign or its supports extend above the highest point of the building supporting the sign. A sign may project from a building beyond the property line and over a public sidewalk providing:

- (1) it does not intrude more than 3/4 of the sidewalk width; and
- (2) it clears the sidewalk by ten (10) feet.

(b) **Wall**:

<u>Wall</u> signs may take any shape or any direction across the facade of a building provided it conforms to the general character of the building to which it is attached. No <u>wall</u>

sign shall project above the cornice of the building to which it is attached.

(c) Free Standing:

No free-standing sign shall be mounted on the roof of a building, nor shall it reach a greater height than the lowest point of the cornice of the building or facility it represents. In no case shall any portion of a free-standing sign be located off the property of the facility it represents. All such signs shall be allowed the same area and be subject to the restrictions outlined for projecting signs.

(d) Roof signs:

Roof signs are permitted on structures in B-4 zoning districts in the Old Village District where a pitched roof overhangs a wall at the ground floor level in a manner which precludes the practical placement of a sign on any part of the wall of the building. The baseline of the roof sign:

- (1) shall be no closer than twelve (12") inches from the gutter line of the roof and the sign, including structure, must be totally contained within the lower fifty percent (50%) of the roof; and
- (2) shall not exceed four feet (4') in height and a maximum of four- and one-half feet (4 ½") from the roof surface as measured vertically from the face of the sign.

In no case shall the roof sign project above the peak of the roof upon which the sign is located.

(4) PHYSICAL CHACTERISTICS OF SIGNS:

(a) Sign Size:

(1) Projecting:

No projecting sign shall be larger than thirty-six (36) square feet total exposed faces, except at the

corner of two (2) public streets. Projecting signs attached to corners of buildings facing two (2) public streets may be of a size equivalent to the total size allowed for projecting signs on the two (2) sides forming the corner (a total 72 square feet), in which case, no other sign shall be permitted. If less than the allowable area is used, the remaining allowable area may be used to establish one (1) other projecting or flat sign on the respective intersecting building faces.

(2)**Wall**:

No sign attached flat against the face of a building or painted thereon, shall be larger than forty (40) square feet and shall not extend more than twelve (12) inches from the building face.

(3) Free Standing:

No free-standing sign shall be larger than thirty-six (36) square feet total exposed faces.

(4) Roof:

No roof sign shall be larger than twenty-four (24) square feet total exposed face.

(5) Color:

Sign colors shall be subdued and compatible with building colors.

(6) Materials:

Allowable materials are stone, wood, and metal. Facsimiles of stone, wood and metal produced from other materials are permissible if deemed acceptable by the Village Review Board.

(7) Lighting:

No sign shall be lighted internally nor shall there be rotating beams, beacons, or flashing illumination. All signs shall be lighted by separate light source set so as to not constitute a hazard to pedestrian or vehicular traffic. No back lighting shall be allowed.

(8) Animation:

No sign shall be permitted which incorporates physical movement or the illusion of physical movement.

(9) Design:

The design (shape, material, colors, lettering) of signs shall in total be compatible with late 19th century architecture characteristic of the Old Village.

(5) ERECTION AND DISPLAY OF SIGNS:

No exterior sign may be erected or displayed within the Old Village without the prior approval of the Community

Development Department. Should the applicant wish to apply for an alternate amount, size, or material of signage permitted in Section 18.20 of the Zoning Code, the applicant shall seek approval from the Old Village District Review Board.

(6) MODIFICATION AND REPAIR OF SIGNS:

(a) Non-Conforming:

Existing signs which do not meet the requirements of this Operdinance shall be considered nonconforming and shall be

subject to the following restriction:

No sign sustaining over fifty percent (50%) damage based

on

current replacement costs shall be repaired. No change in the size, shape, color, or content of a non-conforming sign shall be permitted except a change in ownership identification.

(b) Conforming:

No new sign shall be erected and no change in size, shape, color, or content of an existing conforming sign shall be permitted without prior approval of the Community

Development Department. Should the applicant wish to apply for an alternate amount, size or material of signage permitted in Section 18.20 of the Zoning Code, the applicant shall seek approval from the Old Village District Review Board.

(7) EXCEPTIONS:

(a) General:

Certain unique types of identifying signs or emblems shall be allowed on approval of the Village Review Board. Examples are barber poles, physician's caduceus, community

> bulletin boards, etc., provided they are of a scale suitable for the business they identify and are compatible as to color, materials as otherwise noted herein.

(b) Temporary:

Temporary signs of various types and designs for businesses and public notice shall be allowed provided they are approved by the Village Review Board Community

Development Department and are limited to a total of thirty (30) days display time per quarter. If the applicant wishes to apply for a temporary sign that does not meet the requirements of Section 18.20 of the Zoning Code, the applicant shall seek approval from the Village Review Board.

(c) Awnings Used as Signs:

Signs shall be permitted on awnings provided the awnings are constructed of frame and cloth. The area of the awning used as sign shall be no larger than eighteen (18) square feet.

(8) ADMINISTRATION, ENFORCEMENT AND APPEAL:

- (a) Administration of this section of the Zoning Code
 Regulations shall be the responsibility of the Village
 Review Board. See Section 14.13 for the assembly of the
 Board. It shall also be the duty of the Village Review Board
 to inform the Zoning Inspector Community Development
 Administrator of all approved applications and provide
 the Zoning Inspector Community Development
 Administrator with a copy of the drawings and the
 description contained in the application. Application for
 display or modification shall contain the following
 minimum information:
 - (1) Name and address of the applicants;
 - (2) Address of the property at which the sign will be displayed;
 - (3) Listing of occupants of the buildings on the property where sign is to be displayed;
 - (4) A sketch and/or photograph showing the location of the sign on building or property; and
 - (5) A written description and sketch of the sign containing the wording, colors, material, and dimension of the sign.

(b) Enforcement:

Any person, firm or corporation erecting or modifying a sign not in accordance with an approved application will be subject to penalties as specified in Section 19.12 of the Zoning Ordinance.

(c) Appeal:

Appeal of the decisions of the Village Review Board or the Zoning Inspector Community Development

Administrator shall be made to the Board of Zoning Appeals.

18.21 **NON-CONFORMITIES:**

Within the districts established by this Ordinance or amendments that may later be adopted, there exists lots, structures, and uses of land and structures which were lawful before this Ordinance was passed or amended, but which would be prohibited, regulated, or restricted under the terms of this Ordinance or future amendment. It is the intent of this Ordinance to permit these non-conformities to continue until they are removed but not to encourage their continuance. Such uses are declared by this Ordinance to be incompatible with permitted uses in the districts involved.

It is further the intent of this Ordinance that non-conformities shall not be enlarged upon, expanded, or extended, nor be used as grounds for adding other structures or uses prohibited elsewhere in the same district except by appeal to the Board of Zoning Appeals for approval of specific plans. Expansions of existing non-conforming uses, where allowed by the Board of Zoning Appeals, may be made only on property owned by the applicant as of the effective date of this Ordinance.

A non-conforming use of a structure, a non-conforming use of land, or a non-conforming use of a structure and land shall not be extended or enlarged after passage of this Ordinance by attachment on a building or premises of additional signs intended to be seen from off the premises, or by the addition of other uses of a nature which would be prohibited generally in the district involved.

To avoid undue hardship, nothing in this Ordinance shall be deemed to require a change in the plans, construction, or designated use of any building on which actual construction was lawfully begun prior to the effective date of adoption or

amendment of this Ordinance and upon which actual building construction has been diligently carried on. Actual construction is hereby defined to include the placing of construction materials in permanent position and fastened in a permanent manner, except that where demolition or removal of an existing

building has been substantially begun preparatory to rebuilding such demolition or removal shall be deemed to be actual construction, provided that work shall be diligently carried on until completion of the building involved.

(1) NON-CONFORMING LOTS OF RECORD:

In any district in which single-family dwellings are permitted, notwithstanding other limitations imposed by other provisions of this Ordinance, a single-family dwelling and customary accessory building may be erected on any single lot of record at the effective date of adoption or amendment of this Ordinance subject to the following conditions:

If two (2) or more lots or combinations of lots and portions of lots with continuous frontage in single ownership are of record at the time of passage or amendment of this Ordinance, and if all or part of the lots do not meet the requirements for lot width and area as established by this Ordinance, the lands involved shall be considered to be an undivided parcel for the purposes of this Ordinance, and no portion of said parcel shall be used or sold which does not meet lot width and are requirements established by this Ordinance, nor shall any division of the parcel be made which leaves remaining any lot with width or area below the requirements stated in this Ordinance.

(2) NON-CONFORMING USES OF LAND:

Where at the effective date of adoption or amendment of this Ordinance, lawful use of land exists that is made no longer permissible under the terms of this Ordinance as enacted or amended, such use may be continued, so long as it remains otherwise lawful, subject to the following provisions:

- (a) No such non-conforming use shall be enlarged or increased, nor extended to occupy a greater area of land than was occupied at the effective date of adoption or amendment of this Ordinance, except as provided.
- (b) No such non-conforming use shall be moved in whole or in part to any other portion of the lot or parcel occupied

by such use at the effective date of adoption or amendment of this **O**erdinance.

- (c) If any such non-conforming use of land ceases for any reason for a period of more than thirty (30) days, any subsequent use of such land shall conform to the regulations specified by this Ordinance for this district in which such land is located.
- (d) Where a non-conforming use of the land by the nature of the use requires expansion or enlargement of the land are so used in order to continue in operation such as removal of sand, earth, stone, minerals, etc., continuance of such operations following the adoption or amendment of this Ordinance shall be deemed a violation.

(3) NON-CONFORMING STRUCTURES:

Where a lawful structure exists at the effective date of adoption or amendment of this Ordinance that could not be built under the terms of this Ordinance by reasons of restrictions on area, lot coverage, height, yards or other characteristics of the structure or its location on the lot, such structure may be continued so long as it remains otherwise lawful.

(4) NON-CONFORMING USES OF STRUCTURES:

If a lawful use of a structure or of structure and premises in combination, exists at the effective date of adoption or amendment of this Ordinance, the lawful use may be continued so long as it remains otherwise lawful, subject to the following provisions:

(a) If no structural alterations are made, any non-conforming use of a structure, or structure and premises, may be changed to another non-conforming use provided that the Board of Zoning Appeals, either by general rule or by making findings in the specific case, shall find that the proposed use is equally appropriate or more appropriate to the district than the existing non-conforming use. In permitting such change, the Board of Zoning Appeals in collaboration with the Planning Board may require appropriate conditions and safeguards in accord with the provisions of this Ordinance.

- (b) Any structure, or structure and land in combination, in or on which a non-conforming use is superseded by a permitted use, shall thereafter conform to the regulations for the district in which such structure is located and the non-conforming use may not thereafter be resumed.
- (c) When a non-conforming use of a structure, or structure and premises in combination, is discontinued or abandoned for twenty-four (24) consecutive months or for twenty-seven (27) months during any three (3) year period, the structure or structure and premises in combination shall not thereafter be used except in conformance with the regulations of the district in which it is located.
- (d) When non-conforming use status applies to a structure and premises in combination, removal or destruction of the structure shall eliminate the non-conforming status of the use being conducted at that location and any subsequent use of the lot shall be in conformance with the regulations of the district in which it is located.

(5) REPAIRS AND MAINTENANCE:

On any building devoted in whole or in part to any nonconforming use, work may be done on ordinary repairs or on repairs or replacement of non-bearing walls, fixtures, wiring or plumbing.

(6) RESTORING BUILDINGS:

When a building or structure, the use of which does not conform to the provisions of this $\underline{\mathbf{O}}$ ordinance, has been damaged by explosion, fire, act of God, or the public enemy, to the extent of twice its assessed value for tax purposes, it shall not be restored or reconstructed or in any way used except in conformity with the district regulations in which the building is situated. When a non-conforming use qualifies for such reconstruction, a building permit shall be secured for that purpose within one (1) year from the date of occurrence for that damage and such reconstruction shall be diligently prosecuted and completed without delay. Failure to comply as set forth above shall cause such non-conforming uses to

lapse and the premises shall conform thereafter to the established district regulations therein.

(7) VIOLATIONS NOT RENDERED NON-CONFORMING:

A use, structure or lot which was in violation of the provisions of thise Oordinance which this Oordinance amends shall not be validated or become non-conforming upon the adoption of this Oordinance.

Nothing in this <u>O</u>ordinance shall be deemed to prevent the strengthening or restoring to a safe condition of any building or part thereof declared to be unsafe by any official charge with protecting the public safety, upon order of such official.

18.22 TRAILERS, COMMERCIAL SEMITRAILERS, AND TRUCKS

No section of a commercial tractor, commercial trailer, semi-trailer nor any commercial truck shall be parked in any residential district except for loading or unloading residential goods or furniture; in cases of such loading or unloading, such parking shall not exceed a total of twenty-four (24) hours within any thirty (30) day period beginning with the first instance of commercial semi-trailer or truck parking.

18.23 **DRIVE-IN SERVICE ESTABLISHMENTS**

Establishments that by their nature create periodic lining up of customers in automobiles waiting to be serviced shall provide off-street waiting areas for these customers. This includes such activities as:

- (1) Drive-in banks;
- (2) Auto wash;
- (3) Drive-in retail outlets;
- (4) Drive-in service and repair drop stations for such items as clothing, appliance equipment, etc.

Those establishments that can normally serve their customers in three (3) minutes or less shall provide at least five (5) off-street waiting spaces per window. Auto wash shall provide at least ten (10) off-street waiting spaces. Where normal customer servicing time is greater than three (3) minutes per car, additional spaces shall be provided on the basis of one (1) additional space per additional minute of service time.

18.24 **BARRIERS TO ENCROACHMENT**

Hereafter any lot used for parking, storage or display of vehicles for sale or rent including boats, trailers, mobile homes and trucks where such use is permitted to come within three (3) feet of any property line separating said lot from any property held by any other ownership including public land, such property lines shall be protected from encroachment by the installation of wheel stops, bumper guards or fencing so placed and erected as to prevent vehicles from projecting over said lines except at approved points of ingress and egress.

18.25 **RESERVED**

18.26 **RECREATIONAL VEHICLES**

- (1) One recreational vehicle (RV) up to 25 in length is permitted to be parked at any location on a residential lot provided that the vehicle conforms to the following:
 - (a) RVs parked in front of the principal building line shall be located on a hard surface pad or driveway such as poured concrete, asphalt or brick pavers in a manner so as not to obstruct the view of traffic;
 - (b) Parked or stored RVs shall not be connected to any utilities other than for maintenance purposes;
 - (c) No RV shall be used for living, sleeping or commercial purposes while parked in a residential district or in any area not approved for such use;
 - (d) RVs must be in operable condition and must display a current license and registration, if applicable. There are no restrictions on recreational vehicle storage or parking when the vehicle will be

stored or parked inside an enclosed permanent structure on a residential lot.

(2) One RV 25 feet in length or greater but not more than 40 feet in length or a second RV up to 25 feet in length is permitted to be parked on a residential lot

provided that it conforms to the following:

- (a) RVs must be positioned behind the principal building setback line in a side yard or rear yard;
- (b) Parked or stored RVs shall not be connected to any utilities other than for maintenance purposes;
- (c) No RV shall be used for living, sleeping or commercial purposes while parked in a residential district, or in any area not approved for such use;
- (d) RVs are permitted to be parked in a driveway for loading and/or unloading purposes only not to exceed forty-eight (48) hours per occurrence.
- (e) RVs must be in operable condition and must display a current license and registration, if applicable. There are no restrictions on recreational vehicle storage or parking when the vehicle will be stored or parked inside an enclosed permanent structure on a residential lot; and
- (f) RVs must be parked on a hard surface, driveway, or compacted gravel.
- (3) RVs in excess of 40 feet in length are not permitted to be parked in any residential district. No more than two (2) RVs are permitted to be located on the exterior of any residential lot.

Exception: The parking of a visiting RV is permitted in a residential zoning district provided that the RV is parked on a hard surface pad or driveway such as poured concrete, asphalt, or brick pavers and the vehicle does not restrict the view of traffic. A property owner may only claim this exception one (1) time per calendar year for a period not to exceed thirty (30) days per calendar year.

18.30 **ADULT ENTERTAINMENT FACILITIES**

- (1) No adult entertainment facility shall be established within a radius of 500 feet of any single-family residential zoning district or of any building used for single family residence purposes, whether in this e<u>C</u>ity or in any other political subdivision.
- (2) No adult entertainment facility shall be established within a radius of 500 feet of any school, library or teaching facility, whether public or private, governmental, or commercial, which school, library or teaching facility is attended by persons under 18 years of age, whether in this eCity or in any other political subdivision.
- (3) No adult entertainment facility shall be established within a radius of 500 feet of any park or recreational facility attended by persons under 18 years of age, whether in this e<u>C</u>ity or in any other political subdivision.
- (4) No adult entertainment facility shall be established within a radius of 500 feet of any church, synagogue, or permanently established place of religious service, which is attended by persons under 18 years of age, or day care center or type A or B family day care home as defined by Ohio law, whether in this eCity or in any other political subdivision.
- (5) No adult entertainment facility shall be established within a radius of 500 feet of any other adult entertainment facility or within a radius of 500 feet of any two (2) of the following establishments (or of any one (1) establishment which combines to any degree any two (2) of the following activities), whether in this e<u>C</u>ity or in any other political subdivision:
 - (a) Cabarets, clubs, or other establishments which feature topless or bottomless dancers, go-go dancers, exotic dancers, strippers, male or

female impersonators, or similar entertainers.

- (b) Establishments for the sale of beer or intoxicating liquor for consumption on the premises.
- (c) Pool or billiard halls.

- (d) Pinball palaces or halls.
- (e) Video arcades or establishments known by other descriptions, but which provide video game and/or other games or entertainment attended or participated in by persons under 18 years of age.
- (f) Dance halls or discotheques.
- (6) For the purposes of this section, distances shall be measured from the property lines of any lot or parcel of land which includes, or which is operated or

used in connection with, a building in which an adult entertainment facility is located or in which any activity described or referred to in this section is located.

18.31 <u>LIGHT & AND GLARE</u>

(1) Purpose and Intent

These provisions are intended to control the use of artificial illuminating emitting rays into the night sky and to provide a level of illumination necessary for adequate, safe, and efficient movement of vehicles and persons. It is the intention of this section to accomplish the following:

- a. Encourage lighting systems that are designed to conserve energy;
- b. Minimize glare;
- c. Protect the use and enjoyment of surrounding property; and
- d. Increase nighttime safety, security, and productivity.
- (2) Conformance and Applicable Codes

All outdoor, artificial illuminating devices shall be installed and maintained in conformance with the provisions of this section and the Ohio Building Code.

(3) Nonconforming Lighting – New Uses, Buildings & Major Additions, or Modifications

This section shall apply to all lots of proposed new land uses, developments, buildings, and structures that require a permit.

All building additions or modifications of 25 percent or more in terms of additional dwelling units, gross floor area, or parking spaces, either with a single approval or with cumulative approvals shall invoke the requirements of this sections for the entire property, including previously installed and any new exterior lighting.

Similarly, cumulative modification or replacement of the exterior lighting constituting 25 percent or more of the total exterior light output found on the site, measured in initial lamp lumens, shall invoke the requirements of this section for the entire property, including previously installed and any new exterior lighting.

(4) Nonconforming Lighting – Minor Additions

Additions or modifications of less than 25 percent, as described in 18.31 (3), and that require any type of permit, shall require the submission of a complete inventory and site plan detailing all existing and any proposed new exterior lighting. Any new lighting on the site shall meet the requirements of this Code with regard to shielding and lamp type.

(5) Nonconforming Lighting – Resumption of Use after Abandonment

If a property or use with nonconforming lighting is abandoned for 6 months or more, then all exterior lighting shall be reviewed and brought into compliance with this Code prior to the property's reuse.

(6) Exceptions

The following shall be exempt from the standards of this section:

- a. Public street luminaries:
- b. Emergency lighting: All temporary emergency lighting needed by the police, fire departments, or other emergency services, as well as all vehicular luminaries;
- c. Lighting used for construction crews making improvements of the public right-of-way.

(7) Shielding Standards

All exterior luminaries shall not be aimed, directed, or focused to cause direct light from the luminaire toward residential property or to persons operating motor vehicles on public ways; the luminaire shall be redirected, or its light output controlled as necessary to eliminate such conditions.

In addition, all exterior luminaires having an initial output greater than 2,000 lumens and that are mounted on a pole, building, structure, or tree, shall be full cutoff type luminaires. Such pole, building, structure, or tree mounted luminaires shall not be equipped with adjustable mounting devices permitting alteration of the luminaire's aim in the field.

(8) Light Trespass

All luminaires shall be located, aimed, or shielded to prevent light from trespassing across property boundaries. Light originating on a site shall not exceed the following foot-candles at any property line of said site for the following adjacent uses:

a.	Single Family Residential	0.1 foot-candles
b.	Multi-Family Residential	0.3 foot-candles
c.	Civic	0.5 foot-candles
d.	Office	0.5 foot-candles
e.	Commercial	1.0 foot-candles
f.	Industrial	1.0 foot-candles
g.	Street Rights-of-Way	1.0 foot-candles

18.40 PERSONAL WIRELESS SERVICE SITES

- (1) Personal Wireless Service Sites shall be screened by a solid wood fence, at least six (6) feet high. This fence shall be surrounded by an evergreen hedge, which shall be continuous, except for an entryway. Any equipment other than a Personal Wireless Service Tower or Personal Wireless Service Antenna, which projects above the fence shall be screened from view by a parapet, wall, or screen.
- (2) Personal Wireless Service Sites shall be dismantled, and all Personal Wireless Service Facilities maintained therein shall be removed within six (6) months after cessation of operations therein.
- (3) The Zoning Inspector Community Development Administrator shall not issue a Zoning Permit for a Personal Wireless Service Site until after the applicant has provided written certification that all required state and

federal permits have been obtained and has supplied copies of all such state and federal permits.

(4) If lighting is required by state or federal authorities, it shall be approved only at the minimum number and intensity and shall be shielded from ground

view. If not required by those authorities, no lights, beacons, or strobes shall be mounted on any Personal Wireless Service Antenna or Personal Wireless

Service Tower.

(5) Personal Wireless Service Sites shall be situated on land parcels having a minimum area of at least twenty-five (25) acres, and the base of any Personal

Wireless Service Tower situated thereon shall be set back at the furthest point from property designated for residential use in the $e\underline{Ci}$ ty. The setback from the

nearest point on the parcel shall be at least two hundred (200) feet or a distance equal to the maximum height of the highest antenna supported thereby,

whichever is greater.

- (6) No part of any Personal Wireless Service Facility may have a maximum elevation greater than two hundred (200) feet above ground level.
- (7) Personal Wireless Service Towers shall be constructed in such a manner as to be suitable for supporting at least two (2) differently dedicated Personal Wireless Service Antennas.

ARTICLE 19

ENFORCEMENT AND PENALTIES

19.01

ZONING PERMITS REQUIRED

No building or other structure shall be erected, moved, added to, structurally altered, nor shall any building, structure or land be established or changed in use without a permit issued by the **Community Development Administrator** Zoning Inspector. Zoning permits shall be issued only in conformity with the provisions of this Ordinance unless the Zoning Inspector Community Development Administrator received a written order from the Board of Zoning Appeals or from the City Council, as provided by this Ordinance.

See Commentary - Normal Zoning Flow Chart.

19.02

CONTENTS OF APPLICATION FOR ZONING PERMIT/ZONING COMPLIANCE CERTIFICATE

The application for zoning permit/zoning compliance certificate shall be signed by the owner or applicant attesting to the truth and exactness of all information supplied on the application. Each application shall clearly state that the permit shall expire and shall be revoked if work has not begun within one (1) year or substantially completed within two and one-half (2 ½) years from the date of the issuance of the zoning permit/zoning compliance certificate. For conditional uses, the effective date of the conditional use and zoning permit/zoning compliance certificate shall be that of the conditional use permit. At a minimum, the application shall contain the following information:

- (1) Name, address, and phone number of applicant;
- (2) Legal description of property;
- (3) Existing use;
- (4) Proposed use; if, business, specify use per Zoning Code and describe business;
- (5) Zoning district;

- (6) Plans drawn to scale, showing the actual dimensions and the shape of the lot to be built upon; the exact size and location of existing buildings on the lot, if any; and the location and dimensions of the proposed building(s) or alterations;
- (7) Building heights;
- (8) Application for accessory parking lot (as applicable);
- (9) Number of dwelling units;
- (10) When development is proposed for a single lot which is therefore not subject to standard subdivision rules and regulations, the <u>Community Development Administrator</u> Zoning Inspector shall require the applicant to submit plans meeting those requirements which are listed in the City of Bellbrook Subdivision Regulations under Section IV(B)(2) (a), Final Construction Plans and Estimate, and any additional subdivision regulations which are deemed necessary to allow the proper review and approval of the application for zoning permit/zoning compliance certificate.
- (11) Such other matters as may be necessary to determine conformance with, and provide for the enforcement of this $\underline{\mathbf{O}}$ ordinance, i.e., copy of conditional uses required in conditional use permit.

19.03 <u>SUBMISSION FOR A THOROUGHFARE PLAN REVIEW</u>

Before any zoning permit is issued affecting land within three hundred (300) feet of the centerline of a proposed new highway or a highway for which changes are proposed as described in the Bellbrook and Environs Major Thoroughfare Plan Map of the Comprehensive Plan or any land within a radius of five hundred (500) feet from the point of intersection of said centerline with

any public road or highway, the <u>Community Development Administrator</u> Zoning Inspector shall give notice to the applicant that a zoning permit will not be issued until after a recommendation is made in a period of thirty (30) days from the date notice is received by the Bellbrook Planning Board. After having checked the Comprehensive Plan, the Bellbrook Planning Board may coordinate with the Miami Valley Regional Planning Commission, Transportation Coordinating Committee and Regional Planning and

Coordinating Committee of Greene County so as to ascertain whether or not highway construction or changes are planned. If the Bellbrook Planning Board is notified by any of the above agencies of pending plans for said highway, they shall recommend against the issuance of a zoning permit and shall attempt to resolve any conflict within thirty (30) days. If no plans exist for said highway, or on the expiration of the thirty (30) days, the **Community Development Administrator** Zoning Inspector shall, issue the zoning permit/zoning compliance certificate if the application is conformance with all provisions of this Ordinance.

19.04 <u>APPROVAL OF ZONING PERMIT/ZONING COMPLIANCE</u> <u>CERTIFICATE</u>

Within thirty (30) days after the receipt of an application and required fee, the Community Development Administrator Zoning Inspector shall either approve or disapprove the application in conformance with the provisions of this Ordinance. Zoning permits which require approval by the Village Review Board and/or the Board of Zoning Appeals shall be issued by the Community Development Administrator Zoning Inspector and the appropriate fee(s) paid by the applicant within thirty (30) days of the date of Board approval. Failure to do so will result in the doubling of any permit fees for each thirty (30) day period beyond the initial thirty (30) days following Board approval up to one hundred and eighty (180) days after which the permit may be revoked and a new approval required by the appropriate Board. Reapplication for Village Review Board approval shall be in conformance with Section 14.15; reapplication for variances shall be in conformance with Section 20.17. All zoning permits shall, however, be conditional upon the commencement of work within one (1) year from date of issuance. The copy of the plans may be placed in the official file. The Community Development Administrator Zoning Inspector may issue a placard, to be posted in a conspicuous place on the property in question, attesting to the fact that the use or alteration is in conformance with the provisions of this Oordinance.

19.05 <u>EXPIRATION OF ZONING PERMIT/ZONING</u> <u>COMPLIANCECERTIFICATE:</u>

If the work described in any zoning permit/zoning compliance certificate has not begun within one (1) year from the date of issuance and has not been substantially completed within two and one-half(2½) years of the date of issuance said permit shall expire and written notice thereof shall be given by the **Community Development Administrator** Zoning Inspector to the persons affected, together with notice that further work as described in the

cancelled permit shall not proceed unless and until a new zoning permit/zoning compliance certificate has been obtained or extension granted. One (1) extension shall be granted by the <u>Community Development Administrator</u> Zoning Inspector for a period of six (6) months. The extension will be noted on the zoning permit/zoning compliance certificate and application by the <u>Community Development Administrator</u> Zoning Inspector.

19.06 <u>CERTIFICATE OF HEALTH OFFICER:</u>

In every instance where the zoning permit application/zoning compliance certificate and the attached plan reflect work associated with the water supply and/or disposal of sanitary waste and where a lot is not served with public water and/or the disposal of sanitary wastes by means of public sewers, the application for zoning permit/zoning compliance certificate shall be accompanied by a Certificate of Approval by the County Health Officer of the proposed method of water supply and/or disposal of sanitary wastes.

19.07 <u>APPLICATION FOR ZONING PERMIT/ZONING COMPLIANCE</u> CERTIFICATE

The <u>Community Development Administrator</u> <u>Zoning Inspector shall</u> may sign the application for zoning permit/zoning compliance certificate when he/she is satisfied that the construction or alteration to be done is the same as described and proposed on the zoning permit/zoning compliance certificate. This provision shall apply to new construction, additions, decks, and similar uses, except the raising of crops and other agricultural uses.

19.08 <u>FAILURE TO OBTAIN A ZONING PERMIT/ ZONING COMPLIANCE CERTIFICATE</u>

Failure to obtain a zoning permit/zoning compliance certificate shall be a violation of this **O**erdinance and punishable under Section 19.12 of this **O**erdinance.

19.09 RECORD OF ZONING PERMITS / CERTIFICATES OF ZONING COMPLIANCE

The <u>Community Development Administrator</u> Zoning Inspector shall maintain a record of all zoning permits/zoning compliance certificates and they shall be made available for inspection upon request to any person.

19.10 <u>COMPLAINTS REGARDING VIOLATIONS</u>

Whenever a violation of this <u>O</u>ordinance occurs, or is alleged to have occurred, any person may file a written complaint. Such complaint stating fully the causes and basis thereof shall be filed with the <u>Community</u> <u>Development Administrator</u> <u>Zoning Inspector</u>. The <u>Zoning Inspector</u> <u>Community Development Administrator</u> shall record such complaint, promptly investigate, and take action as provided by th<u>ise</u> <u>O</u>ordinance.

19.11 <u>OTHER ACTION</u>

Nothing herein contained shall prevent the municipality <u>City</u> from taking other lawful action as is necessary to prevent or remedy any violation.

19.12 **PENALTIES**

- (1) Violation of any provision of this $\underline{\mathbf{O}}$ ordinance or any amendment or supplement thereto, or failure to comply with any of the requirements of this Code may be enforced either through the filing of a misdemeanor complaint or assessment of a civil penalty, or both.
 - (a) Minor misdemeanor. Except as otherwise provided herein, any person, firm, or corporation violating any of the provisions of this Code, or any amendment or supplement thereto, shall upon conviction of a minor misdemeanor, be fined not more than one hundred dollars (\$100).
 - (b) Civil penalties. A person who violates any provision set forth in a provision of the Zoning Code has committed a civil offense and is subject to civil penalties. Prior to assessment of a civil penalty, the **Community Development Administrator** Zoning Inspector shall issue written notice of the civil violation to the offending person. The notice shall include:
 - (1) The street address of the property subject to the notice of violation;
 - (2) A description of the violation and the section(s) of **thise O**ordinance violated;

- (3) Corrective action that will eliminate or correct the violations;
- (4) The reasonable time frame required to correct the violation;
- (5) Provide information to the property owner of the right to a hearing before the Board of Zoning Appeals if a request for a hearing is made within twenty (20) days of receipt service of the notice;
- (6) When applicable, include a statement of the right to file a lien.
- (2) The written notice of the violation shall be served by one of the following methods:
 - (a) delivered personally;

(b) <u>Delivered by a commercial carrier service utilizing any form</u> of delivery that requires a signature;

- $(\underline{b}\underline{c})$ sent by certified mail, <u>return receipt requested</u>, to the last known address and receiving a return receipt showing delivery of the mailing; or
- (cd) if the notice <u>sent via certified mail</u> is returned showing that the letter was <u>"refused," or</u> "unclaimed," service by regular mail;
- (d) if the notice is returned showing that the letter was not delivered for reasons other than "refused," or "unclaimed," a copy of the notice shall be posted in a conspicuous place in on or about the structure affected by such notice.
- (3) A person served with a notice of civil violation may request a hearing before the Board of Zoning Appeals to contest the commission of a violation, to offer an explanation of the circumstances, or to offer proof of correction. The request shall be made in writing and received by the e<u>City</u> within 20 days of the person's receipt service of the notice of violation. The hearing shall be conducted in accordance with the provisions of Article 20 of the Zoning Code. If the person desires the presence, at the hearing, of the Community Development Administrator Zoning Inspector, or other City employee who issued the notice of violation, shall be present

at the hearing. the person must request the same in their written response to the notice of civil violation.

If the person served fails to remedy the zoning violation within a reasonable time, if the person admits the offense by written response, or if the person fails to attend a requested hearing, the <u>Community Development Administrator Zoning Inspector</u> may pursue enforcement as set forth in this section, or as otherwise authorized by this Zoning Code. If the person denies the offense by written response, the person must also request a hearing before the Board of Zoning Appeals.

In accordance with Section 19.15 of the Zoning Code, when a notice of civil violation has been issued pursuant to this section, the following <u>civil fines</u> amounts are hereby assessed upon the violator.

- (a) Any violation of the Zoning Code that is a first offense shall be twenty-five dollars (\$25) per day, per offense, or two hundred fifty dollars (\$250) per offense total.
- (b) Any violation of the Zoning Code that is a second offense shall be fifty dollars (\$50) per day, per offense, or five hundred dollars (\$500) per offense total.
- (c) Any violation of the Zoning Code that is a third offense shall be seventy-five dollars (\$75) per day, per offense, or seven hundred fifty dollars (\$750) per offense total.
- (4) Those who erect signs contrary to the provision of this **Zoning** Code shall be fined upon conviction or assessed by civil penalty not more than ten dollars (\$10) per offense, up to a maximum of four (4) offenses in any twelve (12) month period. Illegal signs are subject to confiscation by the City and will not be returned to the owners until all penalties are satisfied; unclaimed signs shall be held for thirty (30) days and then disposed of as surplus property. Convictions or civil penalties for sign violations in excess of four (4) offenses in a twelve-month period shall be fined, upon conviction, or assessed a civil penalty amount not to exceed twenty-five dollars (\$25) per offense.

19.13 **AFFECTED PARTIES**

The owner or tenant of any building, structure, premises, or part thereof and architect, engineer, surveyor, builder, contractor, agent, or other person who

commits, participates in, assists in, or maintains a violation shall be found guilty of a separate offense and suffer the penalties herein provided.

19.14 <u>VIOLATION, NUISANCE PER SE: ABATEMENT</u>

Buildings erected, altered, razed, or converted, or uses carried on in violation of any provision of this $\underline{\mathbf{O}}$ ordinance is hereby declared to be a nuisance per se.

19.15 <u>SCHEDULE OF FEES, CHARGES, AND EXPENSES</u>

The City Council shall establish, by resolution on an annual or semiannual basis, a schedule of fees, charges, and expenses and a collection
procedure for zoning permits/zoning compliance certificates, amendments,
appeals, variances, conditional use permits, plan approvals and other
matters pertaining to the administration and enforcement of this
eOrdinance. This fee scheduled may include, but is not limited to, fees
and expenses incurred for requiring investigations, inspections, legal
advertising, postage, and other expenses. The schedule of fees shall be
posted in the office of the Zoning Inspector Community Development
Administrator and on the City's website and shall be altered or amended
only by approval of the City Council.

19.16 **WAIVER**

Any person charged with a violation of any provision of this chapter for which payment of a prescribed fine may be made, may pay such sum in the manner prescribed on the issued sign ticket. Such payment shall be deemed a plea of guilty, waiver of court appearance and acknowledgement of conviction of the alleged offense and may be accepted in full satisfaction of the prescribed penalty for such alleged violation.

ARTICLE 20

ADMINISTRATION

20.01

OFFICE OF ZONING INSPECTOR COMMUNITY DEVELOPMENT ADMINISTRATOR CREATED

A Zoning Inspector Community Development Administrator designated by the City Council shall administer and enforce this Oordinance. He may be provided with the assistance of such other persons as the Council may authorize.

20.02

DUTIES **OF COMMUNITY DEVELOPMENT** ADMINISTRATOR ZONING INSPECTOR

For the purpose of this **O**ordinance, the **Community Development Administrator** Zoning Inspector shall have the following duties:

- (1) Upon finding that any of the provisions of this Oerdinance are being violated shall notify in writing by certified mail the person responsible for such violation(s) ordering the action necessary to correct such violation;
- (2) Order discontinuance of illegal uses of land, buildings, or structures;
- (3) Order removal of illegal buildings or structures or illegal additions or structural alterations;
- (4) Order discontinuance of any illegal work being done;
- (5) Issue receipts and maintain a record thereof for all fees collected;
- (6) Take any other action authorized by this Ordinance to ensure compliance with or to prevent violation(s) of this Oerdinance. This may include the issuance of and action on zoning permits and certificates of zoning compliance and such similar administrative duties as are permissible under the law, to include the maintenance of records as required by Section 19.09.

20.03 PLANNING BOARD CREATED

A Planning Board shall be created in accordance with Sections 8.01, 8.02 and 8.03 of the Bellbrook Charter. The Planning Board shall have the powers of zoning and all the power and authority conferred upon city planning boards by the Ohio Revised Code and such other duties as may be imposed upon it by the Municipal Administrative Code and by the Bellbrook Charter.

20.04 <u>DUTIES OF PLANNING BOARD</u>

For the purpose of this $\underline{\mathbf{O}}$ ordinance the Planning Board shall have the following duties:

- (1) Initiate proposed amendments to this **O**ordinance;
- (2) Review all proposed amendments to this **O**erdinance and make recommendations to the City Council as specified in Article 21;
- (3) Review all planned unit developments and make recommendations to the City Council provided in Article 17;
- (4) Review all requests for determination of similar uses; and
- (5) Perform other duties specified in the Zoning Ordinance.

20.05 **PROCEEDINGS OF PLANNING BOARD**

The Board shall adopt rules necessary to the conduct of its affairs in keeping with the provisions of this **O**erdinance. Meetings shall be held at the call of the chairman and at such other times as the Board may determine. All meetings shall be open to the public. The Board shall keep minutes of its proceedings showing the vote of each member upon each question, or if absent or failing to vote, indicating such fact, and shall keep records of its examinations and other official actions all of which shall be a public record and be immediately filed in the office of the Board. The official copy of the minutes of all proceedings will be signed by the Board Chairman and Secretary.

20.06 **BOARD OF ZONING APPEALS CREATED**

A Board of Zoning Appeals shall be created in accordance with Sections 8.05 and 8.06 of Bellbrook Charter. The Board of Zoning Appeals shall elect its own officers annually. Three (3) members of the Board of Zoning Appeals shall constitute a quorum for the conducting of business. It shall

require a majority of votes of the sitting Board to pass a motion or take official action.

20.07 <u>DUTIES OF THE BOARD OF ZONING APPEALS</u>

In exercising its duties the Board may, as long as such action is in conformity with the terms of this <u>O</u>erdinance, reverse or affirm, wholly or partly, or modify the order, requirement, decision, or determination as ought to be made, and to that end shall have the powers of the <u>Zoning Inspector Community Development Administrator</u> from whom the appeal is taken. The majority of votes of the sitting Board shall be necessary to reverse any order, requirements, decision or determination of the <u>Community Development Administrator Zoning Inspector</u>, or to decide in favor of the applicant on any matter upon which it is required to pass under this <u>O</u>erdinance or to affect any variation in the application of this <u>O</u>erdinance. For the purpose of this <u>O</u>erdinance the Board has the following specific responsibilities:

- (1) Administrative review;
- (2) Determination of district boundary location;
- (3) Granting conditional use permits; and
- (4) Authorizing variances.

20.08 **PROCEEDINGS OF THE BOARD OF ZONING APPEALS**

The Board shall adopt rules necessary to the conduct of its affairs in keeping with the provisions of this <u>O</u>erdinance. Meetings shall be held at the call of the chairman and at such other times as the Board may determine. The Chairman, or in his absence the acting chairman, may compel the attendance of witnesses. Oral testimony given as evidence must be given under oath. Oaths will be administered by the Board Chair or designee. All meetings shall be open to the public. The Board shall keep minutes of its proceedings, showing the vote of each appeal considered by the Board, and the section of this <u>O</u>erdinance where applicable which the Board has considered in approving or disapproving any petition or other matter brought before the Board and shall keep records of its examinations and other official actions, all of which shall be a public record and be immediately filed in the office

of the Board. The official copy of the minutes of all proceedings will be signed by the Board Chairman and Secretary.

20.09 HEARING OF BOARD OF ZONING APPEALS

The Board of Zoning Appeals shall fix a reasonable time not to exceed 30 days from receipt of application, petition, or appeal, for the hearing. Application, petition, or appeal shall be filed with the Community **Development Administrator Zoning Inspector** at least five days prior to the next regularly scheduled meeting. The Board shall give at least ten (10) days' notice of the time and place of such hearing, to the Community Development Administrator enforcing officer, and to the owners of record of property within three hundred (300) feet of the premises in question, such notice to be delivered personally or by mail addressed to the respective owners at the address given on the last assessment roll. Any party may appear at such hearing in person, by agent, or by attorney. Before holding the public hearing, notice of such hearing shall be posted in full in not less than five (5) of the most public places in the municipality City as determined by Council at least ten (10) days before the date of said hearing. The notice shall set forth the time and place of the public hearing, and the nature of the proposed hearing.

20.10 **ACTION BY THE BOARD OF ZONING APPEALS**

Within thirty (30) days after a public hearing, the Board of Zoning Appeals shall either approve, approve with supplementary conditions, or disapprove the request or application before the Board. If an application for conditional use is approved or approved with modifications, the Board shall direct the Zoning Inspector Community Development Administrator to issue the conditional use permit listing the specific conditions specified by the Board for approval. Appeals from Board decisions shall be made in the manner specified in Section 20.11.

20.11 <u>DUTIES OF COMMUNITY DEVELOPMENT ADMINISTRATOR</u> ZONING INSPECTOR, BOARD OF ZONING APPEALS, LEGISLATIVE AUTHORITY AND COURTS ON MATTERS OF APPEAL

It is the intent of this <u>O</u>ordinance that all questions of interpretation and enforcement shall be first presented to the <u>Community Development Administrator Zoning Inspector</u>, and that such questions shall be presented to the Board only on appeal from the decision of the <u>Community Development Administrator Zoning Inspector</u>, and that recourse from the

decisions of the Board shall be to the courts as provided by law. It is further the intent of this <u>O</u>erdinance that the duties of the City Council in connection with this <u>O</u>erdinance shall not include hearing and deciding questions of interpretation and enforcement that may arise. The procedure for deciding such questions shall be as stated in this section and this <u>O</u>erdinance. Under this <u>O</u>erdinance the City Council shall have only the duties of considering and adopting or rejecting proposed amendments or the repeal of this <u>O</u>erdinance as provided by law, and of establishing a schedule of fees and charges as stated in Ordinance No. 75-1. Nothing in this <u>O</u>erdinance shall be interpreted to prevent any official of the City from appealing a decision of the Board to the courts as provided in Chapters 2505 and 2506 of the Ohio Revised Code. Any such appeal shall be made within ten (10) thirty (30) days of the Board's written decision.

20.12 <u>PROCEDURES AND REQUIREMENTS FOR ADMINISTRATIVE</u> REVIEW

Administrative reviews shall conform to the procedures and requirements of Sections 20.06 through 20.12 of this **O**erdinance. As specified in Section 20.07, the Board of Zoning Appeals has appellate jurisdiction relative to administrative reviews.

20.12.1 <u>ADMINISTRATIVE REVIEW</u>

Appeals will be heard and decided only in such cases where it is alleged there is an error in any order, requirement, decision, or determination made by the enforcing officer Community Development Administrator or designee in the enforcement of this enforcement. Appeals may be taken by any person aggrieved or by any officer of the City affected by any decision of the Community Development Administrator or designee enforcing officer. Such appeal shall be taken within twenty (20) days after the decision. The concurring vote of three (3) members of the Board shall be necessary to reverse any order, requirement, decision, or determination of the Community Development Administrator or designee enforcing officer, or to decide in favor of the applicant on any matter upon which they are required to pass under the terms of this end of this end of the order.

20.12.2 STAY OF PROCEEDINGS

An appeal stays all proceedings in furtherance of the action appealed from, unless the <u>Community Development Administrator</u> <u>Zoning Inspector</u> from whom the appeal is taken certifies to the Board of Zoning Appeals

after the notice of appeal is filed with him, that by reason of facts stated in the application, a stay would, in his opinion, cause imminent peril to life and property. In such case, proceedings shall not be stayed other than by a restraining order which may be granted by the Board of Zoning Appeals or by a court of record on application, on notice to the Zoning Inspector from whom the appeal is taken on due cause shown.

20.13 PROCEDURES AND REQUIREMENTS FOR VARIANCES

To vary the strict application of any of the requirements of this Ordinance in the case of exceptionally irregular, narrow, shallow, or deep lots, or other exceptional physical conditions, whereby such strict application would result in practical difficulty, or unnecessary hardship - not economic in nature, - that would deprive the owner of reasonable use of the land or building involved but in no other case. The fact that another use would be more profitable is not a valid basis for legally granting a variance. No nonconforming use of neighboring lands, structures, or buildings in the same district, and no permitted use of lands, structures or buildings in other districts shall be considered grounds for the issuance of a variance. The variance requested shall not alter the essential character of the locality, nor be in conflict with the Comprehensive Master Plan. In making this determination, the Board of Zoning Appeals shall be advised by the recommendation of the Planning Board. In cases involving properties situated within the Old Village District, as defined by Article 14, Section 14.02 of the Bellbrook Zoning Code, the **Old** Village **District** Review Board shall, when possible, make a recommendation to the Board of Zoning Appeals regarding the disposition of the variance request.

20.13.1 PROCEDURE FOR CONSIDERATION OF PETITIONS FOR VARIANCE:

The procedure for the consideration of petitions for variances shall be:

- (1) The Board of Zoning Appeals shall make a finding that the reasons set forth in the application are valid and justify the granting of the variance. The Board shall also determine if the variance is the minimum variance that will make possible the reasonable use of land, building, or structure.
- (2) Under no circumstances shall the Board of Zoning Appeals grant a variance which will permit a use which is not permitted in the district involved.

- (3) Conditions: The Board of Zoning Appeals may prescribe appropriate conditions and safeguards in conformity with this Ordinance and the recommendation of the Planning Board. The Board of Zoning Appeals shall require a bond or irrevocable letter of credit to assure conformance to such conditions and safeguards as the Board may require.
- (4) Violation of such conditions and safeguards when such are made a part of the terms under which a Variance is granted, shall cause the bond mentioned in (c) above to be forfeited and shall be deemed in violation of this Ordinance and punishable under Section 19.12 of this **O**erdinance.
- (5) Public Hearings: Prior to taking action on a request for a variance the Board of Zoning Appeals shall hold a public hearing and give notice to property owners as in Section 20.09 of this Ordinance.

20.13.2 **VARIANCES:**

APPLICATION AND STANDARDS FOR

A variance from the terms of this Ordinance shall not be granted by the Board of Zoning Appeals unless and until a written application for a variance is submitted to the **Community Development**<u>Administrator Zoning Inspector</u> and the Board of Zoning Appeals containing:

- (1) Name, address, and phone number of applicants;
- (2) Legal description of property; and a site plan based on an accurate survey showing existing and/or future building locations and the locations of buildings on adjacent properties. This site plan should be prepared by a registered surveyor attesting to the accuracy of same;
- (3) Description of nature of variance requested;
- (4) Variances from the terms of the Zoning Ordinance shall be granted only where the property owner shows that the application of a zoning requirement to the property is inequitable causing the property owner practical difficulties

in the use of the property. Factors to consider include, but are not limited to:

- (a) Whether the property in question will yield a reasonable return or whether there can be any beneficial use of the property without the variance;
- (b) Whether the variance is substantial;
- (c) Whether the essential character of the neighborhood would be substantially altered or whether adjoining properties would suffer a substantial detriment as a result of the variance:
- (d) Whether the variance would adversely affect the delivery of governmental services such as water, sanitary sewer, or garbage removal;
- (e) Whether the property owner purchased the property with knowledge of the zoning restriction;
- (f) Whether the property owner's predicament feasibly can be obviated through some method other than a variance; and
- (g) Whether the spirit and intent behind the zoning requirement would be observed and substantial justice done by granting the variance. The Board shall determine, after and weighing the factors described above and any other factors the Board deems relevant,

whether the property owner has shown practical difficulties so inequitable as to justify granting a variance to the property owner.

20.13.3 **SAFEGUARDS:**

SUPPLEMENTARY CONDITIONS AND

In granting any appeal or variance, the Board of Zoning Appeals may prescribe appropriate conditions and safeguards in conformity with this **O**ordinance. Violation of such conditions and safeguards, when made a part of the terms under which the appeal or variance is

granted, shall be deemed a violation of this $\underline{\mathbf{O}}$ ordinance and punishable under Section 19.12 of this $\underline{\mathbf{O}}$ ordinance.

20.14 PROCEDURE AND REQUIREMENTS FOR APPROVAL OF CONDITIONALUSE PERMITS:

It is recognized that an increasing number of new kinds of uses are appearing daily, and that many of these and some other more conventional uses possess characteristics of such unique and special nature relative to location, design, size, method of operation, circulation, and public facilities that each specific use must be considered individually. These specific uses that are conditionally permitted under the provisions of this **O** ordinance shall follow the procedures and requirements set forth in this section.

20.14.1 **GENERAL:**

The following requirements shall be complied with prior to any approval or denial of a conditional use permit by the Board of Zoning Appeals:

- (1) A written application for a conditional use is submitted indicating the section of this **O**rdinance under which the conditional use is sought and stating the grounds on which it is requested per Section 20.14.2.
- (2) A public hearing shall be held as specified in Section 20.09 of this **O**ordinance.
- (3) The Board of Zoning Appeals shall determine:
 - (a) That it has the authority to grant the request.
 - (b) That the granting of the conditional use will not adversely affect the neighborhood in which the conditional use is to be located.
 - (c) That the conditional use is not one which is contrary to the policies of the Comprehensive Master Plan of the Municipality City of Bellbrook. In making this determination the Board shall be advised by the recommendation of the Planning Board.

Item C.Section 8, Item

20.14.2 <u>CONTENTS OF APPLICATION FOR</u>

CONDITIONAL USE PERMIT:

An application for conditional use permit shall be filed with the Chairman of the Board of Zoning Appeals by at least one owner or lessee of **the** property for which such conditional use is proposed. At a minimum, the application shall contain the following information:

- (1) Name, address, and phone number of applicant;
- (2) Legal description of property;
- (3) Description of existing use;
- (4) Zoning district;
- (5) Description of proposed conditional use;
- (6) A plan of the proposed site for the conditional use showing the location of all buildings, parking and loading area, traffic access and traffic circulation, open spaces, landscaping, refuse and service areas, utilities, signs, yards, and such other information as the Board may require determining if the proposed conditional use meets the intent and requirements of this **O** ordinance;
- (7) A narrative statement evaluating the effects on adjoining property; the effect of such elements as noise, glare, odor, fumes, and vibration on adjoining property; a discussion of the general compatibility with adjacent and other properties in the district; and the relationship of the proposed use to the comprehensive plan;
- (8) Such other information as may be required in Section 20.14.

20.14.3 GENERAL STANDARDS APPLICABLE TO ALL CONDITIONAL USES:

In addition to the specific requirements for conditionally permitted uses as specified in Section 20.14.1, the Board shall review the particular facts and

circumstances of each proposed use in terms of the following standards and shall find adequate evidence showing that such use at the proposed location:

- (1) Is in fact a conditional use as established under the provisions of the Zoning Ordinance for the zoning district involved;
- (2) Will be harmonious with and in accordance with the general objectives, or with any specific objective of the Comprehensive Plan and/or the Zoning Ordinance;
- (3) Will be designed, constructed, operated, and maintained so as to be harmonious and appropriate in appearance with the existing or intended character of the general vicinity and that such use will not change the essential character of the same area;
- (4) Will not be hazardous or disturbing to existing or future neighboring uses;
- (5) Will be served adequately by essential public facilities and services such as highways, streets, police and fire protection, drainage structures, refuse disposal, water and sewer, and schools; or that the persons or agencies responsible for the establishment of the proposed use shall be able to provide adequately any such services;
- (6) Will not create excessive additional requirements at public cost for public facilities and services and will not be detrimental to the economic welfare of the community;
- (7) Will not involve uses, activities, processes, materials, equipment and conditions of operation that will be detrimental to any persons property, or the general welfare by reason of excessive production of traffic, noise, smoke, fumes, glare or odors;
- (8) Will have vehicular approaches to the property which shall be so designed as not to create an interference with traffic on surrounding public thoroughfares;
- (9) Will not result in the destruction, loss, or damage of a natural, scenic, or historic feature of major importance.

20.14.4 <u>SUPPLEMENTAL SAFEGUARDS AND CONDITIONS:</u>

In granting any conditional use permit, the Board of Zoning Appeals may prescribe appropriate conditions and safeguards in conformity to the provisions of this $\underline{\mathbf{O}}$ ordinance and the recommendation of the Planning Board. The Board of Zoning Appeals shall require a bond to assure conformance to such conditions and safeguards as may be necessary. Violation of such conditions and safeguards shall cause the performance bond mentioned above to be forfeited and shall be deemed a violation of this $\underline{\mathbf{O}}$ ordinance and punishable under Section 19.12.

20.14.5 **EXPIRATION OF CONDITIONAL USE PERMIT:**

A conditional use permit shall expire in one (1) year after it is issued unless actual construction has taken place or is underway except as provided elsewhere in this $\underline{\mathbf{O}}$ ordinance.

20.15 <u>DETERMINATION OF SIMILAR USES</u>

The Planning Board will determine if uses not specifically mentioned in this **O**ordinance are similar to uses permitted within a district.

20.16 <u>DETERMINATION OF DISTRICT BOUNDARY LOCATION</u>

The Board of Zoning Appeals <u>City Council</u> shall determine the exact location of any district boundary if there is uncertainty as to exact location thereof. In making such determination the Board shall be guided by the provisions of Section 4.03.

20.17 **AMENDMENT LIMITATION:**

Petitions for zoning amendment, conditional use permits or variances concerning any parcel of property, portion thereof, or use thereon shall not be accepted for consideration more than once during any consecutive twelve (12) month period.

ARTICLE 21 <u>AMENDMENT</u>

The Council may amend, supplement, change or repeal this <u>O</u>ordinance in the manner prescribed by Section 5.03 and 5.04 of the Bellbrook Charter after receiving the recommendation from the Planning Board.

21.01 RIGHT OF PETITION - REFERRAL TO PLANNING BOARD

A proposed amendment, supplement, change or repeal may be originated by the Council, the Planning Board, or the owner of the affected property. All such proposals except those originating with the Planning Board shall be referred to that Board for a recommendation thereon. Proposals originating by Council shall be referred to the Planning Board by resolution stating the date, not less than sixty days from the next regularly scheduled meeting of the Planning Board, as to when the Planning Board recommendation is required. Upon its request, the Planning Board shall be granted a thirty-day extension by which its recommendation shall be due. At its discretion, Council may, upon satisfactory petition by the Planning Board, grant a further extension. The Council shall take no final action on any amendment, supplement, change or repeal until it has received and studied the recommendation of the Planning Board; provided, however, if the Planning Board shall fail to forward its recommendation on any Council initiated proposal within the time specified by the resolution proposing the same, then Council may act thereon as though it had received a negative recommendation from the Planning Board.

The Planning Board shall study the proposed amendment, supplement, change or repeal. If the Board decides the proposal has merit it shall hold a public hearing thereon. At least ten (10) days' notice of the time and place of such hearing shall be mailed to the petitioner, and all property owners within three (300) hundred feet of the property included in petitions for rezoning. Public notices of hearings on proposed amendments to Articles of the Zoning Regulations shall be posted in five (5) public places as determined by Council. The Planning Board shall make a report of its findings and a recommendation to the Council.

21.02 <u>CONTENTS OF APPLICATION:</u>

Applications for amendments to the Official Zoning Map adopted as part of this $\underline{\mathbf{O}}$ ordinance by Section 4.02 shall contain at least the following information:

(1) Name, address, and phone number of applicant;

- (2) Proposed amending ordinance, approved as to form by the City Legal Advisor;
- (3) Present use;
- (4) Present zoning district;
- (5) Proposed use;
- (6) Proposed zoning district;
- (7) A vicinity map at a scale approved by the <u>Community</u> <u>Development Administrator</u> <u>Zoning Inspector</u> showing property lines, thoroughfares, existing and proposed zoning and such other items as the <u>Community Development Administrator</u> <u>Zoning Inspector</u> may require;
- (8) A list of all property owners, and their mailing addresses, who are within three hundred(300) feet of the property proposed to be rezoned and others that may have a substantial interest in the case, except that addresses need not be included where more than ten (10) parcels are to be rezoned;
- (9) A statement on how the proposed amendment relates to the Comprehensive Plan; (See Comprehensive Plan map in Community Development Administrator's Zoning Inspector's office).
- (10) A fee as established by Council (See Section 19.15)

21.03 **REVIEWS BY OFFICIALS AND AGENCIES:**

Prior to acceptance, the Board may request and review recommendations for each proposed amendment from the Miami Valley Regional Planning Commission, the Transportation Coordinating Committee, the Regional Planning and Coordinating Commission of Greene County, City Law Director, and other eCity officials, agencies, and adjacent local governments to ascertain the nature

of highway plans, drainage and flooding problems, utility plans and other factors relevant to the proposal.

21.04 <u>COUNCIL ACTION ON RECOMMENDATION:</u>

When the Council receives an adverse recommendation on a proposed amendment, supplement, change or repeal it may concur with the Planning Board and, by a motion of Council stop further action, or if the Council does not agree with the recommendation of the Planning Board, it shall hold a public hearing prior to taking final action on the **O**ordinance.

When the Council receives an affirmative recommendation from the Planning Board on a proposed amendment the Council shall hold a public hearing thereon. When such hearing is held, notice shall be given as prescribed by the Bellbrook Charter, for an ordinance. In addition to such notice the enforcing officer Community Development Administrator or designee shall, upon notification by the Clerk, place upon the involved premises in a prominent position a printed sign bearing the words "rezoning pending from ... classification to ... classification" and the time, date, and place of hearing. Said sign shall be located within ten (10) feet of the right-of-way of a public road or street. The sign shall be placed on the property at three hundred (300) foot intervals. One sign shall suffice for lesser distances. The size of the letters on the sign shall be a minimum of two and one-half inches in height. Such sign shall be posted ten (10) days before the date of the hearing. The Council may adopt such

amendment, supplement, change₂ or repeal without further reference to the Planning Board. If the Planning Board recommended disapproval of the change, however, such amendment shall not be passed except by an affirmative vote of four (4) Council members.

Whenever a written protest against a proposed amendment, supplement, change or repeal is presented duly signed by the owners of twenty percent (20%) or more of the frontage of the block proposed to be altered, or by the owners of twenty percent (20%) or more of the frontage of the block immediately in the rear thereof, or by the owners of twenty percent (20%) of the frontage of the block directly opposite the frontage proposed to be altered, such amendment shall not be passed except by a concurring vote of four (4) Council members.

21.05 <u>SCHEDULING PUBLIC HEARING BY COUNCIL:</u>

Council shall schedule a public hearing not more than forty (40) days from the receipt of the recommendation from the Planning Board.

21.06 **NOTICE TO PROPERTY OWNERS:**

If the proposed amendment is intended to rezone or redistrict ten (10) or less parcels of land as listed on the tax duplicate, written notice of the hearing shall be mailed by the Clerk of Council, by first class mail, at least ten (10) days before the day of the public hearing to all owners of property within, contiguous to, and directly across the street from such area proposed to be rezoned or redistricted to the address of such owners appearing on the County Auditor's current tax list or the Treasurer's mailing list and to such other list or lists that may be specified by Council. The failure to deliver the notification, as provided in this section shall not invalidate any such amendment. The notice shall contain the same information as required of notices published in newspapers and/or posted in public places as required for hearings on ordinances.

21.07 **AMENDMENT LIMITATION:**

Petitions for zoning amendment, conditional use permit or variances concerning any parcel of, portion thereof, or use thereon shall not be accepted for consideration more than once during any consecutive twelve (12) month period.

21.08 **ANNEXATION:**

All land annexed to the City subsequent to the adoption of this **O**erdinance shall remain subject to the previous Township zoning district until such time as the Official Zoning Map is amended according to the provisions of this Article. All land annexed to the City which, prior to annexation, is not subject to Township zoning shall remain un-zoned until the Official Zoning Map is amended according to the provisions of this Article.

21.09 **FEES:**

Fees will be determined by <u>resolution</u> ordinance of the <u>Municipality</u> <u>City</u> of Bellbrook. (See Section 19.15).

ARTICLE 22 <u>VALIDITY AND SEVERABILITY</u>

Should any section or provision of this Ordinance be declared by the courts to be unconstitutional or invalid, such decision shall not affect the validity of the Ordinance as a whole, or any part thereof other than the part so declared to be unconstitutional or invalid.

ARTICLE 23 <u>INTERPRETATION AND CONFLICT</u>

2024-O-6 Exhibit A

In its interpretation and application, the provisions of this Ordinance shall be held to be minimum requirements, adopted for the promotion of the public health, morals, safety, and the general welfare. Wherever the requirements of this Ordinance are at variance with the requirements of any other lawfully adopted rules, regulations, or Ordinances, the most restrictive, or that imposing the high standards, shall govern.

ARTICLE 24

REPEAL OF CONFLICTING ORDINANCES

All ordinances, or parts of ordinances in conflict with this $\underline{\mathbf{O}}$ ordinance or inconsistent with the provisions of this $\underline{\mathbf{O}}$ ordinance are hereby repealed and declared null and void and of no effect.

ARTICLE 25

EFFECTIVE DATE

(1) Date of public hearing
(2) Date of Publication
(3) Date of Adoption by the Village Council: <u>January 24, 1972</u>
(4) Date and time this Ordinance shall take effect

File Attachments for Item:

A. Ordinance No. 2024-O-7 AUTHORIZING THE CITY MANAGER TO SUBMIT THE TAX BUDGET FOR FISCAL YEAR 2025 TO THE GREENE COUNTY AUDITOR, AND DECLARING AN EMERGENCY (Cyphers)



Monday May 27, 2024 7:00 pm City Council Chambers 15 E. Franklin Street

There will be an open Public Hearing by the Bellbrook City Council regarding Ordinance No.2024-O-7

A Public Hearing will be held by Bellbrook City Council, regarding Ordinance No 2024-O-7

Ordinance 2024-O-7 AUTHORIZING THE CITY MANAGER TO SUBMIT THE TAX BUDGET FOR FISCAL YEAR 2025 TO THE GREENE COUNTY AUDITOR, AND DECLARING AN EMERGENCY

Monday May 27, 2024 7:00 pm in the Council Chambers 15 E. Franklin Street. The public is welcome to attend or send comments to the Clerk of Council at clerk@cityofbellbrook.org. A copy of the Ordinance is attached.

Agenda and additional meeting information available at www.cityofbellbrook.org

Ordinance No. 2024-O-7 May 27, 2024

City of Bellbrook State of Ohio

Ordinance No. 2024-O-7

AUTHORIZING THE CITY MANAGER TO SUBMIT THE TAX BUDGET FOR FISCAL YEAR 2025 TO THE GREENE COUNTY AUDITOR, AND DECLARING AN EMERGENCY

WHEREAS, the City is required to prepare and submit the tax budget for 2025 to the Greene County Auditor by July 20, 2024; and

WHEREAS, prior to submitting, there must be a public hearing of the tax budget.

NOW, THEREFORE, THE CITY OF BELLBROOK HEREBY ORDAINS:

- Section 1. The Tax Budget for fiscal year 2025 as set forth in Exhibit A attached hereto and incorporated herein by reference is approved for submission to the Greene County Auditor to guarantee the City's eligibility for the allocation of local government funds in 2025
- Section 2. That there was a public hearing noticed at least ten (10) days prior to approving the tax Budget
- Section 3. This Ordinance is declared to be an emergency measure necessary to meet a public emergency affecting health, safety, morals or the public welfare, or a special emergency in the operation of a Municipal department, and for the further reason that in order to receive Local Government Funding, this Tax Budget must be filed by July 20, 2024; therefore, this Ordinance shall take full force and effect immediately upon its adoption by Council
- Section 4. It is hereby found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council, and that any and all deliberations of this Council that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including, but not limited to Section 121.22 of the Ohio Revised Code.

PASSED BY City Council this 27 th day of May 2024.
Yeas;Nays.
AUTHENTICATION:
Michael W. Schweller, Mayor
Robert Schommer, Clerk of Council
APPROVED AS TO FORM:
Stephen McHugh, Municipal Attorney

GENERAL FUND		For 2022 Actual	For 2023 Actual		Current Year Projected 2024	Budget Year Projected 2025
REVENUE						
Total Local Taxes	\$	709,270	\$ 713,906	\$	961,000	\$ 970,000
Total Intergovernmental Revenues	\$	353,146	\$ 285,146	\$	234,815	\$ 250,000
Charge for Services	\$	99,441	\$ 23,522	\$	19,000	\$ 19,500
Special Assessments	\$	359	\$ 2,887	\$	3,000	\$ 3,000
Fines, Licenses, and Permits	\$	33,515	\$ 113,934	\$	102,600	\$ 103,300
Investment Earnings	\$	70,410	\$ 217,309	\$	150,000	\$ 150,000
Miscellaneous	\$	22,082	\$ 75,730	\$	11,000	\$ 11,000
TOTAL REVENUE	\$	1,288,223	\$ 1,432,434	\$	1,481,415	\$ 1,506,800
EXPENDITURES General Fund						
Total Expenses Other Uses of Funds	\$	558,391	\$ 1,104,511	\$	1,089,359	\$ 1,250,000
Transfer to the Police Fund	\$	-	\$ 110,000	\$	342,160	\$ 385,929
Transfer to the Fire Fund	\$	-	\$ -	\$	- ,	\$ -
Transfer to the Capital Improvement Fund		-	\$ -	, \$	-	\$ -
Total Other Uses of Funds	<u>\$</u> \$	_	\$ 110,000	\$	342,160	\$ 385,929
TOTAL EXPENDITURES	\$	558,391	\$ 1,214,511	\$	1,431,519	\$ 1,635,929

GENERAL FUND	For 2022 Actual	For 2023 Actual	Current Year Projected 2024	Budget Year Projected 2025
Revenues over/(under) Expenditures	\$ 729,832	\$ 217,923	\$ 49,896	\$ (129,129)
Beginning Unencumbered Balance	\$ 2,017,810	\$ 2,748,002	\$ 2,774,895	\$ 2,624,791
Ending Cash Fund Balance	\$ 2,747,642	\$ 2,965,925	\$ 2,824,791	\$ 2,495,662
Estimated Encumbrances (outstanding at year end)	\$ 143,485	\$ 191,030	\$ 200,000	\$ 20,000
Estimated Ending Unencumbered Fund Balance	\$ 2,604,157	\$ 2,774,895	\$ 2,624,791	\$ 2,475,662

POLICE LEVY FUND	For 2022 Actual	For 2023 Actual	Current Year Projected 2024	Budget Year Projected 2025
REVENUE				
Property Tax	\$ 1,403,059	\$ 1,412,090	\$ 1,597,000	\$ 1,616,000
Total Intergovernmental Revenues	\$ 208,318	\$203,600.05	\$ 190,000	\$ 195,700
Charges for Services	\$ 18,375	\$26,784.61	\$ 62,000	\$ 63,860
Fines, Licenses & Permits	\$ 2,323	(\$1,641.08)	\$ 2,250	\$ 2,300
Miscellaneous	\$ 9,735	\$4,221.97	\$ 3,000	\$ 3,000
Transfers-in from the General Fund	\$ -	\$110,000.00	\$ 342,160	\$ 385,929
TOTAL REVENUE	\$ 1,641,810	\$ 1,755,056	\$ 2,196,410	\$ 2,266,789
EXPENDITURES Total Police Expenses	\$ 1,774,171	\$ 1,997,331	\$ 2,200,766	\$ 2,266,789
TOTAL EXPENDITURES	\$ 1,774,171	\$ 1,997,331	\$ 2,200,766	\$ 2,266,789
Revenues over/(under) Expenditures	\$ (132,361)	\$ (242,275)	\$ (4,356)	\$ -
Beginning Unencumbered Balance	\$ 413,990	\$ 281,631	\$ 39,356	\$ -
Ending Cash Fund Balance	\$ 281,629	\$ 281,631	\$ 35,000	\$ -
Estimated Encumbrances (outstanding at year end)	\$ 53,072	\$ 41,150	\$ 35,000	\$ -
Estimated Ending Unencumbered Fund Balance	\$ 228,557	\$ 240,481	\$ -	\$ -

	Fire Fund		Ord	inance 2024-O-7 Ex	khibi	t A I Item A.Section 9, Item
FIRE LEVY FUND	For 2022 Actual	For 2023 Actual	Current Year Projected 2024	Budget Year Projected 2025		
REVENUE						
Property Taxes	\$ 1,119,115	\$ 1,126,408	\$	1,234,000	\$	1,250,000
Total Intergovernmental Revenues	\$ 123,762	\$ 121,853	\$	117,000	\$	120,000
Charges for Services	\$ 101,735	\$ 118,697	\$	100,000	\$	103,000
Miscellaneous	\$ 32,048	\$ 3,379	\$	2,000	\$	2,000
Transfer-in from the General Fund	\$ -	\$ -	\$	-	\$	-
TOTAL REVENUE	\$ 1,376,660	\$ 1,370,337	\$	1,453,000	\$	1,475,000
EXPENDITURES Total Fire Expenses	1,397,287	1,419,898		1,564,755		1,611,698
TOTAL EXPENDITURES	\$ 1,397,287	\$ 1,419,898	\$	1,564,755	\$	1,611,698
Revenues over/(under) Expenditures	\$ (20,627)	\$ (49,561)	\$	(111,755)	\$	(136,698)
Beginning Unencumbered Balance	\$ 535,454	\$ 514,828	\$	465,266	\$	321,011
Ending Cash Fund Balance	\$ 514,827	\$ 465,267	\$	353,511	\$	184,313
Estimated Encumbrances (outstanding at year end)	\$ 93,818	\$ 30,000	\$	32,500	\$	35,000
Estimated Ending Unencumbered Fund Balance	\$ 421,009	\$ 435,267	\$	321,011	\$	149,313

POLICE PENSION LEVY FUND	For 2022 Actual	For 2023 Actual	Current Year Projected 2024	Budget Year Projected 2025
REVENUE				
Total PropertyTaxes	\$ 56,700	\$ 57,111	\$ 81,000	\$ 82,000
Total Intergovernmental Revenues	\$ 8,345	\$ 8,381	\$ 8,000	\$ 8,000
TOTAL REVENUE	\$ 65,045	\$ 65,492	\$ 89,000	\$ 90,000
EXPENDITURES Total Police Pension Expenses	\$ 64,638	\$ 80,590	\$ 89,590	\$ 95,000
TOTAL EXPENDITURES	\$ 64,638	\$ 80,590	\$ 89,590	\$ 95,000
Revenues over/(under) Expenditures	\$ 407	\$ (15,098)	\$ (590)	\$ (5,000)
Beginning Unencumbered Balance	\$ 24,861	\$ 25,268	\$ 10,170	\$ 9,580
Ending Cash Fund Balance	\$ 25,268	\$ 10,170	\$ 9,580	\$ 4,580
Estimated Encumbrances (outstanding at year end)	\$ -	\$ -	\$ -	\$ -
Estimated Ending Unencumbered Fund Balance	\$ 25,268	\$ 10,170	\$ 9,580	\$ 4,580

Item A.Section 9, Item

FUND		Estimated Unencumbered Fund Balance 1/1/2025	Budget Year Projected Receipts	Total Available For Projected Expenditures			Budget Year Projected Expenses	Projected Unencumbered Balance 12/31/2025	
Special Revenue Funds:									
Street Fund	\$	367,000	\$ 435,000	\$	802,000	\$	420,000	\$	382,000
State Highway Fund	\$	98,590	\$ 30,000	\$	128,590	\$	53,000	\$	75,590
Fuel System Fund	\$	3,295	\$ 1,500	\$	4,795	\$	2,500	\$	2,295
Motor Vehicle License Fund	\$	106,819	\$ 49,000	\$	155,819	\$	50,000	\$	105,819
OneOhio Opioid Settlement Fund	\$	6,398	\$ 6,000	\$	12,398	\$	12,000	\$	398
Capital Improvement Fund	\$	301	\$ 440,000	\$	440,301	\$	440,000	\$	301
TOTAL SPECIAL REVENUE FUNDS	\$	582,403	\$ 961,500	\$	1,543,903	\$	977,500	\$	566,403
Enterprise Funds: Waste Collection Fund Water Fund TOTAL ENTERPRISE FUNDS	\$ \$ \$	224,300 4,533,985 4,758,285	\$ 645,000 1,635,000 2,280,000	\$	869,300 6,168,985 7,038,285	\$	650,000 1,640,000 2,290,000	\$ \$ \$	219,300 4,528,985 4,748,285
TRUST AND AGENCY FUNDS									
Performance Bond Fund Agency Fund	\$	78,533 -	\$ 21,000	\$	99,533	\$	25,000	\$	74,533
TOTAL TRUST AND AGENCY FUNDS	\$	78,533	\$ 21,000	\$	99,533	\$	25,000	\$	74,533
TOTAL OTHER FUNDS (MEMO ONLY)	\$	5,419,221	\$ 3,262,500	\$	8,681,721	\$	3,292,500	\$	5,389,221
GRAND TOTAL ALL FUNDS	\$	8,374,603	\$ 8,601,089	\$	16,975,692	\$	8,515,987	\$	8,073,776

Item A.Section 9, Item

State of Ohio Public Works Commission

Loan Amortization Schedule

City of Bellbrook

Loan Nbr: CK15U Upper Hillside Water Main Improvements

Loan Amount: 290,000.00 Interest Rate (percent): 0.0 Loan Term (years): 30.0

Schedule CK15U-0-01

Year	Month	- Per	Per Start	Principal	Interest	Payment	Balance	Mont	h - Per	Per Start	Principal	Interest	Payment	Balance
2019	Loan Ini	tializatio	on				290,000.00	Jul	1	290,000.00	4,833.33	0.00	4,833.33	285,166.67
2020	Jan	2	285,166.67	4,833.33	0.00	4,833.33	280,333.34	Jul	3	280,333.34	0.00	0.00	0.00	280,333.34
2021	Jan	4	280,333.34	4,833.33	0.00	4,833.33	275,500.01	Jul	5	275,500.01	4,833.33	0.00	4,833.33	270,666.68
2022	Jan	6	270,666.68	4,833.33	0.00	4,833.33	265,833.35	Jul	7	265,833.35	4,833.33	0.00	4,833.33	261,000.02
2023	Jan	8	261,000.02	4,833.33	0.00	4,833.33	256,166.69	Jul	9	256,166.69	4,833.33	0.00	4,833.33	251,333.36
2024	Jan	10	251,333.36	4,833.33	0.00	4,833.33	246,500.03	Jul	11	246,500.03	4,833.33	0.00	4,833.33	241,666.70
2025	Jan	12	241,666.70	4,833.33	0.00	4,833.33	236,833.37	Jul	13	236,833.37	4,833.33	0.00	4,833.33	232,000.04
2026	Jan	14	232,000.04	4,833.33	0.00	4,833.33	227,166.71	Jul	15	227,166.71	4,833.33	0.00	4,833.33	222,333.38
2027	Jan	16	222,333.38	4,833.33	0.00	4,833.33	217,500.05	Jul	17	217,500.05	4,833.33	0.00	4,833.33	212,666.72
2028	Jan	18	212,666.72	4,833.33	0.00	4,833.33	207,833.39	Jul	19	207,833.39	4,833.33	0.00	4,833.33	203,000.06
2029	Jan	20	203,000.06	4,833.33	0.00	4,833.33	198,166.73	Jul	21	198,166.73	4,833.33	0.00	4,833.33	193,333.40
2030	Jan	22	193,333.40	4,833.33	0.00	4,833.33	188,500.07	Jul	23	188,500.07	4,833.33	0.00	4,833.33	183,666.74
2031	Jan	24	183,666.74	4,833.33	0.00	4,833.33	178,833.41	Jul	25	178,833.41	4,833.33	0.00	4,833.33	174,000.08
2032	Jan	26	174,000.08	4,833.33	0.00	4,833.33	169,166.75	Jul	27	169,166.75	4,833.33	0.00	4,833.33	164,333.42
2033	Jan	28	164,333.42	4,833.33	0.00	4,833.33	159,500.09	Jul	29	159,500.09	4,833.33	0.00	4,833.33	154,666.76
2034	Jan	30	154,666.76	4,833.33	0.00	4,833.33	149,833.43	Jul	31	149,833.43	4,833.33	0.00	4,833.33	145,000.10
2035	Jan	32	145,000.10	4,833.33	0.00	4,833.33	140,166.77	Jul	33	140,166.77	4,833.33	0.00	4,833.33	135,333.44
2036	Jan	34	135,333.44	4,833.33	0.00	4,833.33	130,500.11	Jul	35	130,500.11	4,833.33	0.00	4,833.33	125,666.78
2037	Jan	36	125,666.78	4,833.33	0.00	4,833.33	120,833.45	Jul	37	120,833.45	4,833.33	0.00	4,833.33	116,000.12
2038	Jan	38	116,000.12	4,833.33	0.00	4,833.33	111,166.79	Jul	39	111,166.79	4,833.33	0.00	4,833.33	106,333.46
2039	Jan	40	106,333.46	4,833.33	0.00	4,833.33	101,500.13	Jul	41	101,500.13	4,833.33	0.00	4,833.33	96,666.80
2040	Jan	42	96,666.80	4,833.33	0.00	4,833.33	91,833.47	Jul	43	91,833.47	4,833.33	0.00	4,833.33	87,000.14
2041	Jan	44	87,000.14	4,833.33	0.00	4,833.33	82,166.81	Jul	45	82,166.81	4,833.33	0.00	4,833.33	77,333.48
2042	Jan	46	77,333.48	4,833.33	0.00	4,833.33	72,500.15	Jul	47	72,500.15	4,833.33	0.00	4,833.33	67,666.82
2043	Jan	48	67,666.82	4,833.33	0.00	4,833.33	62,833.49	Jul	49	62,833.49	4,833.33	0.00	4,833.33	58,000.16
2044	Jan	50	58,000.16	4,833.33	0.00	4,833.33	53,166.83	Jul	51	53,166.83	4,833.33	0.00	4,833.33	48,333.50
2045	Jan	52	48,333.50	4,833.33	0.00	4,833.33	43,500.17	Jul	53	43,500.17	4,833.33	0.00	4,833.33	38,666.84
2046	Jan	54	38,666.84	4,833.33	0.00	4,833.33	33,833.51	Jul	55	33,833.51	4,833.33	0.00	4,833.33	29,000.18
2047	Jan	56	29,000.18	4,833.33	0.00	4,833.33	24,166.85	Jul	57	24,166.85	4,833.33	0.00	4,833.33	19,333.52
2048	Jan	58	19,333.52	4,833.33	0.00	4,833.33	14,500.19	Jul	59	14,500.19	4,833.33	0.00	4,833.33	9,666.86
2049	Jan	60	9,666.86	4,833.33	0.00	4,833.33	4,833.53	Jul	61	4,833.53	4,833.53	0.00	4,833.53	0.00

Item A.Section 9, Item



State of Ohio

Public Works Commission

Loan Amortization Schedule

City of Bellbrook

Loan Nbr: CK04K

Water Treatment Plant Improvement

Loan Amount:

1,123,459.13

Interest Rate (percent):

0.0

Loan Term (years):

20.0

Schedule CK04K-0-01

					lieuule CN04	K-0-01								
Year	Month - Po	er	Per Start	Principal	Interest	Payment	Balance	Mon	nth - Per	Per Start	Principal	Interest	Payment	Balance
2012						_		Loan	n Initializati	on				1,123,459.12
2013	Jan	1	1,123,459.13	28,086.48	0.00	28,086.48	1,095,372.65	Jul	2	1,095,372.65	28,086.48	0.00	28,086.48	1,067,286.17
2014	Jan	3	1,067,286.17	28,086.48	0.00	28,086.48	1,039,199.69	Jul	4	1,039,199.69	28,086.48	0.00	28,086.48	1,011,113.21
2015	Jan	5	1,011,113.21	28,086.48	0.00	28,086.48	983,026.73	Jul	6	983,026.73	28,086.48	0.00	28,086.48	954,940.25
2016	Jan	7	954,940.25	28,086.48	0.00	28,086.48	926,853.77	Jul	8	926,853.77	28,086.48	0.00	28,086.48	898,767.29
2017	Jan	9	898,767.29	28,086.48	0.00	28,086.48	870,680.81	Jul	10	870,680.81	28,086.48	0.00	28,086.48	842,594.33
2018	Jan 1	11	842,594.33	28,086.48	0.00	28,086.48	814,507.85	Jul	12	814,507.85	28,086.48	0.00	28,086.48	786,421.37
2019	Jan 1	13	786,421.37	28,086.48	0.00	28,086.48	758,334.89	Jul	14	758,334.89	28,086.48	0.00	28,086.48	730,248.41
2020	Jan 1	15	730,248.41	28,086.48	0.00	28,086.48	702,161.93	Jul	16	702,161.93	0.00	0.00	0.00	702,161.93
2021	Jan 1	17	702,161.93	28,086.48	0.00	28,086.48	674,075.45	Jul	18	674,075.45	28,086.48	0.00	28,086.48	645,988.97
2022	Jan 1	19	645,988.97	28,086.48	0.00	28,086.48	617,902.49	Jul	20	617,902.49	28,086.48	0.00	28,086.48	589,816.01
2023	Jan 2	21	589,816.01	28,086.48	0.00	28,086.48	561,729.53	Jul	22	561,729.53	28,086.48	0.00	28,086.48	533,643.05
2024	Jan 2	23	533,643.05	28,086.48	0.00	28,086.48	505,556.57	Jul	24	505,556.57	28,086.48	0.00	28,086.48	477,470.09
2025	Jan 2	25	477,470.09	28,086.48	0.00	28,086.48	449,383.61	Jul	26	449,383.61	28,086.48	0.00	28,086.48	421,297.13
2026	Jan 2	27	421,297.13	28,086.48	0.00	28,086.48	393,210.65	Jul	28	393,210.65	28,086.48	0.00	28,086.48	365,124.17
2027	Jan 2	29	365,124.17	28,086.48	0.00	28,086.48	337,037.69	Jul	30	337,037.69	28,086.48	0.00	28,086.48	308,951.21
2028	Jan 3	31	308,951.21	28,086.48	0.00	28,086.48	280,864.73	Jul	32	280,864.73	28,086.48	0.00	28,086.48	252,778.25
2029	Jan 3	33	252,778.25	28,086.48	0.00	28,086.48	224,691.77	Jul	34	224,691.77	28,086.48	0.00	28,086.48	196,605.29
2030	Jan 3	35	196,605.29	28,086.48	0.00	28,086.48	168,518.81	Jul	36	168,518.81	28,086.48	0.00	28,086.48	140,432.33
2031	Jan 3	37	140,432.33	28,086.48	0.00	28,086.48	112,345.85	Jul	38	112,345.85	28,086.48	0.00	28,086.48	84,259.37
2032	Jan 3	39	84,259.37	28,086.48	0.00	28,086.48	56,172.89	Jul	40	56,172.89	28,086.48	0.00	28,086.48	28,086.41
2033	Jan 4	41	28,086.41	28,086.41	0.00	28,086.41	0.00							

File Attachments for Item:

A. Resolution 2024-R-12 DETERMINING IT NECESSARY TO PROCEED WITH SUBMITTING THE QUESTION OF LEVYING AN ADDITIONAL TAX IN EXCESS OF THE TEN-MILL LIMITATION TO THE ELECTORS OF THE CITY, PURSUANT TO OHIO REVISED CODE SECTION 5705.19 AND SUBSECTION 5705.19(JJ), AS AMENDED, WHICH LEVY SHALL BE TWO AND TWO TENTHS (2.2) MILLS AND SHALL RUN FOR A CONTINUING PERIOD OF TIME PURSUANT TO OHIO REVISED CODE SECTION 5705.19, AND REQUESTING THE COUNTY AUDITOR TO CERTIFY MATTERS IN CONNECTION THEREWITH (Harding)

RECORD OF RESOLUTIONS

Resolution No. 2024-R-12

May 13, 2024, 2024

City of Bellbrook State of Ohio

Resolution No. 2024-R-12

DETERMINING IT NECESSARY TO PROCEED WITH SUBMITTING THE QUESTION OF LEVYING AN ADDITIONAL TAX IN EXCESS OF THE TEN-MILL LIMITATION TO THE ELECTORS OF THE CITY, PURSUANT TO OHIO REVISED CODE SECTION 5705.19 AND SUBSECTION 5705.19(JJ), AS AMENDED, WHICH LEVY SHALL BE TWO AND TWO TENTHS (2.2) MILLS AND SHALL RUN FOR A CONTINUING PERIOD OF TIME PURSUANT TO OHIO REVISED CODE SECTION 5705.19, AND REQUESTING THE COUNTY AUDITOR TO CERTIFY MATTERS IN CONNECTION THEREWITH

WHEREAS, the Bellbrook City Council finds that an additional tax levy in excess of the ten-mill limitation is required for the benefit of the City, pursuant to Ohio Revised Code Section 5705.19(JJ), for the purpose of providing and maintaining fire apparatus, mechanical resuscitators, underwater rescue and recovery equipment, or other fire equipment and appliances, buildings and sites therefor, or sources of water supply and materials therefor, for the establishment and maintenance of lines of fire-alarm communications, for the payment of firefighting companies or permanent, part-time, or volunteer firefighting, emergency medical service, administrative, or communications personnel to operate the same, including the payment of any employer contributions required for such personnel under section 145.48 or 742.34 of the Revised Code, for the purchase of ambulance equipment, for the provision of ambulance, paramedic, or other emergency medical services operated by a fire department or firefighting company, or for the payment of related costs, or for providing and maintaining motor vehicles, communications, other equipment, buildings, and sites for such buildings used directly in the operation of the police department, for the payment of salaries of permanent or part-time police, communications, or administrative personnel to operate the same, including the payment of any employer contributions required for such personnel pursuant to section 145.48 or 742.33 of the Revised Code, for the provision of ambulance or emergency medical services operated by a police department, or for the payment of other related costs, at a rate not to exceed two and two tenths (2.2) mills for each \$1 of taxable valuation, which amounts to twenty-two cents (\$0.22) for each \$100 of the taxable property within the City of Bellbrook, County of Greene, State of Ohio, for a continuing period of time, pursuant to Section 5705.19 of the Ohio Revised Code; and

WHEREAS, pursuant to Section 5705.03 of the Ohio Revised Code, this Council is required to certify to the Greene County Auditor a Resolution requesting the County Auditor to certify certain matters in connection with such a tax levy.

NOW, THEREFORE, THE CITY OF BELLBROOK HEREBY RESOLVES:

Section 1. Pursuant to Section 5705.19 of the Ohio Revised Code, the Bellbrook City Council determines it necessary that an additional tax be levied in excess of the ten-mill limitation for the benefit of this City, for the purpose of providing and maintaining fire apparatus, mechanical resuscitators, underwater rescue and recovery equipment, or other fire equipment and appliances, buildings and sites therefor, or sources of water supply and materials therefor, for the establishment and maintenance of lines of fire-alarm communications, for the payment of firefighting companies or permanent, part-time,

Item A.Section 10, Item

RECORD OF RESOLUTIONS

Resolution No. 2024-R-12

May 13, 2024, 2024

or volunteer firefighting, emergency medical service, administrative, or communications personnel to operate the same, including the payment of any employer contributions required for such personnel under section 145.48 or 742.34 of the Revised Code, for the purchase of ambulance equipment, for the provision of ambulance, paramedic, or other emergency medical services operated by a fire department or firefighting company, or for the payment of related costs, or for providing and maintaining motor vehicles, communications, other equipment, buildings, and sites for such buildings used directly in the operation of the police department, for the payment of salaries of permanent or part-time police, communications, or administrative personnel to operate the same, including the payment of any employer contributions required for such personnel pursuant to section 145.48 or 742.33 of the Revised Code, for the provision of ambulance or emergency medical services operated by a police department, or for the payment of other related costs, at a rate not to exceed two and two tenths (2.2) mills for each \$1 of taxable valuation, which amounts to twenty-two cents (\$0.22) for each \$100 of the taxable property within the City of Bellbrook, County of Greene, State of Ohio, for a continuing period of time.

- Section 2. The territory of the City of Bellbrook is located entirely within Greene County, Ohio. The tax is to be levied on the ballot measure submitted to the entire territory of the City of Bellbrook.
- Section 3. Council seeks to have the question of the passage of the tax levy submitted to the electors of the City of Bellbrook at an election to be held on November 5, 2024. If approved by the electors, the tax levy shall first be placed upon the 2024 tax list and duplicate, for first collection in calendar year 2025.
- Section 4. Pursuant to Section 5705.03 of the Ohio Revised Code, the Greene County Auditor is hereby requested to certify to this Council, within ten (10) days after receiving this Resolution, the total current tax valuation of the City and the dollar amount of the revenue that would be generated by the number of mills specified in Section I above, and the Clerk of Council is hereby directed to certify forthwith a copy of this Resolution to the County Auditor at the earliest possible time so that the County Auditor may certify such matters in accordance with said Section 5705.03 of the Ohio Revised Code.
- Section 5. That it is found and determined that all formal actions of the City Council relating to the adoption of this Resolution were adopted in an open meeting of this Council, and that all deliberations of this Council and any of its committees that resulted in such formal action, were in meetings open to the public, in compliance with all legal requirements, including §121.22 of the Revised Code of the State of Ohio.

Section 6. That this resolution	shall take effec	et and be in force forthwith.
PASSED BY City Council this	day of	, 2024.
Yeas; Nays.		
AUTHENTICATION:		
Michael W. Schweller, Mayor		

RECORD OF RESOLUTIONS

Item A.Section 10, Item

Resolution No. 2024-R-12 May 13, 2024, 2024 Robert Schommer, Clerk of Council